# User's Guide 10/2004 Edition

# sinumerik

SINUMERIK 840D/840Di/810D Measuring Cycles



		Part 1: User's Guide	
SIEMENS		Introduction	1
		Parameter Description	2
SINUMERIK 840D/840D	i/810D	Measuring Cycle Auxiliary Programs	3
Measuring Cycles		Measuring in JOG	4
User's Guide		Measuring Cycles for Milling and Machining Centers	5
		Measuring Cycles for Turning Machines	6
		Miscellaneous Functions	7
		Part 2: Description of Functions	
		Hardware, Software and Installation	8
Valid for		Data Description	9
Control SINUMERIK 840D powerline SINUMERIK 840DE powerline (export ve SINUMERIK 840Di SINUMERIK 840DiE (export version)	ersion)	Start-Up	10
SINUMERIK 810D powerline SINUMERIK 810DE powerline (export ve	ersion)	Alarms	11
SIMODRIVE 611 digital		Adaptation of the	40
Software Version Measuring Cycles 6		Measuring Cycles to Previous Software Versions	12
10.04 Edition		Appendix	Α

## SINUMERIK® Documentation

#### **Printing history**

Brief details of this edition and previous editions are listed below.

The status of each edition is indicated by the code in the "Remarks" columns.

Status code in the "Remarks" column:

- **A** .... New documentation.
- **B**.... Unrevised reprint with new Order No.
- C .... Revised edition with new status.If factual changes have been made on the page since the last edition, this is indicated by a new edition coding in the header on that page.

Edition	Order No.	Remarks
09.95	6FC5298-3AA01-0BP0	А
03.96	6FC5298-3AA70-0BP1	С
12.97	6FC5298-4AA70-0BP0	С
12.98	6FC5298-5AA70-0BP0	С
08.99	6FC5298-5AA70-0BP1	С
06.00	6FC5298-5AA70-0BP2	С
10.00	6FC5298-6AA70-0BP0	С
09.01	6FC5298-6AA70-0BP1	С
11.02	6FC5298-6AA70-0BP2	С
01.04	6FC5298-6AA70-0BP3	С
03.04	6FC5298-7AA70-0BP0	С
10.04	6FC5298-7AA70-0BP1	С

#### Trademarks

SIMATIC<sup>®</sup>, SIMATIC HMI<sup>®</sup>, SIMATIC NET<sup>®</sup>, SIROTEC<sup>®</sup>, SINUMERIK<sup>®</sup> and SIMODRIVE<sup>®</sup> are Siemens trademarks. Other product names used in this documentation may be trademarks which, if used by third parties, could infringe the rights of their owners.

More information is available on the Internet at: http://www.siemens.com/motioncontrol	Other functions not described in this documentation might be executable in the control. This does not, however, represent an obligation to supply such functions with a new control or when servicing.
This document has been created using with WinWord V 2000 and Designer V 7.0.	We have checked the contents of this manual for agreement with the hardware and software described. Nonetheless, differences might exist and therefore we cannot guarantee that they are completely identical. The information contained in this document is, however, reviewed regularly and any necessary changes will be included in the next edition. We welcome all recommendations and suggestions.
© Siemens AG, 1995-2004. All rights reserved	Subject to change without prior notice
Order No. 6FC5298-7AA70-0BP1 Printed in Germany	Siemens Aktiengesellschaft



## Preface

## Structure of the Documentation

The SINUMERIK documentation is organized in 3 parts:

- General Documentation
- User Documentation
- Manufacturer/Service Documentation

#### Audience

This Manual is intended for machine-tool users. This publication provides detailed information that the user requires for operating the SINUMERIK 810D and 840D controls.

#### Standard scope

This Operator's Guide describes only the functionality of the standard scope. Extensions or changes made by the machine tool manufacturer are documented by the machine tool manufacturer.

For more detailed information on SINUMERIK 840D, 810D publications and other publications covering all SINUMERIK controls (e.g. universal interface, measuring cycles...), please contact your local Siemens office.

Other functions not described in this documentation might be executable in the control. This does not, however, represent an obligation to supply such functions with a new control or when servicing.

## Validity

This User's Guide is valid for the following controls:

Measuring cycles, version 6.3.

Software versions stated in the User's Guide refer to the 840D and their 810D equivalent, e.g. SW 5 (840D) corresponds to SW 3 (810D).

## SINUMERIK 840D powerline

Since 09/2001

- SINUMERIK 840D powerline and
- SINUMERIK 840DE powerline

have been available with improved performance. A list of the available **powerline** modules can be found in the hardware description /PHD/ in Section 1.1

## SINUMERIK 810D powerline

Since 12/2001

- SINUMERIK 810D powerline and
- SINUMERIK 810DE powerline

have been available with improved performance. A list of the available **powerline** modules can be found in the hardware description /PHC/ in Section 1.1

	Hotline	If you have any questions, please get in touch with our hotline: A&D Technical Support Phone: +49 (0) 180 5050-222 Fax: +49 (0) 180 5050-223
		http://www.siemens.com/automation/support-request
		Please send any queries about the documentation (suggestions or corrections) to the following fax number or email address: Fax: +49 (0) 0131 98-2176 E-mail: motioncontrol docu@erlf sigmens de
		Fax form: See the reply form at the end of the document.
	Internet address	http://www.siemens.com/motioncontrol
	Explanation of symbols	
<b>.</b> ,	Sequence	
=?	Explanation	
	Function	
●	Parameters	
<b>\$</b>	Programming example	
Ē	Programming	
Ť	Other Information	
۳ الس	Cross-references to other doo	cumentation or sections
Δ	Notes and warnings	
	Additional notes or backgrour	nd information





#### Warnings

The following warnings with varying degrees of severity appear in this document.



#### Danger

Indicates an imminently hazardous situation which, if not avoided, **will** result in death or serious injury or in substantial property damage.



## Warning

Indicates a potentially hazardous situation which, if not avoided, **could** result in death or serious injury or in substantial property damage.



## Caution

Used with the safety alert symbol indicates a potentially hazardous situation which, if not avoided, **may** result in minor or moderate injury or in property damage.

#### Caution

Used without safety alert symbol indicates a potentially hazardous situation which, if not avoided, **may** result in property damage.

## Notice

Used without the safety alert symbol indicates a potential situation which, if not avoided, **may** result in an undesirable result or state.

## Principle

Your SIEMENS SINUMERIK 840D, 804Di, 810D is state of the art and is manufactured in accordance with recognized safety regulations, standards and specifications.

## Supplementary devices

The applications of SIEMENS controls can be expanded for specific purposes through the addition of special add-on devices, equipment and expansions supplied by SIEMENS.

#### Personnel

Only **properly trained, authorized, reliable personnel** must be allowed to use this equipment. No-one without the necessary training must be allowed to operate the control, even temporarily.

The **areas of responsibility** assigned to personnel involved in setting up, operating and maintaining the equipment must be clearly **specified** and their compliance **verified**.

#### Procedure

**Before** the control is started up, personnel responsible for its operation must have read and understood the Operator's Guides. The company used this equipment is also obliged to carry out **continuous monitoring** of the overall technical condition of the equipment (with a view to identifying externally visible defects and damage as well as changes in the operating behavior of the control).



#### Servicing

Note

Repairs to equipment may only be carried out by **personnel specially trained and qualified** for the application in question in accordance with the provisions specified in the maintenance and servicing guides. All relevant safety regulations must be followed.

⁄叭

The following are deemed as **improper usage** and **exclude the manufacturer from all liability**:

**Any** usage or application incompatible with or beyond the scope of the items specified above.

Cases where the control is operated **in a technically imperfect condition**, without due provision for safety considerations and/or hazards, or in contravention with any of the instructions in the Instruction Manual.

Cases where faults which could affect safety are not remedied **before** starting up the control.

Any **modification**, **bypassing** or **decommissioning** of equipment on the control whose intended purpose is to ensure proper functioning, unrestricted use of equipment and/or active and passive safety.

Unforeseen danger can arise with reference to:

- life and limb of personnel,
- the control, machine or other assets of the owner and the user.



Notes





## Contents

## Part 1: User's Guide

Introduc	tion	1-17
1.1	Fundamentals	1-18
1.2	General prerequisites	1-19
1.3	Behavior on block search, dry run, program testing, simulation	1-20
1.4	Reference points on the machine and workpiece	1-23
1.5	Definition of the planes, tool types	
1.6	Probes that can be used	1-27
1.7 1.7.1 1.7.2 1.7.3 1.7.4	Probe, gauging block, calibration tool Measuring workpieces on milling machines, machining centers Measuring tools on milling machines, machining centers Measuring workpieces at the lathe Measuring tools at lathes	1-29 1-29 1-30 1-31 1-33
1.8	Measuring principle	1-34
1.9	Measuring strategy for measuring workpieces with tool offset	1-38
1.10	Parameters for checking the measurement result and offset	1-41
1.11	Effect of empirical value, mean value, and tolerance parameters	
1.12 1.12.1 1.12.2 1.12.3 1.12.4 1.12.5 1.12.6 1.12.7 1.12.8 1.13 1.13.1 1.13.2 1.13.3 1.13.4	Overview of measuring cycle functions for milling technology Tool measurement on milling machines, machining centers Calibrating workpiece probes Workpiece measurement at one point Measuring the workpiece parallel to the axis Measuring a workpiece at an angle Measuring a surface at an angle Measuring spheres (from measuring cycles SW 6.3 onwards) Workpiece measurement: Setting-up a corner Overview of measuring cycle functions for turning technology Measuring tools at lathes Calibrating workpiece probes Measuring tools at lathes: 1-point measurement Measuring workpieces at lathes: 2-point measurement	1-47 1-47 1-48 1-49 1-50 1-52 1-53 1-53 1-54 1-55 1-56 1-56 1-58 1-59 1-60
1.13.4	Measuring workpieces at latites: 2-point measurement	
Paramet	er Description	2-61
2.1	Parameter concept of the measuring cycles	
2.2 2 2 1	Parameter overview	
2.2.2	Result parameters	

2.3	Description of the most important defining parameters	2-66
2.3.1	Measurement variant: _MVAR	2-66
2.3.2	Number of the measuring axis: _MA	2-66
2.3.3	Tool number and tool name: _TNUM and _TNAME	2-67
2.3.4	Offset number: _KNUM	2-68
2.3.5	Offset number _KNUM extended for tool offset: Maximum 9 digits	2-71
2.3.6	Correcting setup and additive offset in workpiece measurement: _DLNUM	
	(from measuring cycles SW 6.3 onwards)	2-72
2.3.7	Correcting the tool of a stored tool environment: _TENV	
	(from measuring cycles SW 6.3 onwards)	2-73
2.3.8	Example of automatic tool offset in workpiece measuring cycles	2-74
2.3.9	Variable measuring velocity: _VMS	2-76
2.3.10	Offset angle position: _CORA	2-76
2.3.11	Tolerance parameters: _TZL, _TMV, _TUL, TLL, _TDIF, and _TSA	2-77
2.3.12	Measurement path: _FA	2-78
2.3.13	Probe type, probe number: _PRNUM	2-79
2.3.14	Empirical value, mean value: _EVNUM	2-80
2.3.15	Multiple measurement at the same location: _NMSP	2-81
2.3.16	Weighting factor for mean value calculation: _K	2-81
Measurir	ng Cycle Auxiliary Programs	3-83
3 1	Measuring cycle subroutines	3-84
311		
3.1.2	CYCLE116: Calculation of center point and radius of a circle	
3.2	Measuring cycle user programs	
3.2.1	CYCLE198: User program before measurement	
3.2.2	CYCLE199: User program after measurement	3-87
3.3	Package structure of the measuring cycles	3-88
Measurir	ng in JOG	4-91
4.1	Overview function, procedure, prerequisites	4-92
42	Measuring workpieces (up to measuring cycles SW 6.2)	4-95
4.2.1	Sequence of operation and functions for workpiece measurement	
	(up to measuring cvcles SW 6.2)	4-96
4.2.2	Calibrating workpiece probe (up to measuring cycles SW 6.2)	
4.2.3	Measuring an edge (up to measuring cycles SW 6.2)	4-100
4.2.4	Measuring a corner (up to measuring cycles SW 6.2)	4-101
4.2.5	Measuring a hole (up to measuring cycles SW 6.2)	4-103
4.2.6	Measuring a spigot (up to measuring cycles SW 6.2)	4-104
4.3	Workpiece measurement (from measuring cycles SW 6.3)	4-106
4.3.1	Overview of function and sequence	4-106
4.3.2	Calibrating workpiece probes (from measuring cycles SW 6.3)	4-110
4.3.3	Measuring the edge (from measuring cycles SW 6.3)	4-113
4.3.4	Measuring a corner (from measuring cycles SW 6.3)	4-119
4.3.5	Measuring a pocket, hole or spigot (from measuring cycles SW 6.3)	4-122

4.3.6	Orienting a plane (from measuring cycles SW 6.3)	4-132
4.3.7	Rejecting, repeating, end of measurement (from measuring cycles SW 6.3)	4-134
4.3.8	Information regarding cascaded measurements	4-136
4.3.9	Support of set-up in JOG - after measurement	4-137
4.4	Measuring tools	4-140
4.4.1	Overview of function and sequence	4-140
4.4.2	Calibrating the tool probe	4-140
4.4.3	Measuring milling and drilling tools	4-142
Measurir	ig Cycles for Milling and Machining Centers	5-147
5.1	General prerequisites	5-148
5.2	CYCLE971 tool: Measuring milling tools, drills	5-150
5.2.1	Measuring and compensation strategy	5-152
5.2.2	Calibrating a tool probe	5-156
5.2.3	Calibrating the tool probe automatically (from measuring cycles SW 6.3)	5-162
5.2.4	Measuring the tool	5-166
5.3	CYCLE976 calibrate workpiece probe	5-174
5.3.1	Calibrating a workpiece probe in a hole of known hole center point	5-177
5.3.2	Calibrating a workpiece probe in a hole of unknown hole center point	5-180
5.3.3	Calibration of a workpiece probe on a surface	5-185
5.3.4	Calibrating a workpiece probe in the applicate determining probe length	5-188
5.4 5.4.1 5.4.2 5.4.3	CYCLE977 workpiece: Measure hole/shaft/groove/web/rectangle parallel to axes Measuring contour elements Measurement and tool offset Measurement and ZO determination	5-191 5-195 5-203 5-207
5.5	CYCLE978 workpiece: Measuring a surface parallel to the axis	5-211
5.5.1	Measuring the surface	5-213
5.5.2	Measurement and ZO determination	5-214
5.5.3	Measuring and tool offset	5-217
5.6	CYCLE979 workpiece: Measure hole/shaft/groove/rib at an angle	5-221
5.6.1	Measuring a hole, shaft, groove, web	5-223
5.6.2	Measuring and tool offset	5-227
5.6.3	Measurement and ZO determination	5-232
5.7	CYCLE998 workpiece: Angle measurement and ZO determination	5-236
5.7.1	1-angle measurement	5-239
5.7.2	2-angle measurement	5-246
5.8	CYCLE961 workpiece: Setting up an internal and external corner	5-252
5.8.1	Function overview CYCLE961	5-252
5.8.2	Setting up a corner with definition of distances and angles	5-254
5.8.3	Setting up a corner with 4 points	5-259
5.9	CYCLE997 workpiece: Measuring spheres and ZO determination (from measuring cycles SW 6.3)	5-264



5.9.1	Function overview CYCLE997	5-264
5.9.2	Measurement and ZO determination	5-268
5.9.3	Programming example CYCLE997	5-273
5.9.4	CYCLE119: Arithmetic cycle for calculating position in space	5-274
Measuri	ng Cycles for Turning Machines	6-277
6.1	General prerequisites	6-278
6.2	CYCLE982, CYCLE972 Tool: Measuring turning tools	6-280
6.2.1	Function overview	6-280
6.2.2	Calibrating tool probes (machine-related)	6-282
6.2.3	Determining dimensions of calibration tool	6-285
6.2.4	Measuring a turning tool (machine-related)	6-286
6.3	CYCLE982 tool: Measure turning and milling tools	6-292
6.3.1	Function overview	6-292
6.3.2	Calibrating a tool probe	6-298
6.3.3	Measuring the tool	6-301
6.3.4	Automatic tool measurement	6-311
6.3.5	Incremental calibration	6-318
6.3.6	Incremental measurement	6-321
6.3.7	Milling tool: Suppression of start angle positioning _STA1	6-327
6.3.8	Measuring drills – special applications (from measuring cycles SW 6.3)	6-328
6.4	CYCLE973 calibrate workpiece probe	6-329
6.4.1	Function overview	6-329
6.4.2	Calibrating in the reference groove	6-331
6.4.3	Calibration at a surface	6-334
6.5	CYCLE974 workpiece: 1-point measurement	6-337
6.5.1	Function overview	6-337
6.5.2	1-point measurement and ZO determination	6-340
6.5.3	1-point measurement and tool offset	6-343
6.5.4	1-point measurement with reversal and tool offset	6-348
6.6	CYCLE994 workpiece: 2-point measurement	6-351
6.7	Complex example for workpiece measurement	6-359
Miscella	neous Functions	7-361
7.1	Logging the measurement results	7-362
7.1.1	Overview of the logging cycles	7-362
7.1.2	Log file	7-362
7.1.3	Handling the log cycles	7-363
7.1.4	Variables when logging	7-366
7.1.5	Selecting log content	7-367
7.1.6	Log format	7-369
7.1.7	Log header	7-370
7.1.8	Example: Creating a measurement result log	
7.2	Measuring cycle support in the program editor (up to measuring cycles	7_37/

	7.2.1 7.2.2 7.2.3 7.2.4	Files of the measuring cycle support         Loading the measuring cycle support         Assigning calls and measuring cycles         Description of the parameterizing cycles	7-375 7-375 7-376 7-377
	7.3 7.3.1 7.3.2	Measuring cycle support in the program editor (from measuring cycles Menus, explanation of the cycles Pre-setting of the measuring cycle support	s SW 6.2) 7-387 7-387 7-393
	7.4	Measuring result screens	7-396
F	Part 2 De	escription of Functions	
ł	lardwar	e, Software and Installation	8-399
	8.1 8.1.1 8.1.2 8.1.3 8.2 8.2.1	Hardware prerequisites General hardware prerequisites Measuring probe connection Measuring in JOG Software prerequisites Software versions NC and MMC/HMI	8-400 8-400 8-400 8-402 8-403 8-403
	8.2.2	Measuring in JOG	
	8.3	Function test	
Data Description 9		9-407	
	9.1 9 1 1	Machine data for executing measuring cycles	
	9.1.2 9.2 9.2.1 9.2.2 9.2.3 9.2.3 9.2.4 9.2.5 0.2.6	Additional machine data Cycle data Data blocks for measuring cycles: GUD5.DEF and GUD6.DEF Data adjustment to a specific machine Central values Central bits Central strings Channel gringted values	
	9.1.2 9.2 9.2.1 9.2.2 9.2.3 9.2.4 9.2.5 9.2.6 9.2.7	Additional machine data Cycle data Data blocks for measuring cycles: GUD5.DEF and GUD6.DEF Data adjustment to a specific machine Central values Central bits Central strings Channel-oriented values Channel-oriented bits	9-408 9-412 9-413 9-413 9-413 9-413 9-418 9-420 9-420 9-420 9-420 9-429 9-430 9-430
	9.1.2 9.2 9.2.1 9.2.2 9.2.3 9.2.4 9.2.5 9.2.6 9.2.7 9.3 9.3.1 9.3.2 9.3.3 9.3.4	Additional machine data Cycle data Data blocks for measuring cycles: GUD5.DEF and GUD6.DEF Data adjustment to a specific machine Central values Central bits Central bits Central strings Channel-oriented values Channel-oriented bits Data for measuring in JOG Machine data for ensuring ability to function Modifying the GUD7 data block Settings in data block GUD6 Loading files for measuring in JOG	9-408 9-412 9-413 9-413 9-413 9-413 9-413 9-418 9-420 9-420 9-420 9-420 9-429 9-429 9-430 9-430 9-433 9-440 9-440 9-442 9-446 9-447
ę	9.1.2 9.2 9.2.1 9.2.2 9.2.3 9.2.4 9.2.5 9.2.6 9.2.7 9.3 9.3.1 9.3.2 9.3.3 9.3.4 Start-Up	Additional machine data Cycle data Data blocks for measuring cycles: GUD5.DEF and GUD6.DEF Data adjustment to a specific machine Central values Central values Central bits Central strings Channel-oriented values Channel-oriented bits Data for measuring in JOG Machine data for ensuring ability to function Modifying the GUD7 data block Settings in data block GUD6 Loading files for measuring in JOG	9-408 9-412 9-413 9-413 9-413 9-413 9-413 9-418 9-420 9-420 9-420 9-420 9-420 9-420 9-420 9-420 9-420 9-420 9-430 9-440 9-440 9-440 9-442 9-446 9-447 <b>10-449</b>
Ş	9.1.2 9.2 9.2.1 9.2.2 9.2.3 9.2.4 9.2.5 9.2.6 9.2.7 9.3 9.3.1 9.3.2 9.3.3 9.3.4 <b>Start-Up</b> 10.1	Additional machine data Cycle data Data blocks for measuring cycles: GUD5.DEF and GUD6.DEF Data adjustment to a specific machine Central values Central bits Central strings Channel-oriented values Channel-oriented bits Data for measuring in JOG Machine data for ensuring ability to function Modifying the GUD7 data block Settings in data block GUD6 Loading files for measuring in JOG	9-408 9-412 9-413 9-413 9-413 9-413 9-413 9-413 9-420 9-420 9-420 9-429 9-429 9-430 9-430 9-433 9-440 9-440 9-440 9-442 9-446 9-447 <b>10-449</b>
Ę	9.1.2 9.2 9.2.1 9.2.2 9.2.3 9.2.4 9.2.5 9.2.6 9.2.7 9.3 9.3.1 9.3.2 9.3.3 9.3.4 Start-Up 10.1 10.2	Additional machine data Cycle data Data blocks for measuring cycles: GUD5.DEF and GUD6.DEF Data adjustment to a specific machine Central values Central values Central bits Central strings Channel-oriented values Channel-oriented bits Data for measuring in JOG Machine data for ensuring ability to function Modifying the GUD7 data block Settings in data block GUD6 Loading files for measuring in JOG Commissioning measuring cycles for the first time Upgrading measuring cycles	9-408 9-412 9-413 9-413 9-413 9-413 9-413 9-418 9-420 9-420 9-420 9-420 9-420 9-420 9-420 9-420 9-420 9-420 9-420 9-430 9-440 9-440 9-440 9-442 9-446 9-447 <b>10-449</b> 
ę	9.1.2 9.2 9.2.1 9.2.2 9.2.3 9.2.4 9.2.5 9.2.6 9.2.7 9.3 9.3.1 9.3.2 9.3.3 9.3.4 Start-Up 10.1 10.2 10.3	Additional machine data Cycle data Data blocks for measuring cycles: GUD5.DEF and GUD6.DEF Data adjustment to a specific machine Central values Central values Central bits Central strings Channel-oriented values Channel-oriented bits Data for measuring in JOG Machine data for ensuring ability to function Modifying the GUD7 data block Settings in data block GUD6 Loading files for measuring in JOG Commissioning measuring cycles for the first time Upgrading measuring cycles Sequence for probe installation	9-408 9-412 9-413 9-413 9-413 9-413 9-413 9-413 9-420 9-420 9-420 9-420 9-420 9-420 9-420 9-420 9-420 9-420 9-420 9-430 9-440 9-440 9-440 9-440 9-442 9-446 9-447 <b>10-449</b> 

 $(\mathbf{0})$ 

Alarms		11-461
Adaptatio	on of the Measuring Cycles to Previous Software Versions	12-469
12.1	Adaptation of the measuring cycle version to NC software versions	12-470
12.2	CYCLE103: Cycle for parameter definition (only up to measuring cycles SW 4.5)	12-471
12.3	Measuring cycle sub-routines	12-472
12.4	Start-up of the measuring cycle user interface for MMC 102 (only up to measuring cycles SW 4.4)	12-474
Appendix	(	A-475

А	Overview of measuring cycle parameters	A-476
В	Abbreviations	A-514
С	Terms	A-516
D	References	A-522
E	Commands, identifiers	A-534
F	Index	I-537

## Introduction

1.1	Fundamentals	1-18
1.2	General prerequisites	1-19
1.3	Behavior on block search, dry run, program testing, simulation	1-20
1.4	Reference points on the machine and workpiece	1-23
1.5	Definition of the planes, tool types	1-25
1.6	Probes that can be used	1-27
1.7 1.7.1 1.7.2 1.7.3 1.7.4	Probe, gauging block, calibration tool Measuring workpieces on milling machines, machining centers Measuring tools on milling machines, machining centers Measuring workpieces at the lathe Measuring tools at lathes	1-29 1-29 1-30 1-31 1-33
1.8	Measuring principle	1-34
1.9	Measuring strategy for measuring workpieces with tool offset	1-38
1.10	Parameters for checking the measurement result and offset	1-41
1.11	Effect of empirical value, mean value, and tolerance parameters	1-46
1.12 1.12.1 1.12.2 1.12.3 1.12.4 1.12.5 1.12.6 1.12.7 1.12.8	Overview of measuring cycle functions for milling technology Tool measurement on milling machines, machining centers Calibrating workpiece probes Workpiece measurement at one point Measuring the workpiece parallel to the axis Measuring a workpiece at an angle Measuring a surface at an angle Measuring spheres (from measuring cycles SW 6.3 onwards) Workpiece measurement: Setting-up a corner	1-47 1-47 1-48 1-49 1-50 1-52 1-53 1-54 1-55
1.13 1.13.1 1.13.2 1.13.3 1.13.4	Overview of measuring cycle functions for turning technology Measuring tools at lathes Calibrating workpiece probes Measuring tools at lathes: 1-point measurement Measuring workpieces at lathes: 2-point measurement	1-56 1-56 1-58 1-59 1-60



## 1.1 Fundamentals

Measuring cycles are general subroutines designed to solve specific measurement tasks. They can be adapted to specific problems via parameter settings.

In measurement applications, a general distinction is made between **tool measurement** and **workpiece measurement**.

#### Workpiece measurement

In workpiece measurement, a measuring probe is moved up to the clamped workpiece in the same way as a tool and the measured values are acquired. The flexibility of measuring cycles makes it possible to perform nearly all measurements required on a milling or turning machine. An automatic tool offset or ZO correction can be applied to the workpiece measurement result.

#### **Tool measurements**

In tool measurement, the selected tool is moved up to the probe and the measured values are acquired. The probe is either in a fixed in position or is swung into the working area with a mechanism. The tool geometry measured is entered in the appropriate tool offset data set.







## 1.2 General prerequisites

Certain preconditions must be met before measuring cycles can be used.

These are detailed in Part 2 Description of Functions (Chapter 8 ff.).

The following checklist is useful for checking which preconditions have been met:

#### Machine

- All machine axes are designed in accordance with DIN 66217.
- Machine data have been adapted.

## Existence of cycles, data blocks

The following have been loaded into the control and activated:

 The data blocks into the "Definitions" directory: GUD5.DEF GUD5.DEF and GUD7.DEF (for measuring in JOG)

• The measuring cycles into the standard cycle directory.

#### Starting position

- The reference points have been approached.
- The starting position can be reached by linear interpolation without collision.

#### Display functions of the measuring cycles

An MMC/PCU is required for showing the measuring result displays and measuring cycle support.

# Please observe the following when programming:

- Tool radius compensation is deselected before it is called (G40).
- All parameters for the cycle call have been defined beforehand.
- The cycle is called no later than at the 5th program level.
- The system of units allows measuring in the programmed unit system that deviates from the basic system with switchable technology data. This means in a metric basic system with active G70, G700 and in an inch basic system with active G71, G710.

## 1.3 Behavior on block search, dry run, program testing, simulation

## Function

## Up to meas. cycles SW 6.2

The measuring cycles are skipped during execution if one of the following execution types is active:

- "Block search" (\$P\_SEARCH=1)
- "Dry run" (\$P\_DRYRUN=1)
- "Simulation" (\$P\_SIM=1)
- "Program test" (\$P\_ISTEST=1) (from SW 6.2)

## Function

## From meas. cycles SW 6.3)

The measuring cycles are skipped during execution if one of the following execution modes is active:

- "Dry run" (\$P\_DRYRUN=1)
- "Program test" (\$P\_ISTEST=1)
- "Block search" (\$P\_SEARCH=1), only if \$A\_PROTO=0.

## Simulation

- on HMI Advanced (\$P\_SIM=1) The measuring cycle programs are executed on selection of "Simulation" on the HMI.
- on Jobshop products (\$P\_SEARCH=1 and \$A\_PROTO=1)



The measurements are simulated. No selected tool or zero offset is applied. Active functions such as "measuring result display", "travel with collision monitoring", "measuring cycle logging" are not executed.

#### Suppression of simulation execution

Execution of the measuring cycles in simulation can be suppressed in simulation by resetting the variable **\_MC\_SIMSIM=0** in data block GUD6. The measuring cycles are then skipped.

#### Specifying differences for simulation

The variable **\_MC\_SIMDIFF** of data type REAL permits specification of simulated measurement deviations at measuring points. The value is a dimension in the basic system of the control.

Excessive values of \_MC\_SIMDIFF with corresponding value assignment of the defining parameters cause cycle alarms to be output.



#### Note:

It is not ensured that the correction value contains the correct sign of \_MC\_SIMDIFF. This depends on the measurement or calibration task and the direction of measurement. The sign is influenced in such a way that the overall result makes sense (e.g. "Measure hole"). Changing the sign always causes the sign of the result to change.

#### Demo programs for simulation on HMI Advanced

The measuring cycle software supplied includes programs for demonstrating how the measuring cycles work when simulating on HMI Advanced. The programs are divided into turning technology and milling technology. When a program is loaded, the "workpieces" are created whose directory contains all data and settings required for simulation.

## Examples of demo programs for simulation on HMI Advanced

Example 1:

Measuring a hole

(TESIM\_977BO with CYCLE977, milling technology)



#### Example 2:

Calibrating a tool probe

(TESIM\_982MKS with CYCLE982, turning technology)



#### 1.4 Reference points on the machine and workpiece

## Function

Depending on the measuring task, measured values may be required in the machine coordinate system or in the workpiece coordinate system. e.g.: It may be easier to ascertain the tool length in the machine coordinate system. Workpiece dimensions are measured in the workpiece coordinate system.

The position of tool reference point F in the machine coordinate system is displayed with machine zero M as the machine actual value.

The position of the tool tip (active tool) in the workpiece coordinate system is displayed with workpiece zero W as the workpiece actual value. If a workpiece probe is active, the position usually refers to the center point of the probe ball.

The zero offset (ZO) characterizes the position of the workpiece on the machine. The ZO is the position of workpiece zero W in the machine coordinate system.

In addition to the pure offset, a ZO might also include rotation, mirroring, and a dimension factor. Together these are termed a frame.

SINUMERIK controls use numerous frames: various basic frames, system frames, settable frames (e.g. G54), programmable frames: They interact in a frame chain to produce the overall frame and the workpiece coordinate system.

Measuring cycles do not support any frames with an active scale factor. Rotation or mirroring is not supported in some cycles and measuring variants.

The machine and workpiece coordinate system can be set and programmed separately in the "inch" or "metric" measuring system.





M = Machine zero

W = Workpiece zero

F = Tool reference point

Reference points





If kinematic transformation is active, the control makes a distinction between the **basic coordinate system** and the **machine coordinate system**. If kinematic transformation is deactivated, this distinction is made. All descriptions provided below assume that kinematic transformation is deactivated and therefore refer to the

transformation is deactivated and therefore refer to the machine coordinate system.

## 1.5 Definition of the planes, tool types

Tool radius compensation planes: G17, G18, or G19 can be selected. Depending on the tool type, the tool lengths are assigned to the axes as follows:

- Milling cutter, workpiece probe for milling: 1xy or workpiece probe for milling: 710
- Drill: 2xy
- Turning tool, workpiece probe for turning: 5xy



## Milling

#### G17 plane

Tool type	1xy / 2xy / 710
Length 1	active in Z (applicate)
Length 2	active in Y (ordinate)
Length 3	active in X (abscissa)

## G18 plane

1xy / 2xy / 710
active in Y (applicate)
active in X (ordinate)
active in Z (abscissa)

## G19 plane

Tool type	1xy / 2xy / 710
Length 1	active in X (applicate)
Length 2	active in Z (ordinate)
Length 3	active in Y (abscissa)

Lengths 2 and 3 are used in special cases, for example, if an angle head is attached.





Turning machines generally only use axes Z and X

# and therefore: **G18 plane**

- Tool type	571
roor type	Злу
	(turning tool, workpiece probe)
Length 1	active in X (ordinate)
Length 2	active in Z (abscissa)

G17 and G19 are used for milling on a turning machine. If there is no machine axis Y, milling can be implemented with kinematic transformations. TRANSMIT, TRACYL.

In principle, measuring cycles support kinematic transformations. This is stated more clearly in the individual cycles, measuring variants.

## Notes:

If a drill or milling cutter is measured on a turning machine, in most cases, setting data SD 42950: TOOL\_LENGTH\_TYPE=2 is set. These tools are then length-compensated like a turning tool.

SINUMERIK controls have other machine and setting data that can influence calculation of a tool.







References: /FB1/, /FB2/,

Description of Functions – Basic Machine Description of Functions – Extended Functions

/FB3/, Description of Functions – Special Functions

Workpiece probe

## 1.6 Probes that can be used

## Function

To measure tool and workpiece dimensions, a touchtrigger probe is required that provides a signal change (edge) when deflected.

The probe must operate virtually bounce-free.

Different types of probe are offered by different manufacturers. Probes are distinguished according to the number of measuring directions.

- Monodirectional (mono probe)
- Bidirectional
- Multidirectional (multi probe)

The probe type is defined by a parameter (\_PRNUM) in measuring cycles (see Section 2.10).

## Monodirectional probe

This type of probe can only be used if the spindle can be positioned with NC function SPOS and the switching signal of the probe can be transmitted through 360° to the receiving station (at the machine column).

The probe must be mechanically aligned in the spindle to permit measurements in the following directions at the 0 degree position of the spindle.

X-Y plane G17:	Positive X direction
Z-X plane G18:	Positive Z direction
Y-Z plane G19:	Positive Y direction

The measurement takes longer with a mono probe since the spindle must be positioned in the cycle several times by means of SPOS.





Probe assignment:			
Probe type	Turni	ng machines	Milling and machining centers
	Tool measurements	Workpiece measurement	Workpiece measurement
Multidirectional	Х	Х	Х
Bidirectional	_	Х	Х
Monodirectional	_	-	Х

In workpiece measurement, a bidirectional probe is treated like a mono probe.





If a workpiece probe is used, both the direction of deflection and transmission of switching signal to the machine column (radio, infrared light or cable) must be taken into account. In some versions, transmission is only possible in particular spindle positions or in particular ranges.

This may further limit the use of the probe. In any case, please follow the advice of the probe or machine manufacturer.

## 1.7 Probe, gauging block, calibration tool

## 1.7.1 Measuring workpieces on milling machines, machining centers



## Workpiece probe

On milling machines and machining centers, the probe is classified as tool type 1xy and must therefore be entered as such in the tool memory. As from SW 4, tool type 710 (3D probe) can also be used.

Entry in tool memory

Tool type (DP1):	710 or 1xy
Length 1 – Geometry (DP3):	L1
Radius (DP6):	r
Length 1 – Tool base	Only if necessary
dimension (DP21):	

The wear and other tool parameters must be assigned the value 0.

In \_CBIT[14] you can set whether length L1 refers to the ball center point or the ball circumference.



\_CBIT[14] see Subsection 9.2.4 (central bits)

## Calibration

A probe must be calibrated before it can be used. Calibration involves determining the triggering points (switching points), skew (positional deviation), and active ball radius of the workpiece probe and then entering them in special data fields \_WP[] in data block GUD6.DEF. The default setting has data fields for 3 probes. Up to 99 are possible.

Calibration can be performed on holes of a known size or workpiece surfaces with a sufficient form precision and low surface roughness.

Use of special gauging blocks is not supported on milling and machine centers.

Use the same measuring velocity for calibrating and measuring.

A special cycle is available for calibration.





## Tool probe

The tool probes have dedicated data fields \_TP[] and \_TPW[] in data block GUD6.DEF.

The triggering points (switching points), upper disk diameter and edge length are entered here.

Approximate values must be entered here before calibration – if cycles are used in automatic mode. The cycle will then recognize the position of the probe.

In the default setting there are data fields for 3 probes. Up to 99 are possible.

## Calibration, calibrating tool

A probe must be calibrated before it can be used. Calibration involves precisely determining the triggering points (switching points) of the tool probe and entering them in special data fields. Calibration is performed with a calibration tool. The precise dimensions of the tool are known.

Use the same measuring velocity for calibrating and measuring.

A special cycle is available for calibration.

Entry of calibration tool in tool memory:

Tool type (DP1):	1xy
Length 1 – Geometry (DP3):	L1
Radius (DP6):	r
Length 1 – Tool base	Only if necessary
dimension (DP21):	

The wear and other tool parameters must be assigned the value 0.





## 1.7.3 Measuring workpieces at the lathe



#### Workpiece probe

On turning machines, the workpiece probes are treated as tool type 5xy with permissible cutting edge positions (SL) 5 to 8 and must be entered in the tool memory accordingly.



Lengths specified for turning tools always refer to the tool tip, except in the case of workpiece probes on turning machines where they refer to the probe center.

Probes are classified according to their position:



## Workpiece probe SL 7

entry in tool memory	
Tool type (DP1):	5xy
Tool edge position (DP2):	7
Length 1 – Geometry:	L1
Length 2 – Geometry:	L2
Radius (DP6):	r
Length 1 – Tool base dimension (DP21):	Only if necessary
Length 2 – T. base dim. (DP22):	Only if necessary

The wear and other tool parameters must be assigned the value 0.





#### Workpiece probe SL 8

Entry in tool memory	
Tool type (DP1):	5xy
Tool edge position (DP2):	8
Length 1 – Geometry:	L1
Length 2 – Geometry:	L2
Radius (DP6):	r
Length 1 – Tool base dimension (DP21):	Only if necessary
Length 2 – T. base dim. (DP22):	Only if necessary

The wear and other tool parameters must be assigned the value 0.

Workpiece probe for turning machine Example: tool edge position SL=8





## Workpiece probe SL 5 or SL 6

Entry in tool memory	
Tool type (DP1):	5xy
Tool edge position (DP2):	5 or 6
Length 1 – Geometry:	L1
Length 2 – Geometry:	L2
Radius (DP6):	r
Length 1 – Tool base dimension (DP21):	Only if necessary
Length 2 – T. base dim. (DP22):	Only if necessary

The wear and other tool parameters must be assigned the value 0.



#### Calibration, gauging block

A probe must be calibrated before it can be used. During calibration the triggering points (switching points), skew (positional deviation), and precise ball radius of the workpiece probe are determined and then entered in special data fields \_WP[] in data block GUD6.DEF.

In the default setting there are data fields for 3 probes. Up to 99 are possible.

Calibration of the workpiece probe on turning machines is usually performed with gauging blocks (reference grooves). The precise dimensions of the reference groove are known and entered in the relevant data fields \_KB[] in data block GUD6.DEF.

In the default setting there are data fields for 3 gauging blocks. The gauging block is selected in the program with variable \_CALNUM.

It is also possible to calibrate on a known surface.

Use the same measuring velocity for calibrating and measuring.

A cycle with different measuring versions is provided for calibration.





## 1.7.4 Measuring tools at lathes



#### Tool probe

The tool probes have dedicated data fields \_TP[] and \_TPW[] in data block GUD6.DEF. The triggering points (switching points) are entered here. Approximate values must be entered here before calibration – if cycles are used in automatic mode. The cycle will then recognize the position of the probe.

The default setting has data fields for 3 probes. Up to 99 are possible.

In addition to turning tools, drills and mills can also be measured.

#### Calibration, calibrating tool

A probe must be calibrated before it can be used. Calibration involves precisely determining the triggering points (switching points) of the tool probe and entering them in special data fields.

Calibration is performed with a calibration tool. The precise dimensions of the tool are known.

Use the same measuring velocity for calibrating and measuring.

A special measuring variant in a cycle is available for calibration.

On turning machines, the calibration tool is treated like a turning tool with cutting edge position 3. The lengths refer to the ball circumference, not to the ball center.

#### Entry in tool memory

Tool type (DP1):	5xy
Tool edge position (DP2):	3
Length 1 – Geometry:	L1
Length 2 – Geometry:	L2
Radius (DP6):	r
Length 1 – Tool base dimension (DP21):	Only if necessary
Length 2 – T. base dim. (DP22):	Only if necessary

The wear and other tool parameters must be assigned the value 0.







## 1.8 Measuring principle



## Function

## "On-the-fly" measurement

The principle of "on-the-fly" measurement is implemented in the SINUMERIK control. The probe signal is processed directly on the NC so that the delay when acquiring measured values is minimal. This permits a higher measuring speed for the prescribed measuring precision and time needed for measuring is reduced.

## **Connecting probes**

Two inputs for connecting touch trigger probes are provided on the I/O device interface of the SINUMERIK control systems.

#### Measuring procedure

The procedure is described with **workpiece measurement**. The procedure is the same for tool measurement. In this case, however, the tool is moved and the probe is fixed.

Depending on its design, the actual movements of a machine may be different anyway. Workpiece measurement is described as follows: The workpiece is stationary and the probe moves.

The **starting position** for the measuring procedure is a position **\_FA** in front of the specified **set position** (expected contour).

The starting position is calculated in the cycle based on parameter entries and probe data. The starting position is approached either with rapid traverse G0 or with positioning velocity G1; then from the starting position with **measuring velocity**.

The switching signal is expected along path  $2 \cdot FA$  as from the starting position. Otherwise, an alarm will be triggered or the measurement repeated. The resulting **maximum measuring position** is in the

measuring block of the cycle.

At the instant the switching signal is output by the probe, the current **actual position** is stored internally "on-the-fly" as the actual value, the measuring axis is stopped and then the "**Delete distance-to-go**" function is executed.





The distance-to-go is the path not yet covered in the measuring block. After deletion, the next block in the cycle can be processed. The measuring axis travels back to the starting position. Any measurement repetitions selected are restarted from this point.

## Measurement path \_FA

Measurement path \_FA defines the distance between the starting position and the expected switching position (setpoint) of the probe. (see Chapter 2).

## Measuring velocity

The measuring velocity is dependent on the measurement path \_FA and its default setting is 150 mm/min if \_FA=1; if FA>1: 300 mm/min. Cycles parameter \_VMS is then =0.

Other measuring velocities can be set by the user to a value of >0 in \_VMS (see Chapter 2).

The **maximum permissible measuring velocity** is derived from:

- the deceleration behavior of the axis
- the permissible deflection of the probe
- the signal processing delay

# Deceleration distance, deflection of probe Caution

Safe deceleration of the measuring axis to standstill within the permissible deflection path of the probe must always be ensured.

Otherwise damage will occur!

A delay "t" typical of the control is taken into account in signal processing (IPO cycle) for the time between detection of the switching signal and output of the deceleration command to the measuring axis MD 10050: SYSCLOCK\_CYCLE\_TIME and MD 10070: IPO\_SYSCLOCK\_TIME\_RATIO). This results in the deceleration path component.

The following error of the measuring axis is reduced. The following error is velocity dependent and at the same time dependent on the control factor of the measuring axis (servo gain of the associated machine axis: servo gain factor).

#### Introduction **1.8 Measuring principle**

The deceleration rate of the axis must also be taken into account.

Together they produce an axis-specific velocitydependent deceleration distance.

The servo gain factor is MD 32200: POSCTRL\_GAIN.

Axis acceleration / deceleration rate a is stored in MD 32300: MAX\_AX\_ACCEL. It may have a lesser effect due to other influences. Always used the lowest values of the axes involved in the measurement.

## Calculation of the deceleration path

The deceleration path to be considered is calculated as follows:

$$s_{b} = 1000 \cdot v \cdot t + \frac{1000 \cdot v^{2}}{2a} + \Delta s$$

$$\Delta s_{1} \qquad \Delta s_{2}$$

$\mathbf{S}_{\mathrm{b}}$	Deceleration path	in mm
v	Measuring velocity	in m/s
t	Delay signal	in s
а	Deceleration	in m/s <sup>2</sup>
$\Delta s$ following error		in mm
$\Delta s = v / Kv$		v here in m/min
Kv servo gain factor		in (m/min)/mm

## Example of calculation:

- v = 6 m/min = 0.1 m/ s measuring velocity
- a = 1 m/s<sup>2</sup> deceleration rate
- t = 16 ms signal delay
- Kv = 1 in (m/min)/mm

The deflection of the probe = deceleration distance to zero speed of axis is:  $\underline{s}_{b} = 12.6 \text{ mm}$ .

The deceleration distance components are: $\Delta s = 6/1 = 6 \text{ mm}$ Following error $\Delta s2 = 1000 \cdot 0.01/2 + 6 = 11 \text{ mm}$ Axis spec.<br/>percentage $\Delta s1 = 1000 \cdot 0.1 \cdot 0.016 = 1.6 \text{ mm}$ Percentage due to<br/>signal delay


### Measuring accuracy

A delay occurs between detection of the switching signal from the probe and transfer of the measured value to the control. This is caused by signal transmission from the probe and the hardware of the control. In this time a path is traversed that falsifies the measured value. This influence can be minimized by reducing the measuring speed. The rotation when measuring a mill on a rotating spindle has an additional influence. This can be compensated for by compensation tables. (see Subsection 5.2.1)

The measurement accuracy that can be obtained is dependent on the following factors:

- Repeat accuracy of the machine
- Repeatability of the probe
- Resolution of the measuring system

A test program for determining the overall repeatability of a machine is described in Section 10.4.

### 1.9 Measuring strategy for measuring workpieces with tool offset

The actual workpiece dimensions must be measured exactly and compared with the setpoint values to be able to determine and compensate the actual dimensional deviations on the workpiece. An offset value can then be ascertained for the tool used for machining.



### Function

When taking measurements on the machine, the actual dimensions are derived from the path measuring systems of the position-controlled feed axes. For each dimensional deviation determined from the set and actual workpiece dimensions there are many causes which essentially fall into 3 categories:

 Dimensional deviations with causes that are n o t subject to a particular trend, e.g. positioning scatter of the feedforward axes or differences in measurement between the internal measurement (measuring probe) and the external measuring device (micrometer, measuring machine, etc.).

In this case, it is possible to apply **empirical values**, which are stored in separate memories. The set/actual difference determined is automatically compensated by the empirical value.

- Dimensional deviations with causes that a r e subject to a particular trend, e.g. tool wear or thermal expansion of the leadscrew.
- Accidental dimensional deviations, e.g. due to temperature fluctuations, coolant or slightly soiled measuring points.

Assuming the ideal case, only those dimensional deviations that are subject to a trend can be taken into account for compensation value calculation. Since, however, it is hardly ever known to what extent and in which direction accidental dimensional deviations influence the measurement result, a strategy (sliding averaging) is needed that derives a compensation value from the actual/set difference measured.

10.04

### Mean value calculation

Mean value calculation in conjunction with measurement weighting has proven a suitable method.

The formula of the mean value generation chosen is:

$$Mi_{new} = Mi_{old} - \frac{Mi_{old} - D_i}{k}$$

Mv<sub>new</sub> Mean value new = amount of compensation

Mv<sub>old</sub> Mean value prior to last measurement

k Weighting factor for average value calculation

D<sub>i</sub> Actual/set difference measured (minus empirical value, if any)

The mean value calculation takes account of the trend of the dimensional deviations of a machining series. The **weighting factor k** from which the mean value is derived is selectable.

A new measurement result affected by accidental dimensional deviations only influences the new tool offset to some extent, depending on the weighting factor.

# Computational characteristic of the mean value with different weightings k

- The greater the value of k, the slower the formula will respond when major deviations occur in computation or counter compensation. At the same time, however, accidental scatter will be reduced as k increases.
- The lower the value of k, the faster the formula will react when major deviations occur in computation or counter compensation.
   However, the effect of accidental variations will be that much greater.
- The mean value Mi is calculated starting at 0 over the number of workpieces i, until the calculated mean value exceeds the range of "zero compensation" (cycle parameter \_TZL, see Chapter 2). From this limit on, the calculated mean value is applied as an offset.
- Once the mean value has been used for the offset, it is deleted from the memory. The next measurement then starts again with Mi<sub>old</sub> = 0.

### Example of mean value calculation and offset

	Lower	limit = 40 µm	(TZL= 0.04)
	Di Mean value Mean value		
		k=3	k=2
	[µm]	[µm]	[µm]
1st measurement	30	10	15
2nd measurement	50	23.3	32.5
3rd measurement	60	35.5	46.2 ③
4th measurement	20	30.3	10
5th measurement	40	32.6	25
6th measurement	50	38.4	37.5
7th measurement	50	<b>42.3</b> ①	43.75 ④
8th measurement	30	10	15
9th measurement	70	30	42.5 S
10th measurement	70	43.3 ©	35

In the measurements with marked fields, tool compensation is performed with the mean value (calculated mean value >\_TZL):

If k=3 in the 7th and 10th measurement (  $\stackrel{(1)}{\_}$  and  $\stackrel{(2)}{\_}$  ),

If k=2 in the 3rd, 7th, and 9th meas.((3), (4) and (5)).





### 1.10 Parameters for checking the measurement result and offset



### Explanation

10.04

For constant deviations **not** subject to a **trend**, the dimensional deviation measured can be compensated by an empirical value in certain measuring variants. For other compensations resulting from dimensional deviations, symmetrical tolerance bands are assigned to the set dimension which result in different responses.

### Empirical value/mean value \_EVNUM

The empirical values are used to suppress dimensional deviations that are **not subject to a trend**.

If you do not want to apply any empirical values, \_EVNUM=0 must be set.

The empirical values themselves are stored in data block (GUD5) in field **\_EV[] empirical value**.

\_EVNUM specifies the number of the empirical value memory. The actual/set difference determined by the measuring cycle is corrected by this value **before** any further correction measures are taken.

This is the case

- for workpiece measurement with automatic tool offset
- for single-point measurement with automatic zero offset and
- tool measurement

**Mean value** \_EVNUM is active only for workpiece measurement with automatic tool offset.

When calculating the mean value in a series of machining operations, the mean value determined by the measurement at the same measurement location on the previous workpiece can be taken into account (\_CHBIT[4]=1).

The mean values are stored in data block (GUD5) in field \_MV[] Mean values. \_EVNUM specifies the number of the mean value memory in this field.

### Safe area \_TSA

The safe area is effective for almost all measuring variants and does not affect the offset value; it is used for diagnostics. If this limit is reached then the following can be

assumed:

- a probe defect, or
- an incorrect setpoint position, or
- an illegal deviation from the setpoint position can be assumed.



AUTOMATIC operation is interrupted and the program cannot continue. An alarm text appears to warn the user.

### Dimensional difference control \_TDIF

\_TDIF is active only for workpiece measurement with automatic tool offset and for tool measurement. This limit has no effect on generation of the compensation value either. When it is reached, the tool is probably worn and needs to be replaced.



An alarm text is displayed to warn the operator and the program can be continued by means of an NC start.

This tolerance limit is generally used by the PLC for tool management purposes (twin tools, wear monitoring).

### Tolerance of the workpiece \_TLL, \_TUL

Both parameters are active only for tool measurement with automatic tool offset. When measuring a dimensional deviation ranging between "2/3 tolerance of workpiece" and "Dimensional difference control", this is regarded 100% as tool compensation. The previous average value is erased.

This enables a fast response to major dimensional deviations.



When the tolerance limit of the workpiece is exceeded, this is indicated to the user depending on the tolerance position "oversize" or "undersize".

### 2/3 workpiece tolerance \_TMV

\_TMV is active only for workpiece measurement with automatic tool offset.

Within the range of "Lower limit" and "2/3 workpiece tolerance" the mean value is calculated according to the formula described in Section "Measuring strategy".



Mv<sub>new</sub> is compared with the zero compensation range:

- If Mv<sub>new</sub> is greater than this range, compensation is corrected by Mv<sub>new</sub> and the associated mean value memory is cleared.
- If Mv<sub>new</sub> is less than this range, no compensation is carried out to prevent excessively abrupt compensations.

### Weighting factor for mean value calculation \_K

\_K is active only workpiece measurement with automatic tool offset. The weighting factor can be used to give a different weighting for each measurement.

A new measurement result thus has only a limited effect on the new tool offset as a function of \_K.

### Lower limit (zero compensation area) \_TZL

\_TZL active for

- · Workpiece measurement with automatic tool offset
- Tool measurement and calibration for milling tools and tool probes

This tolerance range corresponds to the amount of maximum accidental dimensional deviations. It has to be determined for each machine.

No tool compensation is made within these limits.

In workpiece measurement with automatic tool offset, however, the mean value of this measuring point is updated and re-stored with the measured actual/set difference, possibly compensated by an empirical value. The tolerance bands (range of permissible dimensional tolerance) and the responses derived from them are as follows:

# • For workpiece measurement with automatic tool offset



In measuring cycles, the workpiece setpoint dimension

is placed in the middle of the permitted

± tolerance range for reasons of symmetry.

See Subsection 2.3.11 "Tolerance parameters..."

### For tool measurement





### For workpiece measurement with zero offset



### • For workpiece probe calibration



### • For tool probe calibration



### 1.11 Effect of empirical value, mean value, and tolerance parameters

The following flowchart shows the effect of empirical value, mean value, and tolerance parameters on workpiece measurement with automatic tool offset.



1.12 Overview of measuring cycle functions for milling technology

### 1.12.1 Tool measurement on milling machines, machining centers



### Function

Measuring cycle CYCLE971 can be used to calibrate a tool probe and measure the tool length and/or radius for milling tools.



### Calibrating a tool probe

### Result:

Probe switching point with reference to machine zero

From measuring cycles SW 6.3 onwards reference to the workpiece zero is possible.





### Measuring the tool

Result: Tool length Tool radius







### 1.12.2 Calibrating workpiece probes



### Function

With cycle CYCLE976 a workpiece probe can be calibrated in a hole (calibration ring) or on a surface for a particular axis and direction.



### Calibrating a workpiece probe

### Result:

Probe switching point (trigger value),

possibly an additional skew, active ball diameter of probe



### 1.12.3 Workpiece measurement at one point



### Function

CYCLE978 permits measurement at one point of a surface.

The measuring point is approached paraxially in the active workpiece coordinate system.

Depending on the measuring variant, the result may influence the selected tool offset or zero offset.



# Workpiece measurement: Blank measurement

Result:

- Position
- Deviation
- zero offset





- Actual dimension
- Deviation
- Tool offset





### Function

The following measuring variants are provided for the paraxial measurement of a hole, shaft, groove, web, or rectangle. They are executed by cycle CYCLE977.



### Workpiece measurement: Measuring a hole

Result:

- Actual dimension, deviation: Diameter, center point
- Deviation: Tool offset of the zero offset



10.04

Workpiece measurement: Measuring a shaft

Result:

- Actual dimension, deviation: Diameter, center point
- Deviation: Tool offset of the zero offset



# Workpiece measurement: Measuring a groove

Result:

- Actual dimension, deviation: Groove width, groove center
- Deviation: Tool offset of the zero offset

Workpiece measurement: Measuring a web

- Actual dimension, deviation: Web width, web center
- Deviation: Tool offset of the zero offset







### Result:

10.04

- Actual dimension, deviation: Rectangle length and width, rectangle center
- Deviation: Tool offset of the zero offset





### Workpiece measurement: Outside rectangle

- Actual dimension: Rectangle length and width, rectangle center
- Deviation: Rectangle length and width, rectangle center
- Deviation: Tool offset of the zero offset



### 1.12.5 Measuring a workpiece at an angle



### Function

The following measuring variants are provided for the measurement of a hole, shaft, groove, or web at an angle. They are executed by CYCLE979.



# Three- or four-point measurement at an angle

Result:

- Actual dimension, deviation: Diameter, center point
- Deviation: Tool offset of the zero offset







### Two-point measurement at an angle

- Actual dimension, deviation: Groove, web width, groove, web center
- Deviation: zero offset



### 1.12.6 Measuring a surface at an angle



### Function

CYCLE998 permits correction of the zero offset after measurement of a surface at an angle. It is still possible the determine the angles on an oblique surface in space.



# Workpiece measurement: Angle measurement

<u>Result:</u> Actual dimension (angle), deviation, zero offset



# Workpiece measurement: Two-angle measurement

### Result:

Actual dimension (two angles), deviation, zero offset



### 1.12.7 Measuring spheres (from measuring cycles SW 6.3 onwards)



### Function

CYCLE997 permits correction of the zero offset after measurement of a sphere or of three identically sized spheres on a common base (workpiece). Either paraxial measurement or measurement at an angle can be selected.



### Workpiece measurement: Sphere

### Result:

Actual dimension (position of center, diameter), deviation, Zero offset (for one sphere translation only, for three

spheres also rotation in space)



10.04



### 1.12.8 Workpiece measurement: Setting-up a corner



### **Function**

Using the CYCLE961 cycle, it is possible to determine the position of a workpiece corner (inner or outer) and use this as zero offset.



### Measuring a corner by specifying clearances and angles

Result:

- Actual position of the corner with angle •
- Zero offset, rotation •



Introduction



### Any corner, example: internal corner Y ト D/ + P2 P3 WI= NVy= ? х Μ NVx=?

## Measuring a corner, specifying 4 points

- Actual position of the corner with angle
- Zero offset, rotation

### Introduction 1.13 Overview of measuring cycle functions for turning technology

### 1.13 Overview of measuring cycle functions for turning technology

### 1.13.1 Measuring tools at lathes



### Function

Cycle CYCLE982 is used to calibrate a tool probe and measure turning, drilling, and milling tools on turning machines.



### Calibrating a tool probe

### Result:

Probe switching point with reference to machine zero

From measuring cycles SW 6.3 onwards reference to the workpiece zero is possible.





### Measuring the tool

Result: Tool length (length1, length2) Milling radius (R - for milling tools)









### 1.13.2 Calibrating workpiece probes



### Function

CYCLE973 permits calibration of a probe on a surface of the workpiece or in a calibration groove.



### Calibrating workpiece probes

### Example:

Calibrate probe with cutting edge position 7, in X axis in both directions in a calibration groove.



### 1.13.3 Measuring tools at lathes: 1-point measurement



### Function

10.04

CYCLE974 is used to determine the actual value of the workpiece in the selected measuring axis with reference to the workpiece zero with 1-point measurement.



### 1-point measurement outside

<u>Result:</u> Actual dimension (diameter, length), deviation, tool offset, zero offset

### 1-point measurement inside

<u>Result:</u> Actual dimension (diameter, length), deviation, tool offset, zero offset

# 1-point measurement outside with 180° reversal spindle

Result:

Actual dimension (diameter, length), deviation, Tool offset

# 1-point measurement inside with 180° reversal spindle

<u>Result:</u> Actual dimension (diameter, length), deviation, Tool offset





### 1.13.4 Measuring workpieces at lathes: 2-point measurement



### Function

CYCLE994 is used to determine the actual value of the workpiece in the selected measuring axis with reference to the workpiece zero with 2-point measurement. This is done automatically by approaching two opposite measuring points on the diameter.



<u>Result:</u> Actual dimension (diameter), deviation, Tool offset

### 2-point measurement on inside diameter

<u>Result:</u> Actual dimension (diameter), deviation, Tool offset



10.04



### **Parameter Description**

Parameter concept of the measuring cycles	2-62
Parameter overview	2-63
Supply parameters	2-63
Result parameters	2-65
Description of the most important defining parameters	2-66
Measurement variant: _MVAR	2-66
Number of the measuring axis: _MA	2-66
Tool number and tool name: _TNUM and _TNAME	2-67
Offset number: _KNUM	2-68
Offset number _KNUM extended for tool offset: Maximum 9 digits	2-71
Correcting setup and additive offset in workpiece measurement: _DLNUM	
(from measuring cycles SW 6.3 onwards)	2-72
Correcting the tool of a stored tool environment: _TENV	
(from measuring cycles SW 6.3 onwards)	2-73
Example of automatic tool offset in workpiece measuring cycles	2-74
Variable measuring velocity: _VMS	2-76
Offset angle position: _CORA	2-76
Tolerance parameters: _TZL, _TMV, _TUL, TLL, _TDIF, and _TSA	2-77
Measurement path: _FA	2-78
Probe type, probe number: _PRNUM	2-79
Empirical value, mean value: _EVNUM	2-80
Multiple measurement at the same location: _NMSP	2-81
Weighting factor for mean value calculation: _K	2-81
	Parameter concept of the measuring cycles Parameter overview Supply parameters Result parameters Description of the most important defining parameters Measurement variant: _MVAR Number of the measuring axis: _MA Tool number and tool name: _TNUM and _TNAME Offset number: _KNUM Offset number _KNUM extended for tool offset: Maximum 9 digits Correcting setup and additive offset in workpiece measurement: _DLNUM (from measuring cycles SW 6.3 onwards) Correcting the tool of a stored tool environment: _TENV (from measuring cycles SW 6.3 onwards) Example of automatic tool offset in workpiece measuring cycles Variable measuring velocity: _VMS Offset angle position: _CORA Tolerance parameters: _TZL, _TMV, _TUL, TLL, _TDIF, and _TSA Measurement path: _FA Probe type, probe number: _PRNUM. Empirical value, mean value: _EVNUM Multiple measurement at the same location: _NMSP Weighting factor for mean value calculation: _K



### 2.1 Parameter concept of the measuring cycles

### Function

Measuring cycles are general subroutines designed to solve specific measurement tasks, which are suitably adapted to the problem at hand with parameter settings. They can be adapted for this purpose via **defining parameters**.

The measuring cycles also return data such as measuring results. They are stored in **result parameters**. These measuring cycle parameters are called **G**lobal **U**ser **D**ata (abbreviated to GUDs).

They are located in the battery-backed memory of the control. These values are therefore not lost when the control is switched off and on.

This data is kept in data blocks.

- GUD5.DEF
- GUD5.DEF and
- GUD7.DEF (for measuring in JOG)

On delivery these data have default settings (see Section 2.2). These must be adapted by the user or machine manufacturer on installation (see Chapter 9). The defining parameters must be assigned values before the measuring cycle is called:

• Either in the program or

• Operator input in the measuring cycle support The data in the operating area "Parameters" can be displayed by means of "User data", "Global user data", "GUD..." or "Channel-specific user data", "GUD...".. As an alternative, parameters that are not assigned values in the program or in the measuring cycles support can be assigned values directly by experts.

Measuring cycles also require **internal parameters** for calculations. Local User Data (abbreviated to LUDs) are used in the measuring cycles as internal arithmetic parameters.

These are set up in the cycle and exist only during runtime.





### 2.2 Parameter overview

### 2.2.1 Supply parameters



### Explanation

The defining parameters of the measuring cycles can be classified as follows:

- Mandatory parameters
- Auxiliary parameters

Mandatory parameters are parameters that have to be adapted to the measuring task at hand (for example, setpoint axis, measuring axis, etc,) **before each** measuring cycle call.

Additional parameters can generally be assigned once on a machine. They are then valid **for each additional measuring cycle call** until modified by programming or operation.

In addition to defining parameters for calculation or character string input, there are also BOOLEAN type variables. These bits can be used to vary planned cycle sequences or enable or disable certain settings. These **cycle bits** are arrays of variables and of two types:

- Central bits: \_CBIT[]
- Channel-oriented bits: \_CHBIT[]

Their name defines their validity and occurrence:

- Central bits --> NCK: Occur once, apply to all channels
- Channel bits --> CHAN: Defined separately for each channel.

An overview of central and channel-oriented bits is given in Section 9.2.

These bits can also be changed by programming or operation.



### Mandatory parameters

Parameters	Туре	Validity	Default:	Meaning
_SETVAL <sup>1)</sup>	REAL	CHAN	_	Setpoint
_SETV[] <sup>1)</sup>	REAL	CHAN	_	Setpoints – additional, e.g. for measuring rectangle
_ID <sup>1)</sup>	REAL	CHAN	-	Incremental infeed depth/offset
_CPA <sup>1)</sup>	REAL	CHAN	_	Center point abscissa for measuring at angle
	REAL	CHAN	_	Center point ordinate for measuring at angle
_SZA <sup>1)</sup>	REAL	CHAN	-	Protection zone in abscissa
_SZO <sup>1)</sup>	REAL	CHAN	-	Protection zone in ordinate
_STA1	REAL	CHAN	_	Start angle
_INCA	REAL	CHAN	-	Indexing angle
_MVAR	INT	CHAN	-	Measuring variant
_MA	INT	CHAN	-	Measuring axis
_MD	INT	CHAN	-	Measuring direction
_TNUM	INT	CHAN	-	T number
_TNAME	STRING[32]	CHAN	-	Tool name (alternative to _TNUM in tool
				management)
_KNUM	INT	CHAN	_	Correction number (D No. or ZO No.)
_RA	INT	CHAN	-	Number of rotary axis at angle measurement
_TENV	STRING[32]	CHAN	-	Name of tool environment
_DLNUM	INT	CHAN	_	DL number for setup or additive offset

Auxiliary pa	aramete	rs		
Parameters	Туре	Validity	Default	Meaning
			<i>(</i> SW 6.3)	
_VMS	REAL	CHAN	0	Variable measuring speed
_RF	REAL	CHAN	0	Feedrate in circular-path programming
_CORA	REAL	CHAN	0	Compensation angle, e.g. for mono probe
_TZL	REAL	CHAN	0.001	Zero offset area
_TMV	REAL	CHAN	0.7	Mean value generation with compensation
_TNVL	REAL	CHAN	1.2	Limit value for distortion of triangle
_TUL <sup>1)</sup>	REAL	CHAN	1.0	Tolerance upper limit
_TLL <sup>1)</sup>	REAL	CHAN	-1.0	Tolerance lower limit
_TDIF	REAL	CHAN	1.2	Dimension difference check
_TSA	REAL	CHAN	2	Safe area
_FA <sup>2)</sup>	REAL	CHAN	2	Measurement path in mm
_СМ[]	REAL	NCK	100, 1000, 1,	Monitoring parameters for tool
			0.005, 20, 4, 10, 0	measurement for a rotating spindle
_PRNUM	INT	CHAN	1	Probe number
_EVNUM	INT	CHAN	0	Number of empirical value memory
_CALNUM	INT	CHAN	0	Calibration block number
_NMSP	INT	CHAN	1	Number of measurements at the same
				location
_K	INT	CHAN	1	Weighting factor for mean value calculation

### Parameters for logging only

Parameters	Туре	Validity	Meaning
_PROTNAME[]	STRING[32]	NCK	[0]: Name of main program the log is from
			[1]: name of log file
_HEADLINE[]	STRING[80]	NCK	6 strings for protocol headers
_PROTFORM[]	INT	NCK	Log formatting
_PROTSYM[]	CHAR	NCK	Separator in the log
_PROTVAL[]	STRING[100]	] NCK	[0, 1]: Log header line
			[2–5]: Specification of the values to be logged
_DIGIT	INT	NCK	Number of decimal places

Н

 All parameters with dimensions, except for those marked 1), must be programmed in the unit of measurement of the basic system. The parameters marked 1) must be programmed in the unit of the active system of units.

2) \_FA is always a value in mm, even when the unit system set is inches.

### 2.2.2 Result parameters

Parameters	Туре	Validity	Meaning
_OVR[ ]	REAL	CHAN	Result parameter – real number:
			Setpoint values, actual values, differences, offset
			values, etc.
_OVI[]	INT	CHAN	Result parameter – integer



#### 2.3 Description of the most important defining parameters

#### 2.3.1 Measurement variant: \_MVAR



### Function

The measuring variant of each individual cycle is defined in parameter \_MVAR. \_MVAR can be assigned certain positive integer values.



Please refer to the individual cycle descriptions!



The value in MVAR is plausibility checked by the cycle. If it does not have a defined value, alarm 61307 is output: "Incorrect measuring variant". The cycle must be interrupted by an NC RESET. \_MVAR must be corrected.

### 2.3.2 Number of the measuring axis: \_MA



### **Function**

In some cycles or measuring variants, number 1, 2, or 3 must be specified in \_MA for the measuring axis. This might by axis X, Y, or Z in the workpiece coordinate system depending on whether G17, G18, or G19 is active.

This always results in:

Measuring axis abscissa	_MA = 1	
Measuring axis ordinate	_MA = 2	
Measuring axis applicate	_MA = 3	

### Example:

Workpiece measurement on milling machine with G17, various measuring directions

With some measuring variants, for example,

CYCLE998, positioning in another axis (which must be defined), called the offset axis, can be performed between measurements in the measuring axis. This is must be defined in parameter \_MA with offset axis/measuring axis. The higher digit codes the offset axis, the lower digit

the measuring axis, the tens digit is 0.



10.04





### Example of \_MA in CYCLE998:

\_MA = 102

- $\Rightarrow$  Offset axis: 1 (abscissa)
- $\Rightarrow$  Measuring axis: 2 (ordinate)

### 2.3.3 Tool number and tool name: \_TNUM and \_TNAME

### Function

The tool to be offset is entered during workpiece measurement in the parameters **\_TNUM** and **\_TNAME**.

The parameter \_TNAME is only relevant if tool management is active. Here it can be used as an alternative to \_TNUM. However, a programmed \_TNUM >0 always has priority.

Example:

- Without tool management: \_TNUM = 12 tool T number 12
  - is corrected
- With tool management:
   \_TNUM = 0 \_TNAME = "DRILL"
   → the tool called
   "DRILL" is corrected

Or

\_TNUM = 13 \_TNAME = "DRILL" → the tool with the internal

T number 13 is corrected



In the case of replacement tools, the tool that was last used is corrected.

However, it is necessary that only one tool in a group be "active" at any one time. Otherwise, the internal tool number of the tool used must be determined and assigned to \_TNUM via the system variable \$P\_TOOLNO during processing.



### 2.3.4 Offset number: \_KNUM



### Parameters

With measuring variant \_MVAR you can select whether **automatic tool offset** will be used or a **zero offset** will be corrected in a **workpiece measuring cycle**.

Parameter \_KNUM contains the

- tool offset memory number (D number) or a
- code for the **zero offset** to be corrected.

### Values of \_KNUM: ≥0, integer

### 1. Specification \_KNUM for tool offset, 7 digits: \_KNUM can accept values with up to 7 digits (for special MD settings, even 9 digit values, see Subsection 2.3.5). \_KNUM=0: no automatic tool offset

	<ul> <li>D number</li> <li>-0/1automatic offset of the length effective in the measuring axis or set-up / sum offset (as of measuring cycles software version 6.3)</li> <li>2automatic radius offset or set-up / sum offset (as of measuring cycles software version 6.3)</li> <li>Standard / inverted offset (sign changed)</li> <li>0Standard</li> <li>1Inverted (sign changed)</li> <li>-0Offset as per 4th digit</li> <li>1Offset of L1</li> <li>2Offset of L3</li> </ul>
	4Radius offset
As of measuring cycles software version 6.3	<ul> <li>-0Offset in length / radius</li> <li>1Correction in set-up / sum correction</li> <li>2Offset in length / radius according to _TENV</li> <li>3Correction in set-up / sum offset according to _TENV</li> <li>Note: Tool environment _TENV -&gt; see parameter _TENV</li> </ul>

Depending on MD 18102: MM\_TYPE\_OF\_CUTTING\_EDGE = 0 and MD 18105: MM\_MAX\_CUTTING\_EDGE\_NO and a value of this MD of between 10...999, the last three digits are read as a D number. If the value is ≥1000, \_KNUM is evaluated as a 5-digit D number (unique D number, as in flat D number structure, see next Section). **Example of \_KNUM=12003:** D3 is correct, calculated as a radius offset, inverted (sign inverted).

In the default setting, the D number has values between 0...9.



- 2. Specification \_KNUM for zero offset:
- \_KNUM=0: No automatic ZO correction
- \_KNUM=1... 99: automatic ZO in settable
  - frame / ZO G54...G57, G505...G599
- \_KNUM=1000: automatic ZO in last channel-specific basic frame according to MD 28081: MM NUM BASE FRAMES.

The offset is calculated to ensure it has the right effect when G500 is activated. The corresponding basic frame must then also be active (relevant bit in \$P\_CHBFRMASK must be set).

- \_KNUM=1011...1026: automatic ZO in 1st to 16th channel basic frame (\$P\_CHBFR[0]...\$P\_CHBFR[15])
- \_KNUM=1051...1066: automatic ZO in 1st to 16th basic frame (NCU global) (\$P\_NCBFR[0]...\$P\_NCBFR[15])
- \_KNUM=2000: automatic ZO in system frame (scratch system frame \$P\_SETFR)
- \_KNUM=99999: automatic ZO in active frame:
  - active settable frame G54...G57, G505...G599 or
  - when G500 is active: last active basic frame according to \$P\_CHBFRMASK (highest set bit).

Only when \_KNUM=99999 is the changed frame immediately activated in the cycle, otherwise **by the user** by writing G500, G54...G5xy.

### Notes:

- The remaining active frame chain must be retained.
- With NCU-global frames, correction for rotation is not possible.

The following must be set during installation (default setting):

- MD 28082: MM\_SYSTEM\_FRAME\_MASK, Bit 0=1 and Bit 5=1 (system frames for scratching and cycles must be available)
- Additionally the MDs for the required basic frames

In the measuring cycles in AUTOMATIC mode the offset for the default setting is corrected additively with **fine offset** (MD 18600: MM\_FRAME\_FINE\_TRANS=1).

Otherwise (when MD 18600=0), or in CYCLE961, or when \_KNUM=2000, or when "measuring in JOG" is active, the offset is implemented in the **coarse offset**. Any existing fine offset is included in the calculation and then deleted.

When measuring workpieces with ZO (offset in CYCLE974, CYCLE977, CYCLE978, CYCLE979, CYCLE997) in AUTOMATIC mode, values can be written either to the coarse or fine offset:



### \_CHBIT[21]:

0: ZO translation additive in FINE

1: ZO translation in COARSE, FINE=0 If the offset is written to the coarse offset any existing fine offset in the offset value is included and the fine offset then deleted.

### Note:

If \_KNUM=2000 (scratch system frame \$P\_SETFR), the offset value is always written to the coarse offset.





### 2.3.5 Offset number \_KNUM extended for tool offset: Maximum 9 digits



### Parameters

For special tool offset structures (D number structures), parameter **\_KNUM** can have up to nine digits.

The **"flat D number"** functionality is implemented as from NCK-SW 4. This function is defined with MD 18102:

MM\_TYPE\_OF\_CUTTING\_EDGE=1. This function allows up to 5-digit D numbers.

"Unique D number" is a second method of implementing a 9-digit \_KNUM: As from NCK-SW 5 and depending on

MD 18102: MM\_TYPE\_OF\_CUTTING\_EDGE = 0 and MD 18105: MM\_MAX\_CUTTING\_EDGE\_NO  $\geq$ 1000 D numbers have 5 digits and \_KNUM has 9 digits.

References: /FB/, W1, "Tool Offset"

The D number is contained in the five lowest digits of \_KNUM. This is automatically recognized in the cycles by the MD settings. The remaining digits of \_KNUM still have the same meaning but have been shifted two places along.





# 2.3.6 Correcting setup and additive offset in workpiece measurement: \_DLNUM (from measuring cycles SW 6.3 onwards)

### Parameters

Setup and additive offsets are assigned to the tool and a D number. Each D number can be assigned up to 6 setup and additive offsets using DL numbers in the program. If DL=0, no setup or additive offset is activated. The existence or number of setup or additive offset is set in the machine data.

References: /FB1/, W1, "Tool Offset"

From measuring cycles SW 6.3 onwards, it is possible to correct a selected setup or additive offset in the measuring cycles when measuring workpieces with automatic tool offset using parameters \_TNUM and \_TNAME, \_KNUM (D-number coded) and additional parameter \_DLNUM. \_DLNUM is an integer. The value range is 0...6 This variable need only be defined by the user if the corresponding digit is also programmed in variable \_KNUM.

Two **channel-oriented bits** are available for selecting additive and setup offsets (see Subsection 9.2.7). The measuring cycles for workpiece measurement with automatic tool offset use channel bit \_CHBIT[6] for selecting length and radius offsets in wear or geometry.

- 0: The offset value is added to the existing wear.
- 1: Geometry(new) = geometry (old) + wear (old) + offset value, wear (new) = 0

\_CHBIT[6] is also used for correcting setup and additive offsets:

- 0: The offset value is added to the existing additive offset.
- 1: Setup offset (new) = setup offset (old) + additive offset (old)

```
+ offset value,
```

additive offset (new) = 0

### additionally \_CHBIT[8]:

- 0: Correct additive/setup offset according to \_CHBIT[6]
- 1: The offset value is added to the existing setup offset, irrespective of \_CHBIT[6]:
# 2.3.7 Correcting the tool of a stored tool environment: \_TENV (from measuring cycles SW 6.3 onwards)



#### Parameters

As from NCK SW 6.3, you can save the operating environment of a particular tool you are using. This is to allow you to correct the tool used to measure a workpiece taking account of the operating conditions (environment: e.g. plane, length assignment, ...). You then no longer have to specify the T, D, DL number in the offset explicitly. These are included in the stored tool environment. The name of a tool environment can have up to 32 characters.

**References:** /PGA/, "Programming Guide Job planning"

Parameter **\_TENV** is used in measuring cycles for workpiece measurement with automatic tool offset to define the tool environment.

\_TENV is type string [32]. \_TENV is only considered if the corresponding digit is programmed in parameter \_KNUM.

#### Note:

\_TENV can only be used if function TOOLENV("NAME") has already been programmed by the user in the workpiece machining program. This sets up the tool environment. The number tool environments that can be created in the SINUMERIK control is set in

MD 18116: MM\_NUM\_TOOL\_ENV.

A tool offset used in conjunction with a tool environment offers many possibilities. These will be shown in more detail using examples in the next Section.



#### 2.3.8 Example of automatic tool offset in workpiece measuring cycles



## Examples of parameter settings for tool offset with and without stored tool environment

If the tool environment was stored during workpiece machining with TOOLENV("NAME"), it is possible to subsequently correct a tool under these stored conditions while measuring a workpiece.

First, the name of the tool environment \_TENV="NAME" must be specified in the offset.

\_TENV is evaluated if the corresponding position in \_KNUM has value 2 or 3 ( offset ... according to \_TENV).

To correct tool T stored in tool environment "NAME", \_TNUM=0 must be set. Otherwise the programmed \_TNUM / TNAME, D (contained in \_KNUM), \_DLNUM is corrected with the conditions of the specified tool environment "NAME". Further alternatives: See examples.

In "flat D number " structures \_TNUM has no significance. Here, only \_KNUM (for D) and \_DLNUM are relevant.

#### Example 1: (without \_TENV)

The wear of length 1 is to be corrected additively for tool T7 with D2. The tool environment is to be the environment currently active (= measuring environment).

Relevant data: \_TNUM=7 \_KNUM=0100002 \_CHBIT[6]=0

#### Example 2: (without \_TENV)

For tool T8 with D3, the wear of the length assigned to measuring axis \_MA for this tool type and setting (G17, G18, or G19 etc.) is to be corrected additively.

The tool environment is to be the environment currently active (= measuring environment).

Relevant data: \_TNUM=8 \_KNUM=3 \_CHBIT[6]=0 \_MA=1

#### Example 3: (without \_TENV)

For tool T5 with D2, the additive offset of DL=3, active in the length as defined for measuring axis \_MA, is to be corrected additively. The tool environment is to be the environment currently active (= measuring environment).

Relevant data: \_TNUM=5 \_KNUM=1001002 (or =1000002) \_DLNUM=3 \_CHBIT[6]=0 \_CHBIT[8]=0 \_MA=1



2

10.04

Example 4: (with \_TENV) The wear of length 1 or the tool and D number stored in tool environment "WZUMG1" is to be corrected additively (T and D not directly known). Relevant data: \_TNUM=0 \_TENV= "WZUMG1" \_KNUM=2100000 \_CHBIT[6]=0

Example 5: (with \_TENV)

For tool T stored in tool environment "WZUMG2" but specific D number D2, the wear of the length assigned to measuring axis \_MA for the stored tool type and stored setting (G17, G18, or G19) is to be corrected additively.

Relevant data: \_TNUM=0 \_TENV= " WZUMG2" \_KNUM=2001002 (or \_KNUM=2000002) CHBIT[6]=0 \_MA=2

Example 6: (with \_TENV)

Tool environment is to be the tool environment stored in "WZUMG3".

However, the following is to be corrected irrespective of T, D, DL stored in it.

For tool T6 with D2 the additive offset of DL=4 that is assigned to the length for tool type T6 and setting (G17, G18, or G19) stored in "WZUMG3" of the measuring axis is to be corrected additively.

Relevant data:

\_TNUM=6 \_TENV="WZUMG3" \_KNUM=3001002 (or \_KNUM=3000002) \_DLNUM=4 \_CHBIT[6]=0 \_CHBIT[8]=0 \_MA=1





#### 2.3.9 Variable measuring velocity: \_VMS



#### Parameters

The measuring velocity can be freely selected by means of **\_VMS**. It is specified in mm/min or inch/min depending on the basic system.

The maximum measuring velocity must be selected to ensure safe deceleration within the probe deflecting path.

When  $_VMS = 0$ , 150 mm/min is the default value for the feedrate (when  $_FA=1$ ). This feedrate value automatically increases to 300 mm/min if  $_FA>1$  is selected.

If the basic system is in inches, 5.9055 inch/min or 11.811 inch/min takes effect.

#### 2.3.10 Offset angle position: \_CORA

#### Function

When using a mono probe, it may be necessary to correct the position of the probe to enable the measurement for machine-specific reasons (e.g. horizontal/vertical millhead).

In CYCLE982 \_CORA is also used to correct the spindle setting after measurement with reversal during milling measurement.



#### Parameters

Correction for mono probe: The incorrect position (see above) can be corrected by means of parameter **\_CORA**. \_CORA is generally set to 90° or a multiple thereof. If the direction of rotation is altered as a result of swiveling the milling head, \_CORA must be preset to  $-360^{\circ}$  (normally 0°).





In Section 1.8 the correction strategy of measuring cycles is explained and a description of the effect of the parameters given.



#### Parameters

_TZL	REAL ≥0	Zero offset <sup>1)2)</sup>
_TMV	REAL >0	Mean value calculation with offset <sup>1)</sup>
_TUL/_TLL	REAL	Workpiece tolerance <sup>1)</sup>
_TDIF	REAL >0	Dimension difference check <sup>1)2)</sup>
_TSA	REAL >0	Safe area

1) For workpiece measurement with automatic tool offset only

2) Also for tool measurement

#### Value range

All of these parameters can have any value. However, only values increasing from \_TZL to \_TSA are meaningful (absolute values). Parameters \_TUL/\_TLL are specified in mm or inches depending on the active dimension system and are signed. All other parameters are programmed in the basic system.

## Making the workpiece tolerance and setpoint symmetrical

If asymmetrical values are chosen for the tolerance parameters \_TUL, \_TLL (workpiece tolerance), the setpoint (\_SETVAL) is corrected internally to place it in the center of a new, symmetrical tolerance band. This changed values appear in the result parameters: OVR[0] – setpoint, OVR[8] – upper tolerance limit, OVR[12] – lower tolerance limit.

The defining parameters themselves (\_TUL, \_TLL, \_SETVAL) remain unchanged.

#### Example:

\_TUL=0.0 \_TLL=-0.004 \_SETVAL=10 The result is: OVR[8]=0.002 OVR[12]=-0.002 OVR[0]=9.998







#### Parameters

Measurement path \_FA defines the distance between the **starting position** and the expected switching position (**setpoint**) of the probe. \_FA is data type REAL with values >0. Values <0 can only be programmed in CYCLE971 Always specify \_FA in **mm**. The measuring cycles automatically generate a Total meas. distance = 2 · \_FA in mm. The **maximum measuring** position is therefore \_FA behind the set position. See also Section 1.7 Measuring principle

#### Example:

The default setting is \_FA=2.0.

As a result a total measurement distance of 4 mm is generated in the measuring cycle, starting 2 mm in front of and 2 mm behind the defined set position.

\_FA is also used as a distance for traveling around workpieces or tool probes.

#### Caution

Even if inches is selected as the measuring system, measurement distance \_FA is always specified in **mm**!

In that case, convert the measurement distance to mm: \_FA [mm] = \_FA' [inch]  $\cdot$  25.4

#### Note

In previous measuring cycle versions, \_FA was the name for "multiplication factor of measurement distance". Its task and function remain the same. \_FA is now directly assigned mm as the unit of measurement.







#### Function

• The data for the workpiece probes are stored in the data block (GUD6) in array

#### \_WP[] – workpiece probe.

 The data for the tool probes are stored in the data block (GUD6) machine-specifically in array
 TPL1 tool probe

### \_TP[] – tool probe.

• From measuring cycles SW 6.3, the data for the tool probes can be stored workpiece-specifically in data block (GUD6) in array

## \_TPW[] – tool probe in any workpiece coordinate system.

Arrays are available for up to 99 probes. In the default setting there are arrays for three probes each.

**\_PRNUM** states the number of the probe. This number is used as the index for the probe's array.

Array type \_WP, \_TP, or \_TPW is selected in the cycle via the measuring variant and measuring task:

workpiece or tool measurement.

#### Note:

Which **measuring input** (1 or 2) is used for workpiece measurement and which is used for tool measurement is defined in \_CHBIT[0] and \_CHBIT[1] (see Section 9.2).

The arrays are configured by the machine manufacturer during installation (see Chapter 9).



#### Parameters

Value of \_PRNUM: >0, integer \_PRNUM can only have three digits in workpiece measurement. In that case the most significant digit is evaluated as the **probe type**. The two least significant digits represent the probe number.

Digit			Meaning
3	2	1	
	-	_	Probe number (two digits)
0			Multi probe
1			Mono probe





Example of w _PRNUM	vorkpiece measuremen = 102	it:
—	→ Probe type:	Mono probe
	→ Probe number:	2
	→ Array index:	_WP[1,n]
Example of to	ool measurement:	
_PRNUM	= 3	
	→ Probe number:	3
	→ Array index:	_TP[2,n]

#### 2.3.14 Empirical value, mean value: \_EVNUM



### **Function**

The effect of empirical and mean values is described in Sections 1.8 and 1.9.

The empirical values and mean values are stored in data block (GUD5) in arrays

- \_EV[] empirical values and
- \_MV[] mean values. •

The unit of measurement is mm in the metric basic system and inch in the inch basic system, irrespective of the active system of units.

The number of existing empirical and mean values is entered in data block (GUD6) \_EVMVNUM[ m,n].

• m: array dimension \_EV[m]

n: array dimension \_MV[n]

The default setting provides 20 values each (array index \_EV, \_MV: 0...19).

#### **Parameters**

Values of \_EVNUM:

= 0: without empirical value, without mean value memory

>0 to 9999: Empirical value memory number = mean value memory number

>9999: the higher 4 digits of \_EVNUM are interpreted as the mean value memory number, the lower 4 digits as the empirical value memory number.

The array index for \_EV and \_MV is formed from the value in \_EVNUM.





Example: \_EVNUM = 11  $\rightarrow$  EV memory: 11  $\rightarrow$  EV[10]  $\rightarrow$  MV memory: 11  $\rightarrow$  MV[10] \_EVNUM = 90012  $\rightarrow$  EV memory: 12  $\rightarrow$  EV[11]  $\rightarrow$  MV memory: 9  $\rightarrow$  MV[8]

#### 2.3.15 Multiple measurement at the same location: \_NMSP



#### Parameters

Parameter **\_NMSP** can be used to determine the number of measurements at the same location. The measured values or distances-to-go Si (I=1...n) are averaged.

measurements

That results, for example, in distance-to-go D:

$$D = \frac{S_1 + S_2 + \dots S_n}{n} \qquad ;n: \text{ Number of}$$

#### 2.3.16 Weighting factor for mean value calculation: \_K



#### Function

The parameter for weighting factor **\_K** can be applied to allow different weighting to be given to an individual measurement.

A detailed description is given in Section 1.8 "Measuring strategy and compensation value definition".



Notes

3

10.04

### Measuring Cycle Auxiliary Programs

3.1	Measuring cycle subroutines	
3.1.1	Overview	
3.1.2	CYCLE116: Calculation of center point and radius of a circle	3-85
3.2	Measuring cycle user programs	
3.2.1	CYCLE198: User program before measurement	
3.2.2	CYCLE199: User program after measurement	3-87
3.3	Package structure of the measuring cycles	3-88



#### 3.1 Measuring cycle subroutines

#### 3.1.1 Overview



#### Function

These measuring cycle subroutines are called directly by the cycles. With the exception of CYCLE100, CYCLE101, and CYCLE116, these subroutines cannot be executed by a direct call.



#### Programming

Cycle	Function	
CYCLE100	Activate logging	
CYCLE101	Deactivate logging	
CYCLE102	Measuring result display	
CYCLE103	Parameter setting in interactive mode	only up to measuring cycles – SW 4.5
CYCLE104	Internal subroutine: Measuring cycle interface	
CYCLE105	Internal subroutine: Log	
CYCLE106	Internal subroutine: Log	
CYCLE107	Output of measuring cycle messages	only up to measuring cycles - SW 6.2
CYCLE108	Output of measuring cycle alarms	only up to measuring cycles – SW 6.2
CYCLE109	Internal subroutine: Data transfer	
CYCLE110	Internal subroutine: Plausibility checks	
CYCLE111	Internal subroutine: Measuring functions	
CYCLE112	Internal subroutine: Measuring functions	
CYCLE113	Internal subroutine: Log	
CYCLE114	Internal subroutine: Load ZO memory,	
	Load tool offset	
	Internal subroutine: Load tool offset	
CYCLE115	Internal subroutine: Load ZO memory	
CYCLE116	Calculation of the center point and radius	
	on a circle	
CYCLE117	Internal subroutine: Prepositioning	
CYCLE118	Internal subroutine: Log	
CYCLE119	Internal subroutine: Arithmetic cycle for	from meas. cycles - SW 6.3
	determining position in space	



#### 3.1.2 CYCLE116: Calculation of center point and radius of a circle



#### Explanation

This cycle calculates from three or four points positioned on one plane the circle they inscribe with center point and radius.

To allow this cycle to be used as universally as possible, its data are transferred via a parameter list.

An array of REAL variables of length 13 must be transferred as the parameter.





#### Programming

CYCLE116 (\_DATE, \_ALM)



#### Parameters

Input data		
Parameters	Туре	Meaning
_DATE [0]	REAL	Number of points for calculation (3 or 4)
_DATE [1]	REAL	Abscissa of first point
_DATE [2]	REAL	Ordinate of first point
_DATE [3]	REAL	Abscissa of second point
_DATE [4]	REAL	Ordinate of second point
_DATE [5]	REAL	Abscissa of third point
_DATE [6]	REAL	Ordinate of third point
_DATE [7]	REAL	Abscissa of fourth point
_DATE [8]	REAL	Ordinate of fourth point



# Measuring Cycle Auxiliary Programs 3.1 Measuring cycle subroutines

The results of the calculation are stored in the last four elements of the same field:

Parameters	Туре	Meaning				
_DATE [9]	REAL	Abscissa of circle center point				
_DATE [10]	REAL	Ordinate of circle center point				
_DATE [11]	REAL	Circle radius				
_DATE [12]	REAL	Status for calculation				
		0 Calculation in progress				
		1 Error occurred				
_ALM	INTEGER	Error number (61316 or 61317 possible)				

This cycle is called as a subroutine by, for example, measuring cycle CYCLE979.

#### Example:

%_N_Circle_MPF			
DEF INT _ALM			
DEF REAL _DATE[13] = (3,0,10,-10,0,0,-10,	;3 points	specified	P1: 0,10
0,0,0,0,0)			P2: -10,0
			P3: 0,-10
CYCLE116(_DATE, _ALM)	;Result:	_DATE[9]=0	
		_DATE[10]=0	
		_DATE[11]=10	
		_DATE[12]=0	
		_ALM=0	
MO			
STOPRE			
M30			

10.04



#### 3.2 Measuring cycle user programs

### Function

Measuring cycle user programs CYCLE198 and CYCLE199 are called in the measuring cycles and can be used to program necessary adjustments before or after a measurement (e.g. activate probe, position spindle).

#### 3.2.1 CYCLE198: User program before measurement



#### Explanation

CYCLE198 is called at the beginning of each measuring cycle. It can be used to program actions necessary before starting a measurement (e.g. activate probe). In the as-delivered state this cycle only contains one CASE

statement that executes a jump to a label with subsequent M17 (end of subroutine) for each measuring cycle.

e.g.: \_M977: ;before measurement in CYCLE977 M17 ;end of cycle

From this label all actions to be executed on each CYCLE977 call must be programmed.

#### 3.2.2 CYCLE199: User program after measurement



#### Explanation

CYCLE199 is called in each measuring cycle when measurement is complete. It can be used to program actions necessary following completion of a measurement (e.g. deactivate probe).

The internal structure of the cycle is that same as that of CYCLE198, i.e. the program lines must be inserted between the label for a particular cycle and M17 (end of subroutine).



Ĵ

The machine data configuration and the software package version determine which programs can be used. It is also possible to partially define these programs in the global cycle data during start-up.

(Please refer to data supplied by the machine manufacturer and Installation and Start-up Guide.)



#### Function

The measuring cycle package supplied consists of:

- Data blocks for defining the global measuring cycle data,
- Measuring cycles,
- Measuring cycle subroutines, and
- Easy-to-use functions.

To ensure that the measuring cycles can be executed in the control, the data blocks must have been loaded into directory "Definitions" and the measuring cycles and measuring cycle subroutines must be stored in the part program memory.



Please note that the control always requires a Power ON between loading and execution of the measuring cycles!



Please refer to the notes on installation in Chapter 10.



### Subpackages

In many applications, not all the measuring cycles are used on one machine. Instead subpackages are used. The following overview shows which subpackages are advisable and executable. This will help you to save

memory.





Notes



74

### Measuring in JOG

4.1	Overview function, procedure, prerequisites	4-92
4.2	Measuring workpieces (up to measuring cycles SW 6.2)	4-95
4.2.1	Sequence of operation and functions for workpiece measurement	
	(up to measuring cycles SW 6.2)	4-96
4.2.2	Calibrating workpiece probe (up to measuring cycles SW 6.2)	4-97
4.2.3	Measuring an edge (up to measuring cycles SW 6.2)	4-100
4.2.4	Measuring a corner (up to measuring cycles SW 6.2)	4-101
4.2.5	Measuring a hole (up to measuring cycles SW 6.2)	4-103
4.2.6	Measuring a spigot (up to measuring cycles SW 6.2)	4-104
4.3	Workpiece measurement (from measuring cycles SW 6.3)	4-106
4.3.1	Overview of function and sequence	4-106
4.3.2	Calibrating workpiece probes (from measuring cycles SW 6.3)	4-110
4.3.3	Measuring the edge (from measuring cycles SW 6.3)	4-113
4.3.4	Measuring a corner (from measuring cycles SW 6.3)	4-119
4.3.5	Measuring a pocket, hole or spigot (from measuring cycles SW 6.3)	4-122
4.3.6	Orienting a plane (from measuring cycles SW 6.3)	4-132
4.3.7	Rejecting, repeating, end of measurement (from measuring cycles SW 6.3)	4-134
4.3.8	Information regarding cascaded measurements	4-136
4.3.9	Support of set-up in JOG - after measurement	4-137
4.4	Measuring tools	4-140
4.4.1	Overview of function and sequence	4-140
4.4.2	Calibrating the tool probe	4-140
4.4.3	Measuring milling and drilling tools	4-142

#### 4.1 Overview function, procedure, prerequisites

#### Function

There are two types of measurement: **Workpiece measurement** and **tool measurement**. Measurement may be

- Automatically executed (cycles for automatic operation: refer to Chapter 5 for milling and Chapter 6 for turning) or
- **semi-automatic** in the JOG mode.

This chapter describes semi-automatic measurement for milling technology: **Measuring in JOG**.

Workpiece measurements may include:

- Calibrating the workpiece probe,
- Measuring the contour elements on a workpiece (edge, corner, hole, spigot, rectangle) and then aligning the workpiece by determining and setting a zero offset.

Tool measurements may include:

- Calibrating the tool probe,
- Determining the tool length or radius of milling tools, or tool length of drills and then setting the appropriate offset in the tool offset memory.

## **.**

#### General sequence

The required function is selected via softkeys on the HMI in JOG mode. Input displays are shown. These have to be completed via screen forms. This describes the measurement tasks and selects the offset.

The user must put the tool or probe into a permissible starting position for the measurement task, e.g. using the traverse keys or handwheel (manual traverse).

Once you have pressed the "NC start" key in JOG mode, the remaining sequence is automatic.

To cancel measurement, press "RESET".

#### Notice

Be sure to select the correct channel! The "Measure in JOG" function is **channel-specific**.

#### Calling CYCLE198, CYCLE199

CYCLE198 and CYCLE199 are called before and after a measurement, see Section 3.2. From meas. cycles SW 6.3: In addition, CYCLE199 is called on "Measure in JOG" after completion of complex measurements with several cycle calls, e.g. after "Center calculation from 3 holes/spigots with offset". CYCLE199 contains labels specially for the various complex measurements.



Users can program their own special actions after the label in question.

Entry for "Measure in JOG" and selection is performed via the softkeys

Measure

10.04

workpiece (calculating and setting workpiece reference points or calibration of the workpiece probe)

Or

Measure tool

(Measurement of milling or drilling tools or calibration of the tool probe)

in the basic display of JOG mode:

Machine	chan1		JOG	\ MPF0					
Channel re:	set								G fct.+
Program ab	orted		_			ROV			uansi.
									Auxiliary
мс	S Positi	on	D	to-go		Master	spindle		func.
X1	0.0	00 mm		0.000		Act.	0.00	0 rpm	
Y1	0.0	00 mm		0.000		Set	0.00	0 rpm	Spindles
Z1	0.0	00 mm		0.000		Pos		0 deg.	
A1	0.0	00 deg	1	0.000			100.	0%	Axis feedrate
B1	0.0	- 00 deg		0.000	÷	Power			recarace
					_				
						Feeulau		°	
						ACL	0.00	0 100.0 %	7
						Set	0.00	0	act. val.
· · · ·						Tool			
						) 			Act. val. MCS
						Presele	cted tool:	(	
						G01			
	Preset	Scrate	h Me	easure Irkniece	Me	asure	Handwheel	INC	



#### Prerequisites

The prerequisites for "Measuring in JOG" are detailed

in Part 2 "Function description" (Chapter 8 ff.).

The following checklist is useful for determining whether the preconditions are fulfilled: **Machine** 

- All machine axes are designed in accordance with DIN 66217.
- All 3 geometry axes are present according to MD 20050 AXCONF\_GEOAX\_ASSIGN\_TAB.
- Reference point approach was completed in all the required machine axes.

#### Measuring in JOG 4.1 Overview function, procedure, prerequisites

• A touch-trigger workpiece probe (3D) is provided for acquiring workpiece dimensions, and a touch-trigger tool probe for acquiring tool dimensions.

#### Control

810D, 840D, 840Di with HMI Advanced, with necessary **option**: "Inter-modal actions" (ASUP's and synchronized actions in all modes)

#### General machine data for machine cycle runs

These machine data are described in Section 9.1.

#### Special machine data and other data for measuring in JOG

These special data and settings are described in Section 9.3.

#### Interrupt for starting the ASUP's

As from PLC-SW 6.1, interrupt number 9, previously 8, applies.

This interrupt must not be used by any other application during "measuring in JOG".

#### Data blocks and measuring cycles in general

• The data blocks:

GUD5.DEF and GUD6.DEF in directory DEFINE of the software supplies has been loaded into the control and activated (directory "Definitions" in the file system).

 The measuring cycles in directory CYCLES of the software supplied have been loaded into the standard cycle directory of the control and a power-on has been performed.

#### Special files for measuring in JOG

- All files in directory JOG\_MEAS\CYCLES of the software supplied have been loaded into the control via "Data in" and a power-on has been performed.
- Data block GUD7\_MC.DEF in directory JOG\_MEAS\DEFINE of the software supplied has been adapted and loaded into the control.
   Data block GUD7.DEF has been activated.

Detailed function description: see Section 9.3 and information on upgrading Section 10.2 and in file SIEMENSD.TXT of the software supplied.

#### 4.2 Measuring workpieces (up to measuring cycles SW 6.2)



#### Introduction

From measuring cycles SW 6.3 onwards, the functional scope has been extended for workpiece measurement. This is described in Section 4.3. The following description only applies up to SW 6.2.



#### Function

With this function you can set reference points on the workpiece using a workpiece probe on the machine.

You call a measuring cycle to set up a workpiece that is clamped on the table. This measuring cycle automatically generates the measurement paths and intermediate positions as a function of the specified setpoints. During the runtime of the measuring cycle, the ZO is active that is defined via a data block (GUD6) in the field and the work plane G17 ... G19 is active that is set in the data block (GUD6) in another field. A field of the data block (GUD6) also defines which array is assigned to the probe in the spindle. The switching behavior parameters determined by probe calibration are stored in this array.

All the measuring points required for the measurement task are approached. Prepositioning can either be performed manually or in a program. Measurement is performed in the workpiece coordinate system (WCS).

When measurement is complete, the result (corner, center point of hole/spigot, edge) is automatically calculated in the measuring cycle according to the type of measurement, and the reference point is set with reference to the basic frame/system frame "set zero point" (\$P\_SETFRAME) or a settable zero offset according to the selection made by correcting the zero offset memory in question. If "Off" is selected, no correction is made.

#### Precondition

The workpiece probe is in the spindle, the tool offset for it has been selected and it has been calibrated.

# 4.2.1 Sequence of operation and functions for workpiece measurement (up to measuring cycles SW 6.2)

#### Sequence

- 1. The workpiece is clamped, the probe is positioned in the spindle and calibrated.
- 2. When you press softkey "Measure workpiece", the following softkey bar is displayed for selection:

Edge >
Corner >
Hole >
Spigots >
Probe calibration

- 3. Selection in the screen form:
- Select the zero offset to which the specified setpoint position refers and for which the offset is to apply:
  - Off (just measure, no offset)
  - Settable zero offset G54, ... G57, ...
  - Basic frame/basis

Up to measuring cycles SW 6.02.20, selection of "basis" puts the offset in the last channelspecific basic frame acc. to MD 28081: MM\_NUM\_BASE\_FRAMES. The offset is calculated to have the right effect when G500 is activated. The corresponding basic frame then be active. From measuring cycles SW 6.02.20, it is possible to correct in the system frame "set zero" (\$P\_SETFRAME). This function must be activated via a setting data in data block GUD6: \_JM\_B[4] =1 : Selecting "Basis" means putting the offset into the system frame "set zero". G500 must be active during measurement.

\_JM\_B[4] =0 : Selecting "basic frame" means offset into the last channel-specific basic frame acc. to MD 28081.

- Enter setpoints if necessary (e.g. approx. diameter of hole/spigot).
- Select the setpoint position in the measuring axis (for edge), the center point (for hole/spigot) or the corner point.
- Select axis and axis direction for edge/corner.
- 4. On "NC start", measurement runs with a preset measuring feed. The measuring probe is triggered. For example: During "measure edge" or "measure corner", the probe is automatically rapid-traverse retracted to its starting position for the measuring point. When a hole or spigot is measured, 4 points are automatically probed one after the other.



The translation offset and also an offset for the rotation around the infeed axis is applied based on the measuring results and the specified setpoint position when determining the corner defined for the selected ZO. Selecting the basic frame always puts the offset in the last channelspecific basic frame according to

MD 28081: MM\_NUM\_BASE\_FRAMES, if several are available.

#### Sequence for "measuring in JOG with active TRAORI"

The subsequent description refers to the following applications:

- Measuring in JOG, with active TRAORI and with the measuring probe "applied"<sup>1)</sup>.
- Measuring in JOG, in the rotated (swiveled) planes
- 1. TRAORI is active
- 2. If a workpiece is measured and the measuring probe is not in the basic (initial) setting of the machine kinematics, then this must be communicated to the NCU as rotation in the active zero offset (ZO).
  - Example 1: The workpiece cannot be clamped to the machine with the longitudinal and transverse sides. In this case, the workpiece coordinate system (WCS) must be rotated around Z through 90 degrees. Entry in the active ZO, rotation around Z.
  - Example 2: The whole to be measured is located on a vertical side of the workpiece. In this case, the WCS must be rotated around X or Y through 90 degrees. Entry in the active ZO rotation around X or Y.
- Aligning the workpiece coordinate system (WCS) in the tool direction in MDA: G0 C3=1 (valid for G17 plane).
- 4. With NC start, traversing motion is initiated and the tool aligns itself in the rotated WCS in the Z axis.

The vertical orientation to the XY plane is kept.

- The edge, corner,... is approached If the edge, corner ... cannot be completely reached with the machine axes, the measuring probe can be advanced using the rotary axes. In this case, the WCS is not updated.
- 6. Start the measuring process!



#### Note

An "applied" measuring probe has an angular offset regarding the tool orientation in the feed axis Z assumed using the NC command C3=1.

#### 4.2.2 Calibrating workpiece probe (up to measuring cycles SW 6.2)

#### Function

With milling machines and machining centers, the probe is usually loaded into the spindle from a tool magazine. This may result in errors when further measurements are taken on account of probe clamping tolerances in the spindle.

In addition, the trigger point must be precisely determined in relation to the spindle center. This is performed by the calibration cycle with which it is possible to calibrate the measuring probe either in any hole or on a surface.

The type of calibration is selected with softkeys "Length" and "Radius".



# Calibrating the workpiece probe in any hole (radius)

This cycle permits probe calibration in any hole of a reference part meeting the geometrical accuracy and surface quality requirements, e.g. on a workpiece or in an adjustment ring. The resulting trigger points are automatically loaded in the corresponding data storage area of the data block (GUD6).



#### **Operating sequence**

#### Precondition

The measuring probe is located in the spindle. The precise radius of the probe ball must be entered in the tool offset block. An adjustment ring with a known radius, for example, is used for calibration.

#### Approaching the reference part

The probe is approximately positioned at the center of and at the calibration depth of the hole.

#### Select the function with softkey

Measure Calibrate probe > Radius

#### Enter details in input form

Enter diameter  $\varnothing$  of the reference part (here: adjustment ring).



Calibration is performed automatically as soon as you press "NC Start". First, the precise position of the center of the adjustment ring is calculated. Then, four trigger points inside the adjustment ring are sampled one after the other.



# Calibrating a workpiece probe on any surface

With this measuring cycle you can calibrate the probe in the infeed axis on any surface, e.g. on the workpiece, to determine the length.





#### **Operating sequence**

#### Precondition

The measuring probe is located in the spindle. The precise radius of the probe ball must be entered in the tool offset block. The precise value of the height of the calibration surface of the active zero offset set with a date in block GUD6 must be known!

#### Approach the workpiece

The probe must be positioned opposite the calibration surface of the workpiece.

#### Select the function with softkey

Measure workpiece	Calibrate probe >	Length

#### Enter details in input form

Known reference point of the calibration surface relative to the active zero offset set in the data block (GUD6) during measurement.



Calibration is performed automatically as soon as you press "NC Start". The measuring probe is triggered.

The calculated length of the probe is written to the tool offset data block.

#### 4.2.3 Measuring an edge (up to measuring cycles SW 6.2)

#### Function

If "Measure edge" is selected, a reference point can be set in any axis of the preset working plane (G17...G19).

Machine	chan1		JOG	\ MPF0					
Channel re	set								
Program ab	ported	_	_			ROV			
									70
M	:s	Position	D	to-go		Master s	pindle		20
X1		0.000	mm	0.000		Act.	0.00	Orpm	~
Y1		0.000	mm	0.000		Set	0.00	Orpm	Ŷ
Z1		0.000	mm	0.000		Pos		0 deg.	
A1		0.000 d	leg	0.000			100.	0%	Y
B1		0.000 d	leg	0.000	•	Power			
<mark>Measure e</mark>	dge						ZO con	npensation in	Z
A Y				Zero	offs		Base (	<mark>)</mark>	
T'				Direc	tion		+	x	
				XU			0.0000	mm	
<b>→</b> 0				Zero	offs			Base	< <
		1		X0 Y0			0.0000 N NNNN	mm mm	
		×							
^									
			Me	easure	M	easure			
			We	orkpiece	to	bl			



#### **Operating sequence**

#### Precondition

The probe is located in the spindle and has been calibrated.

#### Approach the workpiece

Position the probe in the required axis direction in front of the workpiece, e.g. in the +X direction.

#### Select the function with softkey

Measure	Edge	Х	Z
workpiece			

#### Enter details in input form

- Select the zero offset to which the specified setpoint position refers and for which the offset is to apply:
  - Off (just measure, no offset)
  - Basic frame/basis
  - or zero offset taken
    - from the list of zero offsets
- **Direction:** Set the sampling direction of the selected axis for which the reference point has been set, e.g. +X.
- Enter set position of the reference point (edge).

Set the feedrate override switch to the same value as for calibration!

## $\diamond$

10.04

On "NC Start", the measuring operation is automatically performed with a measuring feedrate set via a data block (GUD).

- Probe triggered
- Automatic retraction to starting position in rapid traverse.
- The translation component of the selected ZO is applied based on the measuring results and the specified setpoint position. Selecting the basic frame always puts the offset in the last channel-specific basic frame according to MD 28081: MM\_NUM\_BASE\_FRAMES, if several are available. The offset is applied in the coarse offset, and any fine offset is reset.

### 4.2.4 Measuring a corner (up to measuring cycles SW 6.2)



#### Function

With the selection "Corner", the corner of a workpiece can be measured as the reference point. The probe is positioned at a selected corner of the workpiece.

Machine	chan1		JOG	\ MPF0					
Channel re:	set								
Program ab	orted		_			ROV			
									70
мс	S Position		D	to-go		Master s	pindle		
X1	0.000	mm		0.000		Act.	0.00	)0 rpm	
Y1	0.000	mm		0.000		Set	0.00	)0 rpm	P1
Z1	0.000	mm		0.000		Pos		0 deg.	
A1	0.000	dea		0.000			100.	.0 %	Store
B1	0.000	deg		0.000	•	Power			12
<mark>Measure c</mark>	orner					·	ZO co	mpensation in	Store P3
1 v				Zero	offs		Base	<mark>)</mark> .	
I 1 '	-						Pos. 1	mm	Store
P3_				XU VO			0.0000	mm	P4
				Zero	offs		0.0000	Base	
- <b>-</b> 4	Ta			×0			0.0000	mm	**
•	P1 P2 X			YO			0.0000	mm	
_									
									Proc
									overs



#### **Operating sequence**

#### Precondition

The probe is located in the spindle and has been calibrated.

#### Approach the workpiece

Position the probe at a selected corner of the workpiece.

### Select the function with softkey



#### Enter details in input form

- Select the zero offset to which the specified setpoint position for the corner refers and for which the offset is to apply:
  - Off (just measure, no offset)
  - Basic frame/basis
  - or zero offset taken
  - from the list of zero offsets
- **Position:** Set the corner to be used as the reference point.
- Enter set position of the reference point (corner).

#### Approach sampling point

Position the probe at the first sampling point **P1** of the workpiece edge.

Set the feedrate override switch to the same value as for calibration!



On "NC Start", the measuring operation is automatically performed with a measuring feedrate set via a data block (GUD).

- Probe triggered
- Automatic retraction to starting position in rapid traverse.

Store the position values of sampling point **P1** by pressing softkey "Save P1". Repeat the procedure "approach sampling points" for sampling points **P2...P4** in the same way.

Calculate	
corner	

Press softkey "Calculate corner" to calculate the translation offset and the rotational offset around the infeed axis for the selected zero offset. When the basic frame is selected, the offset is always made in the last channel-specific basic frame according to MD 28081: MM\_NUM\_BASE\_FRAMES, if several are available. The offset is applied in the coarse offset, and any fine offset is reset.

• The order in which sampling points P1...P4 are approached must be maintained.

On a rectangular workpiece, three sampling points are sufficient for the calculation.

SINUMERIK 840D/840Di/810D User's Guide Measuring Cycles (BNM) - 10.04 Edition

#### 4.2.5 Measuring a hole (up to measuring cycles SW 6.2)



#### Function

With "Hole", you can set the center of a hole as the reference point. The probe is approximately positioned at the center of the hole and measuring depth.

Measure hole			ZO co	mpensat	ion in
A HY	Zero		Base	<mark>)</mark>	-
	Diameter Vo		0 0000	mm	
I I I I I I I I I I I I I I I I I I I	YO		0.0000	mm	
	10		0.0000		
	Zero offs	Base			
	X0		0.0000	mm	
×	YO		0.0000	mm	<b>•</b>



#### **Operating sequence**

#### Precondition

The probe is located in the spindle and has been calibrated.

#### Approach the workpiece

Position the probe approximately in the center of the hole.

#### Select the function with softkey



#### Enter details in input form

- Select the zero offset to which the specified setpoint position for the center of the hole refers and for which the offset is to apply:
  - Off (just measure, no offset)
  - Basic frame/basis
  - or zero offset taken
    - from the list of zero offsets
- **Diameter:** Enter approximate diameter of the hole. If no diameter is entered, sampling is started from the starting point at measurement feedrate.
- Enter set position of the hole center.

# Set the feedrate override switch to the same value as for calibration!



Measurement is performed automatically as soon as you press "NC Start". One after the other, the probe samples 4 points on the inner surface of the hole. Once the measurement is complete, the translation offset is determined for the selected zero offset. Selecting the basic frame always puts the offset in the last channel-specific basic frame according to MD 28081: MM\_NUM\_BASE\_FRAMES, if several are available. The offset is applied in the coarse offset, and any fine offset is reset.

#### 4.2.6 Measuring a spigot (up to measuring cycles SW 6.2)

#### Function

With "Spigot", you can set the center of a spigot (shaft) as the reference point. The probe is approximately positioned above the center of the spigot.

Measure spigot			ZO co	mpensat	ion in
	Zero		Base	<u> </u>	-
│ ¶ ┞╸ <sub>╋</sub> ╶╸┥ │	Diameter			mm	
	DZ			mm	
	X0		0.0000	mm	
<b>   +</b> ⊲	YO		0.0000	mm	
	Zero offs	Base			
	X0		0.0000	mm	
	YO		0.0000	mm	-



#### **Operating sequence**

#### Precondition

The probe is located in the spindle and has been calibrated.

#### Approach the workpiece

Position the probe approximately above the center of the spigot.

#### Select the function with softkey

Measure Spigot workpiece



#### Enter details in input form

- Select the zero offset to which the specified setpoint position for the center of the spigot refers and for which the offset is to apply:
  - Off (just measure, no offset)
  - Basic frame/basis
  - or zero offset taken
  - from the list of zero offsets
- **Diameter:** Specify the approximate spigot diameter (check diameter>0, safety clearance, include probe offsets).
- Specify set position of the center of the spigot.
- Enter measurement infeed depth.

Set the feedrate override switch to the same value as for calibration!

## $\mathbf{D}$

Measurement is performed automatically as soon as you press "NC Start". One after the other, the probe samples 4 points on the outside of the spigot.

Once the measurement is complete, the translation offset is determined for the selected zero offset. Selecting the basic frame always puts the offset in the last channel-specific basic frame according to

MD 28081: MM\_NUM\_BASE\_FRAMES, if several are available. The offset is applied in the coarse offset, and any fine offset is reset.



### 4.3 Workpiece measurement (from measuring cycles SW 6.3)

#### 4.3.1 Overview of function and sequence

#### Introduction

From measuring cycles SW 6.3 onwards, the functional scope has been extended for workpiece measurement. This is described in a separate chapter.



#### Function

The "measure workpiece" function permits set-up of a workpiece clamped onto a machine table. Reference points on the workpiece are measured by means of workpiece probes, and necessary ZO compensation is calculated and set. For complete set-up, more than one call may be necessary (several axes, rotated clamping, see Section "Cascaded measurement").

The workpiece probe can be calibrated.



#### Sequence

- 1. The workpiece is clamped; the probe is in the spindle.
- 2. Pressing the "Measure workpiece" softkey displays the following softkey bar for selection:

Machine	CHAN1		Jog MPF0					
Channel reset			ROV				•	→ Measure edge
Work	Position	-	Repos offset	Master sp	indle S1	×	200	→ Measure corner
X Y Z A	-333.3333 -111.1111 222.2222 242.5183	mm mm mm deg	0.0000 0.0000 0.0000 0.0000	Act. Set Pos.	0.000 rpm 0.000 rpm 0 de 100.0 %	n n g.	چ	→ Measure pocket/ hole → Measure spigot
C Workpiece	357.6571 measuring	aeg	0.0000					→ Orient plane
							Calibrate probe	→ Calibrate probe
		-	Measure workpiece	Tool measure	_		~~	→ Back (exit measur ing in JOG)

10.04

- 3. The measurement task is selected by pressing the relevant softkeys.
- 4. The relevant input displays are shown. The input screen forms must be filled out.
- 5. Select the zero offset (ZO) to which the specified setpoint position refers and for which the offset is to apply.
- 6. Input of setpoints
  - e.g. approximate diameter for hole/spigot
  - e.g. specifying the set positions in the meas. axis (at edge)
  - e.g. specifying the center point (for hole/spigot) or the corner point
- 7. Select axis and axis direction for edge/corner.
- 8. For "NC start", measurement is started and completed with a preset measurement feed. The first traversing motion is used to correct the mechanical slope of the measuring probe, determined when calibrating, and is automatically executed by the system!

#### Internal sequence

After pressing "NC start", an automatically generated ASUB (/\_N\_SPF\_DIR/\_N\_JM\_MESS\_SPF) is started to call up a measuring cycle with parameters, supplied from entries made in screen forms. This measuring cycle generates the measurement paths and intermediate positions as a function of the specified setpoints.

While the measuring cycle is running, the following settings, defined in the datablock GUD6, in the field  $\_JM\_I[$ ] are effective:

- Zero offset
- Working plane (G17...G19)

• the data field \_WP[] assigned to the active measuring probe The switching behavior parameters of the probe determined by calculating the probe are stored in this data field (data array) \_WP[].

#### Zero offset

The translation offset and/or an offset for the rotation around the infeed axis in the selected ZO is applied based on the measuring results and the specified setpoint position, depending on the measurement task.

With the default setting  $(\_JM\_I[5]=0)$ , the following offsets are possible:

• "Just measure"

(no ZO correction, just display measurement results)

- "G54...G57, ..." (offset in the settable ZO)
- "Basis reference" (offset in the system frame "set zero point" \$P\_SETFRAME)

In the extended correction selection (\_JM\_I[5]=1), the following options are available:

- Correction in any global basic frame of the frame chain active during measurement. Only possible if no rotation is measured: e.g. during "measure edge", "measure distance", "measure hole", and "measure spigot".
- Correction in any channel-specific basic frame of the frame chain active during measurement.

#### "Function interface of the measuring point softkeys (P1...P4)", in the PLC

#### Function

This functionality includes the transfer of the status "n. measuring point completed and measure value saved" at the PLC interface. This has the same significance as activating a corresponding, vertical measuring point softkey (P1 ... P4) after a successful measurement in one of the workpiece measuring variants from measuring in JOG. The function "reject n. measurement  $\rightarrow$  n. measuring point inactive" can be controlled from the PLC interface equivalent to the same operator sequence at the HMI when measuring in JOG.

#### Description of the PLC user interface

•	Signal flow: PLC $\rightarrow$ measuring in JOG (HMI)							
	DB19.DB43 Bit0=1:	Switches softkey "1st measuring point" inactive> reject measurement,						
		The signal must be displayed on the HMI until DB19.DB43 bit4=0						
	DB19.DB43 bit0=1:	Switches softkey "2nd measuring point" inactive> reject measurement,						
		The signal must be displayed on the HMI until DB19.DB43 bit5=0						
	DB19.DB43 bit2=1:	Switches softkey "3rd measuring point" inactive> reject measurement,						
		The signal must be displayed on the HMI until DB19.DB43 bit6=0						
	DB19.DB43 bit3=1:	Switches softkey "4th measuring point" inactive> reject measurement,						
		The signal must be displayed on the HMI until DB19.DB43 bit7=0						
•	Signal flow: Measuri	ng in JOG (HMI) $\rightarrow$ PLC						
	DB19.DB43 bit4=1:	If softkey "1st measuring point" active and can be seen on the screen,						
		Measuring 1st measuring point completed and measured value saved						
	DB19.DB43 bit5=1:	If SK "2nd measuring point" active and can be seen on the screen,						
		Measuring 2nd measuring point completed and measured value saved						
	DB19.DB43 bit6=1:	If softkey "3rd measuring point" active and can be seen on the screen,						
		Measuring 3rd measuring point completed and measured value saved						
	DB19.DB43 bit7=1:	If softkey "4th measuring point" active and can be seen on the screen,						
		Measuring 4th measuring point completed and measured value saved						
## 4

#### Example

10.04

The functionality shown can be used with a Siemens handheld unit (HHU) – when a measuring point has been successfully completed, an LED on the HHU keyboard is lit. The HHU key, assigned to this LED can then act from the sense as rejecting a measuring point.

The PLC user interface signals should be interlocked with the input and output signals of the HHU using the functionality of the PLc user program.

The use of an HHU should allow the process-related control of a measuring operation.

#### Sequence for "measuring in JOG with active TRAORI"

The subsequent description refers to the following applications:

- Measuring in JOG, with active TRAORI and applied measuring probe.
- Measuring in JOG, in the rotated (swiveled) planes

The sequence corresponds to that described in Subsection 4.2.1!

From measuring cycles SW 6.3 onwards, the descriptive points 2. and 3. can also be executed with the functionality of "swivel in JOG".

#### Note

If the "swiveling" or 5-axis transformation (TRAORI) are used to align the measuring probe, then these must also be appropriately set-up in the machine!

Please carefully observe the information provided by the machinery manufacturer!



#### Note In general:

The translation offset is applied in the coarse offset. Any fine offset is reset.

#### 4.3.2 Calibrating workpiece probes (from measuring cycles SW 6.3)

#### Function

On milling machines and machining centers, the workpiece probe is inserted in the spindle. This can lead to measuring errors caused by the clamping tolerances of the probe in the spindle. The switching points (trigger points) of the measuring probe with reference to the spindle center must therefore be determined for each new probe inserted.

This is achieved with calibration by which the probe is calibrated in a hole or on a surface.

#### Precondition

The workpiece probe is the active tool in the spindle with activated tool offsets. The approximate length and the radius of the measuring probe sphere must be entered in the tool data.

An SPOS-capable spindle is required to calibrate the radius of the measuring probe sphere.

#### Note

The feedrate override should have the same value during calibration and measurement to reduce measurement inaccuracies.

#### Operation

After you have pressed the "probe calibration" softkey, a display image is shown, containing the following options:

- "Length" (calibrating the length of the probe at the surface)
- "Radius" (calibrating the radius of the measuring probe sphere in the hole)

The reference dimension for the diameter of the calibration hole and the height of the calibration surface to be specified can be pre-assigned using the following GUD parameters:

- E\_MESS\_CAL\_D → diameter, calibration hole (calibration ring)
- E\_MESS\_CAL\_L → calibration height in the feed axis (refer to WCS)

#### Calibrating probe length

By selecting "length", it is possible to calibrate the probe in the infeed axis on a suitable and precisely known surface, e.g. on the workpiece. The precise length 1 (L1) of the prove is also determined and entered in the tool offset memory.

Note the setting of \_CBIT[14]: \_CBIT[14]=0: L1 referred to ball center \_CBIT[14]=1: L1 ref. to ball circumference 10.04

Machine	CHAN1	Jog	MPF0			
// Channel	reset		Program ab	orted		
				ROV		
Work	Position	Repos	s offset	Master spindle	S1 🕅	Length
×	-333.3333	mm	0.0000	Act.	0.000 rpm	
Y	-111.1111	mm	0.0000	Set	0.000 rpm	Radius
z	222.2222	mm	0.0000	Pos.	0 deg.	
A	242.5183	deg	0.0000		100.0 %	
с	357.6571	deg	0.0000	Power	0%	
3D-probe o	alibration				Height of ref. piece	
+7			ZO	0.000	0 🖉 mm	
	•					
	×					~~

#### Approaching the calibration surface

The probe must be positioned opposite to the calibration surface of the reference or workpiece.

#### Enter details in input form

Enter the known dimension (workpiece coordinate) of the calibration surface, referred to the WCS active when calibrating.

Note the settings in data block GUD6:

- Active zero offset during measurement: Variable \_JM\_I\_[4]).
- Active working plane during measurement: Variable \_JM\_I\_[3].



Calibration is performed automatically as soon as you press "NC Start". Trigger value and length offset L1 are stored.

#### Calibrating probe radius

With the "radius" selection, the probe can be calibrated in the axes of the working plane in a suitable hole (geometrical accuracy, low surface roughness) and precisely known diameter or in calibration ring.

Machine	CHAN1	Jog	MPF0				
🥢 Channel	l reset		Program a	borted		1	
			_	ROV			
Work	Position	R	epos offset	Master spindle	S1	Ø	Length
×	-333.3333	mm	0.0000	Act.	0.000 rpm		
Y	-111.1111	mm	0.0000	Set	mqr 000.0		Radius
z	222.2222	mm	0.0000	Pos.	0 deg.	-	
A	242.5183	deg	0.0000		100.0 %		
с	357.6571	deg	0.0000	Power	0%		
3D-probe o	calibration			<u></u>	Diameter ref.	piece	
iΥ			D	<mark>50.0</mark>	000 🛞 mm		
	$\overline{\mathbf{b}}$						_
			_				٠٠

#### Approaching the calibration ring

Position the probe roughly in the hole center and at calibration depth.

#### Enter details in input form

Enter the known dimension of the calibration ring diameter.

Note the settings in data block GUD6:

- Active zero offset during measurement: Variable \_JM\_I\_[4].
- Active working plane during measurement: Variable \_JM\_I\_[3].



Calibration is performed automatically as soon as you press "NC Start".

The precise position of the hole center is determined first. After that, in both axes of the plane and in each of the two directions, the 4 switching points within the hole are determined and stored as trigger values in the probe's array.

#### 4.3.3 Measuring the edge (from measuring cycles SW 6.3)



#### Function

After selecting "edge", a selection display is shown containing the following options:

- "Set edge"
- "Orient edge"
- "Distance 2 edges"

Machine	CHAN1	Jog	MPF0					
// Channe	el reset		Program a	R0V			•	→ Set edge
Work	Position	Rana	s offeet	Master spindl	o 91	×		→ Orient edge
X Y Z A	-333.3333 m -111.1111 m 222.2222 m 242.5183 c	im im im leg	0.0000 0.0000 0.0000 0.0000 0.0000	Act. Set Pos.	0.000 rpm 0.000 rpm 0 deg. 100.0 %	~	•	→ Distance 2 edges
Workpiece	e measuring							
								Back → (to selection Measure
								workpiece)

#### Precondition

The workpiece probe is the active tool with activated tool offsets in the spindle and is already calibrated (refer to Subsection 4.3.2).



#### Note:

Set the feedrate override to the same value as for calibration!

#### Setting the edge

This enable measurement of a reference point (paraxial edge) on a workpiece in one of the axes X, Y, or Z and setting as ZO (translation).

#### Approach the workpiece

Position the probe in front of the edge.



### Enter details in input form

- Select of the ZO
- Select an axis and measuring direction
- Enter the set position required for the reference point (edge) for the selected ZO.



Measurement is performed automatically with the set measuring feed as soon as you press "NC Start". Measurement is followed by rapid-traverse retraction to the starting position. The offset is then automatically calculated. If the selected offset is the active ZO, then this is automatically activated.

#### **Display and correction**

After successful offset calculation, the frame components of the selected ZO are re-displayed along with the measured position of the edge in the measuring axis. ZO correction is calculated in such a way that the measured edge assumes the required setpoint after activation of the selected ZO.

#### N

- Notes
- The rejection, repetition, and end of measurement are described in Subsection 4.3.7.
- **Support** for the user is featured for **activation of the zero offset** determined in JOG mode and for necessary probe reorientation. This is shown right after the measurement and is detailed in Subsection 4.3.9.

#### Orienting the edge

By measuring two points on a straight workpiece edge, it is possible to determine the angle position of this edge with respect to a reference axis.

Orientation of the workpiece edge is possible by

- "Coordinate rotation" or
- Rotating the workpiece using a rotary table (specifying the rotary axis).
- 2 measuring points are possible, P1 and P2.

#### Approach the workpiece

Position the probe in front of the edge, first and P1 and after completion of this measurement at P2.

Align edge		Aut. over	ride in <del>w</del> ork offset
Υ	Work offs X 0.0000 deg Y 0.0000 deg Z 0.0000 deg WO dur. measur.: G500 α	<mark>G54</mark> Meas. direct. Ref. axis Angle offset Spec. angle	+ Y X Coord. rot. 0.0000

#### Enter details in input form

- Select of the ZO
- Select axis X, Y, or Z and measurement direction
- Select reference axis for the angle
- Select the required angle offset: Select "coordinate rotation" or the name of the rotary axis
- When entering the "set angle", an orientation deviating from 0 degrees is possible.

#### Orienting the edge, rotating with rotary table:



# $\diamond$

Measurement is performed automatically at **P1** with the set measuring feed as soon as you press "NC Start". Measurement is followed by rapid-traverse retraction to the

starting position. After successful measurement, the measured value is stored internally and the "P1 stored" softkey that was previously switched inactive is activated.

After manual positioning at measuring point **P2** and with softkey "P1 stored" switched active, pressing "NC start" starts measurement of the 2nd measuring point in the selected measuring axis. After successful measurement, softkey "P2 stored" is activated, and the 2nd measured value is stored. A vertical softkey – "Calculate" is then displayed. After this softkey is pressed, then angle between the edge and the reference axis and the selected contour calculation is calculated.

When the "coordinate rotation" angular offset (correction) is selected, then the offset is also activated if it is the active ZO.

When the "rotary axis" angular offset is selected (name of the rotary axis), a help function for the user is displayed to activate the offset (refer to Subsection 4.3.9).

### **Display and correction**

After offset calculation has been completed, the frame components of the selected ZO are redisplayed along with the measured "alpha" angle.

In the case of ZO correction by "rotation of the workpiece coordinate system", the selected ZO is calculated such that the measured edge forms the required angle (set angle) with the selected reference axis after activation of the selected ZO.

In the case of ZO correction in a specified rotary axis, the difference in angle between the measured and desired angle is entered in the selected ZO (translation) of the specified rotary axis.

No check is made for correct assignment of the specified rotary axis!

#### Notes

- The rejection, repetition, and end of measurement are described in Subsection 4.3.7.
- **Support** for the user is featured for **activation of the zero offset** determined in JOG mode and for necessary probe reorientation. This is shown right after the measurement and is detailed in Subsection 4.3.9.

#### **Distance 2 edges**

This means that the distance L of two edges parallel to an axis at a workpiece, e.g.: Slot, rib, or step, can be determined in one of the axes X, Y or Z and their center set as reference point in a ZO. 2 measuring points are possible, P1 and P2.

#### Approach the workpiece

Position the probe in front of the edge, first and P1 and after completion of this measurement at P2.



#### Enter details in input form

- Select of the ZO
- Select axis X, Y, or Z and measurement direction at P1
- Select the measuring direction at P2
- Enter the set position of the two edges for the selected ZO.



Measurement is performed automatically at **P1** with the set measuring feed as soon as you press "NC Start". After successful measurement, the measured value is stored internally and the "P1 stored" softkey that was previously switched inactive is activated.

After manual positioning at measuring point **P2** and with softkey "P1 stored" switched active, pressing "NC start" starts measurement of the 2nd measuring point in the selected measuring axis. After successful measurement, softkey "P2 stored" is activated, and the 2nd measured value is stored. A vertical "Calculate" softkey is then displayed. After this softkey is pressed, the distance and the distance center between the 2 measuring points in the selected measuring axis and the selected offset calculation are calculated. The offset is activated at the same time if it is the active ZO.

#### **Display and correction**

After the offset calculation has been successfully completed, the frame components of the selected ZO are re-displayed so that the calculated center of the measured edge distance after activating the selected ZO assumes the required setpoint position (e.g. X0 in the corrected workpiece coordinate system.



#### Notes

- The rejection, repetition, and end of measurement are described in Subsection 4.3.7.
- Support for the user is featured for activation of the zero offset determined in JOG mode and for necessary probe reorientation. This is shown right after the measurement and is detailed in Subsection 4.3.9.

#### 4.3.4 Measuring a corner (from measuring cycles SW 6.3)



#### Function

After selecting "corner", a selection display is shown containing the following options:

- "right-angled corner"
- "any corner"

Machine CH	HAN1 set	Jog	MPF0 Program a	aborted			1	
Work	Position	Repo	os offset	ROV Master spind	le S1	×	ور مالح م	→ Right-angled corner → Any corner
X Y Z A C	-333.3333 -111.1111 222.2222 242.5183 357.6571	mm mm deg deg	0.0000 0.0000 0.0000 0.0000 0.0000	Act. Set Pos. Power	0.000 rpm 0.000 rpm 0 deg. 100.0 %	]		
							~~	Back → (to selection Measure workpiece)

#### Precondition

The workpiece probe is the active tool with activated tool offsets in the spindle and is already calibrated (refer to Subsection 4.3.2).



#### Note:

Set the feedrate override to the same value as for calibration!

#### **Right-angled corner**

This permits measurement of a right-angled corner of a workpiece as the reference point in the axes of the working plane and setting it as the ZO (translation and rotation). 3 measuring points are required: P1, P2, and P3. The 1st axis of the working plane functions as the reference axis (for G17: X axis). Edge P1, P2 functions as the reference edge on the workpiece. This edge is aligned parallel to the reference axis (G17: X axis).

#### Approaching the workpiece

Position the probe on the corner, first at P1 and after completion of this measurement at P2, and after completion of this measurement at P3 - to measuring depth in each case.

		Aut. override in work offset
ork offs	G54	
10.8180 m	nm Corner	Out.corner
100.0000 m	nm	Pos. 1
-105.0000 m	nm X0	0.0000
) dur. measur.: (	G500 Y0	0.0000
eas X0		
eas YO		
	ork offs 10.8180 n 100.0000 n -105.0000 n O dur. measur.: eas X0 eas Y0	ork offs G54 10.8180 mm Corner 100.0000 mm -105.0000 mm X0 O dur. measur.: G500 Y0 eas X0 eas Y0

#### Enter details in input form

- Select of the ZO
- Select outside or inside corner
- Select the position of the corner
- Enter the set position required for the reference point P0 (corner) for the selected ZO in both axes.

	¢	•
--	---	---

Measurement is performed automatically at **P1** with the set measuring feed as soon as you press "NC Start". Measurement is performed in the axis and axis direction determined by the selected position. After successful measurement, the previously inactive softkey "P1 stored" is activated and the coordinates of the 1st measuring point P1 is stored internally.

After manual positioning in front of the 2nd measuring point **P2**, measuring is performed automatically at this measuring point on pressing "NC start". Proceed in the same way for measuring point **P3**.

If all of the measuring points have been successfully completed and all "Px saved" softkeys activated, then a vertical "calculate" softkey is displayed. After this softkey has been pressed, the corner coordinates P0 and the offset are calculated. The offset is activated at the same time if it is the active ZO.

#### **Display and correction**

After successful calculation and offset application, the coordinates of the corner point determined are displayed in the WCS active during measurement. Display of the translation frame components of the selected ZO is updated. If you have selected "Just measure", the corner point determined and the angle of the reference axis will only be displayed.



- The rejection, repetition, and end of measurement are described in Subsection 4.3.7.
- A support function is available to the user when activating the determined ZO correction in the JOG mode. This is shown right after the measurement and is detailed in Subsection 4.3.9.

#### Any corner

This permits measurement of a corner of a workpiece as the reference point in the axes of the working plane and setting it as the ZO (translation and rotation). The corner does not have to be right-angled. 4 measuring points are required: P1, P2, P3, and P4. The 1st axis of the working plane functions as the reference axis (for G17: X axis). Edge P1, P2 functions as the reference edge on the workpiece. This edge is aligned parallel to the reference axis (G17: X axis).

#### Approaching the workpiece

Position the probe on the corner, first at P1 and after completion of this measurement at P2, and after completion of this measurement at P3, etc.

Measure any corner		Aut. ove	rride in work offset
ŧΥ	Work offs	G54	
T T	X 10.8180 mm	Corner	Out.corner
	Y 100.0000 mm		Pos. 1
	Z -105.0000 mm	X0	0.0000
P3	WO dur. measur.: G500	YO	0.0000
P4	meas X0		
	meas Y0		
	α		

#### Enter details in input form

- Select of the ZO
- Select outside or inside corner
- Select the position of the corner
- Enter the set position required for the reference point (corner) for the selected ZO in both axes.



Measurement is performed automatically at **P1** with the set measuring feed as soon as you press "NC Start". This is the same sequence as for "right-angled corner" but with 4 measuring points.

#### Measuring in JOG **4.3 Workpiece measurement (from measuring cycles SW 6.3)**

#### 4.3.5 Measuring a pocket, hole or spigot (from measuring cycles SW 6.3)



#### Function, pocket/hole

After selecting "pocket/hole", a selection display is shown containing the additional following options:

- "Rectangular pocket"
- "1 hole"
- "2 holes"
- "3 holes"
- "4 holes"





#### Function, spigot

After selecting "spigot", a selection display is shown containing the additional following options:

- "Rectangular spigot"
- "1 circular spigot"
- "2 circular spigots"
- "3 circular spigots"
- "4 circular spigots"

10.04



#### Precondition

The workpiece probe is the active tool with activated tool offsets in the spindle and is already calibrated (refer to Subsection 4.3.2).

#### Note:

Set the feedrate override to the same value as for calibration!

#### Rectangular pocket or 1 hole or 1 spigot

Depending on selection, the **center** 

- of a paraxial rectangular pocket or
- a hole or
- a paraxial rectangular spigot or
- a circular spigot

can be measured on the workpiece as the reference point and as the ZO (translation).

Measurement is performed in both axes of the working plane.

#### Approaching the workpiece

Position the probe roughly at measuring depth in the center of the pocket/hole, or in the case of spigots roughly in the center above the spigot.



#### Enter details in input form

- Select of the ZO
- For rectangular pocket, rectangular spigot: Enter approximate length L (1st axis of the working plane) and width W (2nd axis of the working plane)
- For hole, circular spigot: Enter the approximate diameter
- For spigots only: Enter infeed value DZ (measuring depth as from starting position, value >0)
- For hole, circular spigot only: Enter the probing angle if measurement is not to be paraxial.
- Enter the set position required for the reference point P0 (center) for the selected ZO in both axes.

## $\mathbf{\Phi}$

10.04

Measurement is performed automatically as soon as you press "NC Start". The probe probes 4 points of the inside and outside wall in succession.

The offset is then automatically calculated. If the selected offset is the active ZO, then this is automatically activated.

#### **Display and correction**

After completion of calculation and offset, the diameter or width/length and the coordinates of the center P0 are displayed.

An offset is applied in the translation components of the selected ZO in both axes.



#### Notes

- The rejection, repetition, and end of measurement are described in Subsection 4.3.7.
- Support for the user is featured for activation of the • zero offset determined in JOG mode and for necessary probe reorientation. This is shown right after the measurement and is detailed in Subsection 4.3.9.

#### 2 holes or 2 circular spigots

This permits calculation of the basic rotation (rotation in the working plane) of the clamped workpiece.

- This enables orientation by
- "coordinate rotation" •
- "or rotation of the workpiece with a rotary table (rotary axis).

For an angle offset by "coordinate rotation", it is possible to set the reference point for the calculated center of the 1st hole/spigot.

Measurement is performed in both axes of the working plane.

#### Approaching the workpiece

Position the probe roughly in the center of the 1st hole/spigot P1, after completion of this measurement in the rough center of the 2nd hole/spigot P2, for a hole to measuring depth in each case, for a spigot above the spigot.

Measure 2 holes		Aut. over	ride in <del>w</del> ork offset
λY	Work offs	G54	
	X 0.0000 deg	Diameter	0.0000
P1 P2	Y 0.0000 deg	Angle offset	Coord. rot.
	Z 0.0000 deg	Spec. angle	0.0000
	WO dur. measur.: G500	Set P1	Yes
	α	X1	0.0000
	measX0	Y1	0.0000
►X	measY0		
Meas. 2 circ. spigots		Aut. over	ride in work offset
ιY	Work offs	G54	
	X 0.0000 deg	Diameter	20.0000
P1 +	Y 0.0000 deg	DZ	7.0000
	Z 0.0000 deg	Angle offset	Coord. rot.
	WO dur. measur.: G500	Spec. angle	0.0000
	α	Set P1	Yes
	measX0	X1	0.0000

#### Enter details in input form

- Select of the ZO
- Enter the approximate diameter (hole/spigot). This must be selected so that it is applicable for all holes/spigots.
- For spigots only: Enter infeed value DZ (measuring depth as from starting position, value >0)
- Select the required angle offset:
   "Coordinate rotation" or axis name of the rotary axis
- Under the "set angle" input, an orientation deviating from 0 degrees with respect to the 1st axis of the working plane (e.g., G17: X axis) is possible.
- Only for offset type "coordinate rotation" and if "set P1" is selected:

Enter the set position required for the reference point P1 (center of the 1st hole) for the selected ZO in both axes.



Measurement is performed automatically at **P1** with the set measuring feed as soon as you press "NC Start". The probe probes 4 points of the inside and outside wall in succession. After successful measurement, the center is stored internally and the "P1 stored" softkey that was previously switched inactive is activated.

After manual positioning at point **P2** and with softkey "P1 stored" switched active, pressing "NC start" starts measurement of the 2nd hole/spigot.

10.04

After successful measurement, softkey "P2 stored" is activated, and the 2nd center is stored. If all of the measuring points have been successfully completed and all "Px saved" softkeys activated, then a vertical "calculate" softkey is displayed. After this softkey is pressed, "alpha" is calculated.

When the "coordinate rotation" angular offset (correction) is selected, then the offset is also activated if it is the active ZO.

When the "rotary axis" angular offset is selected (name of the rotary axis), a help function for the user is displayed to activate the offset (refer to Subsection 4.3.9).

#### **Display and correction**

After offset calculation has been completed, the frame components of the selected ZO are redisplayed along with the measured "alpha" angle and the coordinates of the reference point P1.

ZO correction is effected in such a way that the angle determined is in the required set rotation and point P1 is in the set position in the corrected workpiece coordinate system after activation of the selected ZO.



#### Notes

- The rejection, repetition, and end of measurement are described in Subsection 4.3.7.
- **Support** for the user is featured for **activation of the zero offset** determined in JOG mode and for necessary probe reorientation. This is shown right after the measurement and is detailed in Subsection 4.3.9.

#### 3 holes or 3 circular spigots

This permits calculation of the reference point P0 and the "alpha" rotation of the clamped workpiece.

This enables orientation by coordinate rotation and setting the center P0 of the partial circle on which the 3 holes/spigots are located as the reference point.

Measurement is performed in both axes of the working plane.

#### Approaching the workpiece

Position the probe roughly in the center of the 1st hole/spigot P1, after completion of this measurement in the rough center of the 2nd hole/spigot P2, after completion of this measurement in the rough center of the 3rd hole/spigot, for a hole to measuring depth in each case, for a spigot above the spigot.



Meas. 3 circ. spigots		Aut. over	ride in work offset
±Υ	Work offs	G54	
Т 🔪 Р1	X 0.0000 deg	Diameter	50.0000
	Y 0.0000 deg	DZ	7.0000
	Z 0.0000 deg	×0	0.000
	WO dur. measur.: G500	Y0	0.0000
	α	Angle offset	Yes
	meas X0:	Spec. angle	0.0000
⊢ FJ ►X	meas Y0:		

#### Enter details in input form

- Select of the ZO
- Enter the approximate diameter (hole/spigot). This must be selected so that it is applicable for all holes/spigots.
- For spigots only: Enter infeed value DZ (measuring depth as from starting position, value >0)
- Enter the set position required for the reference point P0 (center of the partial circle of the 3rd hole/spigot) for the selected ZO in both axes.
- If angle offset is selected: Under the "set angle" input, an orientation deviating from 0 degrees with respect to the 1st axis of the working plane (e.g., G17: X axis) is possible.

# $\mathbf{0}$

Measurement is performed automatically at **P1** with the set measuring feed as soon as you press "NC Start". The probe probes 4 points of the inside and outside wall in succession. After successful measurement, the measured value is stored internally and the "P1 stored" softkey that was previously switched inactive is activated.

After manual positioning at the next center **P2** and with softkey "P1 stored" switched active, pressing "NC start" starts measurement for this hole/spigot.

The rest of the sequence is as for P1.

If all of the measuring points have been successfully completed and all "Px saved" softkeys activated, then a vertical "calculate" softkey is displayed. After this softkey is pressed, P0 and "alpha" are calculated. 10.04



When the "coordinate rotation" angular offset (correction) is selected, then the offset is also activated if it is the active ZO.

When the "rotary axis" angular offset is selected (name of the rotary axis), a help function for the user is displayed to activate the offset (refer to Subsection 4.3.9).

#### **Display and correction**

After offset calculation has been completed, the frame components of the selected ZO are redisplayed along with the measured "alpha" angle and the coordinates of the reference point P0. ZO correction is effected in such a way that the "alpha" angle determined and point P0 are at the desired setpoints (translation and rotation) in the offset workpiece coordinate system after activation of the selected ZO.

#### Notes

- The rejection, repetition, and end of measurement are described in Subsection 4.3.7.
- Support for the user is featured for activation of the zero offset determined in JOG mode and for necessary probe reorientation. This is shown right after the measurement and is detailed in Subsection 4.3.9.

#### 4 holes or 4 circular spigots

This permits calculation of the reference point P0 and the "alpha" rotation of the clamped workpiece.

This permits orientation by coordinate rotation and setting point P0 (intersection point of the 2 straight lines, whose hole centers are connected diagonally) as the reference point.

Measurement is performed in both axes of the working plane.

#### Approaching the workpiece

Position the probe roughly in the center of the 1st hole/spigot P1, after completion of this measurement in the rough center of the 2nd hole/spigot P2, etc., for a hole to measuring depth in each case, for a spigot above the spigot.

Manager dialate		A	
Measure 4 noies		Aut. over	ride in work offset
λY	Work offs	G54	
T • .	X 0.0000 deg	Diameter	50.0000
	Y 0.0000 deg	X0	0.000
	Z 0.0000 deg	Y0	0.000
Potro	WO dur. measur.: G500	Angle offset	Yes
P1 + ( ) = ( ) + "	α	Spec. angle	0.000
P2	measX0		
►×	measY0		
Meas. 4 circ. spigots		Aut. over	ride in work offset
Meas. 4 circ. spigots	Work offs	Aut. over G54	ride in work offset
Meas. 4 circ. spigots	Work offs X 0.0000 dea	Aut. over G54 Diameter	ride in <del>w</del> ork offset Official 50.0000
Meas. 4 circ. spigots	Work offs X 0.0000 deg Y 0.0000 deg	Aut. over <mark>G54</mark> Diameter DZ	ride in work offset 50.0000 7.0000
Meas. 4 circ. spigots	Work offs X 0.0000 deg Y 0.0000 deg Z 0.0000 deg	Aut. over G54 Diameter DZ X0	ride in work offset
Y P4 P4 P4 P4 P4 P4 P4 P3	Work offs X 0.0000 deg Y 0.0000 deg Z 0.0000 deg W0 dur measur : G500	Aut. over G54 Diameter DZ X0 Y0	ride in work offset 50.0000 7.0000 0.0000 0.0000
Meas. 4 circ. spigots Y P4 $+p_0$ $p_1$ $+p_0$ $+p_0$ $+p_1$ $\alpha$	Work offs X 0.0000 deg Y 0.0000 deg Z 0.0000 deg WO dur. measur.: G500	Aut. over G54 Diameter DZ X0 Y0 Angle offset	ride in work offset 50.0000 7.0000 0.0000 0.0000
Meas. 4 circ. spigots Y P4 $+p_0$ $p_1$ $+p_0$ $+p_1$ $\alpha$	Work offs X 0.0000 deg Y 0.0000 deg Z 0.0000 deg WO dur. measur.: G500 α	Aut. over G54 Diameter DZ X0 Y0 Angle offset	ride in work offset 50.0000 7.0000 0.0000 0.0000 Yes 0.0000
Meas. 4 circ. spigots Y P4 p1 p2 p2 p2	Work offs X 0.0000 deg Y 0.0000 deg Z 0.0000 deg WO dur. measur.: G500 a measX0	Aut. over G54 Diameter DZ X0 Y0 Angle offset Spec. angle	ride in work offset 50.0000 7.0000 0.0000 0.0000 Yes 0.0000

#### Enter details in input form

- Select of the ZO
- Enter the approximate diameter (hole/spigot). This must be selected so that it is applicable for all holes/spigots.
- For spigots only: Enter infeed value (DZ) (measuring depth as from starting position, value >0)
- Enter the set position required for the reference point P0 (intersection point of the diagonals) for the selected ZO in both axes.
- If angle offset is selected: Under the "set angle" input, an orientation deviating from 0 degrees with respect to the 1st axis of the working plane (e.g., G17: X axis) is possible.

## $\mathbf{\Phi}$

Measurement is performed automatically at **P1** with the set measuring feed as soon as you press "NC Start". The probe probes 4 points of the inside and outside wall in succession. After successful measurement, the measured value is stored internally and the "P1 stored" softkey that was previously switched inactive is activated.

After manual positioning at the next center **P2** and with softkey "P1 stored" switched active, pressing "NC start" starts measurement for this hole/spigot.

The rest of the sequence is as for P1. If all of the measuring points have been successfully completed and all "Px saved" softkeys activated, then a vertical "calculate" softkey is displayed. After this softkey is pressed, P0 and "alpha" are calculated.

4-130

10.04



When the "coordinate rotation" angular offset (correction) is selected, then the offset is also activated if it is the active ZO.

When the "rotary axis" angular offset is selected (name of the rotary axis), a help function for the user is displayed to activate the offset (refer to Subsection 4.3.9).

#### **Display and correction**

After offset calculation has been completed, the frame components of the selected ZO are redisplayed along with the measured "alpha" angle and the coordinates of the reference point P0.

ZO correction is effected in such a way that the "alpha" angle determined and point P0 are at the desired setpoints (translation and rotation) in the offset workpiece coordinate system after activation of the selected ZO.

## Notes

- The rejection, repetition, and end of measurement are described in Subsection 4.3.7.
- **Support** for the user is featured for **activation of the zero offset** determined in JOG mode and for necessary probe reorientation. This is shown right after the measurement and is detailed in Subsection 4.3.9.



### 4.3.6 Orienting a plane (from measuring cycles SW 6.3)

### Function

After selection of "orient plane", the following display is shown:

Machine	CHAN1	Jog	MPF0				
🥢 Channel	reset		Program a	aborted			Altor
				ROV			native
Work	Position	Repos	offset	Master sp	indle S1	X	WO
×	0.0000 mi	n	0.0000	Act.	0.000 r	pm	
Y	0.0000 mi	n	0.0000	Set	0.000 r	pm	P1 stored
z	0.0000 mi	n	0.0000	Pos.	0 c	leg.	
A	82.2209 de	g	0.0000		100.0 %	6	P2 stored
с	0.0000 de	g	0.0000	Power	0%		stored
Align plane				<u>_</u>	Aut. override	in work offset	P3 stored
±Υ	Work offs	;		G54			
PI	P3 p2 P3 p2 X WO dur. 1 β	0.0000 de 0.0000 de 0.0000 de neasur.:	9g 9g 9g G500				
$\overline{}$							

#### Precondition

The workpiece probe is the active tool with activated tool offsets in the spindle and is already calibrated (refer to Subsection 4.3.2).



#### Note:

Set the feedrate override to the same value as for calibration!

#### Measuring a plane that is oblique in space

That permits measurement of a workpiece plane that is oblique in space and determination of "alpha" and "beta" rotation.

That enables orthogonal orientation of the infeed axis with respect to this plane by coordinate rotation.

Measurement is performed in the infeed axis (3rd axis) at 3 different points in the working plane.

#### Approaching the workpiece

Position the probe above measuring point P1, after completion of this measurement above measuring point P2, after completion of this measurement above measuring point P3.

#### Enter details in input form

Select of the ZO



Measurement is performed automatically at **P1** with the set measuring feed as soon as you press "NC Start". After successful measurement, the measured value is stored internally and the "P1 stored" softkey that was previously switched inactive is activated.

After manual positioning at the next measuring point **P2** and with softkey "P1 stored" switched active, pressing "NC start" starts measurement for this measuring point.

The rest of the sequence is as for P1.

If all of the measuring points have been successfully completed and all "Px saved" softkeys activated, then a vertical "calculate" softkey is displayed. After this softkey is pressed, "alpha" and "beta" are calculated.

A support function for the user is displayed to activate the offset (refer to Subsection 4.3.9).

#### **Display and correction**

After offset calculation has been completed, the frame components (rotation) of the selected ZO are redisplayed along with the measured "alpha" and "beta" angle. ZO correction is performed in such a way that the plane with the calculated points P1 to P3 after activation of the selected ZO is parallel to the new working plane.

### Notes

- The rejection, repetition, and end of measurement are described in Subsection 4.3.7.
- Support for the user is featured for activation of the zero offset determined in JOG mode and for necessary probe reorientation. This is shown right after the measurement and is detailed in Subsection 4.3.9.



### Function

#### **Rejection and repetition of measurements**

The last measurement (Px) can be declared invalid any number of times by pressing the assigned softkey "Px stored". The softkey is then deactivated (grayed out). Pressing "NC start" again repeats this measurement and the "Px stored" softkey is then reactivated (black text).

Machine	CHAN1	Jog	MPF0			
🥢 Channel	reset		Program a	borted		
			-	ROV		Alter- native
Work	Position	Repo	s offset	Master spindle	S1 🕅	l wo
×	0.0000 m	m	0.0000	Act.	0.000 rpm	
Y	0.0000 m	Im	0.0000	Set	0.000 rpm	P1 stored
7	0.0000 m	m	0 0000	Pos.	0 deq.	Stored
-	0.0000 11		0.0000		-	
A	82.2209 d	eg	0.0000	_	100.0 10	stored
С	0.0000 d	eg	0.0000	Power	0%	
Meas. 4 cir	rc. spigots			Aut.	override in work offse	t stored
. •	Work off	s		G54		
l † .	×	0.0000	dea	Diameter	50.0000	P4
	Y Y	0.0000	deq	DZ	7.0000	stored
	$P^3$ z	0.0000	deg	×0	0.0000	
	WO dur.	measur.:	G500	Y0	0.0000	
P1++				Angle offset	Yes	
L-	P2 measX0			Spec. angle	0.0000	
	— → x measY0					

Example of vertical softkeys P1 to P4 - inactive:

#### End of measurements

If all "Px saved" softkeys are active, then a vertical "calculate" softkey is displayed.

When this softkey is pressed, the measuring points are finally confirmed/acknowledged and the translatory and rotary offset values are calculated. The result is immediately entered into the preselected "zero offset". If this is identical with "ZO when measuring", the offset values become immediately effective. This means that the measurements have been completed.

For offsets/corrections, that required e.g. a new alignment/ orientation of the measuring probe or the rotary table, the user is provided with a **support function** to activate the determined ZO offset in the JOG mode. This is displayed after pressing the "calculate" softkey and is described in detail in Subsection 4.3.9. 10.04



4

10.04

The "Px stored" softkeys are then switched inactive and it is possible to start new measurement. The number of softkeys Px is determined with the measurement task.

When the input screen form is exited, a measurement is also completed/interrupted.



#### Notes

A mode change is only possible after exiting the operator area "measuring in JOG".

Example for vertical softkeys P1 to P4 active, the "calculate" softkey is displayed:

Machine	CHAN1	Jog	MPF0			
🥢 Channel	reset		Program a	borted		Alter-
				ROV		native
Work	Position	Repos	s offset	Master spindle S	n 🕺	wo
×	0.0000	mm	0.0000	Act. 0.0	100 rpm	
Y	0.0000	mm	0.0000	Set 0.0	mq1 001	P1 stored
7	0 0000	mm	0 0000	Pos.	0 deq.	stored
-	0.0000			10	0.0 %	D2
A .	82.2209	deg	0.0000			stored
С	0.0000	deg	0.0000	Power	0%	
Meas. 4 cir	c. spigots			Aut. ove	rride in work offset	P3 stored
	Work o	offe		C54		Stored
ļ †Υ ,	X	0.0000 (	dea	Diameter	50.0000	D/
ني ا	Y IN	0.0000 0	deg	DZ	7.0000	stored
P4	$P^3 z$	0.0000 (	deg	X0	0.0000	
	WO du	r. measur.:	G500	Y0	0.0000	
	a a			Angle offset	Yes	
	P2 measX	:0		Spec. angle	0.0000	
	→X measY	<i>'</i> 0				
$\frown$						

### 4.3.8 Information regarding cascaded measurements

### Function

Often, it is not possible to completely set-up a workpiece using just one single measurement; instead, a series of measurements has to be made. This results in certain interdependencies regarding the selected measuring series.

Example:

- Orienting an oblique plane
- Orienting an edge, reference in X axis or Orienting an edge, reference in Y axis
- Setting a reference point in X, Y, Z

A zero offset can be applied after every measurement. The "measuring in JOG" function supports activation of another or changed ZO in JOG mode by displaying an additional activation screen form with queries after measurement.

If so selected, the new zero offsets are activated and the probe possibly also re-oriented.

The changed zero offset is offered as the preferred setting for the next measurement. This setting can therefore deviate from the basic setting \_JM\_I[4] in data block GUD6.

Function

#### 4.3.9 Support of set-up in JOG - after measurement

After an offset has been applied in the selected zero offset by the measuring function, this ZO is activated in JOG mode and the probe possibly oriented in the new WCS, e.g. for the following measurements.

For "measuring in JOG", an offset in the active ZO (the ZO that applies during measurement) is preset. This ZO is updated after measurement and is therefore active.

This is sufficient in cases where translation offset of the workpiece coordinate system and / or rotation about the infeed axis has occurred.

Positioning of the rotary axes for orienting the workpiece after measurement or orienting the probe because of rotation about the axes of the working plane has so far only been possible in MDA or AUTOMATIC mode.

To support set-up in JOG after measurement, an additional **activation screen form** is automatically shown **afterward**, if at least one of the following reasons applies:

- Offset in a ZO not active during measurement
- An offset causes rotation of the WCS about at least one **plane axis** following by a necessary reorientation of the probe
- An offset demands **rotary axis positioning** for workpiece alignment without affecting the WCS

The system detects whether the measuring probe must be aligned/oriented using the function "swiveling" or the 5-axis transformation (TRAORI). The relevant screen form is displayed. If axial movements are required, the user is informed about this in the screen form. It is possible to decide whether pressing "NC start" will start the swivel cycle for orienting the probe or whether the selected rotary axis will move to orient the workpiece.

#### Notes

If the function "swiveling" or the 5-axis transformation (TRAORI) is used to align/orient the measuring probe, then this must also be set-up in the machine!

Please carefully observe the information provided by the machinery manufacturer!

#### Example 1

4 holes were measured. An offset is applied in a ZO that was not active during measurement. Reorientation of the probe is not necessary. The following selection display appears with information:

Machine CHAN1 Jog		MPF0				
// Channel reset			Program a	Program aborted		
				ROV		native
Work	Position	Repos	s offset	Master spindle	S1 🛛 🕅	wo
×	1.3232 n	ım	0.0000	Act.	0.000 rpm	
Y	61.6800 n	ım	0.0000	Set	0.000 rpm	
z	-55.0000 n	ım	0.0000	Pos.	11 deg.	
A	11.0000 d	eg	0.0000		100.0 %	
с	0.0000 d	eq	0.0000	Power	0%	
Measure 4	holes			You have correct	ted in an inactive WO!	
÷Υ	Work of	s G54	A	ctivate this		
ΙŤ΄ κ	x	0.0000	deg we	ork offset?	Yes	
	T m Y	0.0000	deg Tl	he work is rotated		
P4	Z Z	8.6263	deg ro	ound the tool axis!		
	WO dur.	measur.:	G500 PI	lease note when		
PITC	a a	1	8.6263 tra	aversing in work!		
	P2 meas XI	1: 2:	2.5557			
	→ X meas Y	): 51	2.9536			
NV korrig	NV korrigiert, zum Aktivieren bitte NC-Start betätigen					

#### Explanation

The display for the measurement function "measure 4 holes" is shown in the left part of the screen form. The display of the measurement result is also shown in the WCS that was active during measurement (G500) and display of the translation values of the offset ZO.

The operator is informed about the selection made in the toggle field (yes) with a text or in an upper image line:

"You have applied an offset in a non-active ZO!".

A message text also appears for the special case:

"The WCS is about the ...".

An additional text is output on the HMI "dialog line with user information and instructions. It provides instructions on action to take:

"Please press NC start to activate!"

After pressing "NC start", the ZO with the applied offset is activated. It is not necessary to traverse axes. The previous measurement selection display is then shown automatically. Measurement can be performed again. To avoid activating the ZO with the applied offset, exit the display by pressing the softkey "<<" (return). 10.04

#### Example 2

10.04

An oblique plane was measured. An offset is applied in a ZO that was not active during measurement. "Swiveling" is active. This requires re-orientation of the probe with respect to the plane via the swivel cycle.

The following selection display appears with information:

Align plane			You have corrected	l in an inactive WO!
Y P1 P1 P2 P2 P2 P1 P2 P2 P1 P3 P2 P2	Work offs X -6.106 Y -14.303 Z 0.000 WO dur. measur.: α β	G54 9 deg 0 deg 0 deg G500 -6.1069 -14.3030	Activate this work offset? Set probe perpen- dicular to plane? Alignment made by swivelling! Retract:	Yes () Yes
∠►X				Z

#### Explanation

The display for the measurement function "orient plane" is shown in the left part of the screen form. The display of the measurement result is also shown in the WCS that was active during measurement and display of the translation values of the offset ZO.

The operator is informed about the selection made in the toggle field 1 (yes) with a text or in an upper image line:

"You have applied an offset in a non-active ZO!" Because of the answer with "yes" in toggle field 2, the text in the upper display line is changed to:

"The new WCS has new orientation!"

With toggle field 3, retraction is supported.

If all screen forms have been filled out, a further text appears on a lower line of the display. It provides instructions on action to take:

"Please press NC start to activate!"

After pressing "NC start", execution is performed according to the entries that the user made in the activation screen form:

- Activating the corrected ZO or
- Retracting the measuring probe and re-aligning/orienting it using the swivel cycle.

The previous measurement selection screen is then re-displayed. After the plane has been successfully aligned/oriented, the workpiece can continue to be gauged with measuring "edge", "holes", "spigots" etc. Refer to Subsection 4.3.8.

If the corrected ZO is not to be activated, exit the screen by pressing the softkey "<<" (return).



#### 4.4 Measuring tools

#### 4.4.1 Overview of function and sequence

# Function

The "tool measurement" function permits the following functions.

- Calibrating the tool probe
- Determining the tool length or radius of milling tools, or tool length of drills and then setting the appropriate offset in the tool offset memory.

The tools are measured on the machine.



#### **General sequence**

When you press softkey "Measure tool", the following selection and more appears on the vertical softkey bar:

Length
Diameter >
Calibrate probe >
<< back

- 1. The measuring variant is selected and the input screen form is filled with values.
- 2. Position the tool near the tool measuring probe with the axis traverse keys.
- 3. "NC start" starts measurement and the remaining sequence with offset entry runs automatically.

#### 4.4.2 Calibrating the tool probe



#### Function

The function "Calibrate tool measuring probe" calculates the current distance (switching points) between the machine zero and the tool probe (calibration relative to machine) using a **calibration tool** and automatically writes them to a data storage area as trigger values. The type 120 (end mill) can be defined as the tool type for the calibration tool. There is no special "calibration

tool" type.





#### Operating sequence

#### Precondition

- The tool probe is mounted in the machining space of the machine (usually on the machine table) and align with the machining axes (machine axes).
- All necessary data of the tool probe (form, dimensions, ...) are entered in the intended variables for "measuring in JOG" E\_MESS\_MT\_... in data block GUD7 (see Section 9.3).
- The precise length 1 and the radius of the calibration tool must be stored in a tool offset data set (geometry).
- The calibration tool is in the spindle and is activated with the offset data set.

#### Approaching the tool measuring probe

Position the calibration tool roughly over the center of the measuring surface of the tool probe.

#### Select the function with softkey



In the input screen form, select the type of calibration by pressing the "alternative" softkey.

Calibrate length only:

Calibrate	lengun	only.	

Tool probe calibr.		
× ×	Length	O

Or

Calibrate length and diameter:



# $\diamond$

On "NC start", calibration runs automatically with the measuring feed entered in \_E\_MESS\_MT\_FM in data block GUD7.

The current distance dimensions between the machine zero and tool probe (trigger points) are determined using the calibration tool and stored in a data area. The tool probe is now prepared for measurement of tools.

### 4.4.3 Measuring milling and drilling tools

## **Function**

This allows you to determine the tool length or radius of milling tools, or tool length of drills and then set the appropriate offset in the tool offset memory.

The tool offset mode for "measure in JOG" can be set via a variable in data block GUD6:

• \_JM\_B[0]=0: Offset in geometry component

• \_JM\_B[0]=1: Offset in wear component For milling tools, cutting edges with a special shape, e.g. rounded cutting edges, can be taken into account with additional entries.

Measurement is performed with a motionless or rotating spindle:

- Radius measurement is performed with a rotation spindle.
- Length measurement is performed with a rotating spindle if the tool radius is greater than the upper radius of the too probe. Otherwise, measurement is with a motionless spindle.

Corresponding to the user specifications regarding the required measuring accuracy, it is possible that extremely small measurement feeds are applied. This is the reason that the tool gauging is automatically sub-divided into two measuring operations (corresponding to the default setting of E\_MESS\_... parameterization in GUD7\_MC).

- The first measurement is always made from a starting position selected by the user with a high measurement velocity. This is used to detect the principle position of the tool to be gauged.
- The 2nd measurement is realized from an optimized position with a short measuring distance and a low measurement feed corresponding to the required measuring accuracy.



The number of probing operations, the required/specified measurement accuracy as well as various limit values can be adapted to the real conditions in the GUD parameters associated with tool measurement (E\_MESS\_...) (refer to Subsection 9.3.2).

The mathematical interrelationship between the measuring accuracy and measurement feed can be taken from Subsection 5.2.1.

#### Preconditions

- The specific GUD parameters used to measure tools are adapted to the real user conditions.
- The reference points are approached.
- The tool probe is functional.
- The tool probe has been calibrated before measurement.
- The tool to be measured is located in the spindle and is activated.
- The tool offset data (length radius) are entered as approximate values and activated.

**Operating sequence** 

•

Measure tool

Diameter

Approaching the tool probe Position the active tool:

length measurement) or

Select the function with softkey

measuring the radius

Over the measuring surface of the tool probe (for

from the side over the measuring probe when



Continue with selection:

The system displays:



Or:

Length

#### The system displays:

Measure tool			
↓ <sup>z</sup>		<mark>Measure length</mark> ∀ 0.0000	€

#### Length measurement of milling tools

If the tool diameter (entered tool radius x 2) is greater than the entered upper diameter of the tool probe, the milling tool is placed in the center of the probe, offset by the tool radius and measured with rotating spindle. Otherwise, the tool is place in the center and measurement is performed with a motionless spindle.


# Measurement of milling tools with special cutting edge

If a cutting edge is rounded, for example, the tool must be measured with a probe offset. This offset (V) must be entered additionally for diameter determination.





#### Enter details in input form

Enter offset (V, positive value) when required.

 $\mathbf{\Phi}$ 

Measurement is performed automatically as soon as you press "NC Start".

The tool offsets "radius" or "length 1" are calculated in and entered in the active tool offset data in accordance with the setting (in geometry or wear).



Notes



5



10.04

# Measuring Cycles for Milling and Machining Centers

General prerequisites	5-148
CYCLE971 tool: Measuring milling tools, drills Measuring and compensation strategy Calibrating a tool probe Calibrating the tool probe automatically (from measuring cycles SW 6.3) Measuring the tool	
CYCLE976 calibrate workpiece probe Calibrating a workpiece probe in a hole of known hole center point Calibrating a workpiece probe in a hole of unknown hole center point Calibration of a workpiece probe on a surface Calibrating a workpiece probe in the applicate determining probe length	5-174 5-177 5-180 5-185 5-188
CYCLE977 workpiece: Measure hole/shaft/groove/web/rectangle parallel to axes Measuring contour elements Measurement and tool offset Measurement and ZO determination	5-191 5-195 5-203 5-207
CYCLE978 workpiece: Measuring a surface parallel to the axis Measuring the surface Measurement and ZO determination Measuring and tool offset	5-211 5-213 5-214 5-217
CYCLE979 workpiece: Measure hole/shaft/groove/rib at an angle Measuring a hole, shaft, groove, web Measuring and tool offset Measurement and ZO determination	5-221 5-223 5-227 5-232
CYCLE998 workpiece: Angle measurement and ZO determination 1-angle measurement 2-angle measurement	5-236 5-239 5-246
CYCLE961 workpiece: Setting up an internal and external corner Function overview CYCLE961 Setting up a corner with definition of distances and angles Setting up a corner with 4 points	
CYCLE997 workpiece: Measuring spheres and ZO determination (from measuring cycles SW 6.3) Function overview CYCLE997 Measurement and ZO determination Programming example CYCLE997 CYCLE119: Arithmetic cycle for calculating position in space	
	General prerequisites CYCLE971 tool: Measuring milling tools, drills Measuring and compensation strategy Calibrating a tool probe Calibrating the tool probe automatically (from measuring cycles SW 6.3) Measuring the tool CYCLE976 calibrate workpiece probe Calibrating a workpiece probe in a hole of known hole center point Calibrating a workpiece probe in a hole of unknown hole center point Calibrating a workpiece probe in a hole of unknown hole center point Calibrating a workpiece probe in the applicate determining probe length CYCLE977 workpiece: Measure hole/shaft/groove/web/rectangle parallel to axes Measuring contour elements Measurement and tool offset Measurement and ZO determination CYCLE978 workpiece: Measuring a surface parallel to the axis Measuring the surface Measuring and tool offset. CYCLE979 workpiece: Measure hole/shaft/groove/rib at an angle Measuring and hole offset. CYCLE979 workpiece: Angle measurement and ZO determination CYCLE979 workpiece: Angle measurement and ZO determination CYCLE998 workpiece: Angle measurement and ZO determination CYCLE998 workpiece: Angle measurement and ZO determination CYCLE998 workpiece: Angle measurement and ZO determination CYCLE961 workpiece: Setting up an internal and external corner Function overview CYCLE961 Setting up a corner with definition of distances and angles. Setting up a corner with 4 points. CYCLE997 workpiece: Measuring spheres and ZO determination (from measuring cycles SW 6.3) Function overview CYCLE997 Measurement and ZO determination Programming example CYCLE997 CYCLE119: Arithmetic cycle for calculating position in space

#### 5.1 General prerequisites

The measuring cycles below are intended for use on milling machines and machining centers. Under certain conditions, workpiece measuring cycles CYCLE976, CYCLE977, and CYCLE978 can also be used on turning machines.

To be able to run the measuring cycles described in this Chapter, the following programs must be stored in the part program memory of the control.

#### Overview of measuring cycles

CYCLE961	Workpiece: Setup inside and outside corner
CYCLE971	Tool measurement for milling tools, calibrate tool probe
CYCLE976	Calibrate workpiece probe in a hole or on a surface
CYCLE977	Paraxial measurement of hole, shaft, groove, web or ZO calculation
CYCLE978	1-point measurement or ZO determination on surface
CYCLE979	Measurement of hole, shaft, groove, web, or ZO determination at an angle
CYCLE997	Measuring spheres or ZO determination (from measuring cycles SW 6.3)
CYCLE998	Angle measurement (ZO determination only)

#### Overview of the auxiliary programs required

CYCLE100	Log ON
CYCLE101	Log OFF
CYCLE102	Measurement result display selection
CYCLE103	Pre-assignment of input data
CYCLE104	Internal subroutine: Measuring cycle interface
CYCLE105	Generate log contents
CYCLE106	Log sequential controller
CYCLE107	Output of message texts (up to measuring cycles SW 6.2)
CYCLE108	Output of alarm messages (up to measuring cycles SW 6.2)
CYCLE109	Internal subroutine: Data transfer
CYCLE110	Internal subroutine: Plausibility checks
CYCLE111	Internal subroutine: Measuring functions
CYCLE112	Internal subroutine: Measuring functions
CYCLE113	Internal subroutine: Log
CYCLE114	Internal subroutine (tool offset)
CYCLE115	Internal subroutine (zero offset)
CYCLE116	Calculate circle center point
CYCLE118	Format real values
CYCLE119	Internal subroutine: Determine position in space
	(from measuring cycles SW 6.3)
The measuring	avala data ara dafinad in tha data blaaka:

The measuring cycle data are defined in the data blocks:

- GUD5.DEF
- GUD6.DEF

10.04

10.04



# Call and return conditions

The following general call and return conditions must be observed:

- The D offset must be activated with the data of the probe for • workpiece measurement before the cycle is called. Tool type 1x0 or 710 (3D probe) is permitted. Dimension factors <> 1 may be active.
- The workpiece measuring cycles can also be used on turning • machines if the following conditions are fulfilled:
  - The 3rd geometry axis exists.
  - Tool type of probe 5xy with cutting edge position 5 to 8
  - The tool length compensation is specific to turning machines (SD TOOL LENGHT TYPE=2)
  - If cutting edge position 5 or 7 is used, measurement is performed in the G17 plane If cutting edge position 6 or 8 is used, measurement is performed in the G19 plane
- Coordinate rotation is permitted for workpiece measuring . cycles.
- Mirroring for the workpiece measuring cycles is permitted with the exception of calibration (condition: MD 10610=0).
- When using a multidirectional probe the best measurement • results are achieved if, during calibration and measurement, the probe in the spindle is mechanically oriented to have one and the same point on the probe ball point, for example, in the + direction of the abscissa (+X with active G17) in the active workpiece coordinate system.
- The G functions active before the measuring cycle is • called remain active after the measuring cycle call, even if they have been changed inside the measuring cycle.
- Measurements must always be performed under the • same conditions as applied when the probe was calibrated.

Measuring cycles from measuring cycles SW 6.2 and higher can only be used with NCK-SW 6.3 and higher.



# 5.2 CYCLE971 tool: Measuring milling tools, drills



# Programming

CYCLE971



# Function

Measuring cycle CYCLE971 implements:

- Calibration of a tool probe
- Measurement of the tool length with motionless or rotating spindle for drills and milling tools
- Measure tool radius with motionless and rotating spindle for milling tools





#### **Measuring variants**

Measuring cycle CYCLE971 permits the following measuring variants which are specified via parameter

_MVAR.	
Value	Meaning
0	Calibrate tool probe (machine-related)
1	Measure tool with motionless spindle (length or radius, machine related)
2	Measure tool with rotating spindle (length or radius, machine related)
10000	Calibrate tool probes incrementally (machine-related)
10	Calibrate tool probes (workpiece-related)
11	Measure tool with motionless spindle (length or radius, workpiece related)
12	Measure tool with rotating spindle (length or radius, workpiece related)
10010	Calibrate tool probes incrementally (workpiece-related)
1) from mea	is. cycles SW 6.3

# **Result parameters**

The measuring cycle CYCLE971 returns the following values in the data block GUD5 for the measuring

variant calibration:

_OVR [8]	REAL	Trigger point in minus direction, actual value of 1st geometry axis
_OVR [10]	REAL	Trigger point in plus direction, actual value of 1st geometry axis
_OVR [12]	REAL	Trigger point in minus direction, actual value of 2nd geometry axis
_OVR [14]	REAL	Trigger point in plus direction, actual value of 2nd geometry axis
_OVR [16]	REAL	Trigger point in minus direction, actual value of 3rd geometry axis
_OVR [18]	REAL	Trigger point in plus direction, actual value of 3rd geometry axis
_OVR [9]	REAL	Trigger point in minus direction, difference of 1st geometry axis
_OVR [11]	REAL	Trigger point in plus direction, difference of 1st geometry axis



_OVR [13]	REAL	Trigger point in minus direction, difference of 2nd geometry axis
_OVR [15]	REAL	Trigger point in plus direction, difference of 2nd geometry axis
_OVR [17]	REAL	Trigger point in minus direction, difference of 3rd geometry axis
_OVR [19]	REAL	Trigger point in plus direction, difference of 3rd geometry axis
_OVR [27]	REAL	Zero offset area
_OVR [28]	REAL	Safe area
_OVI [2]	INTEGER	Measuring cycle number
_OVI [3]	INTEGER	Measuring variant
_OVI [5]	INTEGER	Probe number
_OVI [9]	INTEGER	Alarm number

# Measuring cycle CYCLE971 returns the following values

in the data b	lock GUD5 fo	r tool measurement:
_OVR [8]	REAL	Actual value length L1
_OVR [10]	REAL	Actual value radius R
_OVR [9]	REAL	Difference length L1
_OVR [11]	REAL	Difference radius R
_OVR [27]	REAL	Zero offset area
_OVR [28]	REAL	Safe area
_OVR [29]	REAL	Permissible dimensional deviation
_OVR [30]	REAL	Empirical value
_OVI [0]	INTEGER	D number
_OVI [2]	INTEGER	Measuring cycle number
_OVI [3]	INTEGER	Measuring variant
_OVI [5]	INTEGER	Probe number
_OVI [7]	INTEGER	Number of empirical value memory
_OVI [8]	INTEGER	T number
_OVI [9]	INTEGER	Alarm number

5

#### 5.2.1 Measuring and compensation strategy

#### **Measuring strategy**

#### Prepositioning the tool

The tool must be aligned vertically with the probe before the measuring cycle is called. Tool axis parallel to center line of probe.

It must be prepositioned in such as way that collision-free approach to the probe is possible. First the measuring cycle generates traverse paths to the position where measuring starts with a reduced rapid traverse velocity (\_SPEED[0]), or with active collision monitoring at the position feedrate set in \_SPEED[1] or \_SPEED[2].

#### Tool measurement with motionless spindle

Before the cycle call for measurement of **milling tools** the tool and spindle must be moved such that the selected cutting edge can be measured (length or radius).

The measurement feedrate is defined in \_VMS.

#### Tool measurement with rotating spindle

Typically, measurements of the radius of **milling tools** are executed with rotating spindle, that is the largest edge determines the measuring result. Length measurement of milling tools with rotating spindle might also be practical.

Points to bear in mind:

- Is the tool probe permissible for measuring with rotating spindle with length and/or radius calculation? (Manufacturer documentation)
- Permissible peripheral speed for the tool to be measured.
- Maximum permissible speed.
- Maximum permissible feedrate for probing.
- Minimum feedrate for probing.
- Selection of the rotation direction depending on the cutting edge geometry to prevent hard impacts when probing.
- Required measuring accuracy.







When measuring with rotating tool the relation between measuring feedrate and speed must be taken into account. One cutting edge is taken into account. With multiple cutters the longest edge is used for the measuring result.

The following connections have to be taken into account:

$$n = \frac{S}{2 \cdot \pi \cdot r \cdot 0.001}$$
  
F = n · A

Basic system

		<u>Metric</u>	<u>inch</u>
n	Speed	rpm	rpm
S	Max. permissible peripheral speed	m/min	feet/min
r	Tool radius	mm	inch
F	Measuring feedrate	mm/min	inch/min
$\Delta$	Measuring accuracy	mm	inch

#### Example:

Given a peripheral speed of S = 90m/min, speeds of n = 2865 to 143 rpm will result for milling tools with a radius of r = 5 to 100 mm. Given a specified measuring accuracy of  $\Delta$  = 0.005 mm

measuring feedrates of

F= 14 mm/min to F= 0.7 mm/min will result.



# **Compensation strategy**

The tool measuring cycle is provided for various applications:

# • First-time measurement of a tool ( CHBIT[3]=0):

The tool offset values in geometry and wear are replaced.

Compensation is written in the geometry component of length or radius. The wear component is reset.

# Remeasurement of a tool

(\_CHBIT[3]=1):

The resulting difference is calculated into the wear component (radius or length).

Further, for tool measurement, the measured values can be corrected by empirical values.





Compensation of length 1 or the tool radius only occurs in **tool measurement** if the measured difference lies in the tolerance band between \_TZL and \_TDIF!



Compensation of the tool probe trigger points \_TP[] and. \_TPW[] only occurs when the tool probe is **calibrated** if the measured difference lies in the tolerance band between \_TZL and \_TSA!

# Compensation with correction table when measuring with rotating spindle

When measuring tools with a rotating spindle, the measuring precision can be compensated for by additional compensation values during measurement of the cutter radius or cutter length. These compensation values are stored in tables in the dependency peripheral speed / cutter radius. Users can also create their own compensation values in dedicated tables in data block GUD6.

This offset is activated with variable of data type INTEGER \_**MT\_COMP** >0.

\_MT\_COMP= 0: No compensation

_MT_COMP= 1:	automatic compensation, i.e. internal
	compensation when TT130 is used
	(Heidenhain) or TS27R (Renishaw)

\_MT\_COMP= 2: compensation with user-defined compensation data, i.e. **even** if (Heidenhain) or Renishaw specified (alternate probe)

With the variables of data type INTEGER\_**TP\_CF** it is possible to activate the pre-prepared compensation tables of several tool probe models.

\_TP\_CF= 0: No status

\_TP\_CF= 1: TT130 (Heidenhain)

\_TP\_CF= 2: TS27R (Renishaw)

The user can enter his own compensation values in two arrays of data type REAL:

\_MT\_EC\_R[6,5] for radius measurements and \_MT\_EC\_L[6,5] for length measurement.

10.04

#### Structure of user arrays

_MT_EC_R	m=0	m=1	m=2	m=3	m=4
_MT_EC_L					
[n,m]					
n=0	0	1. Radius	2. Radius	3. Radius	4. Radius
n=1	1. Peripheral	Compensation	Compensation	Compensation	Compensation
	speed	value for 1st	value for 2nd	value for 3rd	value for 4th
		radius/	radius/	radius/	radius/
		1. Peripheral	1. Peripheral	1. Peripheral	1. Peripheral
		speed	speed	speed	speed
n=2	2. Peripheral	Compensation	Compensation	Compensation	Compensation
	speed	value for 1st	value for 2nd	value for 3rd	value for 4th
		radius/	radius/	radius/	radius/
		2. Peripheral	2. Peripheral	2. Peripheral	2. Peripheral
		speed	speed	speed	speed
n=3	3. Peripheral	Compensation	Compensation	Compensation	Compensation
	speed	value for 1st	value for 2nd	value for 3rd	value for 4th
		radius/	radius/	radius/	radius/
		3. Peripheral	3. Peripheral	3. Peripheral	3. Peripheral
		speed	speed	speed	speed
n=4	4. Peripheral	Compensation	Compensation	Compensation	Compensation
	speed	value for 1st	value for 2nd	value for 3rd	value for 4th
		radius/	radius/	radius/	radius/
		4. Peripheral	4. Peripheral	4. Peripheral	4. Peripheral
		speed	speed	speed	speed
n=5	5. Peripheral	Compensation	Compensation	Compensation	Compensation
	speed	value for 1st	value for 2nd	value for 3rd	value for 4th
		radius/	radius/	radius/	radius/
		5. Peripheral	5. Peripheral	5. Peripheral	5. Peripheral
		speed	speed	speed	speed

Units: mm or inch m/min or ft/min for tool radius and compensation value for peripheral speed



#### Function and notes:

In the as-delivered state of the measuring cycles the default setting of the of the arrays is 0. The radii and peripheral speeds must be entered in ascending order.

Access to these arrays in automatic mode is only possible with \_MT\_COMP = 2. In tool measurement with rotating spindle a compensation value from these tables is calculated based on the tool radius of the tool to be measured. The value for the next lowest table peripheral speed and the next lowest table radius are always the values used. In radius measurement the corresponding compensation value in array \_MT\_EC\_R [n,m] is subtracted from the measured tool radius. In length measurement the corresponding compensation value in array \_MT\_EC\_L [n,m] is subtracted from the measured tool length.

#### 5.2.2 Calibrating a tool probe



# Calibration

The cycle uses the calibration tool to ascertain the current distance dimensions between machine zero (machine-related calibration) and workpiece zero (workpiece-related calibration) and the tool probe trigger points, and automatically loads them into the appropriate data area in data block GUD6. Values are corrected without empirical and mean values.

#### Preconditions

The approximate coordinates of the tool probe must be entered before calibration starts in array \_TP[\_PRNUM 1, 0] to \_TP[\_PRNUM-1, 9] (machine related) or

\_TPW[\_PRNUM-1, 0] to \_TPW[\_PRNUM-1, 9] (workpiece related).

The precise length and radius of the calibration tool must be stored in a tool offset data block. This tool offset must be active when the measuring cycle is called. Tool type 120 can be entered. There is no special "calibration tool" type.

Machining plane G17, G18, or G19 must be defined before the cycle is called.

All the necessary parameters have been assigned values.

	Calibration tool
-	Calibration tool

	_
•	

# **Parameters**

10       Calibrate tool probes (workpiece-related)         10000       Calibrate tool probes incrementally (machine-related)         10010       Calibrate tool probes incrementally (workpiece-related)	
10000Calibrate tool probes incrementally (machine-related)10010Calibrate tool probes incrementally (workpiece-related)_MA13Number of the measuring axis103, 203Number of the offset and measuring axis	
10010Calibrate tool probes incrementally (workpiece-related)_MA13Number of the measuring axis103, 203Number of the offset and measuring axis	
_MA13Number of the measuring axis103, 203Number of the offset and measuring axis	
103, 203 Number of the offset and measuring axis	
102, 201 (not for _MVAR=10000 and _MVAR=10010)	
_FA >0 Measurement path	
<pre>&lt;0 For incremental calibration (_MVAR=1000x0) the travel is also defined via _FAFA &gt; 0: Travel direction + _FA &gt; 0: Travel direction -</pre>	direction



10.04

_ID	REAL, ≥0	Offset
		The offset affects calibration of 3rd measuring axis if the
		calibration tool diameter is larger than the upper diameter of the
		probe. Here the tool is offset by the tool radius from the center
		of the probe, minus the value in _ID. The offset axis is also
		specified in _MA. Parameter _ID should usually set to 0.



The following additional parameters are also valid \_**VMS**, \_**TZL**, \_**TSA**, \_**PRNUM**, and \_**NMSP**. See Sections 2.2 and 2.3.

\_TZL, \_TSA not for incremental calibration!



# Sequence

#### Position before measuring cycle call

The calibration tool must be prepositioned as shown in the figure and for the selected variant. The tool must have reached a permissible starting position.

With **incremental calibration**, there is no generation of traversing movements before the actual measured block. The calibration tool must be positioned in front of the tool probe such that the calibration tool traverses to the tool probe when the measuring axis and a signed incremental measuring path up to the expected edge are entered in FA.



In normal **calibration** the measuring cycle calculates the approach path to the probe independently from the starting position and then generates the appropriate traverse blocks.

# Note on calibrating in the 3rd measuring axis (\_MA=3, \_MA=103, \_MA=203):

If the tool diameter (2x \$TC\_DP6) is smaller than the upper diameter of the probe (\_TP[i,6]), the calibration tool is always positioned in the center of the probe.

If the tool diameter is larger, the calibration tool is offset by the tool radius toward the center onto the probe. The value of ID is subtracted.

The axis in which the offset is applied (offset axis) is also specified in \_MA (\_MA=103 or MA=203). If no offset axis is specified (\_MA=3), the offset is applied in the abscissa, if necessary, (for G17: X axis).

#### Sequence on additional offset axis specification

Additionally specifying the offset axis in \_MA (\_MA= 102 or \_MA= 201) first causes the exact center of the tool probe to be detected in the offset axis before calibration takes place in the measuring axis.

An entry in the array is only made for the measuring axis in the selected direction of measurement.

#### Position after end of measuring cycle

On completion of calibration, the calibration tool (radius) is \_FA from the measuring surface.







Ø.

# Programming example 1

# Complete calibration of tool probes (machine-related)

Values of the calibration tool T7 D1:

Tool type (DP1):	120
Length 1 - Geometry (DP3):	L1= 20.000
Radius - Geometry (DP6):	R = 5.000

Values of the tool probe 1 in data block GUD6, approximate values before calibration begins (machine-related):

_TP[0,0] = 50
_TP[0,1] = 28
_TP[0,2] = 42
_TP[0,3] = 20
_TP[0,4] = 80
_TP[0,6] = 20 (upper diameter)
_TP[0,9] = 4



%_N_CALIBRATE_MTT_MPF		
N05 G0 G17 G94 G90 ;Machining plane, define feedrate ty		
N10 T7 D1	;Select calibration tool	
N15 M6	;Change calibration tool and activate offset	
N30 SUPA G0 Z100	;Position infeed axis over tool on hole center	
	;point	
N35 SUPA X70 Y90	;Position in plane on tool on hole center	
	;point	
N40 _TZL=0.005 _TSA=5 _VMS=0 _NMSP=1	;Parameters for calibration in the Y axis	
_PRNUM=1 _FA=6	;with previous calculation of the probe	
N41 _MVAR=0 _MA=102	;center in X. The array applied is	
	;tool probe 1: _TP[0,i]	
N50 CYCLE971	;Calibration in minus Y direction	
N55 SUPA Z100	;Traverse up in infeed axis in rapid traverse	
N60 SUPA Y0	;Move in plane to position from which	
N65 _MA=2	;calibration can start in the plus Y direction	
N70 CYCLE971	;Calibration in plus Y direction (probe	
	;positioned on center in X)	
N80 SUPA X70 Z100	;In X axis and Z axis, rapid traverse	
	;retraction of probe	
N85 _MA=1	;Calibration in the X axis	
N90 CYCLE971	;Calibration in minus X direction	
N100 SUPA Z100	;In Z axis rapid traverse probe retraction	

#### Measuring Cycles for Milling and Machining Centers 5.2 CYCLE971 tool: Measuring milling tools, drills

N110 SUPA X10;In X axis move to position from which<br/>;calibration can start in the plus directionN120 CYCLE971;Calibration in plus X directionN130 SUPA Z100;Traverse up in infeed axisN140 \_MA=3;Calibration in the Z axis on G17N150 CYCLE971;Calibration in minus Z directionN160 M2;End of program



# Explanation

The new trigger values in -X, +X, -Y, +Y, and -Z are stored in the global data of tool probe 1 (\_PRNUM=1) \_TP[0,0...4] if they deviate by more than 0.005 mm (\_TZL=0.005) from the old values. Deviations of up to 5 mm (\_TSA=5) are permissible.



# Programming example 2

Calibrating tool probe in minus X

(workpiece-related) (from measuring cycles SW 6.3)

Values of the calibration tool T7 D1:

Tool type (DP1):	120
Length 1 - Geometry (DP3):	L1= 20.000
Radius - Geometry (DP6):	R = 5.000

Values of the settable ZO for G54: Displacement: X=60, Y=15, Z= 30

Rotation through: X=0, Y=0, Z=18 degrees

Values of the tool probe 1 in data block GUD6, approximate values before calibration begins (workpiece-related):

> \_TPW[0,0] = 50 \_TPW[0,1] = 28 \_TPW[0,2] = 42 \_TPW[0,3] = 20 \_TPW[0,4] = 80 \_TPW[0,9] = 4





10.04

%_N_CALIBRATE_MTT_X_MPF		
N05 G0 G17 G94 G54	;Define machining plane, zero offset and	
	;feed type	
N10 T7 D1	;Select calibration tool	
N15 M6	;Change calibration tool and activate offset	
N30 G0 Z100	;Position infeed axis over tool on hole center	
	;point	
N35 X70 Y90	;Position in plane on tool on hole center	
	;point	
N40 _TZL=0.005 _TSA=5 _VMS=0 _NMSP=1	;Parameters for calibration in the X axis	
_PRNUM=1 _FA=6	;The array of tool probe 1 is effective:	
N41 _MVAR=10 _MA=1	;_TPW[0,i]	
N50 CYCLE971	;Calibration in minus X direction	
N55 Z100	;Traverse up in infeed axis in rapid traverse	
N60 M2	;End of program	

#### Explanation

The calibration tool moves with its point from the starting position at N35 (X70, Y90, H100) in Y to the center of the probe Y31 ((\_TPW[0,2] + (\_TPW[0,3]) / 2 = (42+20) / 2=31)); then in the measuring axis X (\_MA=1, G17) to position X61 (\_TPW[0,0] + \_FA + R = 50 + 6 + 5 = 61 Here it is lowered to position Z76 (\_TPW[0,4] - \_TPW[0,9] = 80 - 4 = 76). Then measuring (calibration) is performed in the minus X direction. At the end the calibration tool is again at position X61.

The new trigger values in minus X are stored in the data of tool probe 1 (\_PRNUM=1) \_TP[0,0] if they deviate by more than 0.005 mm (\_TZL=0.005) from the old values. Deviations of up to 5 mm (\_TSA=5) are permissible.

In block N55 the calibration tool is moved up to position Z100 and the program ends with block N60.

# 5.2.3 Calibrating the tool probe automatically (from measuring cycles SW 6.3)



#### Automatic calibration

Measuring variants

\_MVAR=100000 (machine-related)

\_MVAR=100010 (workpiece-related) are used to calibrate the tool probe automatically. The cycle uses the calibration tool to determine the tool probe trigger points in **all axes** and loads them into the relevant data area of data block GUD6. A measuring axis does not have to be specified in \_MA. Otherwise, the same parameters apply as for calibration of an axis. Values are corrected without empirical and mean

values.

# Preconditions

The approximate coordinates of the tool probe must be entered before calibration starts in array \_TP[\_PRNUM-1, 0] to \_TP[\_PRNUM-1, 9] (machine

related) or \_TPW[\_PRNUM-1, 0] to

\_TPW[\_PRNUM-1, 9] (workpiece related). These values must be so precise that the parameter values of \_TSA and \_FA are fulfilled.

The precise length and radius of the calibration tool must be stored in a tool offset data block. This tool offset must be active when the measuring cycle is called. Tool type 120 can be entered. There is no special "calibration tool" type. Machining plane G17, G18, or G19 must be defined before the cycle is called.

All the necessary parameters have been assigned values.



# Parameters

_MVAR	100000	Calibrate tool probe automatically (machine related)		
	100010	Calibrate tool probe autor	natically (workpiece related)	
_FA	>0	Measurement path		
The following additional parameters are also valid				
_VMS, _TZL, _TSA, _PRNUM, _ID and _NMSP. Set offset _ID=0 as standard				
See Sections 2.2 and 2.3.				

10.04

### Sequence

#### Position before measuring cycle call

The position before measuring cycle call can be anywhere, but:

The cycle must be able to position the 1st calibration point at distance \_FA above the center of the probe without collision.

The cycle approaches this point in the axis sequence: applicate (tool axis) followed by axis of the plane.

All subsequent traversing movements are also performed by the measuring cycle with "automatic calibration" using the values entered in array \_TP[] or TPW[] of the probe and the dimensions of the active calibration tool.

Calibration is performed in this sequence:

 –applicate, +abscissa, –abscissa, +ordinate, –ordinate;

then finally again in

–applicate, but now in the located center.

For G17, for example, these are the axes: -Z, +X, -X, +Y, -Y, -Z.

Calibration in the plus direction of the abscissa is performed after ascertaining the center of the probe in the ordinate. Additional movements are performed in the plane.

This sequence applies to

\_TP[\_PRNUM-1, 7]=133 or

\_TPW[\_PRNUM-1, 7]=133:

probe in Z axis can only be calibrated in minus direction, X, Y, in both directions.

Value \_TP[k, 7] or \_TPW[k, 7] =133 is the default value.

If some axes or axis directions on the probe cannot be approached the value must be changed. Meaning: Decimal point units: 1. geometry axis (X)

tens: 2. geometry axis (Y)

hundreds: 3. geometry axis (Z)

Value 0: axis not possible

- 1: only minus direction possible
- 2: only plus direction possible
- 3: both directions possible







5

Example \_TP[k, 7]=123:

X in both directions,

Y only in plus direction,

Z can only be calibrated in minus direction. It must always be possible to approach the tool axis (applicate, for example, Z axis for G17) in the minus direction. Otherwise automatic calibration is not possible.

The sequence described above changes according to the value of  $_{TP[k, 7]}$  or  $_{TPW[k, 7]}$ .

#### Position after end of measuring cycle

On successful completion of the calibration process, the calibration tool is positioned distance \_FA above the center of the probe.



### Programming example

# Calibrate tool probe automatically, machine-related for G17

Values of the calibration tool T7 D1:

Tool type (DP1):	120	
Length 1 - Geometry (DP3):	L1= 70.123	
Radius - Geometry (DP6):	R = 5.000	



Values of tool probe 1 in block GUD6 before calibration:

\_TP[0,0] = 50 (minus X axis) \_TP[0,1] = 28 (plus X axis) \_TP[0,2] = 42 (minus Y axis) \_TP[0,3] = 20 (plus Y axis) \_TP[0,4] = 80 (minus Z axis) \_TP[0,6] = 21 (wheel diameter at upper edge)

\_TP[0,7] = 133 (calibratable: minus Z axis in both directions in X and Y)

- \_TP[0,8] = 101 (wheel in X/Y)
- \_TP[0,9] = 4 (distance to upper edge, calibration depth)

%_N_AUTO_CALIBRATE_MPF		
N10 G17 G0 G90 G94		
N20 T7 D1	;Preselect calibration tool	
N30 M6	;Change calibration tool and activate offset	
N40 SUPA X39 Y31 Z100	;Take up start position	

10.04

N20 _MVAR=100000 _FA=6 _TSA=5 _TZL=0.001	;Parameters for calibration cycle	
_PRNUM=1 _VMS=0 _NMSP=1		
N30 CYCLE971	;Automatic calibration (complete)	
N99 M2		
Explanation:		
The tool probe is calibrated from the starting position		
in the –Z. +XX. +YY and again -Z axes. The		

values are entered in the \_TP field and \_OVR field if the results (values of the differences) are within the limits:

>\_TZL, <\_TSA.

5

#### 5.2.4 Measuring the tool



# Function

The cycle determines the new tool length or the new tool radius and checks whether the difference can be corrected with an empirical value to the old tool length or radius within a defined tolerance range (upper limits: Safe area

\_TSA and dimensional deviation check \_TDIF, lower limit: zero offset range \_TZL, see Section 1.9). If this range is not violated, the new tool length or radius is accepted, otherwise an alarm is output. Violation of the lower limit is not corrected.

Measuring is possible either with

- motionless spindle
- rotating spindle.

#### Precondition

- The tool probe must be calibrated.
- The tool geometry data (approximate values) must be entered in a tool offset data record.
- The tool must be active.
- The desired machining plane must be activated.
- The tool must be prepositioned in such as way that collisionfree approach to the probe is possible in the measuring cycle.

#### Special features of measurement with rotating spindle

- An additional compensation can be activated with variable \_MT\_COMP>0.
   (see Subsection 5.2.1 Measurement and compensation strategy)
- By default, the cycle-internal calculation of feed and speed is executed from the limit values defined in array \_CM[] for peripheral speed, rotation speed, minimum feed, maximum feed and measuring accuracy, as well as the intended direction of spindle rotation for measurement.
   Measuring is conducted by probing twice; the first probing action causes a higher feedrate. A maximum of three probing operations are possible for measuring.
   If probing is performed several times the speed is additionally reduced on the last probing operation. This speed reduction can be suppressed by setting channelrelated bit \_CHBIT[22].



10.04

- The operator can deactivate the cycle-internal calculation • via the measuring cycle bit CBIT[12]=1 and specify his or her own values for feed and speed. Array \_MFS[] is for entering the values. If the bit is set, the values from MFS[0/1] are valid for the first probing and the values from \_MFS[2/3] (speed/feedrate) for the second. If \_MFS[2] = 0 only one probing action is performed. If \_MFS[4]> 0 and \_MFS[2]> 0, probing is performed in three probing actions; the values from \_MFS[4/5] are valid in the third action. The monitoring functions from array \_CM[] are not active!
- If the spindle is motionless when the measuring cycle is • called, the direction of rotation is determined from \_CM[5].

Monitoring for measuring with	n rotating spindle and	cycle-internal calculation
-------------------------------	------------------------	----------------------------

_CM[0]	REAL	Maximum permissible peripheral speed [m/min]/[feet/min]
		Default setting: 100 m/min.
_CM[1]	REAL	Maximum permissible speed for measuring with rotating spindle [rpm]
		(if it is exceeded, the speed is automatically reduced)
		Default setting: 1000 rpm
_CM[2]	REAL	Minimum feedrate for first probing operation [mm/min]/[inch/min]
		(prevents feeds from being too low with large tool radii)
		Default setting: 1mm/min.
_CM[3]	REAL	Required measuring accuracy [mm]/[inch]
		is effective with the last probing action
		Default setting: 0.005 mm
_CM[4]	REAL	Maximum feedrate for probing [mm/min]/[inch/min]
		Default setting: 20 mm/min.
_CM[5]	REAL	Direction of spindle rotation during measuring
		Default setting: 4 = M4
_CM[6]	REAL	Feed factor 1
		<ol> <li>Only one probing action at calculated feed (but at least value of _CM[2])</li> </ol>
		≥1: 1. Probing with calculated feedrate (but at least value of
		_CM[2]) · feedrate factor 1
		Default setting: 10
_CM[7]	REAL	Feed factor 2
		0: 2. Probing with calculated feedrate (only active when _CM[6]>0)
		≥1: 2. Probing action with calculated feed · Feed factor 2
		3. Probing with calculated feed
		Feed factor 2 should be smaller than feed factor 1.
		Default setting: 0
Notice		

If the spindle is already rotating when the measuring cycle

is called, this direction of rotation remains independent of

\_CM[5]!



	_	_	
F.	=	_	
	_		

# Parameters

_MVAR	1	Measure with motionless spindle, machine related
	2	Measure with rotating spindle, machine related
	11	Measure with motionless spindle, workpiece related <sup>1)</sup>
	12	Measure with rotating spindle, workpiece related <sup>1)</sup>
_MA		Number of the measuring axis
	1	Measure radius in abscissa direction (milling tool)
	2	Measure radius in ordinate direction (milling tool)
	3	Measure length at center point of the tool probe (drill or
		milling tool)
	103	Measure length, offset by radius in abscissa direction
		(milling tool)
	203	Measure length, offset by radius in ordinate direction
		(milling tool)
_ID	$REAL, \geq 0$	Offset
		Parameter is usually set to 0.
		With multiple cutters the offset of tool length and the
		highest point of the tool edge must be specified in _ID for
		radius measurement and the offset from the tool radius to
		the highest point of the tool edge must be specified for
		length measurement.
_MFS[0]	REAL	Speed 1st probing (only with _CBIT[12]=1)
_MFS[1]	REAL	Feed 1st probing
_MFS[2]	REAL	Speed 2nd probing
		0: Measurement terminated after 1st probing
_MFS[3]	REAL	Feed 2nd probing
_MFS[4]	REAL	Speed 3rd probing
		0: Measurement terminated after 2nd probing
_MFS[5]	REAL	Feed 3rd probing

See Sections 2.2 and 2.3.

1) from meas. cycles SW 6.3

Ę

# Length measuring variants

Example: G17, machine related **Note:** 

If the tool diameter (2x \$TC\_DP6) is smaller than the upper diameter of the probe (\_TP[i,6]), the tool is always positioned in the center of the probe, if the tool diameter is larger, the tool is offset by the tool radius toward the center onto the probe. The value of \_ID is subtracted. The axis in which the offset is applied (offset axis) is also specified in \_MA (\_MA=103 or MA=203). If no offset axis is specified (\_MA=3), the offset is applied in the abscissa, if necessary, (for G17: X axis).

# Radius measuring variants (milling tool)

Example: G17, machine related, \_MA=1





#### Sequence

#### Position before measuring cycle call

Before cycle call a starting position must be taken up from which approach to the probe is possible without collision. The measuring cycle calculates the continued approach path and generates the necessary travel blocks.

#### Position after end of measuring cycle

On completion of the cycle, the tool nose or tool radius is positioned facing the measuring surface at a distance corresponding to \_FA.





# Programming example 1

# Measuring the length and radius or a milling tool (machine related)

Milling tool F3 with D1 is to be measured for the first time along length L1 and radius R (to determine geometry).

The length measurement is to be performed with motionless spindle. Radius measurement is to be performed with rotating spindle – in the X axis. The tool has a specially shaped cutting edge and therefore requires an offset for measurement.

A measured value deviation of < 1.6 mm compared with the entered values is expected.

Values of tool T3, D1, before measurement:

Tool type (DP1):	123
Length 1 - Geometry (DP3):	L1= 70
Radius - Geometry (DP6):	R = 18
Length 1 - Wear (DP12):	0
Radius - Wear (DP15):	0

Tool probe 1 is used. It has already been calibrated under the same conditions.

Values: See programming example 1 "Calibration"



%_N_T3_MEAS_MPF	
N01 G17 G90 G94	
N05 T3 D1	;Selection of the tool to be measured
N10 M6	;Insert tool, offset active
N15 G0 SUPA Z100	;Position infeed axis with probe
N16 SUPA X70 Y90 SPOS=15	;Position X/Y, align cutting edge
	;(if required)
N20 _CHBIT[3]=0 _CBIT[12]=0	;Offset of tool geometry, cycle calculation of
	;feedrate and speed during measurement
	;with rotating spindle
N30 _TZL=0.04 _TDIF=1.6 _TSA=2 _PRNUM=1 _VMS=0	;Parameters for cycle
_NMSP=1_FA=3_EVNUM=0	
N31_ID=2.2_MVAR=1_MA=103	;Offset in X axis for length measurement
N40 CYCLE971	;Measure length with motionless spindle
N50 SUPA X70	;Retract from probe in X
N70_ID=2.4 _MA=1 _MVAR=2	;New offset for radius measurement
N80 CYCLE971	;Measure radius in minus X direction with
	;rotating spindle
N90 SUPA Z100 M2	;Raise in Z, end of program

# =?

# Explanation

Length 1 (derived in block N40) and the radius (derived in block N80) of the active tool (T3, D1) are entered in the relevant geometry memory (\_CHBIT[3]=3) if they

deviate by more than 0.04 mm (\_TZL=0.04) and less than 1.6 mm ( TDIF=1.6)

from entered values L1, R.

If the differences are  $\geq$  \_TDIF or \_TSA alarms are output.

Values are corrected without empirical values (\_EVNUM=0).

Wear values L1 and R of the tool are reset (\_CHBIT[3]=0).





# Programming example 2

# Measuring the radius of a milling tool (workpiece-related) (from measuring cycles SW 6.3)

Milling tool T4 with D1 is to be remeasured in radius R (to ascertain wear).

Radius measurement is to be performed with rotating spindle – in the X axis.

A measured value deviation of < 0.6 mm compared with the entered values is expected.

Tool type (DP1):	120
Length 1 - Geometry (DP3):	L1= 70
Radius - Geometry (DP6):	Rg = 18.0
Radius - Wear (DP15):	Rv= 0.024
	R = Rg + Rv

Values of the settable ZO for G54:

Displacement: X=60, Y=15, Z= 30 Rotation through: X=0, Y=0, Z=18 degrees

Tool probe 1 is used. It has already been calibrated under the same conditions (G17, G54,  $\dots$ ).

Values: See programming example 2 "Calibration"

Measuring radius, workpiece related
(_MVAR=2, _MA=1) Starting position
Y1 YA
20+R
FA
TPW [0,3]
TPW[0,0]
M <sup>4</sup> X1

%_N_T4_MEASURE_MPF	
N01_PRNUM=1	;Select tool probe 1
N02 G17 G54 G94 G90	;Plane, ZO, feedrate type, dimensioning
N05 T4 D1	;Selection of the tool to be measured
N10 M6	;Insert tool, offset active
N15 G0 Z=_TPW[_PRNUM-1,4]+20	;Position in infeed axis above the tool probe
N16 X=_TPW[_PRNUM-1,0]+\$P_TOOLR+20	;Position X/Y plane: tool edge 20 mm next to
Y=_TPW[_PRNUM-1,2] +\$P_TOOLR +20	;probe edge +X, +Y
N20 _CHBIT[3]=1 _CBIT[12]=0	;Offset in wear, internal cycle calculation
	;of feedrate and speed during measurement
	;with rotating spindle
N30 _TZL=0.04 _TDIF=0.6 _TSA=2 _VMS=0 _NMSP=1	;Remaining parameters for cycle
_FA=3 _EVNUM=0	
N31 _ID=0 _MVAR=2 _MA=1	;Without offset
N40 CYCLE971	;Measurement with rotating spindle
N50 Z=_TPW[_PRNUM-1,4]+20	;Raise from probe in Z
N60 M2	

# 2 Explanation

10.04

The tool moves in N40 (in cycle) with its point from the starting position in N16 in Y to the center of the probe (\_TPW[0,2] + (\_TPW[0,3]) / 2); then in the measuring axis X (\_MA=1, G17) to position (\_TPW[0,0] + \_FA + R). Here it is lowered to the position in Z (\_TPW[0,4] – \_TPW[0,9]. Then measuring is performed in the minus X direction. At the end the tool (radius) is again positioned distance \_FA in X in front of the probe. In block N50 the tool is raised 20 mm in Z above the probe. Then the program is ended (N60). The difference in radius (derived in block N40) of the

active tool (T4, D1) is subtracted from the wear and entered (\_CHBIT[3]=1), if they deviate

by more than 0.04 mm (\_TZL=0.04) and less than 0.6 mm (\_TDIF=0.6).

If the difference is  $\geq$  \_TDIF or \_TSA, alarms are output.

Values are corrected without empirical values (\_EVNUM=0).



# 5.3 CYCLE976 calibrate workpiece probe



# Programming

CYCLE976



# Function

With milling machines and machining centers, the probe is usually loaded into the spindle from a tool magazine.

This may result in errors when further measurements are taken on account of probe clamping tolerances in the spindle.

The probe trigger points must be determined in the axis directions that are dependent on:

- the probe ball diameter
- the mechanical design of the probe
- speed at which the probe hits an obstacle

With this calibration cycle it is possible to calibrate a workpiece probe **in a hole** (axes in the plane) or **on suitable surfaces** for a particular axis and direction. **Determining the skew of the workpiece probe** A real workpiece probe can deviate from its ideal vertical position even when not deflected. This positional deviation (skew) can be determined with measuring variants in this cycle and then entered in the intended array of the workpiece \_WP[i, 7] for abscissa and \_WP[i, 8] for ordinate (for detailed data: see Section 9.2).

These values are taken into account in precise probe positioning in subsequent measurements with a similarly calibrated workpiece probe.





# Workpiece probe types that can be used

- Multidirectional probe (\_PRNUM=xy)
- Monodirectional, bidirectional probe (\_PRNUM=1xy)



10.04

# =?

# Measuring variants

Measuring cycle CYCLE976 permits the following **calibration** variants which are specified via parameter \_MVAR.

# • Calibration in hole (axes of the plane)

Digit					Meaning	
6	5	4	3	2	1	
					1	Hole (for measurement in the plane), center of the hole known
					8	Hole (for measurement in the plane), center of the hole not known
				0		With any data in the plane (workpiece-related)
			0			Without including probe ball in calculation
			1			Including probe ball in calculation (for measurement in plane)
		0				4 axis directions
		1				1 axis direction (also specify measuring axis and axis direction)
		2				2 axis directions (also specify measuring axis)
	0					Without determining skew of probe
	1					Determining skew of probe
0						Paraxial calibration (in the plane)
1						Calibration at an angle (in the plane)

#### Note:

When \_MVAR=xx1x0x calibration is only performed in one direction. It is not possible to determine skew or calculate probe ball.

#### • Calibration on surface

Digit					Meaning	
6	5	4	3	2	1	
					0	Calibration on surface (workpiece-related)
	1	0	0	0	0	Calibration on a surface in the applicate with calculation of probe
						length (workpiece-related, only permissible when _MA=3)

#### **Result parameters**

Measuring cycle CYCLE976 returns the following values in data block GUD5 for **calibration**:

_OVR [4]	REAL	Actual value probe ball diameter
_OVR [5]	REAL	Difference probe ball diameter
_OVR [6] <sup>1)</sup>	REAL	Center point of the hole in the abscissa
_OVR [7] <sup>1)</sup>	REAL	Center point of the hole in the ordinate
_OVR [8]	REAL	Trigger point in minus direction, actual value, abscissa
_OVR [10]	REAL	Trigger point in plus direction, actual value, abscissa
_OVR [12]	REAL	Trigger point in minus direction, actual value, ordinate
_OVR [14]	REAL	Trigger point in plus direction, actual value, ordinate
_OVR [16]	REAL	Trigger point in minus direction, actual value, applicate
_OVR [18]	REAL	Trigger point in plus direction, actual value, applicate
_OVR [9]	REAL	Trigger point in minus direction, difference, abscissa
_OVR [11]	REAL	Trigger point in plus direction, difference, abscissa
_OVR [13]	REAL	Trigger point in minus direction, difference, ordinate
_OVR [15]	REAL	Trigger point in plus direction, difference, ordinate
_OVR [17]	REAL	Trigger point in minus direction, difference, applicate
_OVR [19]	REAL	Trigger point in plus direction, difference, applicate
_OVR [20]	REAL	Positional deviation abscissa (skew of probe)
_OVR [21]	REAL	Positional deviation ordinate (skew of probe)
_OVR [24]	REAL	Angle at which the trigger points were determined
_OVR [27]	REAL	Zero offset area
_OVR [28]	REAL	Safe area
_OVI [2]	INTEGER	Measuring cycle number
_OVI [5]	INTEGER	Probe number
_OVI [9]	INTEGER	Alarm number

1) For calibration variant with unknown

Hole center point



# Function

With measuring cycle and measuring variant \_MVAR= xxxx01

it is possible to calibrate the probe in the axes of the plane (G17, G18, or G19) in a calibration ring. A hole that is perpendicular to the selected plane and with the same quality requirements regarding geometrical accuracy and surface roughness can also be used.

The center point (CP) of the hole and its diameter (D) must be known for this calibration variant!

The calculated trigger points are automatically loaded in the relevant data area \_WP[] of block GUD6.DEF if the calculated difference from the stored trigger points lies within the tolerance band between \_TZL and \_TSA. If \_TSA is exceeded an error message is output.

Calibration is performed either **paraxially** with the axes of the active workpiece coordinate system **or at an angle** to these axes.

The number of axes and axis directions can be selected in \_MVAR. If fewer than four axis directions are selected (\_MVAR= xx1xx01, xx2xx01), additional information must be supplied in \_MA and possibly in \_MD.







# Parameters

_MVAR	xxxx01	Calibration variant		
_SETVAL	REAL, >0	Calibration setpoint = diameter of hole		
_MA	1, 2	Meas. axis, only for _MVAR= xx1xx01, = xx2xx01		
		(only 1 axis or only 1 axis direction)		
_MD	0 positive axis direction	Meas. axis, only for _MVAR= xx1x01		
	1 negative axis direction	(calibrate one axis direction only)		
_PRNUM	>0	Probe number		
_STA1	REAL	Starting angle, only for _MVAR= 1xxx01		
		(calibration performed at this angle)		
The following additional parameters are also valid _CORA only relevant for monodirectional				
_VMS, _CORA, _TZL, _TSA, _FA, and _NMSP. probe				
See Sections 2.2 and 2.3.				

5



#### Measuring Cycles for Milling and Machining Centers 5.3 CYCLE976 calibrate workpiece probe



#### Precondition

The probe must be called with tool length offset. Tool type, preferably: 710.

#### Notice

The first time calibration is performed the default setting in the array of probe \_WP[] is still "0". For that reason \_TSA> probe ball radius must be programmed to avoid alarm "Safe area violated".

# Sequence

#### Position before measuring cycle call

The probe must be positioned at the center of hole (MP) in the abscissa and the ordinate of the selected measuring plane and at the calibration depth in the hole.

#### Axis sequence, axis direction sequence

Paraxial, two axis directions:

Calibration starts in the positive axis direction. If \_MVAR=xx0xx1 (all four directions), calibration starts in the abscissa. This is followed by the ordinate.

#### At an angle:

The axes travel in combination acc. to starting angle \_STA1 plus steps of 90 degrees. Otherwise the same principle as for "paraxial" applies.

#### Position after end of measuring cycle

When calibration is complete the probe is again positioned at calibration depth in the center of the hole.



# Programming example

Calibrating a workpiece probe in the X-Y plane, known hole center point

Workpiece probe 3, used as tool T9, D1, is to be recalibrated in a known hole with

MPx=100,000, MPy=80,000, D=110,246 mm in axes X and Y in both axis directions with G17 and paraxially (to redetermine trigger values \_WP[i,1] to \_WP[i,4].

The skew (positional deviation \_WP[i,7], \_WP[i,8]) and precise ball diameter \_WP[i,0] of the probe is also to be ascertained.

The radius of the probe ball and length 1 must be entered in the tool offset memory under T9 D1, before the cycle is called.

Tool type (DP1):	710	
Length 1- Geometry (DP3):	L1= 50.000	
Radius (DP6):	R = 3.000	

Length 1 (L1) must refer to the center of the probe ball: \_CBIT[14]=0.

Careful when positioning! Radius R in length (L1) is ignored. But the desired calibration height can be entered directly.

Zero offset, with settable ZO G54:

NVx, NVy, ...

Arrays for workpiece probe 3: \_WP[2, ...] (already contains approximate values)

%_N_CALIBRATE_IN_X_Y_MPF	
N10 G54 G90 G17 T9 D1	;ZO selection, select probe as tool and
	;plane
N20 M6	;Insert probe and activate tool offset
N30 G0 X100.000 Y80.000	:Position probe at center of on hole center
	;point
N40 Z10	;Position probe in hole at calibration height
N50 _CBIT[14]=0	;Length 1 relative to probe ball center



5

10.04

N60 _TSA=1 _PRNUM=3 _VMS=0 _NMSP=1 _FA=1 _TZL=0 N61 _MVAR=010101 _SETVAL=110.246	;Assign parameters for calibration cycle: ;Calibrate probe 3 in ;4 axis directions with calculation of skew ;and active diameter of probe ball
N70 CYCLE976	;Measuring cycle call, calibrate paraxially
N80 Z40	;Position probe above workpiece
N100 M2	;End of program

#### Explanation

The new trigger values in -X, +X, -Y and +Y are stored in the global data of measuring probe 3 \_WP[2,1...4]. The positional deviation calculated in the X and Y direction is stored in \_WP[2,7], \_WP[2,8], the active probe ball diameter in \_WP[2,0].

# 5.3.2 Calibrating a workpiece probe in a hole of unknown hole center point

# Function

With measuring cycle and measuring variant \_MVAR= xx0x08

it is possible to calibrate the probe in the axes of the plane (G17, G18, or G19) in a calibration ring. A hole that is perpendicular to the selected plane and with the same quality requirements regarding geometrical accuracy and surface roughness can also be used.

The center (CP) of the hole is **unknown** in the precise position. But diameter (D) is known.

In this measuring variant first the hole center and then the positional deviation (skew) of the probe is calculated. Then the trigger points in all 4 axis directions on the plane are calculated. In addition to the values in array \_WP[], the measuring cycle also provides the determined hole center point in result array \_OVR[6], \_OVR[7]. Calibration can be performed paraxially or at an angle to the active workpiece coordinate system. All 4 axis directions are always calibrated.




10.04



# Parameters

_MVAR	xx0x08	Calibration in hole, center unknown
_SETVAL	REAL, >0	Calibration setpoint = diameter of hole
_PRNUM	>0	Probe number
_STA1	REAL	Starting angle, only for _MVAR= 1xxx08
		(calibration performed at this angle)

The following additional parameters are also valid \_VMS, \_CORA, \_TZL, \_TSA, \_FA, and \_NMSP. See Sections 2.2 and 2.3.

\_CORA only relevant for monodirectional probe



# Precondition

- The probe must be called with tool length offset.
- Tool type, preferably: 710.
- The exact diameter of the hole is known.
- The spindle must be SPOS-capable.
- Probe in spindle can be positioned 0...360 degrees (all-round coverage).

#### Notice

The first time calibration is performed the default setting in the array of the probe is still "0". For that reason, \_TSA> probe ball radius must be programmed to avoid alarm "Safe area violated".

# $\rightarrow$

•

### Sequence

### Position before measuring cycle call

The probe must be positioned near the hole center in the abscissa and the ordinate of the selected measuring plane and at the calibration height in the hole.

#### Axis sequence, axis direction sequence

- Paraxial: Calibration always starts in the positive axis direction, first in the abscissa, then in the ordinate.
  - At an angle: The axes always travel in combination acc. to starting angle \_STA1, \_STA1+180 degrees, STA1+90 degrees, and STA1+270 degrees.

Otherwise the same principle as for "paraxial" applies.



The cycle performs two measurement operations to determine the skew of the probe and the center point of the hole.

- 1. Spindle positioned 180 degrees from initial position with SPOS and all axis directions traversed.
- 2. Spindle positioned at initial position and all axis directions traversed again.

#### Position after end of measuring cycle

When calibration is complete the probe is again positioned at calibration height in the center of the hole.

#### Note

Repeating calibration using the determined hole center is advisable if the starting position at the beginning is severely eccentric or measuring accuracy demands are high. 10.04

# Programming example

Calibrating a workpiece probe in the X-Y plane, unknown hole center point

Workpiece probe 2, used as tool T10, D1, and not precisely known center point (CP) is to be calibrated in axes X and Y in both axis directions with G17 and paraxially (to redetermine trigger values \_WP[i,1] to \_WP[i,4].

The skew (positional deviation \_WP[i,7], \_WP[i,8]) and precise ball diameter \_WP[i,0] of the probe is also to be ascertained.

The radius of the probe ball and length 1 must be entered in the tool offset memory under T10, D1, before the cycle is called.

Tool type (DP1):	710	
Length 1- Geometry (DP3):	L1= 50.000	
Radius (DP6):	R = 3.000	



Careful when positioning! Radius R in length (L1) is ignored. But the desired calibration height can be entered directly.

Zero offset, with settable ZO G54:

NVx, NVy, ...

Arrays for workpiece probe 2: \_WP[1, ...] (already contains approximate values)

%_N_CALIBRATE2_IN_X_Y_MPF	
N10 G54 G90 G17 T10 D1	;ZO selection, select probe as tool and
	;plane
N20 M6	;Insert probe and activate tool offset
N30 G0 X100 Y80	;Position probe in hole
N40 Z10	;Position probe in hole at calibration height
N50 _CBIT[14]=0	;Length 1 relative to probe ball center



N60 _TSA=1 _PRNUM=2 _VMS=0 _NMSP=1 _TZL=0 N61 _MVAR=010108 _SETVAL=110.246 _FA=_SETVAL/2	;Assign parameters for calibration cycle: ;calibrate probe 2 in 4 axis directions with ;calculation of skew and calculation of ;diameter of probe ball
N70 CYCLE976	;Measuring cycle call, calibrate paraxially
N80 Z40	;Position probe above workpiece
N100 M2	;End of program

# \_?

# Explanation

The hole center is determined twice, the spindle with the probe being rotated through 180° between each measurement if a multi probe is used, in order to record any positional deviation (skew) of the measuring probe. Triggering is then determined in all 4 axis directions.

The new trigger values in -X, +X, -Y and +Y are stored in the global data of probe 2\_WP[1,1...4], the positional deviation in the X and Y direction in \_WP[1,7], \_WP[1,8], the active probe ball diameter in \_WP[1,0]. The calculated hole center is entered in OVR[6], OVR[7].

### 5.3.3 Calibration of a workpiece probe on a surface



#### Function

With this measuring cycle and measuring variant \_\_MVAR=0

a workpiece probe can be calibrated in one axis and one direction on a known surface with sufficiently good surface roughness and which is perpendicular to the measuring axis . This can be done on a workpiece, for example.

The trigger point of the relevant axis and axis direction is calculated and entered in the workpiece probe array \_WP[i,1] to \_WP[i,5] provided.

#### Precondition

The probe must be called as a tool with a tool length offset. Tool type, preferably: 710

When using the cycle on a turning machine:

set type and \_CBIT[14]=0.





# Parameters

_MVAR	0	Calibration variant: Calibration on surface
_SETVAL	REAL	Calibration setpoint (position of surface)
_MA	1, 2, or 3	Measuring axis
_MD	0 positive axis direction	Measuring direction
	1 negative axis direction	
_PRNUM	INT, >0	Probe number



The following additional parameters are also valid \_VMS, \_CORA, \_TZL, \_TSA, \_FA and \_NMSP. See Sections 2.2 and 2.3.

\_CORA only relevant for monodirectional probe.



#### Notice

The first time calibration is performed the default setting in the array of the probe is still "0". For that reason \_TSA>probe ball radius must be programmed to avoid alarm "Safe area violated".

# Sequence

#### Position before measuring cycle call

The probe must be positioned facing the calibration surface. Recommended distance: >\_FA.

#### Position after end of measuring cycle

When calibration is complete the probe (ball radius) is distance \_FA from the calibration surface if \_MA=3, if \_MA=1 or \_MA=2 it is at the starting position.

#### Programming example

#### Calibrating a workpiece probe on the workpiece.

Workpiece probe 1 is to be calibrated in the Z axis on the surface at position Z=20,000 mm of a clamped workpiece. Determine trigger value in minus direction \_WP[0.5].

Clamping for workpiece: Zero offset, with settable ZO G54: NVx, NVy, ...

The workpiece probe is to be inserted as tool T9 with offset D1.

The radius of the probe ball and length 1 must be entered in the tool offset memory under T9 D1, before the cycle is called.

Tool type (DP1):	710
Length 1- Geometry (DP3):	L1= 50.000
Radius (DP6):	R = 3.000

Length 1 (L1) must refer to the center of the probe ball: \_CBIT[14]=0.

Careful when positioning! Radius R in length (L1) is ignored. But the desired calibration height can be entered directly when calibrating in the abscissa or ordinate (\_MA=1, =2).

Arrays for workpiece probe 1: \_WP[0, ...]



10.04



10.04

%_N_CALIBRATE_IN_Z_MPF	
N10 G54 G90 G17 T9 D1	;ZO selection, select probe as tool and
	;plane
N20 M6	;Insert probe and activate tool offset
N30 G0 X100 Y80	;Position probe above surface
N40 Z55	;Lower probe, distance > (_FA + R)
N50 _CBIT[14]=0	;Length 1 relative to probe ball center
N60 _TSA=4 _TZL=0 _PRNUM=1 _VMS=0 _NMSP=1	;Define parameter for calibration cycle,
_FA=2	calibrate probe 1 in minus Z axis;
N61 _MVAR=0 _SETVAL=20 _MA=3 _MD=1	· ·
N70 CYCLE976	;Measuring cycle call
N80 Z55	;Position probe above workpiece
N100 M2	;End of program



# Explanation

The new trigger value in the minus Z direction is entered in the global data of workpiece probe 1 in \_WP[0,5].

© Siemens AG, 2004. All rights reserved SINUMERIK 840D/840Di/810D User's Guide Measuring Cycles (BNM) – 10.04 Edition

## 5.3.4 Calibrating a workpiece probe in the applicate determining probe length



#### Function

With this measuring cycle and measuring variant \_\_MVAR=10000

the workpiece probe can be calibrated in the tool axis (applicate) on a known surface with sufficiently good surface roughness and which is perpendicular to the measuring axis.

This can be done on a workpiece, for example. The trigger point of the relevant axis and axis direction is calculated and entered in the workpiece probe array \_WP[i,5] provided.

At the same time length 1 (L1) of the probe is calculated according to the setting of \_CBIT[14]:

\_CBIT[14]=0: L1 referred to ball center \_CBIT[14]=1: L1 ref. to ball circumference and entered in the tool offset memory.

#### Precondition

The probe must be called as a tool with a tool length offset.

Tool type, preferably: 710

When using the cycle on a turning machine: set type and \_CBIT[14]=0.

#### Notice

If you want to position with the tool in the program, the approximate probe length should be known and entered in the tool offset memory. Otherwise position the probe with JOG in front of the calibration surface.

The exact ball radius must be known and entered. The first time calibration is performed the default setting in the array of the probe is still "0". Therefore:

\_TSA> probe ball radius must be programmed to avoid alarm "Safe area violated".





#### Parameters

MVAR	10000	Calibration in applicate with length calculation	
	10000		
_SETVAL	REAL	Calibration setpoint (position of surface)	
_MA	3	Measuring axis, only tool axis (applicate) possible	
_MD	0 positive axis direction	Measuring direction	
	1 negative axis direction		
_PRNUM	>0	Probe number	
The following	ng additional parameters are also	_CORA only relevant for monodirectional probe.	
valid _VMS	, _CORA, _TSA, _FA and	_TSA is only evaluated with reference to the trigger	
_NMSP.		value, not with reference to the tool length.	

See Sections 2.2 and 2.3.



**\$**:

#### Sequence

#### Position before measuring cycle call

The probe must be positioned facing the calibration surface such that it is deflected within the max. measuring path of  $2 \cdot FA$  [mm] on cycle start.

#### Position after end of measuring cycle

When the calibration procedure is completed the probe is positioned on the starting position.

#### Programming example

**Calibration of a workpiece probe in the Z axis on the workpiece with length calculation** Workpiece probe 1 is to be calibrated in the Z axis on the surface at position Z=20,000 mm of a clamped workpiece. Determine trigger value in minus direction \_WP[0,5] and length 1 (L1).

Clamping for workpiece: Zero offset, with settable ZO G54: NVx, NVy, ...

The workpiece probe is inserted as tool T9 with offset D1.

The radius of the probe ball must be entered in the tool offset memory under T9, D1 before the cycle is called.

Tool type (DP1):	710
Length 1- Geometry (DP3):	L1= ?
Radius (DP6):	R = 3.000





#### Measuring Cycles for Milling and Machining Centers 5.3 CYCLE976 calibrate workpiece probe

Length 1 (L1) must refer to the center of the probe ball: \_CBIT[14]=0. Careful when positioning! Radius R in length (L1) is ignored. Equally, L1 is only an approximate value or even unknown or zero.

Arrays for workpiece probe 1: \_WP[0, ...]

%_N_CALIBRATE_Z_L_MPF	
N10 G54 G90 G17 T9 D1	;ZO selection, select probe as tool and
	;plane
N20 M6	;Insert probe and activate tool offset
;	;The probe is prepositioned in front of the
	;calibration surface within distance $2 \cdot \_FA$
N50 _CBIT[14]=0	;Length 1 relative to probe ball center
N60 _TSA=25 _TZL=0 _PRNUM=1 _VMS=0 _NMSP=1	;Define parameter for calibration cycle,
_FA=12	;calibrate probe 1 in minus Z axis
N61 _MVAR=10000 _SETVAL=20 _MA=3 _MD=1	;with determination of length 1
N70 CYCLE976	;Measuring cycle call
	determine calibration in minus Z and L1;
N100 M2	;End of program

# Explanation

On cycle call the probe travels in the minus Z direction max. 24 mm (\_FA=12) at measuring feedrate 300 mm/min (\_VMS=0, \_FA>1). If the probe is triggered within this measuring path of 24 mm, length 1 (geometry) is calculated and entered in tool offset memory T9, D1, D3. The trigger value of probe 1 in the minus Z direction is used as the ball radius of the probe from T9, D1, DP6 and then entered in \_WP[0,5] – if \_CBIT[14]=0. If \_CBIT[14]=1, this value is =0.

# 5.4 CYCLE977 workpiece: Measure hole/shaft/groove/web/rectangle parallel to axes

5.4 CYCLE977 workpiece: Measure hole/shaft/groove/web/rectangle



#### Programming

CYCLE977

10.04



#### Function

With this measuring cycle you can measure the dimensions of the following contour elements on a workpiece using different measuring variants:

- Hole
- Shaft
- Groove
- Web
- Inside rectangle
- Outside rectangle

Measurement is performed paraxially to the workpiece coordinate system.

In some measuring variants defined **safety zones** are taken into account during the measuring operation.

#### CYCLE977 can

© Siemens AG, 2004. All rights reserved

- measure the contour elements and additionally either
- perform an automatic tool offset
- for a specified tool based on the differences in diameter or width, or

• correct a **zero offset** (ZO) based on the differences between the center positions.

#### Workpiece probe types that can be used

- Multidirectional probe (\_PRNUM=xy)
- Monodirectional, bidirectional probe (\_PRNUM=1xy)

SINUMERIK 840D/840Di/810D User's Guide Measuring Cycles (BNM) - 10.04 Edition



Measuring Cycles for Milling and Machining Centers

X

#### **Measuring variants**

Measuring cycle CYCLE977 permits the following measuring variants that are specified with parameter \_MVAR: Value Measuring variant Measure hole with tool offset 1 2 Measure shaft with tool offset 3 Measure groove with tool offset 4 Measure web with tool offset 5 Measure rectangle inside with tool offset 6 Measure rectangle outside with tool offset 101 ZO calculation in hole with ZO compensation 102 ZO calculation on shaft with ZO compensation 103 ZO calculation in groove with ZO compensation 104 ZO calculation on web with ZO compensation 105 ZO determination in inside rectangle with ZO correction 106 ZO determination in outside rectangle with ZO correction 1001 Measure hole traveling around a safety zone and tool offset 1002 Measure shaft taking account of a safety zone and tool offset 1003 Measure groove traveling around a safety zone and tool offset 1004 Measure web taking account of a safety zone and tool offset 1005 Measure inside rectangle with safety zone and tool offset 1006 Measure outside rectangle with safety zone and tool offset 1101 ZO determination, hole traveling around a safety zone with ZO correction 1102 ZO determination, shaft taking account of a safety zone with ZO correction 1103 ZO determination, groove traveling around a safety zone with ZO correction 1104 ZO determination, web taking account of a safety zone with ZO correction 1105 ZO determination, inside rectangle with safety zone with ZO correction 1106 ZO determination, outside rectangle with safety zone with ZO correction

> © Siemens AG, 2004. All rights reserved SINUMERIK 840D/840Di/810D User's Guide Measuring Cycles (BNM) – 10.04 Edition



# 

# **Result parameters**

Depending on the measuring variant \_MVAR=xxx1 to \_MVAR=xxx4, measuring cycle CYCLE977 supplies the following values as results in data block GUD5: (**not** with rectangle measurement, see next table):

_OVR [0]	REAL	Setpoint diameter/width hole, shaft, groove, web
_OVR [1]	REAL	Setpoint center point/center hole, shaft, groove, web in abscissa
_OVR [2]	REAL	Setpoint center point/center hole, shaft, groove, web in ordinate
_OVR [4]	REAL	Actual value diameter/width hole, shaft, groove, web
_OVR [5]	REAL	Actual value center point/center hole, shaft, groove, web in abscissa
_OVR [6]	REAL	Actual value center point/center hole, shaft, groove, web in ordinate
_OVR [8] <sup>1)</sup>	REAL	Upper tolerance limit for diameter/width hole, shaft, groove, web
_OVR [12] <sup>1)</sup>	REAL	Lower tolerance limit for diameter/width hole, shaft, groove, web
_OVR [16]	REAL	Difference diameter/width hole, shaft, groove, web
_OVR [17]	REAL	Difference center point/center hole, shaft, groove, web in abscissa
_OVR [18]	REAL	Difference center point/center hole, shaft, groove, web in ordinate
_OVR [20] <sup>1)</sup>	REAL	Offset value
_OVR [27] <sup>1)</sup>	REAL	Zero offset area
_OVR [28]	REAL	Safe area
_OVR [29] <sup>1)</sup>	REAL	Dimensional difference
_OVR [30] <sup>1)</sup>	REAL	Empirical value
_OVR [31] <sup>1)</sup>	REAL	Mean value
_OVI [0]	INTEGER	D number or ZO number
_OVI [2]	INTEGER	Measuring cycle number
_OVI [4] <sup>1)</sup>	INTEGER	Weighting factor
_OVI [5]	INTEGER	Probe number
_OVI [6] <sup>1)</sup>	INTEGER	Mean value memory number
_OVI [7] <sup>1)</sup>	INTEGER	Empirical value memory number
_OVI [8]	INTEGER	Tool number
_OVI [9]	INTEGER	Alarm number
_OVI [11] <sup>2)</sup>	INTEGER	Status offset request
_OVI [13] <sup>1)</sup>	INTEGER	DL number (from measuring cycles SW 6.3)



1) for workpiece measurement with tool offset only

2) For ZO correction only

10.04 5

Measuring cycle CYCLE977 returns the following		
values depending on measuring variant		
rectangle measurement (_MVAR= xxx5, =xxx6)		
in data block	GUD5:	
_OVR [0]	REAL	Setpoint value rectangle length (in the abscissa)
_OVR [1]	REAL	Setpoint value rectangle length (in the ordinate)
_OVR [2]	REAL	Setpoint for rectangle center point, abscissa
_OVR [3]	REAL	Setpoint for rectangle center point, ordinate
_OVR [4]	REAL	Actual value for rectangle length (in the abscissa)
_OVR [5]	REAL	Actual value for rectangle length (in the ordinate)
_OVR [6]	REAL	Actual value for rectangle center point, abscissa
_OVR [7]	REAL	Actual value for rectangle center point, ordinate
_OVR [8] <sup>1)</sup>	REAL	Upper tolerance limit for rectangle length (in the abscissa)
_OVR [9] <sup>1)</sup>	REAL	Upper tolerance limit for rectangle length (in the ordinate)
_OVR [12] <sup>1)</sup>	REAL	Lower tolerance limit for rectangle length (in the abscissa)
_OVR [13] <sup>1)</sup>	REAL	Lower tolerance limit for rectangle length (in the ordinate)
_OVR [16]	REAL	Difference of rectangle length (in the abscissa)
_OVR [17]	REAL	Difference of rectangle length (in the ordinate)
_OVR [18]	REAL	Difference of rectangle center point, abscissa
_OVR [19]	REAL	Difference of rectangle center point, ordinate
_OVR [20] <sup>1)</sup>	REAL	Offset value
_OVR [27] <sup>1)</sup>	REAL	Zero offset area
_OVR [28]	REAL	Safe area
_OVR [29] <sup>1)</sup>	REAL	Dimensional difference
_OVR [30] <sup>1)</sup>	REAL	Empirical value
_OVR [31] <sup>1)</sup>	REAL	Mean value
_OVI [0]	INTEGER	D number or ZO number
_OVI [2]	INTEGER	Measuring cycle number
_OVI [4] <sup>1)</sup>	INTEGER	Weighting factor
_OVI [5]	INTEGER	Probe number
_OVI [6] <sup>1)</sup>	INTEGER	Mean value memory number
_OVI [7] <sup>1)</sup>	INTEGER	Empirical value memory number
_OVI [8]	INTEGER	Tool number
_OVI [9]	INTEGER	Alarm number
_OVI [11] <sup>2)</sup>	INTEGER	Status offset request
_OVI [13] <sup>1)</sup>	INTEGER	DL number (from measuring cycles SW 6.3)
1) for workpiece measurement with tool offset only		
2) For ZO correction only		

#### 5.4.1 Measuring contour elements

#### Function

Using this measuring cycle and various \_MVAR measuring variants the following contour elements can be measured:

\_MVAR = xxx1 - hole \_MVAR = xxx2 - shaft \_MVAR = xxx3 - groove \_MVAR = xxx4 - web \_MVAR = xxx5 - rectangle, inside \_MVAR = xxx6 - rectangle, outside

If no tool offset or no ZO correction is to be applied, set

**\_KNUM = 0** Detailed information on the parameters: Refer to Section 2.3.

#### Measuring principle for hole or shaft

Two points each are measured in the abscissa and ordinate. The actual position of the center point (CP) in relation to workpiece zero is calculated from these four measured values. The actual diameter is calculated from the two points in the ordinate.

The center of the abscissa is calculated from the two points in the abscissa. Then the probe is positioned on this calculated center and the two points on the ordinate measured. The hole and shaft center points are now known and the results entered in array \_OVR[].

The positive direction of an axis is measured first.

#### Measuring principle for groove or web

The groove or web lies parallel to the axes of the workpiece coordinate system.

2 measuring points are measured with specified measuring axis \_MA.

The actual value of the groove width and web width and the actual position of the groove center and web center in relation to workpiece zero are calculated from the two measured values. The results are entered in array \_OVR[].





# Measuring principle for inside and outside rectangle

The rectangle lies parallel to the axes of the workpiece coordinate system.

The measuring cycle ascertains 2 measuring points in both axes and determines the actual rectangle center and both actual values of the rectangle lengths. The procedure is the same as for hole and shaft.

The results are entered in array \_OVR[ ]. The positive direction of an axis is measured first.

# Procedure for specifying a safety zone

#### If required,

#### \_MVAR = 1xxx

can take account of a safety zone (\_SZA, \_SZO) in the travel movement. The safety zone refers to the centerpoint or center line of the hole, shaft, groove, web, rectangle. The starting point in the height is always above the hole, shaft, groove, web, or rectangle.

# Options for hole and shaft diameter, groove or web width, and tool compensation.

- An empirical value from data block GUD5 can be included with the correct sign.
- A mean value can be derived from several workpieces, measurement calls.

#### Precondition

The probe must be called as a tool with a tool length offset. Tool type, preferably: 710

when using the cycle on a turning machine: set type and \_CBIT[14]=0.





# 10.04 Measuring Cycles for Milling and Machining Centers 5.4 CYCLE977 workpiece: Measure hole/shaft/groove/web/rectangle

_MVAR	xxx1	Measure hole			
	xxx2	Measure shaft			
	xxx3	Measure groove			
	xxx4	Measure web	Measure web		
	xxx5	Measure rectangle, inside	9		
	xxx6	Measure rectangle, outsic	de		
_SETVAL	REAL, >0	Setpoint (acc. to drawing)			
		(only for hole/shaft/groove	e/web)		
_SETV[0]	REAL, >0	Setpoint value rectangle l	ength (in the abscissa)		
_SETV[1]		Setpoint value rectangle l	ength (in the ordinate)		
		(only when measuring a r	ectangle)		
_ID	REAL	Incremental infeed in app	Incremental infeed in applicate, direction before sign		
		(for shaft, web or rectangl	le measurement and for hole/groove/shaft/web		
		measurement traveling ar	ound and considering a safety zone)		
_SZA	REAL, >0	Diameter or width of plant	Diameter or width of protection zone		
		(inside for hole/groove, outside for shaft/web)			
		<ul> <li>Length of safety zone in abscissa</li> </ul>			
		(only when measuring	a rectangle)		
_SZO	REAL, >0	Length of the protection zone in the ordinate (only for measuring			
		rectangle)			
_MA	1, 2	Number of measuring axis	Number of measuring axis (only for measuring a groove or a web)		
_KNUM	0	0: Without automatic tool offset, without ZO determination			
The follow	ing additional pa	arameters are also valid	With _TSA, the diameter or width is		
_VMS, _C	ORA, _TSA, _F	A, _PRNUM, _EVNUM,	monitored for "tool compensation", the		
_NMSP.			center for "ZO determination".		
The follow	ing also applies	for measuring variants	_CORA only relevant for monodirectional		
with tool compensation (even when KNUM=0):			probe.		
TZL. TN	IV. TUL. TLL	. TDIF.	•		
_·, _·, _·, _·, ', ',					

See Sections 2.2 and 2.3.

5

#### Sequence

#### Specification of setpoints

The setpoint for diameter or width is specified in \_SETVAL and The setpoint for lengths of rectangle in SETV[0], SETV[1].

The position of the probe in the abscissa, ordinate at the beginning of a cycle is evaluated for the setpoint of the center point of a hole, shaft, or rectangle, or for the center of a groove, web. This value is also entered in result array

\_OVR[1], \_OVR[2] (for hole, shaft, groove, web), \_OVR[2], \_OVR[3] (for rectangle).

#### Specification of measuring axis:

The measuring axis in MA only has to be specified for web or groove:

MA=1: measurement in abscissa \_MA=2: measurement in ordinate.

For the remaining contour elements measurement is always performed in both axes of the plane and in both directions.

#### Position before measuring cycle call for shaft, web, rectangle - outside

_MVAR	Pre-positioning			
	in the plane	in applicate		
2/102	Shaft center point	Above shaft		
4/104	Web center, meas. axis	Above web		
6/106	Rectangle center point	Above rect.		

The probe must be positioned at the center point in the plane and the probe ball positioned above the upper edge such that when infeed of value \_ID (sign) is applied, measurement depth is reached.







# Position before cycle call for hole, groove, rectangle - inside

The probe must be positioned at the center point in the plane. The probe ball must be positioned at measurement height inside the hole/groove/rectangle.

_MVAR	Pre-positioning			
	in the plane	in applicate		
1/101	Hole center point	At meas. height		
3/103	Groove center, meas. ax.	At meas. height		
5/105	Rectangle center point	At meas. height		

#### Position before measuring cycle call when measuring with safety zone

_MVAR	Pre-positioning		lowered
	in the plane	in applicate	Example: _CBIT[14
1001/1101 1003/1103 1005/1105	Hole center point Groove center, meas. ax. Rectangle center point	Above hole Above groove Above rect.	Startin height

1002/1102	Shaft center point	Above shaft
1004/1104	Web center, meas. axis	Above web
1006/1006	Rectangle center point	Above rect.

#### Note:

If the value selected for FA is so large that the safety zone is violated the distance is automatically reduced in the cycle. However, there must be sufficient room for the probe ball.



Starting position on cycle call:



Starting position on cycle call, measurement with safety zone: In the center above shaft, web, or outside rectangle Measurement depth: lowered by \_ID (applicate) Z⋪ Example: G17, CBIT[14]=0 Starting Safety zone position Starting height

\_FA

Safety zone

W

2

FA

≙

ment

depth → X

(abscissa)

Measure

#### Specification of safety zone:

The safety zone (diameter or width) for shaft, hole, web, groove is defined in \_SZA.

For a rectangle, the safety zone (length) is defined with \_SZA in the abscissa and with \_SZO in the ordinate.



When measurement is complete the probe is positioned above the **calculated** center point or center at starting position height.

#### Notice

Precise measurement is only possible with a probe calibrated under the measurement conditions, i.e. working plane, orientation of the spindle in the plane and measuring velocity are the same for both measurement and calibration. Deviations can cause additional measuring errors.

The range of positions of the center or diameter, or groove, web width, rectangle length, must be within the value specified in \_FA for all workpieces to be measured.

Otherwise there is danger of collision or the measurement cannot be performed!









# **Programming example**

#### Measuring a web – paraxial

In the G17 plane, a web is to be measured with desired width 132 mm.

The assumed center is X=220.

The maximum possible deviation of the center is taken as 2 mm, the width 1 mm. To obtain a minimum measuring path of 1 mm, the measuring path is programmed as \_FA=2+1+1=4 mm (max. measuring path FA=8 mm).

However a measured deviation of the web center of >1.2 mm is not permissible.

Clamping for workpiece:

Zero offset, with settable ZO G54: NVx, NVy, ...

Workpiece probe 1, used as tool T9, D1, is to be used.

The probe is already calibrated. Arrays for workpiece probe 1: \_WP[0, ...]

The following is entered under T9, D1 in the tool offset memory:

Tool type (DP1):	710	
Length 1- Geometry (DP3):	L1= 50.000	
Radius (DP6):	R = 3.000	

Length 1 (L1) must refer to the center of the probe ball (\_CBIT[14]), as for calibration. Careful when positioning! Radius R in length (L1) is ignored.

# % N WEB MEASURE MDE

%_N_WEB_WEASURE_WPF	
N10 G54 G17 G90 T9 D1	;ZO, select tool as probe
N20 M6	;Insert probe,
	;Activate tool offset
N30 G0 X220 Y130	;Position probe in X/Y plane cent web center
	;in X and measuring position Y
N40 Z101	;Position Z axis above web



N60 _TSA=1.2 _PRNUM=1 _VMS=0 _NMSP=1 _FA=4 N61 _MVAR=104 _SETVAL=132 _MA=1 _ID=-40 _KNUM=0	;Set parameters for measuring cycle call ;Without ZO or tool compensation, ;Observe negative sign in _ID! ;Probe lowered in Z axis! ;Measuring variant with ZO determination ;selected (MVAR=X1xxx), because web ;center to be monitored (with _TSA), ;but _KNUM=0: without execution ;of ZO determination and ZO correction
N70 CYCLE977	;Call measuring cycle
N80 G0 Z160	;Traverse up Z axis
N100 M2	;End of program

#### Explanation

The measuring results of web width, web center in X, and associated differences are entered in result array \_OVR[]. If the difference in the position of the web center is >1.2mm (\_TSA=1.2) an alarm is output. Only cancellation with NC RESET is then possible.

The setpoint of the web center is the position of the probe in the workpiece coordinate system at the beginning of the cycle in the X axis.

#### 5.4.2 Measurement and tool offset



#### Function

With this measuring cycle and measuring variants \_MVAR = x0xx

a hole, shaft, groove, a web, or a rectangle can be measured paraxially. Automatic tool offset is also possible. This tool is

specified in **\_TNUM** and **\_TNAME**. The D number and type of offset are specified in coded form in variable **\_KNUM**.

From measuring cycles SW 6.3, extended tool offset is available. With this function a tool from a particular stored tool environment **\_TENV**, and additive, setup offsets can be corrected by specifying the DL number in **\_DLNUM**.

For more detailed information about parameters see Section 2.3.

If the dimensions of a tool are corrected to this extent, the next workpiece can be manufactured with lower tolerances.



# Parameters

_MVAR	1	Measure hole with tool offset
	2	Measure shaft with tool offset
	3	Measure groove with tool offset
	4	Measure web with tool offset
	5	Measure rectangle inside with tool offset
	6	Measure rectangle outside with tool offset
	1001	Measure hole by contouring a protection zone with tool offset
	1002	Measure shaft by including a protection zone with tool offset
	1003	Measure groove by contouring a protection zone with tool offset
	1004	Measure web by including a protection zone with tool offset
	1005	Measure rectangle inside with protection zone with tool offset
	1006	Measure rectangle outside with protection zone with tool offset

# Measuring Cycles for Milling and Machining Centers 10.04 5.4 CYCLE977 workpiece: Measure hole/shaft/groove/web/rectangle

_SETVAL	REAL, >0	Setpoint (acc. to drawing) (only for hole/shaft/groove/web)		
_SETV[0]	REAL, >0	Setpoint value r	Setpoint value rectangle length (in the abscissa)	
_SETV[1]		Setpoint value r	ectangle length (in the ordinate)	
		(only when mea	asuring a rectangle)	
_ID	REAL	Incremental infe	eed of applicate with sign (only for measuring shaft, web,	
		or rectangle, an	d for measuring hole/groove/shaft/web traveling around	
		or taking accou	nt of a safety zone)	
_SZA	REAL, >0	Diameter/w	idth of the protection zone	
		(inside for h	ole/groove, outside for shaft/web)	
		Length of th	ne safety zone in the abscissa	
		(only for me	easuring rectangle)	
_szo	REAL, >0	Length of the pr	otection zone in the ordinate (only for measuring	
		rectangle)		
_MA	12	Number of measuring axis (only for measuring a groove or a web)		
_KNUM	0, >0	0: without automatic tool offset		
>0: with automatic tool offset		atic tool offset		
		(individual value	es: See Section 2.3, parameter _KNUM)	
_TNUM	INT, ≥0	Tool number for automatic tool offset		
_TNAME	STRING[32]	Tool name for automatic tool offset		
		(alternative to _	TNUM with tool management active)	
_DLNUM	INT, ≥0	DL number for additive, setup offset		
		(from measuring	g cycles SW 6.3)	
_TENV	STRING[32]	Name of tool environment for automatic tool compensation		
		(from measuring	g cycles SW 6.)	
The followi	ing additional pa	rameters are	_CORA only relevant for monodirectional probe.	
also valid			With _TSA, the diameter or width is monitored for	
_VMS, _CORA, _TZL, _TMV, _TUL,		/IV, _TUL,	"tool compensation". The remaining parameters	
_TLL, _TDIF, _TSA, _FA, _PRNUM,		_PRNUM,	must also be assigned if _KNUM=0 because they	

refer to the workpiece.



# Programming example

\_EVNUM, \_NMSP and \_K.

See Sections 2.2 and 2.3.

#### Measuring a hole – paraxially with tool offset

The diameter of a hole in a workpiece is to be measured in the G17 plane and the radius of a tool corrected accordingly.

Clamping for workpiece: Zero offset, with settable ZO G54: NVx, NVy, ...

The probe is to be workpiece probe 1, inserted as tool **T9**, **D1**. The probe is already calibrated. Arrays for workpiece probe 1: \_WP[0, ...]

# 10.04 Measuring Cycles for Milling and Machining Centers 5.4 CYCLE977 workpiece: Measure hole/shaft/groove/web/rectangle

5

٦F

The following is entered under T9, D1 in the tool offset memory:

Tool type (DP1):	710
Length 1- Geometry (DP3):	L1= 50.000
Radius (DP6):	R = 3.000

Length 1 (L1) refers to the center of the probe ball (\_CBIT[14]=0), as for calibration.

Careful when positioning! Radius R in length (L1) is ignored.

The hole was machined with milling tool **T20**, **D1** in the same environment as for measuring (G17, ...) with circular milling.

The radius of this tool is to be corrected for wear, according to the measuring result for the hole diameter difference (actual value – set value). This tool offset will therefore affect the production of the next workpieces or possible remachining.

The offset must take the empirical value in memory \_EV[9] into consideration. Mean value calculation \_MV[9] and inclusion in calculation are also to be used.

A maximum deviation of the diameter from the setpoint of 1 mm is expected.

	Z1	7	(applicate)			(Y)
			(applicate)			T
_			Actual	center Setpoint o	center	
		35—		Sta	rting position	n for cycle
is	M	₩ NVx€	180		(abscis	ম sa) X / X1
	Y1,	• • 1	Y (ordinate	 })		
)			Actual hole	<sub>Setpoir</sub>	nt hole	
		stual MPy	2		Center fou	nd in X
S		1: 1		ETVAL	(start positi measurem	tion for ent in Y)
e		Ž W	Actual MP	x	(abscis	sa) X
ry	M	<b>*</b>				X1

Measuring a hole

%_N_DRILL_MEASURE_MPF	
N10 G54 G17 G90 T9 D1	;ZO, select tool as probe
N20 M6	;Insert probe,
	;Activate tool offset
N30 G0 X180 Y130	;Position probe in X/Y plane to
	;hole center point
N40 Z20	;Position Z axis to measuring depth
N50 _CHBIT[4]=1	;Include average value
N60 _TUL=0.03 _TLL=-0.03 _EVNUM=10 _K=3	;Set parameters for measuring cycle call,
_TZL=0.01 _TMV=0.02 _TDIF=0.06 _TSA=1 _PRNUM=1	;Probe 1 (multidirectional),
_VMS=0 _NMSP=1 _FA=1	;Measure hole, setpoint diameter
N61 MVAR=1_SETVAL=132_TNUM=20_KNUM=2001	;132 mm, compensation in radius of T20, D1
N70 CYCLE977	;Call measuring cycle
N560 G0 Z160	;Retract Z axis from hole
N570 M2	;End of program

#### Explanation

The difference calculated from the actual and setpoint diameter is compensated for by the empirical value in the empirical value memory \_EV[9] and compared with the tolerance parameter.

- If it is more than 1 mm (\_TSA), alarm "Safe area violated" is output and the program is halted.
- Cancel with NC RESET on the control!
- If it is more than 0.06 mm (\_TDIF), no compensation is performed and alarm "Permissible dimensional difference exceeded" is output and the program continues.
- If ±0.03 mm (\_TUL/\_TLL) is exceeded, the radius in T20 D1 is compensated 100% by this difference/2.
- Alarm "Oversize" or "Undersize" is displayed and the program is continued.
- If 0.02 mm (\_TMV) is exceeded, the radius in T20, D1 is compensated 100% by this difference/2.
- If it is less than 0.02 mm (\_TMV), the mean value is calculated from the mean value in mean value memory \_MV[9] and inclusion of weighting factor \_K=3 (only for \_CHBIT[4]=1! with mean value memory).
  - If the mean value obtained is >0.01 (\_TZL), the reduced compensation of the radius for T20 D1 is the mean value/2 and the mean value is deleted in \_MV[9].
  - If the mean value is <0.01 (\_TZL) the radius in T20 D1 is not compensated but is stored in mean value memory \_MV[9].

The results are entered in result array \_OVR[]. The wear of the radius of T20, D1 is included if a change is necessary.

5

#### Measuring Cycles for Milling and Machining Centers 10.04 5.4 CYCLE977 workpiece: Measure hole/shaft/groove/web/rectangle

#### Measurement and ZO determination 5.4.3

#### Function

With this measuring cycle and measuring variants \_MVAR=x1xx

a hole, shaft, groove, a web, or a rectangle can be measured paraxially. The zero offset (ZO) of the associated workpiece can also be determined and corrected. A possible rotation of the workpiece is kept without changing it.

The angular position cannot be determined with this cycle.

Compensation of the ZO is executed in such a way that the actual center (position of center on the machine, e.g.: MPx1, MPy1) includes the desired setpoint position in the workpiece coordinate system when the compensated ZO (frame) is applied.

Mirroring can be active in a frame of the frame sequence. Dimension factors must never be active.

The ZO to be corrected is specified in coded form with variable KNUM >0.

The ZO can be specified and corrected by various methods, e.g. in various settable frames, in various basic frames, system frames, fine offset, or coarse offset, etc.

For detailed information on specifying KNUM for the zero offset: Refer to Section 2.3.

#### ZO determination in a hole, on a shaft, or rectangle

ZO correction of a workpiece is applied in the abscissa and ordinate using the actual value/setpoint difference of the position of the center point.

#### ZO determination in a groove or on a web

ZO correction of a workpiece is applied in measuring axis \_MA (abscissa and ordinate) using the actual value/setpoint difference of the position of the center.



Y1

MP<sub>Y1</sub>

NVy

Example: G17

Y act.

Wne



Principle of ZO calculation and correction

set

hole

Xse

X

X1



### **Parameters**

MVAR	101	ZO calculation in hole with ZO compensation	
-	102	ZO determination on a shaft with ZO correction	
	103	ZO determination in a groove with ZO correction	
	104	ZO determination on a web with ZO correction	
	105	ZO determination in inside rectangle with ZO correction	
	106	ZO determination in outside rectangle with ZO correction	
	1101	ZO determination in hole traveling around a safety zone with ZO correction	
	1102	ZO determination of shaft taking account of a safety zone with ZO correction	
	1103	ZO determination in groove traveling around a safety zone, ZO correction	
	1104	ZO determination on web taking account of a safety zone with ZO correction	
	1105	ZO determination, inside rectangle with safety zone with ZO correction	
	1106	ZO determination, outside rectangle with safety zone with ZO correction	
_SETVAL	REAL, >0	Setpoint (acc. to drawing)	
		(only for hole/shaft/groove/web)	
_SETV[0]	REAL, >0	Setpoint value rectangle length (in the abscissa)	
_SETV[1]		Setpoint value rectangle length (in the ordinate)	
		(only when measuring a rectangle)	
_ID	REAL	Incremental infeed of applicate with sign (only for measuring shaft, web	
		or rectangle, and for measuring hole/groove/shaft/web traveling around	
		or taking account of a safety zone)	
_SZA	REAL, >0	Diameter/width of the protection zone	
		(inside for hole/groove, outside for shaft/web)	
		<ul> <li>Length of the safety zone in the abscissa</li> </ul>	
		(only for measuring rectangle)	
\$70		Length of the protection zone in the ordinate (only for measuring	
_320	REAL, 20	rectangle)	
_MA	12	Number of measuring axis (only for measuring a groove or a web)	
_KNUM	0, >0	0: No automatic ZO correction	
		>0: With automatic ZO correction	
		(individual values: See Section 2.3, parameter _KNUM)	
The followi	ing additional p	arameters are also valid CORA only relevant for monodirectional	

VMS, \_CORA, \_TSA, \_FA, \_PRNUM, and \_NMSP. probe. With \_TSA the center is See Sections 2.2 and 2.3. monitored with ZO determination.



# Programming example

#### ZO determination on a rectangle with CYCLE977

In the G17 plane, an outside rectangle is to be measured with the

desired lengths in X=100.000 and in Y=200.00 mm. The settable ZO G54 is to be corrected in such a way that the center of the rectangle is at X=150,000 and Y=170,000 mm.

Measurement is also performed at G54. After measurement is complete, the changed ZO is activated.

The maximum possible deviation of the center is taken to be 2 mm; the maximum possible deviation in the lengths, 3 mm. To obtain a minimum measuring path of 1 mm, the measuring path is programmed as \_FA=2+3+1=6 mm (max. total measuring path

\_FA=12 mm).

A measured deviation of the center of the rectangle from the setpoint of >1.8 mm is however not permitted in either axis.

Height of rectangle and measuring height in Z: see Fig.

Clamping for workpiece:

Zero offset, with settable ZO G54:

NVx, NVy, ... (values when measuring)

The probe is to be workpiece probe 1, inserted as tool **T9**, **D1**. The probe is already calibrated. Arrays for workpiece probe 1: \_WP[0, ...] The following is entered under T9, D1 in the tool offset memory:

Tool type (DP1):	710
Length 1- Geometry (DP3):	L1= 50.000
Radius (DP6):	R = 3.000

Length 1 (L1) refers to the center of the probe ball (\_CBIT[14]=0), as for calibration. Careful when positioning! Radius R in length (L1) is ignored.



%_N_ZO_RECTANGLE_MPF	
N10 G54 G17 G90 T9 D1	;ZO, select tool as probe
N20 M6	;Insert probe,
	;Activate tool offset
N30 G0 X150 Y170	;Position probe in X/Y plane to
	;rectangle center (setpoint position)
N40 Z120	;Position Z axis above rectangle
N60 _KNUM=1 _TSA=1.8 _PRNUM=1 _VMS=0	;Set parameters for measuring cycle call,
_NMSP=1_FA=6	
N61_MVAR=106_SETV[0]=100_SETV[1]=200_ID=-30	;Measuring height lowered by 30 mm in Z
N70 CYCLE977	;Call measuring cycle
N80 G54	;Repeat call of ZO G54
	;This activates the altered ZO offset!
N90 G0 Z160	;Traverse up Z axis
N100 M2	;End of program

#### Explanation

Automatic compensation is performed in G54 – translation in axes X and Y by the calculated difference between actual value and setpoint of the rectangle centerpoint, if it is less than 1.8 mm (\_TSA=1.8) in both axes. Otherwise alarm "Safe area violated" is output and program execution cannot be continued. If the values are inside the tolerance, the setpoint and actual values for center point and length of rectangle in the abscissa and ordinate as well as the differences are entered in result array OVR[]. The zero offset (ZO) for G54 is entered in the data management (\$P\_UIFR[1]) and is activated by programming G54 again in block N80.



10.04

### 5.5 CYCLE978 workpiece: Measuring a surface parallel to the axis



#### Programming

CYCLE978



#### Function

This measuring cycle determines the position of a **paraxial surface** in the workpiece coordinate system. This is done with 1-point measurement If differences are ascertained, different measuring variants will also allow correction with

- automatic **tool compensation** for a specified tool, or
- zero offset (ZO).

Special measuring variants allow you to perform the measurement as a **differential measurement**. The special procedure for this measurement permits use of an **uncalibrated** multidirectional probe.



Workpiece measuring probe types that can be used

- Multidirectional probe (\_PRNUM=xy)
- Monodirectional, bidirectional probe (\_PRNUM=1xy)

#### Notice

A monodirectional or bi-directional probe must be calibrated! These probes cannot be used for differential measurements!

#### Preconditions for differential measurement

- Spindle orientation (with SPOS command) by means of NC
- Bidirectional/multidirectional probe
- Positioning of probe in spindle between  $0^\circ$  and  $360^\circ$  (at least every  $90^\circ)$  (all-round coverage).

# Measuring variants

Measuring cycle CYCLE978 permits the following measuring variants that are specified with parameter \_MVAR.

Value	Measuring variant
0	Measure surface and tool offset
100	ZO determination on surface and ZO correction
1000	Measure surface with differential measurement and tool offset
1100	ZO determination on surface with differential measurement and ZO correction

#### **Result parameters**

 Depending on the measuring variant, CYCLE978

 makes the following values available as results in

 the block GUD5:

 \_OVR [0]
 REAL

 Setpoint value for measuring axis

 \_OVR [1]
 REAL

 Setpoint for abscissa

 \_OVR [2]
 REAL

 Setpoint for ordinate

 \_OVR [3]
 REAL

 Setpoint for applicate

 \_OVR [4]
 REAL

 Actual value for measuring axis

 \_OVR [8]<sup>10</sup>
 REAL

 Upper tolerance limit for measuring

 \_OVR [12]<sup>10</sup>
 REAL

_OVR [4]	REAL	Actual value for measuring axis
_OVR [8] <sup>1)</sup>	REAL	Upper tolerance limit for measuring axis
_OVR [12] <sup>1)</sup>	REAL	Lower tolerance limit for measuring axis
_OVR [16]	REAL	Difference for measuring axis
_OVR [20] <sup>1)</sup>	REAL	Offset value
_OVR [27] <sup>1)</sup>	REAL	Zero offset area
_OVR [28]	REAL	Safe area
_OVR [29] <sup>1)</sup>	REAL	Dimensional difference
_OVR [30]	REAL	Empirical value
_OVR [31] <sup>1)</sup>	REAL	Mean value
_OVI [0]	INTEGER	D number or ZO number
_OVI [2]	INTEGER	Measuring cycle number
_OVI [4] <sup>1)</sup>	INTEGER	Weighting factor
_OVI [5]	INTEGER	Probe number
_OVI [6] <sup>1)</sup>	INTEGER	Mean value memory number
_OVI [7]	INTEGER	Empirical value memory number
_OVI [8]	INTEGER	Tool number
_OVI [9]	INTEGER	Alarm number
_OVI [11] <sup>2)</sup>	INTEGER	Status offset request
_OVI [13] <sup>1)</sup>	INTEGER	DL number (from measuring cycles SW 6.3)

1) For 1-point measurement with automatic tool offset only

2) For ZO correction only

#### 5.5.1 Measuring the surface

#### Sequence

10.04

#### Position before measuring cycle call

The probe is positioned in relation to the surface to be measured in such a way that during traversal of the specified measuring axis \_MA in the direction of the setpoint \_SETVAL, the intended measuring point on the surface will be reached. Recommended distance from surface: > FA.

The absolute value of the positional deviation from the setpoint must not be greater than the measuring path \_FA. Otherwise, no measurement will be performed.

#### Position after end of measuring cycle

After the end of the measuring cycle, the probe (ball circumference) is at a distance \_FA from the measuring surface.

#### Notice

Precise measurement is only possible with a probe calibrated under the measurement conditions, i.e. working plane, orientation of the spindle in the plane and measuring velocity are the same for both measurement and calibration. Deviations can cause additional measuring errors.

#### Precondition

The probe must be called as a tool with a tool length offset. Tool type, preferably: 710 when using the cycle on a turning machine: set type and \_CBIT[14]=0.

#### Special procedure for differential measurement

The measuring point is measured twice during differential measurement.

- with spindle rotated through 180 degrees compared with the position at the beginning of the cycle (rotation of the probe by 180 degrees).
- 2. with the spindle position that applied at the beginning of the cycle

This is how the position deviation of the probe in the measuring axis is determined.

The tool radius of the probe + R or - R is defined as the trigger point defined for the axis direction.



5





10.04

The values are entered and stored in the global user data (GUD6) in the array for the associated workpiece probe \_WP[].

A **multidirectional** probe does not have to be calibrated at the beginning of the cycle for measuring variants

### \_MVAR= 1000 or \_MVAR=1100.

These measuring variants with differential measurement are only useful with the measuring axes \_MA=1 or \_MA=2.

#### Notice

In the case of great measurement accuracy demands, differential measurement is not recommended!

### 5.5.2 Measurement and ZO determination

#### Function

With this measuring cycle and measuring variants \_\_MVAR=100, \_MVAR=1100 it is possible to determine the position of a surface that is parallel with the axis in the workpiece coordinate system.

The zero offset (ZO) of the associated workpiece can also be determined and corrected. The offset is corrected in such a way that the real position of the surface (actual value) adopts the required set angle (\_SETVAL) in the workpiece coordinate system when the corrected ZO (frame) is used.

Mirroring can be active in a frame of the frame sequence. Dimension factors must never be active.

The ZO to be corrected is specified in coded form with variable **\_KNUM** >0.

The ZO can be specified and corrected by various methods, e.g. in various settable frames, in various basic frames, system frames, fine offset, or coarse offset, etc.

If \_KNUM=0, there is no ZO correction.

For detailed information on specifying \_KNUM for the zero offset: Refer to Section 2.3.

#### Measuring Cycles for Milling and Machining Centers 5.5 CYCLE978 workpiece: Measuring a surface parallel to the axis

An empirical value stored in data block GUD5 in array \_EV[] can be included in calculation of the result after measurement is completed.

This is activated in \_EVNUM (see Section 2.3).



# Parameters

10.04

_ <b>MVAR</b> 100		ZO determination on surface and ZO correction	
	1100	ZO determination on surface with differential measurement and ZO	
		correction	
_SETVAL	REAL	Setpoint with respect to workpiece zero	
_MA	13	Number of the measuring axis	
_KNUM	0, >0	0: without automatic ZO correction	
		>0: with automatic ZO correction	
		(individual values: See Section 2.3, parameter _KNUM)	
The followi	ing additional	parameters are also valid	

Ĩ

The following additional parameters are also valid VMS, \_CORA, \_TSA, \_FA, \_PRNUM, \_EVNUM and \_CORA

A, \_FA, \_PRNUM, \_EVNUM and \_CORA only relevant for monodirectional probe.

Ζ1

Ň

See Sections 2.2 and 2.3.

NMSP.

# Programming example

### ZO calculation at a workpiece with CYCLE978

A rectangular workpiece is clamped in the G17 plane. The ZO in axes X and Y is to be checked. Any deviation from the active values should be automatically corrected in settable ZO G54. The corrected ZO should also be activated so that machining of the workpiece can start directly after.

The empirical value entered in array \_EV[9] (data block GUD5) for the X axis and \_EV[10] for the Y axis are to be included in the measuring results.

The permissible deviation is 3 mm from the setpoint value is assumed. To obtain a minimum measurement path of 1 mm to the surface, the measurement path is programmed with \_FA=3+1=4. (max. total measurement path=8 mm).

The value of the positional deviation should not be monitored. Therefore \_TSA > \_FA is set. Clamping for workpiece: Zero offset, with settable ZO G54: NVx, NVy, ... (values when measuring)



Calculating ZO - rectangular workpiece

Z (applicate)

Ξ

X1

10.04

Workpiece probe 1, used as tool <b>T9</b> , <b>D1</b> , is to be	
used.	
The probe is already calibrated. Arrays for	
workpiece probe 1: _WP[0,]	
The following is entered under T9, D1 in the tool	
offset memory:	
Tool type (DP1): 710	
Length 1- Geometry (DP3): L1= 50.000	
Radius (DP6): R = 3.000	
Length 1 (L1) refers to the center of the probe	
ball (_CBIT[14]=0), as for calibration.	
Careful when positioning! Radius R in length (L1) is	
ignored.	
%_N_ZO_DETERMINING_1_MPF	
N10 G54 G17 G90 T9 D1	;ZO, select tool as probe
N20 M6	;Insert probe,
	;Activate tool offset
N30 G0 G90 X-20 Y25	;Position probe in X/Y plane in front of
	;measuring surface
N40 Z10	;Position probe at measuring height
N60_TSA=6 _PRNUM=1 _VMS=0 _NMSP=1 _FA=4	;Set parameters for measuring cycle call
N61_MVAR=100_SETVAL=0_MA=1_KNUM=1	
_EVNUM=10	
N70 CYCLE978	;Measuring cycle for ZO determination in
N80 G0 X-20	;Retract in X axis
N90 Y-20	;Position in Y axis
N100 X22	;Position in X axis
N110_EVNUM=11_MA=2	;Set parameters for measuring cycle call
N120 CYCLE978	;ZO determination in Y axis
N130 G54	;Repeat call of ZO G54
	;This activates the changes!
N140 G0 Y-20	;Retract in Y axis
N150 Z50	;Retract in Z axis
N160 X-40 Y80	;Retract in X/Y
N200 M2	;End of program

# \_?

#### Explanation

Automatic compensation is performed in G54 – translation of axes X and Y by the calculated difference between actual value and setpoint. The setpoints and actual values as well as the differences are entered in result array OVR[]. At the end of the program the values for the Y axis (ordinate) are in the result array as these were the last to be measured. The zero offset (ZO) for G54 is entered in the data management (\$P\_UIFR[1]) and is activated by programming G54 again in block N130.
### 5.5.3 Measuring and tool offset

### Function

10.04

With this measuring cycle and measuring variants \_\_MVAR=0, \_MVAR=1000 it is possible to determine the dimension (position) of a surface that is parallel with the axis in the workpiece coordinate system.

Automatic tool offset is also possible. This tool is specified in **\_TNUM** and **\_TNAME**. The D number and type of offset are specified in coded form in variable **\_KNUM**.

From measuring cycles SW 6.3, extended tool offset is available. With this function a tool from a particular stored tool environment **\_TENV**, and additive, setup offsets can be corrected by specifying the DL number in **\_DLNUM**.

Detailed information on the parameters: Refer to Section 2.3.

### Empirical values and mean values

An empirical value stored in data block GUD5 in array \_EV[] can be included in calculation of the result after measurement is completed.

Optionally, averaging is performed over a number of parts (array \_MV[ ]) and the tolerance bands are checked.

Both are activated in \_EVNUM (see Section 2.3).

### Parameters

_MVAR	0	Measure surface and tool compensation		
	1000	Measure surface with dif	ferential measurement and tool offset	
_SETVAL	REAL	Setpoint (acc. to drawing	a)	
_MA	13	Number of the measurin	g axis	
_KNUM	0, >0	0: without automatic too	l offset	
		>0: with automatic tool o	ffset	
_TNUM	INT, ≥0	Tool number for automatic tool offset		
_TNAME	STRING[32]	Tool name for automatic tool offset		
		(alternative to _TNUM with tool management active)		
_DLNUM	INT, ≥0	DL number for additive, setup offset		
		(from measuring cycles SW 6.3)		
_TENV	STRING[32]	Name of tool environment for automatic tool compensation		
		(from measuring cycles	SW 6.3)	
The followi	ng additional pa	rameters are also valid	_CORA only relevant for monodirectional	
_VMS, _C	_VMS, _CORA, _TZL, _TMV, _TUL, _TLL, _TDIF, probe.			
_TSA, _FA	., _PRNUM, _E\	NUM, NMSP, and K.	The parameters must also be assigned if	

See Sections 2.2 and 2.3.

The parameters must also be assigned if \_KNUM=0 because they refer to the workpiece.

# **\$**

### Programming example

1-point measurement in X axis with tool compensation

A surface parallel with the Y axis has been machined with milling tool T20, D1 on a set-up workpiece. This surface should be a distance of exactly 100,000 mm in the X axis from the defined workpiece zero and be measured.

If the absolute value of the difference determined is >0.01, the radius of this tool is to be automatically offset in the wear.

1 mm is assumed to be the maximum permissible deviation of the position of the surface.

To obtain a minimum measuring path of 1 mm, the measuring path is programmed as \_FA= 1+1=2 mm (max. measuring path=4 mm).

The offset must take the empirical value in memory \_EV[19] into consideration. Mean value calculation \_MV[19] and inclusion in calculation are also to be used.



# 10.04 Measuring Cycles for Milling and Machining Centers 5.5 CYCLE978 workpiece: Measuring a surface parallel to the axis

5

This tool offset will therefore affect the production of the next workpieces or possible remachining.

Clamping for workpiece: Zero offset, with settable ZO G54: NVx, NVy, ...

The probe is to be workpiece probe 1, inserted as tool **T9**, **D1**. The probe is already calibrated. Arrays for workpiece probe 1: \_WP[0, ...] The following is entered under T9, D1 in the tool offset memory:

Tool type (DP1):	710
Length 1- Geometry (DP3):	L1= 50.000
Radius (DP6):	R = 3.000

Length 1 (L1) refers to the center of the probe ball (\_CBIT[14]=0), as for calibration. Careful when positioning! Radius R in length (L1) is ignored.



%_N_ONE_POINT_MEASURE_MPF	
N10 G54 G17 G90 T9 D1	;ZO, select tool as probe
N20 M6	;Insert probe,
	;Activate tool offset
N30 G0 G90 X120 Y150	;Position probe in X/Y plane in front of
	;measuring surface
N40 Z40	;Position probe at measuring height
N50 _CHBIT[4]=1	;with mean value calculation
N60 _TUL=0.03 _TLL=-0.03 _TNUM=20 _EVNUM=20 _K=3	;Set parameters for measuring cycle call
_TZL=0.01 _TMV=0.02 _TDIF=0.06 _TSA=1 _PRNUM=1	
_VMS=0 _NMSP=1 _FA=2	
N61 _MVAR=0 _SETVAL=100 _MA=1 _KNUM=2001	
N70 CYCLE978	;Measuring cycle for single-point
	;Measurement in X axis
N80 G0 Z160	;Traverse up Z axis
N100 M2	;End of program

10.04

### Explanation

The difference calculated from the actual and setpoint value (position of surface) is compensated for by the empirical value in the empirical value memory \_EV[19] and compared with the tolerance parameter.

- If it is more than 1 mm (\_TSA), alarm "Safe area violated" is output and program execution cannot be continued.
- If it is more than 0.06 mm (\_TDIF), no compensation is performed and alarm "Permissible dimensional difference exceeded" is output and the program continues.
- If 0.03 mm (\_TUL/\_TLL) is exceeded, the radius in T20 D1 is compensated 100% by this difference. Alarm "Oversize" or "Undersize" is displayed and the program is continued.
- If 0.02 mm (\_TMV) is exceeded, the radius in T20 D1 is compensated 100% by this difference.
- If it is less than 0.02 mm (\_TMV), the mean value is calculated from the mean value in mean value memory \_MV[19] and inclusion of weighting factor \_K=3 (only for \_CHBIT[4]=1! with mean value memory).
  - If the mean value obtained is >0.01 (\_TZL), the reduced compensation of the radius for T20 D1 is the mean value/2 and the mean value is deleted in \_MV[19].
  - If the mean values is < 0.01 (\_TZL), there is no radius offset for T20 D1, but it is stored in the mean value memory \_MV[19] if the mean value storage (\_CHBIT[4]=1) is active.

The results are entered in result array \_OVR[]. The wear of the radius of T20, D1 is included if a change is necessary.

### 5.6 CYCLE979 workpiece: Measure hole/shaft/groove/rib at an angle

5.6 CYCLE979 workpiece: Measure hole/shaft/groove/rib at an angle



### Programming

CYCLE979

10.04



### Function

With this measuring cycle you can measure the dimensions of the following contour elements on a workpiece using different measuring variants:

- Hole
- Shaft
- Groove
- Web

Measurement is performed at a specified starting angle to the abscissa of the workpiece coordinate system.

For hole, shaft, additional measurements are performed at an indexing angle, added to the previous angle.

This allows you to measure **circle segments** of a workpiece contour whose center points lie outside the machine.

### CYCLE979 can

- measure the contour elements
- and additionally either
- perform an automatic tool offset

for a specified tool based on the differences in diameter or width, or

• correct a **zero offset** (ZO) based on the differences between the center positions.

### Workpiece probe types that can be used

- Multidirectional probe (\_PRNUM=xy)
- Monodirectional, bidirectional probe (\_PRNUM=1xy)

When measuring contour elements hole, shaft, 3 or 4 point measurement can be used. This is also specified in \_PRNUM (in this cycle only): PRNUM=0zxy  $\rightarrow$  3-point measurement

 $PRNUM=1zxy \rightarrow 4-point measurement$ 



Measuring Cycles for Milling and Machining Centers





### **Measuring variants**

Measuring cycle CYCLE979 permits the following measuring variants that are specified with parameter MVAR

Value	Measuring variant
1	Measure hole with tool offset
2	Measure shaft with tool offset
3	Measure groove with tool offset
4	Measure web with tool offset
101	ZO calculation in hole with ZO compensation
102	ZO calculation on shaft with ZO compensation
103	ZO calculation in groove with ZO compensation
104	ZO calculation on web with ZO compensation



### **Result parameters**

Depending on the measuring variant, CYCLE979

makes the following values available as results in the block GUD5:

_OVR [0]	REAL	Setpoint diameter/width hole, shaft, groove, web
_OVR [1]	REAL	Setpoint center point/center in abscissa
_OVR [2]	REAL	Setpoint center point/center in ordinate
_OVR [4]	REAL	Actual value diameter/width hole, shaft, groove, web
_OVR [5]	REAL	Actual value center point/center in abscissa
_OVR [6]	REAL	Actual value center point/center in ordinate
_OVR [8] <sup>1)</sup>	REAL	Upper tolerance limit for diameter of hole, shaft or width of groove, web
_OVR [12] <sup>1)</sup>	REAL	Lower tolerance limit for diameter of hole, shaft or width of groove, web
_OVR [16]	REAL	Difference diameter/width hole, shaft, groove, web
_OVR [17]	REAL	Difference center point/center in abscissa
_OVR [18]	REAL	Difference center point/center in ordinate
_OVR [20] <sup>1)</sup>	REAL	Offset value
_OVR [27] <sup>1)</sup>	REAL	Zero offset area
_OVR [28] <sup>1)</sup>	REAL	Safe area
_OVR [29] <sup>1)</sup>	REAL	Permissible dimensional deviation
_OVR [30] <sup>1)</sup>	REAL	Empirical value
_OVR [31] <sup>1)</sup>	REAL	Mean value
_OVI [0]	INTEGER	D number or ZO number
_OVI [2]	INTEGER	Measuring cycle number
_OVI [4] <sup>1)</sup>	INTEGER	Weighting factor
_OVI [5]	INTEGER	Probe number
_OVI [6] <sup>1)</sup>	INTEGER	Mean value memory number
_OVI [7] <sup>1)</sup>	INTEGER	Empirical value memory number

 $$\$  Siemens AG, 2004. All rights reserved SINUMERIK 840D/840Di/810D User's Guide Measuring Cycles (BNM) – 10.04 Edition



10.04

_OVI [8]	INTEGER	Tool number
_OVI [9]	INTEGER	Alarm number
_OVI [11] <sup>2)</sup>	INTEGER	Status offset request
_OVI [13] <sup>1)</sup> INTEGER DL number (from measuring cycles SW 6.3)		
1) For workpiece measurement with tool offset only		

2) For ZO correction only

### 5.6.1 Measuring a hole, shaft, groove, web

### Function

Using this measuring cycle and various \_MVAR measuring variants the following contour elements can be measured at an angle:

\_MVAR=x01 → hole \_MVAR=x01 → shaft \_MVAR=x01 → groove \_MVAR=x01 → web

If no tool offset or ZO correction is to be applied,

set KNUM=0.

Detailed information on the parameters: Refer to Section 2.3.

### Measuring principle for hole or shaft

Inside the hole or outside the hole when traveling around a shaft. the measuring cycle measures points P1, P2, P3, or also P4.

The position of the points is determined by starting angle \_STA1, indexing angles \_INCA, the diameter and the set center point.

These four measured values are used to calculate the actual value of the diameter and position of the center point in the abscissa and ordinate relative to the workpiece zero.

Measurement is performed in the radial direction:

- toward the set center point in the case of shaft,
- away from the set center point in the case of a hole.

The sum of the starting angle plus all incremental angles must not exceed 360 degrees.



### Measuring principle for groove or web

The measuring cycle measures points P1 and P2 inside the groove and outside the web. The actual value of the groove width and web width and the position of the groove center and web center in relation to workpiece zero are calculated from the measured values.

# Options for hole and shaft diameter, groove or web width, and tool compensation

- An empirical value from data block GUD5 can be included with the correct sign.
- A mean value can be derived from several workpieces, measurement calls.

### Precondition

The probe must be called as a tool with a tool length offset.

Tool type, preferably: 710 when using the cycle on a turning machine: set type and \_CBIT[14]=0.

The probe must have been calibrated with "Determine active probe ball diameter". Calibration with an additional "Determine position deviation" of the workpiece probe improves the measuring precision.

### Notice

Precise measurement is only possible with a probe calibrated under the measurement conditions, i.e. working plane, orientation of the spindle in the plane and measuring velocity are the same for both measurement and calibration. Deviations can cause additional measuring errors.





### Sequence

### Specification of setpoints

The setpoint for diameter or width is specified in \_SETVAL.

The setpoint for the center point of the hole, shaft, or for the measured center of the groove, web, is specified by

\_CPA for the abscissa and

\_CPO for the ordinate.

### Measuring axes

Measuring axes are not specified. As a rule, both axes of the plane are included in the measurements, depending on the angle.

### Position before measuring cycle call

In all measuring variants the probe must be positioned at the required **measuring height** in the applicate (tool axis) close to the first measuring point P1.

It must be possible to approach the first measuring point P1 via intermediate point P1' from this position without collision using linear interpolation.

Recommended distance from contour: >\_FA.

### Procedure for hole, shaft

The intermediate positions of the measuring points are approached along a circular path (G2, G3). The distance between the probe ball (ball circumference) and the hole or shaft is \_FA. The travel direction G2 or G3 is derived from the sign of \_INCA: G3 is angle is positive.

The velocity along the circular path is programmed with  $\_\textbf{RF}.$ 

### Procedure for groove

The probe is in the groove and approaches both measuring points one after the other in the selected measuring height along an oblique straight line as defined by angle \_STA1 and which travels through CPA, CPO.







In the measuring variants for a web

\_MVAR=4, \_MVAR=104

additional information for traveling over the web with \_ID is required.

\_ID specifies the distance (with sign) from the measuring height.

### Caution

If \_CBIT[14]=0, length 1 (L1) of the probe refers to the ball center. Radius R is then not taken into account in the length and must be included in \_ID!

Measuring point P2 is approached via P2' along an oblique straight line according to angle \_STA1 and which runts through \_CPA, \_CPO.

P1', P2' are both distance \_FA (path) from the contour.

### Position at end of measuring cycle

At the end of the measuring cycle, the probe (ball circumference) is distance FA (path) from the last measuring point (setpoint) at measuring height.





### Notice

The range of positions of the center or diameter, or groove, web width, must be within the value specified in \_FA for all workpieces to be measured. Otherwise there is danger of collision or the measurement cannot be performed!

# Measuring Cycles for Milling and Machining Centers 5.6 CYCLE979 workpiece: Measure hole/shaft/groove/rib at an angle

### 5.6.2 Measuring and tool offset



### Function

With this measuring cycle and measuring variants \_MVAR=1...4 contour elements hole, shaft, groove, web, can be measured at an angle.

Automatic tool offset is also possible. This tool is specified in **\_TNUM** and **\_TNAME**. The D number and type of offset are specified in coded form in variable **\_KNUM**.

From measuring cycles SW 6.3, extended tool offset is available. With this function a tool from a particular stored tool environment **\_TENV**, and additive, setup offsets can be corrected by specifying the DL number in **\_DLNUM**.

Detailed information on the parameters: Refer to Section 2.3.

### Empirical values and mean values

An empirical value stored in data block GUD5 in array \_EV[] can be included in calculation of the result after measurement is completed.

Optionally, averaging is performed over a number of parts (array \_MV[ ]) and the tolerance bands are checked.

Both are activated in \_EVNUM (see Section 2.3).



Ê	Parameters

_MVAR	1	Measure hole with tool offset	
	2	Measure shaft with tool offset	
	3	Measure groove with tool offset	
	4	Measure web with tool offset	
_SETVAL	REAL, >0	Setpoint diameter, width (acc. to drawing)	
_CPA	REAL	Center point of abscissa	
		(with reference to workpiece zero)	
_CPO	REAL	Center point of ordinate	
		(with reference to workpiece zero)	
_STA1	-360 to	Start angle	
	+360 degrees		
_ID	REAL	Incremental raise of the applicate with sign	
		(only measure for rib, raise to pass)	
_INCA	-360 to	Indexing angle (only for measuring hole or shaft)	
	+360 degrees	Useful values for three-point measurement: -120 + 120 degrees	
		Useful values for four-point measurement: -90 + 90 degrees	
_RF	REAL, >0	Feedrate for circular interpolation (mm/min)	
		(only for measuring hole/shaft)	
_KNUM	0, >0	0: without automatic tool offset	
		>0: with automatic tool offset	
		(individual values: See Section 2.3, parameter _KNUM)	
_TNUM	INT, ≥0	Tool number for automatic tool offset	
_TNAME	STRING[32]	Tool name for automatic tool offset	
		(alternative to _TNUM with tool management active)	
_DLNUM	INT, ≥0	DL number for additive/setup offset	
		(> meas. cycle SW 6.3)	
_TENV	STRING[32]	Name of tool environment for automatic tool compensation	
		(from measuring cycles SW 6.3)	
The followi	ing additional para	meters are also valid	
_VMS, _C	ORA, _TZL, _TM\	/, _TUL, _TLL,	
_TDIF, _TSA, _FA, _PRNUM, _EVNUM,			
_NMSP, ar	nd <b>_K</b> .		
See Sections 2.2 and 2.3.			

\_CORA only relevant for monodirectional probe. The remaining parameters must also be assigned if \_KNUM=0 because they refer to the workpiece. With \_TSA, the diameter or width difference is monitored.

5

### Programming example

### Measuring a hole with CYCLE979

The trueness of a circular segment in plane G17 (semi-circle, contour element "hole") is to be checked. Machining was performed with milling tool T20, D1.

If the amount deviates by <0.01 mm from the set diameter \_SETVAL= 130 mm the tool radius of this tool is to be automatically corrected in terms of wear. The maximum permissible deviation is taken as max. 1 mm. To obtain a minimum measurement path of 1 mm to the contour, the measurement path is programmed with

\_FA = 1+1=2 mm (max. total measurement path=4 mm).

The center point of the circular segment (setpoint) is X=180 mm, Y=0 mm (\_CPA, \_CPO). Measurement is to be performed with three-point

measurement at a measuring height of Z= 20 mm at initial angle 15° and following angles 80°. Traversing between points is carried out with a circular feed of RF= 900 mm/min.

The offset must take the empirical value in memory \_EV[19] into consideration. Mean value calculation \_MV[19] and inclusion in calculation are also to be used.

This tool offset will therefore affect the production of the next workpieces or possible remachining.

Clamping for workpiece: Zero offset, with settable ZO G54: NVx, NVy, ...

The probe is to be workpiece probe 1, inserted as tool **T9**, **D1**. The probe is already calibrated. Arrays for workpiece probe 1: \_WP[0, ...] The following is entered under T9, D1 in the tool offset memory:





5

Tool type (DP1): 710	
Length 1- Geometry (DP3): L1= 50.000	
Radius (DP6): R = 3.000	
Length 1 (L1) refers to the center of the probe ball (CBIT[14]=0) as for calibration	
Careful when positioning Radius R in length (I 1) is	
isseered	
Ignorea.	
%_N_DRILL_SEGMENT_MPF	
N10 G54 G17 G90 T9 D1	;ZO, select tool as probe
N20 M6	;Insert probe,
	;Activate tool offset
N30 G0 X210 Y-20	;Position probe in X/Y plane close to P1
N40 Z20	;Position probe at measuring height
N50 _CHBIT[4]=1	;With mean value calculation
N60 _TUL=0.03 _TLL=-0.03 _EVNUM=20 _K=3	;Set parameters for measuring cycle call
_TZL=0.01 _TMV=0.02 _TDIF=0.06 _TSA=1 _PRNUM=1	;Three-point measurement with probe 1
_VMS=0 _NMSP=1 _FA=2	
N61_MVAR=1_SETVAL=130_STA1=15_INCA=80	
_RF=900 _TNUM=20 _KNUM=2001 _CPA=180 _CPO=0	
N70 CYCLE979	;Call measuring cycle for hole
	;measurement in X/Y
N80 G0 Z160	;Traverse up Z axis
N100 M2	;End of program



### Explanation

The difference calculated from the actual and setpoint diameter is compensated for by the empirical value in the empirical value memory \_EV[19] and compared with the tolerance parameter.

- If it is more than 1 mm (\_TSA), alarm "Safe area violated" is output and program execution cannot be continued.
- If it is more than 0.06 mm (\_TDIF), no compensation is performed and alarm "Permissible dimensional difference exceeded" is output and the program continues.
- If 0.03 mm is exceeded (\_TUL/\_TLL), the radius in T20 D1 is compensated 100% by this difference/2. Alarm "oversize" or "undersize" is displayed and the program continues.
- If 0.02 mm (\_TMV) is exceeded, the radius in T20 D1 is compensated 100% by this difference/2.

10.04

- If it is less than 0.02 mm (\_TMV), the mean value is • calculated from the mean value in mean value memory \_MV[19] and inclusion of weighting factor \_K=3 (only for \_CHBIT[4]=1! with mean value memory).
  - If the mean value obtained is >0.01 (\_TZL), the radius value in parameter T20 D1 is divided by 2 and corrected – and the mean value in \_MV[19] is deleted.
  - If the mean value is < 0.01 (\_TZL), the radius</li> value in T20 D1 is not corrected, but if the mean value memory is active (\_CHBIT[4]=1) it is saved in the mean value memory \_MV[19].

The results are entered in result array \_OVR[].

### 5.6.3 Measurement and ZO determination

### Function

With this measuring cycle and measuring variants \_MVAR=10x

a hole, shaft, groove, or web can be measured at an angle. The zero offset (ZO) of the associated workpiece can also be determined and corrected. A possible rotation of the workpiece is kept without changing it.

The angular position cannot be determined with this cycle.

Compensation of the ZO is executed in such a way that the actual center (position of center on the machine, e.g.: MPx1, MPY1) includes the desired setpoint position in the workpiece coordinate system when the compensated ZO (frame) is applied. Mirroring can be active in a frame of the frame sequence.

Dimension factors must never be active.

The ZO to be corrected is specified in coded form with variable **\_KNUM** >0.

The ZO can be specified and corrected by various methods, e.g. in various settable frames, in various basic frames, system frames, fine offset, or coarse offset, etc.

For detailed information on specifying \_KNUM for the zero offset: Refer to Section 2.3.

# The following applies to all measuring variants with ZO determination in CYCLE979:

The difference between the setpoint (\_CPA and \_CPO) and the actual value of the **center point** derived by the cycle determines the ZO correction (offset). This value is monitored here with \_TSA.



### .04 Measuring Cycles for Milling and Machining Centers 5.6 CYCLE979 workpiece: Measure hole/shaft/groove/rib at an angle 10.04

_	

Parameter	5
-----------	---

Ð	Paramete	rs	
	_MVAR	101	ZO calculation in a hole with ZO compensation
		102	ZO calculation on a shaft with ZO compensation
		103	ZO calculation in a groove with ZO compensation
		104	ZO calculation on a web with ZO compensation
	_SETVAL	REAL, >0	Setpoint diameter, width (acc. to drawing)
	_CPA	REAL	Center point of abscissa
			(with reference to workpiece zero)
	_CPO	REAL	Center point of ordinate
			(with reference to workpiece zero)
	_STA1	-360 to	Start angle
		+360 degrees	
	_ID	REAL	Incremental raise of the applicate with sign
			(only measure for rib, raise to pass)
	_INCA	-360 to	Indexing angle (only for measuring hole or shaft)
		+360 degrees	Useful values for three-point measurement: -120 + 120 degrees
			Useful values for four-point measurement: -90 + 90 degrees
	_RF	REAL, >0	Feedrate for circular interpolation (mm/min)
			(only for measuring hole/shaft)
	_KNUM	0, >0	0: without automatic ZO correction
			>0: with automatic ZO correction
			(individual values: See Section 2.3, parameter _KNUM)
The following additional parameters are also valid		meters are also valid	
_₹	_VMS, _C0	ORA, _TSA, _FA,	_PRNUM and _NMSP
	See Sectio	ns 2.2 and 2.3.	
	_CORA on	ly relevant for mo	nodirectional probe.
	With TCA the conter point difference is manitared		

With \_TSA, the center point difference is monitored.

5

### Measuring a groove and determining the ZO with CYCLE979

The groove width on a workpiece is to be measured in plane G17 and measuring height Z=40 mm. The groove lies at an angle of 70° in its width from the X axis (STA1).

The resulting center of the slot in the measured path should lie in the corrected workpiece coordinate system at

X=150 mm, Y=130 mm (\_CPA, \_CPO).

Any deviation from the selected ZO must be compensated for automatically in G55 by means of additive ZO.

Measurement is also performed with G55.

The maximum conceivable deviation of the groove center is taken as 1 mm. The measuring path is therefore specified as \_FA= 2 mm (max. measuring path = 4 mm) and ensures that there is still a minimum measuring path of 1 mm up to the edge of the groove.

A deviation of < 0.8 mm of the center is permitted. This should be monitored with TSA.

Clamping for workpiece:

Zero offset, with settable ZO G55:

NVx, NVy, (values before the correction [offset])		
%_N_OFFSET_GROOVE_MPF		
N10 G55 G17 G90 T9 D1	;ZO, select tool as probe	
N20 M6	;Insert probe,	
	;Activate tool offset	
N30 G0 X150 Y130	;Position probe in X/Y plane	
	;in setpoint center	
N40 Z40	;Position probe at measuring height	
N60 _TSA=0.8 _PRNUM=1 _VMS=0 _NMSP=1 _FA=2	;Set parameters for measuring cycle call	
N61 _MVAR=103 _SETVAL=130 _CPA=150 _CPO=130		
_STA1=70 _KNUM=2		
N70 CYCLE979	;Call measuring cycle for ZO	
	;determination in X/Y plane	
N80 G0 Z160	;Traverse up Z axis	
N90 G55	;Repeat call of zero offset G55	
	;This activates the changes!	
N100 M2	;End of program	





# =?

### Explanation

Automatic compensation is performed in G55, offset in X and Y by the calculated difference between the actual value and set position of the groove center point, should it be less than 1 mm (\_TSA) in both axes. Otherwise alarm "Safe area violated" is output and program execution cannot be continued. It must be interrupted with NC RESET.

The corrected ZO G55 is activated in block N90.

The results are entered in result array \_OVR[ ].



### 5.7 CYCLE998 workpiece: Angle measurement and ZO determination



### Programming

CYCLE998



### Function

This measuring cycle enables you to determine the angular position of surfaces of a workpiece. This can be used to close the workpiece clamping and correct the ZO as regards **angular position**.

### With 1-angle measurement:

- When a workpiece is clamped rotated in the plane: The angular offset is applied in the rotation component of the geometry axis that is perpendicular to the measurement plane.
- If a workpiece is on a rotary table: The angular offset is applied additively in the translation component of the rotary axis (table axis).

### With 2-angle measurement:

If a workpiece has a plane that is inclined in space:

The angular offsets are applied in the rotation part of the geometry axes.

The angular position is corrected, taking account of set angles in the specified frame (ZO).

### Note:

In this cycle, only the **rotation** components of the frame are determined and corrected (except for rotary table). To complete correction of the ZO (frame), a further measuring cycle is required to determine the translation component (e.g. CYCLE977 or CYCLE978).

An empirical value \_EV[] stored in the GUD5 data block can be included in the measurement result with the correct sign. This is activated in \_EVNUM (see Section 2.3).

A special measuring variant permits **differential measurement** with the axes of the plane. The special procedure for this measurement permits use of an **uncalibrated** multidirectional probe.







### Workpiece probe types that can be used

- Multidirectional probe (\_PRNUM=xy)
- Monodirectional, bidirectional probe
   (\_PRNUM=1xy)

### Notice

A monodirectional or bi-directional probe must be calibrated! These probes cannot be used to take differential measurements!

### Preconditions for differential measurement

- Spindle orientation (with SPOS command) by means of NC
- Bidirectional/multidirectional probe
   Positioning of probe in spindle between 0° and 360°
   (at least even; 00°) (all round experses)

(at least every 90°) (all-round coverage).

### Maximum measurement angle

The cycle is capable of measuring a maximum angle of -45 ...+45 degrees.

### Measuring variants

Measuring cycle CYCLE998 permits the following measuring variants that are specified with parameter \_MVAR.

Value	Measuring variant
105	Angle measurement and ZO determination,
	positioning at an angle from measuring point to measuring point
1105	Angle measurement with differential measurement and ZO determination,
	positioning at an angle from measuring point to measuring point
100105	Angle measurement and ZO determination,
	paraxial positioning from measuring point to measuring point in the offset axis
106	2-angle measurement and ZO determination,
	positioning at an angle from measuring point to measuring point in height
100106	2-angle measurement and ZO determination,
	paraxial positioning from measuring point to measuring point in the height





### **Result parameters**

CYCLE998 makes the following values available as results in the GUD5 data block:

_OVR [0]	REAL	Setpoint angle or
		setpoint angle between workpiece area and 1st axis of the plane
		(abscissa) of the active WCS <sup>1)</sup>
_OVR [1] <sup>1)</sup>	REAL	Setpoint angle between workpiece area and 2nd axis of the plane
		(ordinate) of the active WCS
_OVR [4]	REAL	Actual value angle or
		actual value angle between workpiece area and 1st axis of the plane
		(abscissa) of the active WCS <sup>1)</sup>
_OVR [5] <sup>1)</sup>	REAL	Actual value angle between workpiece area and 2nd axis of the plane
		(ordinate) of the active WCS
_OVR [16]	REAL	Difference angle or
		difference angle about 1st axis of the plane <sup>1)</sup>
_OVR [17] <sup>1)</sup>	REAL	Difference angle about 2nd axis of the plane
_OVR [20]	REAL	Offset value angle
_OVR [21] <sup>1)</sup>	REAL	Offset value angle about 1st axis of the plane
_OVR [22] <sup>1)</sup>	REAL	Offset value angle about 2nd axis of the plane
_OVR [23] <sup>1)</sup>	REAL	Offset value angle about 3rd axis of the plane
_OVR [28]	REAL	Safe area
_OVR [30]	REAL	Empirical value
_OVI [0]	INTEGER	ZO number
_OVI [2]	INTEGER	Measuring cycle number
_OVI [5]	INTEGER	Probe number
_OVI [7]	INTEGER	Empirical value memory number
_OVI [9]	INTEGER	Alarm number
_OVI [11] <sup>1)</sup>	INTEGER	Status offset request

1) For measuring variant \_MVAR=x00106 only

## 10.04 Measuring Cycles for Milling and Machining Centers 5.7 CYCLE998 workpiece: Angle measurement and ZO determination

### 5.7.1 1-angle measurement



### Function

With this measuring cycle and measuring variants \_MVAR=x0x105

it is possible to determine the angular position of a surface in the plane of the workpiece coordinate system.

This can be used to determine and correct the rotation component in the zero offset (ZO, frame) of the workpiece in the plane.

The rotation is corrected in such a way that the real position of the surface (actual value) adopts the required set angle (\_STA1) in the workpiece coordinate system when the corrected ZO (frame) is used.

The ZO to be corrected is specified in coded form with variable **\_KNUM** >0.

The ZO can be specified and corrected by various methods, e.g. in various settable frames, in various basic frames, system frames.

If \_KNUM=0, there is no ZO correction.

For detailed information on specifying \_KNUM for the zero offset: Refer to Section 2.3.

In addition to \_KNUM, another item of data is required to determine the type of angular offset in parameter \_**RA**:

- \_RA=0: Correction of the rotation around the 3rd axis, that is not contained in \_MA (neither measuring nor offset axis)
- \_RA>0: Channel axis number of the rotary table. The angular correction is made in the translatory part of the channel axis \_RA.











# Parameters For 1-angle measurement:

_MVAR	105	Angle measurement and ZO determination, positioning at an angle from measuring point to measuring point
	1105	Angle measurement with differential measurement and ZO determination, positioning at an angle from measuring point to measuring point
	100105	Angle measurement and ZO determination, paraxial positioning from measuring point to measuring point in the offset axis
_SETVAL	REAL, >0	Setpoint (axis position) at measuring point 1 in the meas. axis (for _MVAR=105 and _MVAR=1105 only)
_STA1	REAL,	Setpoint angle
	-45 to +45 deg.	
_MA	201	Measuring axis: 1 (abscissa), offset axis: 2 (ordinate)
	102	Measuring axis: 2 (ordinate), offset axis: 1 (abscissa)
	301	Measuring axis: 1 (abscissa), offset axis: 3 (applicate)
	302	Measuring axis: 2 (ordinate), offset axis: 1 (applicate)
	203	Measuring axis: 3 (applicate), offset axis: 2 (ordinate)
	103	Measuring axis: 3 (applicate), offset axis: 1 (abscissa)
_ID	REAL, >0	Distance between measuring points P1 and P2 in offset axis
_RA	0	Offset of rotation about axis that is not contained in _MA
	>0	Channel axis number of the rotary table
		The angle offset is applied in the translation component of the
		channel axis number (rotary axis).
_MD	INT, 0 or1	0: Positive measuring direction
		1: Negative measuring direction
		MVAR=10x10x)
_KNUM	0, >0	0: without automatic ZO correction
		>0: with automatic ZO correction
		(individual values: See Section 2.3, parameter _KNUM)
The follow	ing additional	CORA only relevant for monodirectional probe.
parameter	s are also valid	With _TSA, the difference of the angle is monitored and this
_VMS, _C	ORA, _TSA, _FA,	value is additionally traversed to _STA1 with intermediate
_PRNUM, _EVNUM and		positioning at an angleTSA has the dimension unit degrees
_NMSP.		in this case!

See Sections 2.2 and 2.3.

### Notice

Precise angle definition requires a minimum surface finish at least at the measuring points. The distances between the measuring points must be selected as large as possible.

## 3

### Sequence

### Measurement axis \_MA

In this cycle, not only the measuring axis but also the offset axis are specified in **\_MA**. The offset axis is the 2nd axis of the measuring plane. Intermediate positioning to the measuring point is performed in this axis for paraxial positioning; for positioning at an angle it is performed in both axes. It is also possible to specify the applicate as the measurement or offset axis.

# Distance measuring point 1 to measuring point 2 in the offset axis: \_ID

Parameter **\_ID** is used to define the distance between P1 and P2 in the offset axis. Only positive values are permissible for \_ID. P1 must therefore be selected in the offset axis before the cycle begins.

### \_STA1 set angle

The setting in \_MA makes all 3 measurement planes possible. The set angle **\_STA1** therefore refers to the positive direction of the offset axis and is negative in the clockwise direction, positive in the counterclockwise direction.

The set angle \_STA1 specifies the required angle between the edge and the positive direction of the offset axis. In the case of \_STA1=0, the edge is aligned paraxially with regards to the offset axis after correction.

With measuring variants "Positioning at an angle" (\_MVAR=00x105), \_STA1 is also used for positioning. The positioning angle is formed together with **\_TSA**. \_STA1 should therefore deviate only a little from the measured angle.



5

# Procedure with MVAR=00x105: Intermediate positioning at an angle

### Position before measuring cycle call

The probe is positioned with respect to the surface to be measured in such a way that during traversal of the measuring axis \_MA specified in the direction of the setpoint \_SETVAL, measuring point 1 on the surface will be reached.

Recommended distance from \_SETVAL: >\_FA. The measuring operation then starts with the measuring feed at distance **\_FA** in front of \_SETVAL.

The absolute value of the positional deviation from the setpoint must not be greater than the measuring path \_FA. Otherwise, no measurement will be performed.

In the other two axes, the positions are retained for the measurement in measuring point 1 at the beginning of a cycle.

### Intermediate positioning at an angle

The starting point for measurement 2 is approached at an angle. The angle comprises \_STA1 and \_TSA. \_TSA contains the value for a permissible angle deviation and leads away from the setpoint.

# Procedure with MVAR=10x105: paraxial intermediate positioning

### Position before measuring cycle call

The probe is positioned with respect to the surface to be measured in such a way that during traversal in the specified measuring axis \_MA and direction of the measurement in \_MD both measuring points on the surface within the

total measuring path: 2 · \_FA in mm

will be reached.

Otherwise no measurement or complete measurement will result.

In the other two axes, the positions are retained for the measurement in measuring point 1 at the beginning of a cycle.

The starting point for measurement 2 is approached **paraxially** in the offset axis. Measuring point 2 is also approached with the measuring axis in direction \_MD.







After the end of measurement, the probe is at the last measuring point at distance \_FA from the measuring surface.

### Notice

Precise measurement is only possible with a probe calibrated under the measurement conditions, i.e. working plane, orientation of the spindle in the plane and measuring velocity are the same for both measurement and calibration. Deviations can cause additional measuring errors.

### Precondition

The probe must be called as a tool with a tool length offset.

Tool type, preferably: 710 when using the cycle on a turning machine: set type and \_CBIT[14]=0.

### Special procedure for differential measurement

**Measuring point 1** is measured twice during differential measurement.

- with spindle rotated through 180 degrees compared with the position at the beginning of the cycle (rotation of the probe by 180 degrees).
- 2. with the spindle position that applied at the beginning of the cycle

The tool radius of the probe + R or - R is defined as the trigger point defined for the axis direction. The values are entered and stored in the global user data (GUD6) in the array for the associated workpiece probe \_WP[].

A **multidirectional** probe does not have to be calibrated at the beginning of the cycle for measuring variant **\_MVAR=1105**. This measuring variant with differential measurement is only useful with the measuring axes \_MA=x01 or \_MA=x02.

### Notice

In the case of great measurement accuracy demands, differential measurement is not recommended!

### Programming example

### 1-angle measurement with CYCLE998

A rectangular workpiece (60 x 40 mm) is clamped in the G17 plane on a rotary table. The intention is to orient it with its edges running parallel with axes X and Y.

An angular deviation detected is to be compensated automatically through additive ZO compensation of the rotary axes. The maximum possible angular deviation is taken as \_TSA=5°. The measuring path is programmed as \_FA=8 mm (max. total measurement path = 16 mm). The measuring points should be 40 mm apart. Intermediate positioning is to be at an angle.

The rotary table is the 4th axis in the channel (C axis).

Clamping for workpiece: Zero offset, with settable ZO G54: NVx, NVy, NVz, NVc

Workpiece probe 1, inserted as tool T9, D1,

is to be used as probe.

The probe is already calibrated. Arrays for

workpiece probe 1: \_WP[0, ...]

The following is entered under T9, D1 in the tool offset memory:

Tool type (DP1):	710
Length 1- Geometry (DP3):	L1= 50.000
Radius (DP6):	R = 3.000

Length 1 (L1) refers to the center of the probe ball (\_CBIT[14]=0), as for calibration. Careful when positioning! Radius R in length (L1) is ignored.

%_N_ANGLEMEAS_MPF	
N10 G54 G17 G90 T9 D1	;Select T No. probe
N20 M6	;Insert probe as tool, activate offset
N30 G0 C0	;Position rotary table at 0°
N40 X-20 Y-40	;Position probe in X/Y plane opposite
	;measuring point
N50 Z40	;Z axis at measurement height
N60 _PRNUM=1 _VMS=0 _NMSP=1 _EVNUM=0	;Set parameters for measuring cycle call
N61 _MVAR=105 _SETVAL=-18 _MA=102 _ID=40	
RA=4 KNUM=1 STA1=0 TSA=5 FA=8	



## 10.04

N70 CYCLE998	;Measuring cycle for angle measurement
N80 G0 Z160	;Traverse up Z axis
N90 G54 C0	;Repeat call of ZO G54
	;This activates the changes!
	;Position rotary table at 0°
	;(edge is now setup).
N100 M2	;End of program

### Explanation

Measurement is performed in the Y direction, offset in the X direction, intermediate positioning at an angle.

The cycle determines the measuring direction from the actual position in the Y direction and \_SETVAL. Automatic correction is performed in G54, offset in the C axis (4th channel axis) with the calculated angle (\_STA1=0).

In block N90, the corrected ZO G54 is activated and

The results are entered in result array \_OVR[].

```
the C axis is moved from position zero to position
zero; the ZO difference is eliminated. The workpiece
is then paraxial.
```

### 5.7.2 2-angle measurement



### Function

Measuring variants

### \_MVAR=106 and \_MVAR=100106

can calculate and correct the angular position of a plane oblique in space on a workpiece by measuring three points The angles refer to rotation about the axes or the active plane G17 to G19. Otherwise, the same conditions apply as for simple angle measurement. Additional data are required for the setpoint input of the 2nd angle. A ZO is implemented in the rotary part of the

set ZO memory (coordinate rotation)

### Parameters

For 2-angle measurement:

· • • = •		
_MVAR	106	2-angle measurement and ZO determination,
		Intermediate positioning at an angle
	100106	2-angle measurement and ZO determination,
		Intermediate positioning parallel to contour
_SETVAL	REAL, >0	Setpoint (axis position): Expected position on workpiece surface
		in measuring point P1 in the applicate (for _MVAR=106 only)
_STA1	REAL	Setpoint angle about 1st axis of the plane
_INCA	REAL	Setpoint angle about 2nd axis of the plane
_MD	0, 1	0: Positive measuring direction
		1: Negative measuring direction
		(for _MVAR=100106 only)
_ID	REAL, >0	Distance between measuring points P1 and P2 in the 1st axis of
		the plane (abscissa)
_SETV[0]	REAL, >0	Distance between measuring points P1 and P3 in the 2nd axis
		of the plane (ordinate)
_KNUM	0, >0	0: without automatic ZO correction
		>0: with automatic ZO correction
		(individual values: See Section 2.3, parameter _KNUM)
The followi	ng additional	_CORA only relevant for monodirectional probe.
parameters	s are also valid	With _TSA, the difference of the angle is monitored and this
_VMS, _C	ORA, _TSA, _FA,	value is additionally traversed to _STA1 with intermediate
_PRNUM,	_EVNUM and _NMSP.	positioning at an angleTSA has the dimension unit degrees
See Sectio	ons 2.2 and 2.3.	in this case!
Notice		
Precise angle definition requires a minimum surface		
finish at least at the measuring points. The distances		
between the measuring points must be selected as		
large as possible.		

### Procedure for 2-angle measurement

### Position before measuring cycle call

Before the cycle is called, the probe must be positioned over the 1st measuring point (P1) in the plane and at the appropriate depth in the applicate. The measuring axis is always the applicate. Measuring point P1 must be selected in the plane such that \_ID and \_SETV[0] result in positive values.

# Procedure for variant "intermediate positioning at an angle" (MVAR=106):

After completion of the measurement in P1 the probe is positioned at P2 in the abscissa and applicate (X and Z in G17) taking angle \_INCA and maximum deviation in \_TSA into account. After the measurement has been performed in P2, repositioning to P1 is performed by the same path. Then the probe is positioned from P1 to P3 in the ordinate and applicate (Y and Z in G17) taking account of angle \_STA1 and maximum deviation in \_TSA and then measured.

# Procedure for variant "intermediate positioning parallel to axis" (MVAR=100106):

Positioning from P1 to P2 is performed in the abscissa, from P1 to P3 in the ordinate. It must also be possible to reach P2 and P3 from starting position P1 in the applicate (in Z in G17) without collision.

### Position after end of measuring cycle

After completion of the measuring operation, the probe will always be amount \_FA (MVAR=106) above the 3rd measuring point in the applicate or, if \_ MVAR= 100106, at the initial height (positioning height).



### Programming example 2

### 2-angle measurement with CYCLE998

(determining an oblique plane in space)

The task is to check the angular position of a machined oblique surface on a workpiece. The result is taken from the result parameters \_OVR[] for evaluation.

A measuring point 1(P1) must be selected where P2 in the ordinate (Y axis in G17) has the same value as P1 and the abscissa value (\_ID) is positive. P3 must still have the same value in the abscissa (X axis in G17) as P1. The ordinate value (\_SETV[0]) must be positive.

Positioning in the applicate must be performed as far as possible parallel with the oblique plane (set angle). The machined oblique plane has set angle about Y: 12 degrees (\_INCA) and about X: 8 degrees (\_STA1), maximum deviation \_TSA= 5 degrees.

Workpiece probe 1, used as tool **T9**, **D1**, is to be used as probe.

The probe is already calibrated. Arrays for workpiece probe 1: \_WP[0, ...]

The following is entered under T9, D1 in the tool offset memory:

*	
Tool type (DP1):	710
Length 1- Geometry (DP3):	L1= 50.000
Radius (DP6):	R = 3.000

Length 1 (L1) refers to the center of the probe ball (\_CBIT[14]=0), as for calibration.

Careful when positioning! Radius R in length (L1) is ignored.

%_N_INCLINEDMEAS_MPF	
N10 G54 G17 G90 T9 D1	;Select T No. probe
N20 M6	;Activate offsets
N30 X70 Y-10	;Position probe in X/Y plane above
	;measuring point
N40 Z40	;Position Z axis at measuring point level and
	;select tool offset
N60 _MVAR=106 _SETV[0]=30 _ID=40 _KNUM=0	;Set parameters for measuring cycle call
_RA=0 _STA1=8 _INCA=12 _TSA=5 _PRNUM=1	
_VMS=0 _NMSP=1 _FA=5 _EVNUM=0	
N520 CYCLE998	;Measuring cycle for measuring the
	;oblique plane
N530 G0 Z160	;Traverse up Z axis
N540 M30	;End of program





### Explanation

Both measured angles are entered in result field \_OVR[]. A ZO correction is not applied (\_KNUM=0).



### Programming example 3

Orientation of an oblique workpiece surface for remachining using CYCLE800

### Initial state

- The workpiece is clamped on the swivel table (swiveling workpiece holder) and aligned roughly paraxially to the machine axes.
- The swivel table is in its home position.
- The probe is in inserted as T9 and positioned in JOG approximately 20 mm above the front left corner of the workpiece to be set up.
- The scratch function is used to define the zero point of the ZO G56 at which the 2-angle measurement is to be performed and the G17 machining plane is defined as X0 Y0 Z20.

### Task

Remachining will be performed with G57 active. The workpiece should be aligned so that for G17, the tool is located vertically on the previously inclined surface, the workpiece zero (G57) is the lefthand corner and the workpiece edges run in parallel to axes X and Y (G57). CYCLE978 should be used to set the 3 edges. To determine the angle, CYCLE998 (1 and 2 angle measurement).



%_N_PLANE_SETUP_MPF	
N500 G56 G17 G90	;Select ZO and machining plane
N505 T9 D1	;Select probe
N506 M6	;Activate tool compensation for probe
N510 CYCLE800(1,"",0,57,0,0,0,0,0,0,0,0,0,-1)	;Align swivel table
N520 \$P_UIFR[4] = \$P_UIFR[3]	;Copy the data of ZO memory G56 to G57
N530 G1 F500 X20 Y25	;Approach of the 1st MP for
	;2 angle measurement in the plane
N540 Z40	;Positioning height in Z, in which all 3 MPs
	;can be approached

5

N550_VMS=0_PRNUM=1_TSA=20_EVNUM=0	;Measuring velocity 300 mm/min, data
_NMSP=1 _FA=40 _STA1=0 _INCA=0 _MVAR=100106	;field 1 for probe, safe area 20°, without
_MD=1 _ID=50 _SETV[0]=35 _KNUM=4	empirical value, number of measurements
	;at same position =1, measurement path
	;40 mm, angles 1 and 2 =0, 2 angle
	;measurement with paraxial positioning,
	;measurement in the minus direction,
	;distance in X between MP1 and MP2 50 mm,
	;distance in Y between MP1 and MP3 35 mm,
	;ZO correction in G57
N560 CYCLE998	;Call measuring cycle
N570 G57	;Activate ZO G57
N580 CYCLE800(1,"",0,57,0,0,0,0,0,0,0,0,0,-1)	;Align swivel table, probe is perpendicular
	;above oblique surface
N590 X20 Y25	;Approach 1st MP in the plane
N600 Z20	;Lower in Z' about 20 mm above surface
N610 _MVAR=100 _SETVAL=0 _MA=3 _TSA=10	;ZO determination on surface, setpoint 0,
_FA=20 _KNUM=4	;meas. axis Z', safe area 10 mm,
	;meas. path 20 mm before and after expected
	;switching position, ZO correction in G57
N620 CYCLE978	;ZO determination on surface in Z' axis for
	;setting the zero in Z'
N625 G57	;Activate the changed zero offset
N630 X20 Y-20	;Place in plane before the front edge
N640 Z-5	;Lower in Z' direction to align the front edge
	;in the X' direction
N650 _MVAR=105 _MA=102 _SETVAL=0 _RA=0	;Angle measurement measuring axis Y',
_STA1=0	;displacement in X' axis, distance between
	;measuring points 50 mm; offset in the
	;rotation part of the ZO memory G57, set
	;angle between edge and X' direction 0
N660 CYCLE998	;Angle measurement by measuring in Y'
	;and displacement between the 2 measuring
	;points in X' with offset in G57
N665 G57	;Activate the changed ZO G57
N680 X20 Y-20	
N690 Z-5	;Position at measuring height before the front
	;edge
N700 _MVAR=100 _MA=2 _SETVAL=0 _FA=10	;ZO determination on surface, meas. in
	;Y' direction, measurement path 10 mm in
	;front of to 10 mm behind expected edge
N710 CYCLE978	;ZO determination on surface with meas.
	;in +Y' direction and ZO in G57 for
	;setting the zero in Y'
N720 G57	;Activate the changed ZO G57
N730 X-20 Y-20	

### 10.04

N740 Y25	;Place in front of the left edge
N750 _MA=1	;Measure in +X'
N760 CYCLE978	;ZO determination on surface, measurement
	;in ;+X' direction, and ZO correction in G57
	;memory. Measurement path 10 mm in front
	;of up to 10 mm behind expected edge to
	;set zero in X'
N770 G57	;Activate the changed ZO G57
N780 Z20	;Raise in Z
	;The oblique surface is now completely set up
<u>.</u>	
N1000 M2	;End of program



### **Comment about CYCLE800**

The swivel cycle CYCLE800 is used to measure and operate on any surface by converting the active workpiece zero and the active tool offset to the oblique surface in the cycle by calling the relevant NC functions, taking account of the kinematic chain of the machine, and positioning the rotary axes. Cycle CYCLE800 is not part of the "measuring cycle package" but of the "standard cycles".

### Explanation

- CYCLE998 (2 angle measurement) measures the oblique workpiece surface and an offset is entered in the rotation part of the ZO memory G57.
- After CYCLE800 has been called, axes X, Y, and Z • and the rotary axes involved are positioned such that the probe is perpendicular above the oblique workpiece surface.
- Subsequent measurement with ZO in the -Z' • direction with CYCLE978 zeroes the workpiece surface in the Z' direction.
- Determining the angular position of the front • workpiece edge with respect to the X' direction and offset in the ZO memory G57 with CYCLE998 aligns the front edge paraxially with the X' direction.
- Then the workpiece zero is precisely defined in the • plane by measuring with the ZO in the +X' direction and +Y' direction with CYCLE978.
- After that, remachining can begin on the setup . surface.



### 5.8 CYCLE961 workpiece: Setting up an internal and external corner

### 5.8.1 Function overview CYCLE961



### Programming

CYCLE961



### Function

The cycle can measure the **position of an internal or external corner** of a workpiece in the selected plane with different measuring variants The position of this corner can also be used as the workpiece zero in a defined zero offset (ZO).

In certain measuring variants an additional offset can be defined.

The measurements are performed with different specified values depending on the measuring variant used:

### Specification of distances and angles

- The workpiece is a rectangle: 3-point measurement
- Unknown workpiece geometry:
   4-point measurement

### **Specification of 4 points**

 Unknown workpiece geometry: 4-point measurement

### Compensation of the zero offset

The ZO correction is applied in the **coarse offset**. If a fine offset is available (MD18600: MM\_FRAME\_FINE\_TRANS=1), it is reset. If \_KNUM=0, there is no zero offset (ZO). When \_KNUM 0, the corresponding ZO for the abscissa and ordinate is calculated in such a way that the calculated corner point becomes the workpiece zero. The rotary component for the applicate (rotation about Z for G17) is offset in such a way that the workpiece coordinate system lies in the plane parallel to the reference edge.

### Workpiece probe type that can be used

Multidirectional probe (\_PRNUM=xy)




#### Notice

Precise measurement is only possible with a probe calibrated under the measurement conditions, i.e. working plane, orientation of the spindle in the plane and measuring velocity are the same for both measurement and calibration. Deviations can cause additional measuring errors.

#### Precondition

The probe must be called as a tool with a tool length offset. Tool type, preferably: 710 when using the cycle on a turning machine: set type and \_CBIT[14]=0.



#### **Measuring variants**

Measuring cycle CYCLE961 permits the following measuring variants that are specified with parameter \_MVAR.

Value Measuring variant

105	Setting-up an internal corner of a square/rectangle, specifying the angle and distances
106	Setting-up an external corner of a square/rectangle, specifying the angle and distances
107	Set up internal corner, specify angle and distances
108	Set up external corner, specify angle and distances
117	Set up internal corner, specify 4 points
118	Set up external corner, specify 4 points

# 5.8.2 Setting up a corner with definition of distances and angles

# Function

With this measuring cycle and measuring variants \_MVAR=105, \_MVAR=106

it is possible to measure and set up the internal and external corners of a **rectangle**,

with measuring variants

# \_MVAR=107, \_MVAR=108

it is possible to measure and set up the internal and external corners of an **unknown workpiece** geometry.

The cycle approaches either 3 (for a rectangle) or 4 measuring points (if workpiece geometry is not known) and calculates the point of intersection of the resulting straight lines and the angle of rotation to the positive abscissa axis of the current plane. If the workpiece geometry is known (rectangle) the corner to be calculated can be offset.

The result, the position of the corner, is stored as an absolute value in the specified ZO (offset and rotation) and in the result parameters \_OVR[].

The measuring points are derived from the specified angle and distances. Measurement is performed paraxially to the existing workpiece coordinate system (WCS).

# Notes:

When **Set up internal corner** the cycle only traverses in the plane at measuring height. With **Set up external corner**, the corner can either be **passed over** the shortest path (lift in applicate) or **traveled around** in the plane.



# .04 Measuring Cycles for Milling and Machining Centers 5.8 CYCLE961 workpiece: Setting up an internal and external corner 10.04

MVAR	105	Set up internal corner of a rectangle		
-		(geometry known, 3 measuring points)		
	106	Set up external corner of a rectangle		
		(geometry known, 3 measuring points)		
	107	Set up internal corner (geometry unknown, 4 measuring points)		
	108	Set up external corner (geometry unknown, 4 measuring points)		
_FA	REAL	Measuring path, only included if _FA larger than internally calculated		
_KNUM	0, >0	0: without automatic ZO correction		
		>0: with automatic ZO correction		
		(individual values: See Section 2.3, parameter _KNUM)		
_STA1	REAL	Approximate angle between positive direction of abscissa to		
		reference edge of workpiece in MCS <sup>1)</sup> (precision: <10 degrees):		
		Negative value in clockwise direction		
		Positive value in counterclockwise direction		
_INCA	REAL	Approximate angle between reference edge and 2 <sup>nd</sup> edge of workpiece		
		(precision: <10 degrees):		
		Negative value in clockwise direction		
		Positive value in counterclockwise direction		
_ID	REAL	Incremental retraction in applicate when measuring external corner,		
		used to pass over the corner		
		special <b>_ID=0:</b> The corner is traveled around – not passed over.		
_SETV[0]	REAL, >0	Distance between starting point and measuring point 2 STA1		
		(P1 is at _SETV[0] / 2)		
_SETV[1]	REAL, >0	Distance between starting point and measuring point 4 in direction STA?		
		INCA (P3 is at _SETV[1] / 2)		
For measur	ing variants	105 and 106 only (rectangle):		
_SETV[2]	REAL	Offset of zero offset WCS (corrected) in abscissa		
_SETV[3]	REAL	Offset of zero offset WCS (corrected) in ordinated		
_SETV[4]	REAL,	Selection of offset:		
	possible va	lues: 1: Measured corner entered as zero point		
	1, 2, 3, 4	2: Measured corner is entered as zero point offset by _SETV[3		
		3: Measured corner is entered in both axes offset by		
		_SETV[2] (abscissa) and _SETV[3] (ordinate).		
		4: Measured corner is entered as zero point		
		offset by _SETV[3].		
1) Transformation deactivated, otherwise basic coordinate system				

See Sections 2.2 and 2.3.

5

# Result parameters

Measuring cycle CYCLE961 places the following values in data block GUD5 are results:

_OVR [4]	REAL	Angle to abscissa axis) in the workpiece coordinate system (WCS)
_OVR [5]	REAL	Actual value for corner point in abscissa in WCS
_OVR [6]	REAL	Actual value for corner point in abscissa in WCS
_OVR [20]	REAL	Angle to abscissa axis in the workpiece coordinate system (WCS) <sup>1)</sup>
_OVR [21]	REAL	Actual value for corner point in abscissa in MCS <sup>1)</sup>
_OVR [22]	REAL	Actual value for corner point in abscissa in MCS <sup>1)</sup>
_OVI [2]	INTEGER	Measuring cycle number
_OVI [3]	INTEGER	Measuring variant
_OVI [5]	INTEGER	Probe number
_OVI [9]	INTEGER	Alarm number
1) Transformation deactivated, otherwise basic coordinate system		

# Sequence

# Defining distances and angles





10.04

#### Position before measuring cycle call

The probe is positioned at measuring depth opposite the corner to be measured. It must be possible to approach the measuring points from here without collision. The measuring points are derived from the programmed distance between the starting point and

\_SETVAL[0] (measuring point 2), or

\_SETVAL[1] (meas. point 4) in direction of angle \_STA1, \_INCA.

The starting points for measuring point 1 and 3 are each located at half the distance.

Measurement is performed paraxially with the existing workpiece coordinate system.

\_STA1 refers to the machine coordinate system. The measuring cycle generates the required traversing blocks and performs the measurements at the measuring points.

During travel an additional tolerance angle of 10 degrees is added to the programmed angles in the cycle.

First measuring point P 2, then P 1, P 3, and depending on parameterization, P 4 is approached.

#### Traversing between P 1 and P 3 on outside edge:

\_ID=0: The corner is traveled around

ID>0: At P 1, the probe is lifted in the applicate after measurement of ID and approached via corner P.

#### Position after end of measuring cycle

The probe is again positioned at the starting point (at measuring depth opposite the measured corner).

#### Selection of offset with \_SETV[4]

With measuring variants MVAR=105, =106 (rectangle) the measured corner can be selected offset as workpiece zero.

The offset is specified in SETV[2] (abscissa) and \_SETV[3] (ordinate).

SETV[4] can assume values 1 to 4.







# Programming example

The coordinates of the external corner of a workpiece with unknown geometry are to be determined.

Zero offset G55 is to be corrected in such a way that this corner is workpiece zero for G55.

The reference edge lies approximately at \_STA1=-35 and the 2nd edge approximately at \_INCA= 80 degrees in addition. The distance to measuring points 2 and 4 is 100 mm.

The corner is to be passed over from P1 to P3 at distance \_ID= 30 mm above measuring height. The starting point opposite the corner that is to be set

up is reached before the measuring cycle is called.

Workpiece probe 1, inserted as tool **T9**, **D1**, is to be used as probe. The probe is already calibrated. Arrays for workpiece probe 1: \_WP[0, ...] The following is entered under T9, D1 in the tool

offset memory:

Tool type (DP1):	710
Length 1- Geometry (DP3):	L1= 50.000
Radius (DP6):	R = 3.000

Length 1 (L1) refers to the center of the probe ball (\_CBIT[14]=0), as for calibration. Careful when positioning! Radius R in length (L1) is ignored.

# %\_N\_CORNER\_SETUP\_MPF

N10 G500 G17 G90 T9 D1	;Select probe, offset active	
N20 _PRNUM=1 _VMS=0 _NMSP=1	;The probe is at the starting position,	
N21 _MVAR=108 _FA=20 _KNUM=2 _STA1=-35	;e.g. by traversing in JOG	
_INCA=80 _ID=30 _SETV[0]=100 _SETV[1]=100	;Parameterize CYCLE961	
N30 CYCLE961		
N40 G55	;Call corrected ZO G55	
N100 M2		



10.04

#### 5.8.3 Setting up a corner with 4 points



### Programming

CYCLE961



# Function

With this measuring cycle and measuring variants \_MVAR=117, \_MVAR=118

it is possible to measure and set up the internal and external corners of an unknown workpiece geometry.

One after the other, points P2, P1, P3, P4 are traversed in the cycle with positioning feedrate at positioning height. At each of these points the probe is lowered to measuring depth and then traversed at measuring feedrate parallel to the axis against the workpiece corner.

The cycle uses the relative positions of points P1 to P4 to determine the approach directions and the measuring axis. The cycle calculates the corner point and the angle between the reference edge and the positive abscissa axis of the current plane from the results of measurement.

The angle is calculated by measuring P2 and P1 (reference edge).

The position of corner, corner point coordinates, and rotation are stored in result parameter \_OVR[].

If KNUM>0, absolute correction to the coarse offset in the specified ZO (translation and rotation) is performed.

The measuring points are derived from the specified 4 points. Measurement is performed paraxially to the existing workpiece coordinate system (WCS).



Ô

### Parameters

_MVAR	117	Set up internal corner, specify 4 points	
	118	Set up external corner, specify 4 points	
_FA	REAL	Measurement path	
_KNUM	0, >0	0: without automatic ZO correction	
		>0: with automatic ZO correction	
		(individual values: See Section 2.3, parameter _KNUM)	
_ID	REAL	Incremental infeed in applicate for measuring depth	
_SETV[0]	REAL	Abscissa P1 in active WCS	
_SETV[1]	REAL	Ordinate P1 in active WCS	
_SETV[2]	REAL	Abscissa P2 in active WCS	
_SETV[3]	REAL	Ordinate P2 in active WCS	
_SETV[4]	REAL	Abscissa P3 in active WCS	
_SETV[5]	REAL	Ordinate P3 in active WCS	
_SETV[6]	REAL	Abscissa P4 in active WCS	
_SETV[7]	REAL	Ordinate P4 in active WCS	
The fellow	بمنابا املم مسميا		

The following additional parameters are also valid

\_VMS, \_PRNUM, and \_NMSP.

See Sections 2.2 and 2.3.

# **Result parameters**

Measuring cycle CYCLE961 places the following

values in data block	GUD5 are results:
----------------------	-------------------

_OVR [4]	REAL	Angle to abscissa axis) in the workpiece coordinate system (WCS)
_OVR [5]	REAL	Actual value for corner point in abscissa in WCS
_OVR [6]	REAL	Actual value for corner point in abscissa in WCS
_OVR [20]	REAL	Angle to abscissa axis in the workpiece coordinate system (WCS) <sup>1)</sup>
_OVR [21]	REAL	Actual value for corner point in abscissa in MCS <sup>1)</sup>
_OVR [22]	REAL	Actual value for corner point in abscissa in MCS <sup>1)</sup>
_OVI [2]	INTEGER	Measuring cycle number
_OVI [3]	INTEGER	Measuring variant
_OVI [5]	INTEGER	Probe number
_OVI [9]	INTEGER	Alarm number
1) Transformation deactivated, otherwise basic coordinate system		

### Sequence

#### Defining the 4 points

The position of points P1 and P2 in relation to each other determines the direction of the abscissa axis (X axis in G17) of the new coordinate system. A negative offset between P1 and P2 in the abscissa (X axis in G17) results in an additional rotation about 180°! The position of the corner is selected with all 4 points. So for a rectangle, for example, different corners can be selected as the zero point depending on whether measuring variant internal or external corner is applied.

The individual points or FA must be selected such that the contour is reached within a

total measuring path: 2 · \_FA in mm. Otherwise, no measurement will be performed.

A minimum total measurement path of

2 • 20 mm is generated internally.

#### Position before measuring cycle call

The probe is above the workpiece at positioning height. It must be possible to reach all points without collision.

The measuring cycle generates the traversing blocks and performs the measurements at the measuring points from points P1 to P4. The measuring depth is derived from the positioning height lowered by the value in \_ID (negative sign). After measurement at one point the probe is again raised to positioning height and traversed to the next point and then lowered again to measuring depth.

Point P2 is approached first, followed by P1, P3, and P4.

#### Position after end of measuring cycle

The probe is at the positioning height at point P4.











# Programming example

The coordinates of the corner of a workpiece are to be determined by outside measurement. ZO G55 must be corrected in such a way that the corner point is workpiece zero when G55 is selected.

Measurement is performed in plane G17 with active G54. The values in mm of the coordinates of points P1...P4 from which the workpiece can be approached parallel to the axis are:

P1.y=20
P2.y=20
P3.y=40
P4.y=80.

The probe is to be positioned at a height of 100 mm. The measuring depth is 60 mm lower. The workpiece corner is expected to be at a

distance less than 200 mm at each point (\_FA=100[mm]).

Workpiece probe 1, used as tool **T9**, **D1**, is to be used. The probe is already calibrated. Arrays for workpiece probe 1: \_WP[0, ...]

The following is entered under T9, D1 in the tool offset memory:

710
L1= 50.000
R = 3.000

Length 1 (L1) refers to the center of the probe ball (\_CBIT[14]=0), as for calibration. Careful when positioning! Radius R in length (L1) is ignored.

%_N_CORNER_SETUP_1_MPF	
N10 G54 G17 G90 T9 D1	;Select ZO, plane, probe, …
N20 G0 Z100	;Position probe at measuring height
N30 X100 Y70	;Position probe in X/Y plane above
	;workpiece
N50 _MVAR=118 _SETV[0]=50 _SETV[1]=20	;Measuring variant external corner
_SETV[2]=150 _SETV[3]=20 _SETV[4]=15 _SETV[5]=40	;Coordinates from P1 to P4
_SETV[6]=15 _SETV[7]=80 _ID=-60	;Measurement path 100 mm to expected
N51_VMS=0_NMSP=1_PRNUM=1_FA=100_KNUM=2	;edge (max. measurement path=200 mm)
N60 CYCLE961	;Cycle call





N70 G55	;Call corrected ZO G55
N80 G0 X0 Y0	;Position probe in X/Y plane above
	;corner (new zero point)
N100 M2	;End of program

5.9 CYCLE997 workpiece: Measuring spheres and ZO determination (from measuring cycles SW 6.3)

# 5.9.1 Function overview CYCLE997



# Programming

CYCLE997



# Function

With measuring cycle CYCLE997, different measuring variants can be used to measure

- a sphere or
- 3 equal sized spheres, fixed to a common base (workpiece).

Measurement is performed **paraxially** with the existing workpiece coordinate system (WCS) **or at an angle** in the plane.

The center point (position of sphere) is derived from 4 or 5 measuring points around the circumference with a known diameter. With an additional measurement the diameter can also be determined. Intermediate positioning for measuring points P1 to P3 and P4 (determining circle in plane) is performed with measuring variant "at an angle" on a circular path, otherwise parallel with the axis. Positioning in the infeed axis and between the spheres is always a linear movement.

CYCLE997 can measure the sphere and in addition automatically correct a **zero offset (ZO)** on the basis of the position of the center of the sphere.

With "Measure 3 spheres" the angles in space of the sphere grouping can also be determined. The target of the ZO correction can be selected with \_KNUM.



# Workpiece measuring probe type that can be used

Multidirectional probe (\_PRNUM=xy)

Measurement is performed in all three coordinate axes.

Different lengths can be specified for the probe in \_CBIT[14]:

- \_CBIT[14]=0: L1 referred to probe ball center
- \_CBIT[14]=1: L1 is the total length

In measurement the same setting as for calibration of the workpiece probe must be used.

### Notice

Precise measurement is only possible with a probe calibrated under the measurement conditions, i.e. working plane, orientation of the spindle in the plane and measuring velocity are the same for both measurement and calibration. Deviations can cause additional measuring errors.

# Н

#### Precondition

The probe must be called as a tool with a tool length offset. Tool type, preferably: 710

For "Measuring at an angle" (\_MVAR=xx1109) the probe must have been calibrated with "Determine active probe ball diameter". Calibration with an additional "Determine position deviation" improves the measuring precision.

In ZO (frame) the approximate values for the position of the spheres in offset and rotation are entered and activated.

Only small deviations are expected. The sphere diameter must be much larger than the probe ball diameter.

### Important

The user must select measuring points for the particular measuring variant such that during measurement or intermediate positioning a collision with a sphere fixture or other obstacle is ruled out.



10.04



# **Measuring variants**

Measuring cycle CYCLE997 permits the following measuring variants which are specified via parameter \_MVAR.

Va	lue						Measuring variant
7	6	5	4	3	2	1	
				1		9	Measuring a sphere and ZO determination
					0		No measurement repetition
					1		With measurement repetition (with derived values)
			0				Measurement paraxial (to axes of the WCS)
			1				Measurement at an angle (intermediate positioning on
							circular path)
		0					Measure 1 sphere
		1					Measure 3 spheres
	0		1				3 circular measuring points (for "Measuring at an angle" only)
	1		1				4 circular measuring points (for "Measuring at an angle" only)
0							Without diameter determination (ball diameter known)
1							With diameter determination

### Parameters

_SETVAL	REAL	Setpoint sphere diameter	
_SETV[0]	REAL	Setpoint center abscissa – 1st sphere	
_SETV[1]	REAL	Setpoint center ordinate – 1st sphere	
_SETV[2]	REAL	Setpoint center applicate – 1st sphere	
_SETV[3]	REAL	Setpoint center abscissa – 2nd sphere <sup>1)</sup>	
_SETV[4]	REAL	Setpoint center ordinate – 2nd sphere <sup>1)</sup>	
_SETV[5]	REAL	Setpoint center applicate – 2nd sphere <sup>1)</sup>	
_SETV[6]	REAL	Setpoint center abscissa – 3rd sphere <sup>1)</sup>	
_SETV[7]	REAL	Setpoint center ordinate – 3rd sphere <sup>1)</sup>	
_SETV[8]	REAL	Setpoint center applicate – 3rd sphere <sup>1)</sup>	
_RF	REAL	Velocity for intermediate paths on circular path (G2 or G3)	
		(for _MVAR=xx11x9, - "Measuring at an angle" only)	
_KNUM	INT	0: without automatic ZO correction	
	0, >0	>0: With automatic ZO correction	
		(individual values: See Section 2.3, parameter _KNUM)	
_STA1	REAL	Starting angle (for _MVAR=xx11x9, – "Measuring at an angle" only)	
_INCA	REAL	Incremental angle (for _MVAR=xx11x9, - "Measuring at an angle" only)	
TNVL	REAL	Limit value for distortion of triangle (sum of deviations)	
		ZO is only corrected if the calculated distortion is below this limit value.	
		(for _MVAR=x1x1x9 – "Measuring 3 spheres" and _KNUM>0 only)	
1) For measuring variants _MVAR=x1x1x9, measuring 3 spheres, only			
The followi	The following additional parameters are also valid		

\_FA, \_TSA, \_VMS, \_PRNUM and \_NMSP.

See Sections 2.2 and 2.3.

#### **Result parameters**

The measuring cycle CYCLE997 supplies the following values as results in the GUD5 block:

_OVR [0]	REAL	Setpoint sphere diameter, 1st sphere
_OVR [1]	REAL	Setpoint center point coordinate abscissa 1st sphere
_OVR [2]	REAL	Setpoint center point coordinate ordinate 1st sphere
_OVR [3]	REAL	Setpoint center point coordinate applicate 1st sphere
_OVR [4]	REAL	Actual value sphere diameter 1st sphere
_OVR [5]	REAL	Actual value center point coordinate abscissa 1st sphere
_OVR [6]	REAL	Actual value center point coordinate ordinate 1st sphere
_OVR [7]	REAL	Actual value center point coordinate applicate 1st sphere
_OVR [8]	REAL	Difference sphere diameter 1st sphere
_OVR [9]	REAL	Different center point coordinate abscissa 1st sphere
_OVR [10]	REAL	Difference center point coordinate ordinate 1st sphere
_OVR [11]	REAL	Difference center point coordinate applicate 1st sphere
_OVR [12]	REAL	Actual value sphere diameter 2nd sphere <sup>1)</sup>
_OVR [13]	REAL	Actual value center point coordinate abscissa 2nd sphere <sup>1)</sup>
_OVR [14]	REAL	Actual value center point coordinate ordinate 2nd sphere <sup>1)</sup>
_OVR [15]	REAL	Actual value center point coordinate applicate 2nd sphere <sup>1)</sup>
_OVR [16]	REAL	Difference sphere diameter 2nd sphere <sup>1)</sup>
_OVR [17]	REAL	Difference center point coordinate abscissa 2nd sphere <sup>1)</sup>
_OVR [18]	REAL	Difference center point coordinate ordinate 2nd sphere <sup>1)</sup>
_OVR [19]	REAL	Difference center point coordinate applicate 2nd sphere <sup>1)</sup>
_OVR [20]	REAL	Actual value sphere diameter 3rd sphere <sup>1)</sup>
_OVR [21]	REAL	Actual value center point coordinate abscissa 3rd sphere <sup>1)</sup>
_OVR [22]	REAL	Actual value center point coordinate ordinate 3rd sphere <sup>1)</sup>
_OVR [23]	REAL	Actual value center point coordinate applicate 3rd sphere <sup>1)</sup>
_OVR [24]	REAL	Difference sphere diameter 3rd sphere <sup>1)</sup>
_OVR [25]	REAL	Difference center point coordinate abscissa 3rd sphere <sup>1)</sup>
_OVR [26]	REAL	Difference center point coordinate ordinate 3rd sphere <sup>1)</sup>
_OVR [27]	REAL	Difference center point coordinate applicate 3rd sphere <sup>1)</sup>
_OVR [28]	REAL	Safe area
_OVI [0]	INTEGER	ZO number
_OVI [2]	INTEGER	Measuring cycle number = 997
_OVI [5]	INTEGER	Probe number
_OVI [9]	INTEGER	Alarm number
_OVI [11]	INTEGER	Status offset request
_OVI [12]	INTEGER	Additional error information on alarm output, internal measurement
		evaluation
1) =		



1) For measuring variants \_MVAR=x1x1x9, measure 3 spheres, only

5

### 5.9.2 Measurement and ZO determination

#### Measurement and calculation strategy

At the beginning of the cycle the probe must be in the infeed axis at safety height. It must be possible to reach all spheres from here without collision. The cycle starts with measurement of the 1st sphere. Active G17 to G19 defines the plane with abscissa, ordinate. The applicate is the infeed axis. 4 or 3 measuring points are approached at the height of the center point setpoint of the applicate. The actual center point of the circle in the plane is calculated internally from these measured values (center of sphere in plane). For measurement "at an angle" the auxiliary cycle CYCLE116 is used for calculation. The last measuring point is located exactly above the calculated sphere center in the plane and is approached using the applicate as the measuring axis.

The actual **sphere center point** in abscissa, ordinate, applicate is calculated from all these measuring points.

If measuring variant "Measuring 3 spheres" is selected with \_MVAR, these spheres are then measured in the order sphere 2, sphere 3 in the same way.

#### Selecting the measuring variant

With variant "Measure at angle" (\_MVAR=0x1109) fast calculation of the sphere position is possible if the sphere diameter is known (low number of measuring points and few intermediate positioning actions) "Paraxial measurement" (\_MVAR=0x0109) always requires 5 measuring points with more intermediate positioning actions.

In both types of measurement it is possible to **repeat measurement** with the located sphere center point (\_MVAR=xxx119). Repeating measurement improves the measuring result.

It is also possible to calculate the **sphere diameter** (\_MVAR=10xx1x9). In this case an additional measurement is taken parallel to the axis in the plus direction of the abscissa at the height of the sphere center calculated in the first measurement. Calculation of the sphere diameter and measurement repetition can be combined (\_MVAR=10xx119). Here the diameter is calculated after each position measurement.





### Safe area

All setpoint/actual differences are checked for compliance with the safe area (parameter \_TSA). If this value is exceeded, alarm "61303 Safe area violated"

is output and an NCE RESET is necessary. Measurement is then canceled.

If \_CBIT[0]=1 the measurement is first repeated.

# ZO correction when measurement of one sphere only (\_MVAR=x0x1x9):

The actual value/setpoint differences of the center point coordinates are included the calculation of the translatory component of the ZO. The offset acts such that the calculated sphere center point in the offset ZO includes the specified setpoint position (workpiece coordinates, 3 axes).

The offset is applied to the ZO number as defined in \_KNUM. When KNUM=0, there is no offset. No scaling factors must be active in the ZOs / frames.

Settings in \_CHBIT[21]:

You can set whether a ZO offset should be FINE or COARSE in the translation component.

• \_CHBIT[21]=0:

Offset is additive in FINE (if FINE is available as set in the MD, otherwise in COARSE).

• \_CHBIT[21]=1:

Offset is COARSE, FINE is included in calculation and then reset (if FINE is available as set in the MD).



#### **ZO** correction when

#### measuring 3 spheres (\_MVAR=x1x109):

Compensation of the entire active frame with its translational and rotary components is performed after 3 spheres have been measured with cycle CYCLE119 (see Subsection 5.9.4 "CYCLE119..."). No mirroring or scaling factor may be active. The offset acts such that the triangle formed by the 3 sphere center points includes the specified setpoint center positions (workpiece coordinates). The sum of the deviations of the spheres in relation to each other (distorsion) must lie within the value in **\_TNVL**. Otherwise no offset is performed and an alarm is output.

Offset compensation is always performed in COARSE (as described for \_CHBIT[21]=1).

#### Note:

In this measuring variant (Measure 3 spheres) compensation in an NCU-global basic frame is not possible (\_KNUM=1051 to 066). This frame has no rotation component.



#### Sequence

#### Position before measuring cycle call

Before measuring cycle CYCLE997 is called the probe must be positioned at safety height above the set sphere center point (setpoints in \_SETV[...]) of the 1st sphere.

#### General

The measuring cycle generates the travel movements for approaching the measuring points itself and executes the measurements according to the selected measuring variant.

#### Note

The value selected for parameter **\_FA** should be so large that all spheres can be reached within total measurement path 2 · \_FA. Otherwise, no measurement will be performed or they will be incomplete.



# 10.04 Measuring Cycles for Milling and Machining Centers 5.9 CYCLE997 workpiece: Measuring spheres and ZO determination

#### **Further procedure**

# Measuring variant "Paraxial measurement" (MVAR=x01x9):

All intermediate positioning actions and measurement movements are paraxial with the active workpiece coordinate system. The measuring points are approached at distance \_FA from the sphere lateral surface (setpoint sphere diameter). Measuring point P1 is approached first. After positioning in the abscissa, ordinate, the applicate is lowered to the height of the center point setpoint and the 1st measurement is taken Then P2 to P4 are approached and measured parallel to the axis.

P2 is approached via positioning of the applicate at distance \_FA above the sphere (setpoint diameter) and lowering to measuring height again (setpoint center point of applicate). P3 and P4 are approached in the same way.

P3 and P4 lie at the center derived from P1 and P2 (actual value center point of abscissa). The applicate is positioned from P4 to a distance \_FA above the sphere and then approached in the abscissa, ordinate of the calculated actual value point (P5). The last measurement is taken: in the minus direction of the applicate.

After this measurement, the applicate is positioned to safety height (height as beginning of the cycle). When 3 spheres are measured, the abscissa, ordinate are positioned toward the **set center point of the next sphere**. Procedure continues as above.

No sphere fixture or other obstacle must be located in this entire traversing range. It may be necessary to select this measuring variant with intermediate positioning on a circular path (\_MVAR=xx1109). The position and number of measuring points is then variable.



#### **Further procedure**

Measuring variant "Measure at an angle" (\_MVAR=xx11x9):

With \_STA1 (starting angle) the angle position of P1 is defined, with \_INCA, the incremental angle after P2 and then after P3. If the measuring variant is selected with 4 measuring points on a circle (\_MVAR=1x1109), \_INCA is also valid from P3 to P4.

The measuring points are approached at distance \_FA from the sphere lateral surface (setpoint sphere diameter). Measuring point P1 is approached first. After joint positioning of abscissa and ordinate, the applicate is lowered to the height of the center point setpoint of the applicate and the 1st measurement is taken radially in the direction center point setpoint abscissa/ordinate.

Then P2 to P3 and P4 are measured on a **circular path** with feedrate \_RF and measured in the same way as P1.

The applicate is positioned from P4 or P3 to a distance \_FA above the sphere and then approached in the abscissa and ordinate of the calculated actual value point (P5). The last measurement is taken: in the minus direction of the applicate.

After this measurement, the applicate is positioned to safety height (height as beginning of the cycle). When 3 spheres are measured the abscissa and ordinate are positioned simultaneously **toward measuring point P1 of the next sphere** and continued as described above.

No sphere fixture or other obstacle must be located in this entire traversing range.

The sum of the starting angle \_STA1 and all incremental angles \_INCA may not exceed 360 degrees.

#### Position after end of measuring cycle

At the end of the cycle the probe is located above the first calculated actual center point of the 3rd or only sphere at safety height (height same as at beginning of cycle).



### 5.9.3 Programming example CYCLE997



### Programming example

**Determining positional deviations in space Three spheres** each with a diameter of 50 mm are measured. Sphere center points 1 to 3 are expected for (X,Y,Z)=(100, 100, 100), (600, 100, 100) and (1100, 1100, 100).

ZO correction of the active frame is to be performed in accordance with the measured values. The exact diameter of the sphere is known.

A measured sphere lateral surface with a maximum deviation of  $\pm$  5 mm is expected (--> \_FA=5).

Fixture of workpiece (ZO) with G54: NVx, NVy, NVz

Workpiece probe 1, inserted as tool **T20**, **D1**, is to be used as probe.

The probe is already calibrated. Arrays for workpiece probe 1: \_WP[0, ...]

The following is entered under T20, D1 in the tool offset memory:

Tool type (DP1):	710
Length 1- Geometry (DP3):	L1= 50.000
Radius (DP6):	R = 3.000

Length 1 (L1) refers to the center of the probe ball (\_CBIT[14]=0), as for calibration. Careful when positioning! Radius R in length (L1) is ignored.



%_N_ DETERMINE_KS _MPF	
N10 G17 G54	;X-Y plane, active ZO
N20 T20 D1	;Select and activate
	;probe with tool offset D1 (M6)
N30 G0 G90 Z200	;Approach position Z at safety height
N40 X100 Y100	;Approach position X, Y of 1st sphere
N50 _SETVAL=50 _SETV[0]=SET(100,100,100, 600,	;Setpoint parameters for measuring cycle
100, 100,1100, 1100, 100)	;call
N60 _MVAR=010109 _KNUM=99999 _TNVL=1.2	;Measure 3 spheres parallel to axis
	;Offset in active frame,
	;The offset is applied only if calculated
	;distortion is less than 1.2 mm
N70 _VMS=200 _NMSP=1 _FA=5 _PRNUM=1	;Measuring velocity 200 mm /min
	;Measurement at same measuring point
	;Measurement path 5 mm in front of to 5 mm
	;behind setpoint position (sphere lateral
	;surface)
	;Probe array _WP[0,09])
N100 CYCLE997	;Call measuring cycle
N200 M2	;End of program

# 5.9.4 CYCLE119: Arithmetic cycle for calculating position in space



# **Function**

This auxiliary cycle calculates the deviations in position and angle to the active frame from 3 defined setpoint positions in space (reference triangle) and 3 actual positions, and corrects a selected frame if necessary.

CYCLE119 is called as a subroutine by measuring cycle CYCLE997.

To allow this cycle to be used universally, its data are transferred via parameters.



# 10.04 Measuring Cycles for Milling and Machining Centers 5.9 CYCLE997 workpiece: Measuring spheres and ZO determination





# Programming

CYCLE119(\_SETPOINT,\_MEASPOINT,\_ALARM,\_RES, \_REFRAME,\_COR,\_RESLIM)

# Parameters

Input data	Data type	Meaning	
SETPOINTI3 31	REAL	Field for 3 setpoint positions in the sequence 1st 2nd 3rd geometr	
		axis $(X \vee Z)$	
		These points are the reference triangle	
		Field for 3 massured positions in the sequence	
	NEAL	1 2 2 Coomptry avia (X X Z) This is the real position in anose of	
		(x, y, z). This is the real position in space of the described triangle	
COR	INT	0: No compensation	
_0011		1 99 <sup>.</sup> 70 correction in G54 G57 G505 G599	
		1000: ZO correction of last active channel basic frame	
		acc. to MD 28081	
		1011 to 1026: 70 correction in channel basic frame	
		2000: ZO correction in scratch system frame \$P_SETEP	
		2000. ZO correction in active frame, actually frame	
		9999. 20 correction in active frame, settable frame	
		active basic frame as per \$P_CHBFRMASK	
_RESLIM	REAL	Limit value for distortion (only relevant if _COR >0)	
		If _RES is below this limit value, ZO is offset,	
		otherwise an alarm is output.	
Output data	Data type	Meaning	
The results of calcu	lation are st	tored in these transfer parameters	
		Cycle clerm number for foodback	
	INT	(transfer value must be 0 on cycle call)	
RES	REAL	Result of calculation	
		< 0. No frame was calculated An alarm (ALARM > 0) is	
		returned.	
		>=0: Calculation was successful. The size of the value is a	
		measure of the distortion of the triangle, for example, due to	
		inaccurate measurements. It is the sum of the deviations	
		of the individual points in mm.	
_REFRAME	FRAME	Result frame, difference from active frame	
		It this result frame is linked to the active frame, the measured	
		triangle position is given the desired setpoint position (workpiece	
		COORDINALES).	

### Notes for compensation:

The frame to be corrected must not contain any mirroring or scaling factors. If no channel basic frame exists for G500, a cycle alarm (\_ALARM>0) is output. The new frame data with renewed programming of the G command of the associated settable frame

is activated (G500, G54 to ...) outside this cycle by the user.



- 24	

# Programming example Application of CYCLE119

%_N_ Check _MPF		
;Calculate new frame according to transferred po	oints and correct in active frame	
;Apply (_COR=9999) if distortion is _RES < 1.2 mm:		
DEF REAL _SETPOINT[3,3],_MEASPOINT[3,3	]	
DEF REAL _RES, _RESLIMIT		
DEF INT _ALARM		
DEF FRAME _REFRAME		
N10 G17 G54 T1 D1		
N20 _SETPOINT[0,0]=SET(10,0,0)	;Setpoint coordinates 1st point (X1,Y1,Z1)	
N30 _SETPOINT[1,0]=SET(0,20,0)	;Setpoint coordinates 2nd point (X2,Y2,Z2)	
N40 _SETPOINT[2,0]=SET(0,0,30)	;Setpoint coordinates 3rd point (X3,Y3,Z3)	
;Program section for determining actual workpie	ce coordinates of the 3 points:	
;Assignment of derived values:		
N100_MEASPOINT[0,0]=SET(11,0,0)	;Actual value coordinates 1st point	
	(X1,Y1,Z1)	
N110 _MEASPOINT[1,0]=SET(1,20,0)	;Actual value coordinates 2nd point	
	(X2,Y2,Z2)	
N120 _MEASPOINT[2,0]=SET(1,0,30)	;Actual value coordinates 3rd point	
	(X3,Y3,Z3)	
;Calculation with compensation in G54:		
N200 CYCLE119(_SETPOINT,_MEASPOINT,		
_ALARM,_RES,_REFRAME,9999,1.2)		
IF (_ALARM==0) GOTOF _OKAY		
MSG ("Error: " <<_ALARM)		
МО	;Alarm occurred	
GOTOF_END		
_OKAY: G54	;Activate corrected frame (ZO)	
N400 G0 X Y Z	;Traverse in corrected frame	
N500 _END: M2		

10.04

# Measuring Cycles for Turning Machines

6.1	General prerequisites	6-278
6.2	CYCLE982, CYCLE972 Tool: Measuring turning tools	6-280
6.2.1 6.2.2	Calibrating tool probes (machine-related)	6-280 6-282
6.2.3	Determining dimensions of calibration tool	6-285
6.2.4	Measuring a turning tool (machine-related)	6-286
6.3	CYCLE982 tool: Measure turning and milling tools	6-292
6.3.1	Function overview	6-292
6.3.2	Calibrating a tool probe	6-298
6.3.3	Measuring the tool	6-301
635		0-311 6-318
6.3.6	Incremental measurement	6-321
6.3.7	Milling tool: Suppression of start angle positioning _STA1	6-327
6.3.8	Measuring drills – special applications (from measuring cycles SW 6.3)	6-328
6.4	CYCLE973 calibrate workpiece probe	6-329
6.4.1	Function overview	6-329
6.4.2	Calibrating in the reference groove	6-331
6.4.3	Calibration at a surface	6-334
6.5	CYCLE974 workpiece: 1-point measurement	6-337
6.5.1	Function overview	6-337
6.5.2	1-point measurement and ZO determination	6-340
6.5.3	1-point measurement and tool offset	6-343
6.5.4	1-point measurement with reversal and tool offset	6-348
6.6	CYCLE994 workpiece: 2-point measurement	6-351
6.7	Complex example for workpiece measurement	6-359

The measuring cycles below are intended for use on turning machines. To be able to run the measuring cycles described in this Chapter, the following programs must be stored in the part program memory of the control.

#### Overview of measuring cycles

(CYCLE972) <sup>1)</sup>	Calibrate tool probe, measure turning tools
CYCLE973	Calibrate workpiece probe in the reference groove or on surface
CYCLE974	1-point measurement with automatic tool offset or ZO determination
CYCLE982	Calibrate tool probe, measure turning and milling tools
CYCLE994	2-point measurement on diameter with automatic tool offset

#### Overview of the auxiliary programs required

CYCLE100	Logging on
CYCLE101	Logging off
CYCLE102	Measurement result display selection
CYCLE103	Preassignment of input data
CYCLE104	Internal subroutine: Measuring cycle interface
CYCLE105	Generate log contents: Log
CYCLE106	Logging the sequential controller Log
CYCLE107	Output of message texts (up to measuring cycles SW 6.2)
CYCLE108	Output of alarm messages (up to measuring cycles SW 6.2)
CYCLE109	Internal subroutine: Data transfer
CYCLE110	Internal subroutine: Plausibility checks
CYCLE111	Internal subroutine: Measuring functions
CYCLE113	Read system date and time Log
CYCLE114	Internal subroutine (tool offset)
CYCLE115	Internal subroutine (zero offset)
CYCLE117	Internal subroutine: Measuring functions
CYCLE118	Format real values: Log
1) 0) (0) 5000 1 ()	

1) CYCLE982 is the preferred cycle to use. CYCLE972 does not support graphic measuring cycles.

The measuring cycle data are defined in the data blocks:

- GUD5.DEF
- GUD6.DEF

The measuring cycles with measuring cycles SW 6.2 can only be used with NCK-SW 6.3 and higher.

10.04



# Call and return conditions

- D compensation with the data of the calibration tool or the workpiece probe or the tool to be measured must be activated in accordance with the measuring variant before a measuring cycle is called.
- The permissible tool type for the workpiece probe is type 5xy with cutting edge positions SL 5 to 8. Lengths refer to the center of the probe ball.
- No scaling factors <>1 must be active in the frames. Mirroring is permissible in the workpiece measuring cycles except for calibration

(condition: MD 10610: MIRROR\_REF\_AX =0).

• The G functions active before the measuring cycle call are reactivated at the end of the cycle.



## Level definition

The measuring cycles work internally with the abscissa and ordinate of the current plane G17 to G19.

The default setting for turning machines is G18.



# Notes on the spindle

Spindle commands in the measuring cycles always refer to the active **master spindle** of the control. When using measuring cycles on machines with several spindles, the spindle concerned before the cycle call must be defined as the master spindle.

References: /PG/ "Programming Guide Fundamentals"



#### 6.2.1 Function overview



### Programming

CYCLE982 CYCLE972



# Function

Cycles CYCLE982, CYCLE972 implement the

• Calibration of a tool probe and the

• Measuring turning tools (machine-related, probe arrays \_TP[ ]).

Tool lengths L1, L2 of turning tools with cutting edge positions SL = 1 to 8 are measured.

It is only possible to measure tools with one calibrated tool probe.

#### Notes

**CYCLE982** is the **preferred** cycle to use. CYCLE972 does not provide graphics measuring cycle support. Measuring cycle CYCLE982 provides extended measuring and calibration features over CYCLE972.



These variations are described in Section 6.3.



### Measuring variants

Measuring cycles CYCLE982, CYCLE972 permit the following measuring variants that are specified with parameter \_MVAR.

Value	Meaning	
0	Calibrate tool probe (machine-related)	
1	Measure tool (machine-related)	





10.04

# **Result parameters**

Measuring cycles CYCLE982, CYCLE972 return the following values in data block GUD5 for measuring variant **calibration**:

_OVR [8]	REAL	Trigger point in minus direction, actual value, abscissa
_OVR [10]	REAL	Trigger point in plus direction, actual value, abscissa
_OVR [12]	REAL	Trigger point in minus direction, actual value, ordinate
_OVR [14]	REAL	Trigger point in plus direction, actual value, ordinate
_OVR [9]	REAL	Trigger point in minus direction, difference, abscissa
_OVR [11]	REAL	Trigger point in plus direction, difference, abscissa
_OVR [13]	REAL	Trigger point in minus direction, difference, ordinate
_OVR [15]	REAL	Trigger point in plus direction, difference, ordinate
_OVR [27]	REAL	Zero offset area
_OVR [28]	REAL	Safe area
_OVI [2]	INTEGER	Measuring cycle number
_OVI [3]	INTEGER	Measuring variant
_OVI [5]	INTEGER	Probe number
_OVI [9]	INTEGER	Alarm number

Measuring cycles CYCLE982, CYCLE972 return the following values in data block GUD5 for measuring

variant tool measurement:		
_OVR [8]	REAL	Actual value length L1
_OVR [9]	REAL	Difference length L1
_OVR [10]	REAL	Actual value length L2
_OVR [11]	REAL	Difference length L2
_OVR [27]	REAL	Zero offset area
_OVR [28]	REAL	Safe area
_OVR [29]	REAL	Permissible dimension difference
_OVR [30]	REAL	Empirical value
_OVI [0]	INTEGER	D number
_OVI [2]	INTEGER	Measuring cycle number
_OVI [3]	INTEGER	Measuring variant
_OVI [5]	INTEGER	Probe number
_OVI [7]	INTEGER	Empirical value memory number
_OVI [8]	INTEGER	Tool number
_OVI [9]	INTEGER	Alarm number

6

# 6.2.2 Calibrating tool probes (machine-related)

# Calibration

The cycle uses the calibration tool to ascertain the current distance dimensions between the **machine zero** and the probe trigger point and automatically loads them into the appropriate data area in data block GUD6 (\_TP []fields).

Values are corrected without empirical and mean values.

### Note

If no special calibration tool is available, a turning tool with cutting edge position SL=3 can be used instead for calibration of 2 sides of the probe (see Subsection 6.2.3).

## Preconditions

Length 1 and 2 and the radius of the calibration tool must be known exactly and stored in a tool offset data block.

This tool offset must be active when the measuring cycle is called. A turning tool must be specified as the tool type (type 5xy). The cutting edge position must be SL= 3.

The lateral surfaces of the probe cube must be aligned parallel to the machine axes Z1, X1 (abscissa and ordinate).

The approximate coordinates of the tool probe PRNUM with respect to the machine zero must be entered in array

\_TP[\_PRNUM-1,0] to \_TP[\_PRNUM-1,3] before calibration starts.

These values are used for automatic approach of the probe with the calibration tool and their absolute value must not deviate from the actual value by more than the value in parameter \_TSA. It is also necessary for the probe to be reached within the

total measurement path  $2 \cdot FA$  will be reached.









10.04



# ₿

_MVAR	0	Calibrate tool probe (machine-related)
_MA	1, 2	Measuring axis
_PRNUM	INT	Probe number

The following additional parameters are also valid

\_VMS, \_TZL, \_TSA, \_FA, and \_NMSP.

See Sections 2.2 and 2.3.

# Sequence

**Parameters** 

### Position before measuring cycle call

The calibration tool must be prepositioned as shown in the figure.

The measuring cycle calculates the center of the probe and the approach paths automatically and generates the necessary traverse blocks.

## Position after end of measuring cycle

On completion of calibration, the calibration tool is \_FA from the measuring surface.

Calibrating a probe with calibration tool, starting position suitable for both axes (machine related)







# **Programming example**

# Calibrating tool probes

# (machine-related)

Tool probe 1 is stationary but provides a switching signal. The calibration tool is inserted in the turret as tool T7.

Values of the calibration tool T7 D1:

Tool type (DP1):	500
Cutting edge position (DP2)	3
Length 1 - Geometry (DP3):	L1 = 10
Length 2 - Geometry (DP4):	L2 = 40
Radius - Geometry (DP6):	R = 5

This radius must be taken into account in selecting the starting position for calibration of  $_TP[0,1]$ ,  $_TP[0,3]$  (increase distance from probe by  $2 \cdot R$ ).

Values of tool probe 1 in data block GUD6 which were determined manually to 5 mm accuracy beforehand (relative to the machine zero):

\_TP[0,0] = 50 \_TP[0,1] = 20 \_TP[0,2] = 70 \_TP[0,3] = 40

To obtain a minimum measurement path of 1 mm, the measurement path is programmed with

\_FA = 1+5= 6 mm (max. total measurement path

= 12 mm).

%\_N\_CALIBRATE\_MTT\_MPF

N05 G94 G90 DIAMOF	
N10 T7 D1	;Calibration tool
N15 G0 SUPA Z300 X240	;Starting position in minus X direction,
	;Traversal with ZO deactivated
N20 _TZL=0.001 _PRNUM=1 _VMS=0 _NMSP=1	;Parameters for calibration cycle
N21 _MVAR=0 _MA=2 _TSA=5 _FA=6	
N30 CYCLE982	;Calibration in minus X direction
N35 G0 SUPA Z60	;Approach new starting position
N38 _MA=1	;Select another measuring axis
N40 CYCLE982	;Calibration in minus Z direction
N45 G0 SUPA X20	;Approach new starting position
N48 _MA=2	
N50 CYCLE982	;Calibration in plus X direction
N55 G0 SUPA Z0	;Approach new starting position
N58 _MA=1	
N60 CYCLE982	;Calibration in plus Z direction
N65 G0 SUPA X240	;Approach change position in each axis
N70 SUPA Z300	
N99 M2	



Measuring Cycles for Turning Machines

# 6.2.3 Determining dimensions of calibration tool

# Function

10.04

If no special calibration tool is available, a turning tool with cutting edge position SL=3 can be used instead for calibration of 2 sides of the probe ( TP[i,0], TP[i,2]). With the following procedure it is possible to determine the dimensions as the calibration tool. Example: X axis, probe PRNUM=1 (\_TP[0,2])

- 1. Approximate probe data in the data block GUD6: Parameters \_TP[0,0]...\_TP[0,3]
- 2. Measure the turning tool at the presetting location.
- 3. Enter all tool data in the tool offset (incl. e.g.: L1=60.000) and use the tool in the revolver.
- 4. Machine a test part (turn to X dimension), e.g.: Set diameter: 200.000 mm Actual diameter: 200.100 mm.
- 5. Adjust tool offset (L1=59.950).
- 6. Finish-turn the test part again, e.g.: Set diameter: 195.000 mm Actual diameter: 195.000 mm, The setpoint must be equal to the actual value, then:
- 7. Calibrate tool probe in the X axis (see example program in Subsection 6.2.2).
- 8. Measure tool (see following Section). The aim is to determine value L1=59.950 (see Item 5.).

Another tool can then be measured and used as the calibration tool. Calibrate the probe; the subsequent tool measurement must result in the same tool length.



# 6.2.4 Measuring a turning tool (machine-related)



# Function

The cycle determines the new tool length (L1 or L2) and checks whether the difference from the old tool length can be corrected within a defined tolerance range:

Upper limits: Safe area \_TSA and

Dimensional difference check \_TDIF, Lower limit: Zero offset range \_TZL.

If this range is not violated, the new tool length is accepted, otherwise an alarm is output. Violation of the lower limit is not corrected.



### **Compensation strategy**

The tool measuring cycle is provided for various applications:

#### • First-time measurement of a tool

(\_CHBIT[3]=0):

The tool offset values in geometry and wear are replaced.

The offset is applied in the geometry component of the length.

The wear component is deleted.

• Remeasurement of a tool (\_CHBIT[3]=1): The resulting difference is calculated into the wear component (length).

Empirical values may optionally be included. The mean value is not calculated.

#### Precondition

The tool probe must be calibrated. The approximate tool dimensions must be entered in the tool offset data: Tool type 5xy, cutting edge position, tool nose radius, length 1, length 2. The tool to be measured must be active with its tool offset values when the cycle is called.



10.04



## Parameters

_MVAR	1	Measure tool (machine-related)
_MA	1, 2	Measuring axis

The following additional parameters are also valid \_VMS, \_TZL, \_TDIF, \_TSA, \_FA, \_PRNUM, \_EVNUM, and \_NMSP.

See Sections 2.2 and 2.3.



#### Sequence

#### Position before measuring cycle call

Before the cycle is called, the tool must be moved to the tool tip starting position, as shown in the figure.

The measuring cycle calculates the center of the probe and the associated approach paths automatically. The necessary traverse blocks are generated.

The tool nose center (S) is positioned at the center of the probe.

#### Position after end of measuring cycle

On completion of the cycle, the tool nose is positioned facing the measuring surface and \_FA from it.

Measuring turning tool with different tool tip directions and starting positions suitable for both axes.







# Programming example

Calibrating the tool probe with subsequent measurement of turning tool (machine-related)

**Calibration tool T7, D1** is to be used to calibrate all 4 sides of probe 1.

After that, **turning tool T3, D1** is to be remeasured in both lengths L1 and L2 (wear calculation).

The dimensions of the calibration tool T7 are in lengths L1, L2 and the radius **R=5.0 mm** are known precisely and entered in offset field D1.

The cutting edge position is SL=3.

The default values of probe 1 as applied in data block GUD6 with a tolerance of approx. 1 mm:

\_TP[0,0] = 220 \_TP[0,1] = 200 \_TP[0,2] = 400 \_TP[0,3] = 380

After calibration, the measured value (calibration value) is set.

The lengths for the tool to be measured T3, D1 are known, remeasurement in wear:

Tool type (DP1):	500
Cutting edge position (DP2):	3
Length 1 – Geometry (DP3):	L1=100.654
Length 2 – Geometry (DP4):	L2=60.321
Radius (DP6):	R=2.000
Length 1 – Wear (DP12):	0
Length 2 – Wear (DP13):	0






;Call calibration tool
;Starting position for calibration
;Parameter definition
Calibration in minus X direction
:Now starting position
, New starting position
;Set other measuring axis (2)
;New starting position
;Set other measuring axis (X)
;Calibration in plus X direction
;New starting position
;Set other measuring axis (Z)
;Calibration in plus Z direction
;Go to tool change position in each axis
;Traverse
;Selection of the tool to be measured
;Starting position for measurement
;Change of parameter definition for
;measurement, otherwise calibration
;Offset in wear (remeasuring)
;Tool measurement in minus X direction (L1)
;New starting position
;Set other measuring axis (Z)
;Tool measurement in minus Z direction (L2)
;Retraction axis by axis

N300 M2

# =?

## Explanation

#### N10 to N180, calibrate:

The "tip" of the calibration tool T7 is positioned in measuring axis X from the starting position at distance \_FA=1 mm (dimension  $\rightarrow$  with reference to the radius) before the probe. In axis Z, the probe tip center is centered with respect to the probe. The measuring process is initiated in the negative X direction (\_MA=2, starting position) with measuring velocity 150 mm/min (\_VMS=0, \_FA=1).  $\mathbf{O}$ 

#### Measuring Cycles for Turning Machines 6.2 CYCLE982, CYCLE972 Tool: Measuring turning tools

The switching signal is expected by the probe 1 (\_PRNUM=1) within a distance of  $2 \cdot _FA = 2$  mm. Otherwise, an alarm will be triggered. Measurement is performed once (\_NMSP=1). After successful measurement, the "tip" of T7 is \_FA=1 mm in front of the probe in the X direction. The calculated probe value is entered in \_TP[0,2]. Calibration with the measuring process has been completed in minus X.

Calibration is then performed in the other measuring directions/axes.

#### Explanation

#### N200 to N300, measure:

The probe is completely calibrated. The "nose" of the turning tool T3 is positioned in measuring axis X from the starting position at distance \_FA=1 mm (dimension  $\rightarrow$  with reference to the radius) in front of the probe. In axis Z, the center of the cutting edge is centered with respect to the probe. If the cutting edge radius = 0, it is the tool nose.

The measuring process is initiated in the negative X direction (\_MA=2, starting position) with measuring velocity 150 mm/min (\_VMS=0, \_FA=1). The switching signal is expected by the probe 1 (\_PRNUM=1) within a distance of  $2 \cdot _FA = 2$  mm. Otherwise, an alarm will be triggered.

Measurement is performed once (\_NMSP=1).

After successful measurement, the "tip" of T3 is \_FA=1 mm in front of the probe in the X direction.

The calculated length difference of L1 (tool type 5xy, \_MA=2, \_MVAR=1) is summated and entered in D1 from T3 in the wear (\_CHBIT[3]=1).

Measurement and wear offset are then performed in L2 in the minus Z direction.

10.04



٠

•

#### **Recommended parameters**

The following parameters are recommended so that this programming example runs reliably:

- **Calibration:** TZL = 0.001 Zero offset range \_TSA = 1 Safe area
- \_FA=1 Measurement path First-time measurement of a tool: \_TZL = 0.001 Zero offset range TDIF=3 Dimension difference check \_TSA=3 Safe area FA=3 Measurement path Remeasure the tool: •

## \_TZL = 0.001 Zero offset range

\_TDIF=0.3 Dimensional difference check Safe area \_TSA=1 \_FA=1 Measurement path



#### 6.3 CYCLE982 tool: Measure turning and milling tools

#### 6.3.1 Function overview

_	
	٠
	٠
	-

Programming

CYCLE982

#### Function

Cycle CYCLE982 permits

- calibration of a tool probe,
- measurement of tool lengths L1 and L2 for turning tools with cutting edge positions 1 to 8,
- the tool lengths for milling tools and drills on turning machines,
- the radius for milling tools.

Measurement of milling cutters/drills requires NCK software version SW 5 or higher.

# The following measurement and calibration tasks are supported by CYCLE982:

Calibration as the preparation for measurement / automatic measurement

The 4 switching positions of the probe are roughly known and entered in the array of the associated tool probe. Positioning of the calibration tool with respect to the probe is performed in the cycle. It is only possible to determine the switching position that is in the measuring axis \_MA and measuring direction according to starting position.

 Incremental calibration as preparation for incremental measurement

The switching positions of the probe are not known. The calibration tool must have been positioned in front of the probe manually (in JOG mode) before the cycle is called.

It is only possible to determine the switching position that is in the measuring axis \_MA and the stated measuring direction \_MD.

Only the probe switching position in which the axis and direction will subsequently be measured incrementally have to be calibrated.





#### Measuring Cycles for Turning Machines 6.3 CYCLE982 tool: Measure turning and milling tools

# Measuring

10.04

Only measured values that are in the measurement axis \_MA can be calculated. The geometry of the tool to be measured is roughly known and entered in the tool offset. Positioning of the tool with respect to the calibrated probe is performed in the cycle. The geometry must be determined precisely or wear (initial measurement or remeasurement of a tool).

#### Incremental measurement

The geometry of the tool to be measured is not known. The tool must have been positioned in front of the probe manually before the cycle is called.

The geometry is to be determined exactly. Only one measured value that is in the measurement axis \_MA can be calculated. The cycle approaches the probe in the measuring axis in the specified measuring direction \_MD.

#### Automatic measurement

All values that can be determined are determined automatically according to the active tool type. The geometry of the tool to be measured is roughly known and entered in the tool offset. Positioning of the tool with respect to the calibrated probe is performed in the cycle. The geometry must be determined precisely or wear (initial measurement or remeasurement of a tool).

• Machine-related measurement, calibration The switching positions of the tool probe refer to the machine zero.

The array for the tool probe \_PRNUM is used: \_**TP**[PRNUM-1,...].









Measurin 82 tool: Measure turnin

#### Measuring Cycles for Turning Machines 6.3 CYCLE982 tool: Measure turning and milling tools

Machine-related measurement, calibration

The switching positions of the tool probe refer to

7



Note for turning machines with a Y axis

Before CYCLE982 is called, the Y axis (applicate in G18) must be put in a position corresponding to the center of the probing surface of the tool probe in this axis.

The Y axis is not positioned in the cycle itself.



Tool probe workpiece related

TPW[i,2]

\_TPW[i,3]

w

\_TPW[] data field (trigger values) for probe \_PRNUM = i+1 in workpiece coordinate system

1

\_TPW[i,1] \_TPW[i,0]



#### **Measuring variants**

Measuring cycle CYCLE982 permits the following measuring variants that are specified with

parameter \_MVAR.

Digit						Meaning	
6 5	5 4	Ļ	3	2	1		
					0	Calibrate tool probe with ca	alibration tool
					1	Measure turning and millin Measurement axis in _MA (i • Turning tools: Cutting ec • Milling tools: Points 3 to	ng tool/drill, s specified for dge position 18, 5 in _MVAR)
				0	2	Automatic measurement (determine both lengths, for The following is specified: • Turning tools: Of edge p • Milling tools: Points 3 to Relative to machine	milling cutter, the radius tool.) positions 18, 5 in _MVAR)
				1		Relative to the workpiece (fr	om meas. cycles SW 6.3)
			0			Significance for measuring r Measuring without reversal Measuring with reversal	nilling tools only, also automatically:
						Significance for measuring r	nilling tools only, also automatically:
	C C	) )			1 2	Correct length only Measure milling cutter autor	(for measuring only) or natically
	1				1	Correct radius only	(for measuring only)
	2	2			1	Correct length and radius	(only for measurement, not for incremental measurement)
	3	\$			2	Measure upper tool edge au Correct length and radius, tr starting position side (for au groove mill)	itomatically: avel round measuring cube opposite tomatic measurement only, e.g.
	4				2	Measure upper tool edge au Correct length and radius, m length opposite to traversing _MVAR=x3x02 but with diffe (for automatic measurement	Itomatically: neasuring direction for determining direction, measuring sequence as for erent traversing motion t only, e.g. groove mill)
						Significance for measuring r	milling tools only, also automatically:
0						Axial position of milling tool/ (radius in ordinate, for G18:	drill X axis, SD 42950: value = 2)
0						Radial position of milling too (radius in abscissa, for G18: Measurement and calibratio	I/drill ∶Z axis, SD 42950: value = 2) n
1 1					0 1	Incremental calibration or Incremental measurement (limited variants, no automatic	tic measurement)



- The following measuring variants are not possible for incremental measurement: 1xxxx2; 102xx1; 112xx1
  - The following measuring variants are permitted if \_CHBIT[20]=1 (suppression of the starting angle position with \_STA1) on a milling tool: xxx0x1 (with x: 0 or 1, no other values)
  - A measuring variant can also be impermissible if it cannot be performed with the specified measuring axis \_MA, e.g. determining the milling cutter radius. However, with this position of the milling cutter it is not in the measuring axis.

#### **Result parameters**

The measuring cycle CYCLE982 returns the following values in the data blockGUD5 for the measuring

variant calil	bration:			
_OVR [8]	REAL	Trigger point in minus direction, actual value, abscissa		
_OVR [10]	REAL	Trigger point in plus direction, actual value, abscissa		
_OVR [12]	REAL	Trigger point in minus direction, actual value, ordinate		
_OVR [14]	REAL	Trigger point in plus direction, actual value, ordinate		
_OVR [9]	REAL	Trigger point in minus direction, difference, abscissa		
_OVR [11]	REAL	Trigger point in plus direction, difference, abscissa		
_OVR [13]	REAL	Trigger point in minus direction, difference, ordinate		
_OVR [15]	REAL	Trigger point in plus direction, difference, ordinate		
_OVR [27]	REAL	Zero offset area		
_OVR [28]	REAL	Safe area		
_OVI [2]	INTEGER	Measuring cycle number		
_OVI [3]	INTEGER	Measuring variant		
_OVI [5]	INTEGER	Probe number		
_OVI [9]	INTEGER	Alarm number		

# The measuring cycle CYCLE982 provides

the following values in data block GUD5 as results in

tool measu	irement:	
_OVR [8]	REAL	Actual value length L1
_OVR [9]	REAL	Difference length L1
_OVR [10]	REAL	Actual value length L2
_OVR [11]	REAL	Difference length L2
_OVR [12]	REAL	Actual value for radius
_OVR [13]	REAL	Difference for radius
_OVR [27]	REAL	Zero offset area
_OVR [28]	REAL	Safe area
_OVR [29]	REAL	Permissible dimensional deviation



Measuring Cycles for Turning Machines 6.3 CYCLE982 tool: Measure turning and milling tools

_OVR [30]	REAL	Empirical value	
_OVI [0]	INTEGER	D number	
_OVI [2]	INTEGER	Measuring cycle number	
_OVI [3]	INTEGER	Measuring variant	
_OVI [5]	INTEGER	Probe number	
_OVI [7]	INTEGER	Empirical value memory	
_OVI [8]	INTEGER	T number	
_OVI [9]	INTEGER	Alarm number	



#### Note on tool types

During measurement or calibration, the tool type (tool parameter DP1 in the tool offset data) of the active tool is evaluated.

- Type 5xy: Turning tool or calibrating tool
- Type 1xy: Milling tool
- Type 2xy: Drill

Use of tool types 711 to 799 is also possible. These are treated as a milling tool (type 1xy).

From measuring cycles SW 6.3, drill (Typ 2xy) also with SD 42950: TOOL\_LENGTH\_TYPE=0 can be gauged (refer to Subsection 6.3.8). Otherwise this is only possible for drills and milling tools with SD 42950: TOOL\_LENGTH\_TYPE=2.





# Function

#### Calibrating tool probes - machine-related

Measuring variant **\_MVAR=0** permits calibration of a tool probe relative to the calibration tool relative to the machine.

This variant is already described in detail in Subsection 6.2.2.

# Calibrating a tool probe relative to the workpiece (from measuring cycles SW 6.3)

Measuring variant **\_MVAR=10** permits calibration of a tool probe relative to the calibration tool relative to the workpiece.

The switching positions of the tool probe refer to the workpiece zero. The array for the tool probe \_\_PRNUM is used: \_\_TPW[PRNUM-1,...].

Transformations can be activated in workpiece-related measurement, calibration.

The requirements and procedures are as for machinerelated calibration (see Subsection 6.2.2).

#### Parameters

_MVAR	0	Calibrate tool probe (machine-related)			
	10	(see Subsection 6.2.2)			
		Calibrate tool probe (workpiece-related)			
		(> meas. cycle SW 6.3)			
_MA	1, 2	Measuring axis			
_PRNUM	INT	Probe number			
The following additional parameters are also valid					
_VMS, _TZL, _TSA, _FA, and _NMSP.					

See Sections 2.2 and 2.3.





The calibration tool is inserted in the turret as tool T7.

system.

Values of the calibration tool T7 D1:

**Programming example** 

Calibrating a tool probe (relative to workpiece)

Tool type (DP1):	500
Cutting edge position (DP2)	3
Length 1 - Geometry (DP3):	L1 = 10
Length 2 - Geometry (DP4):	L2 = 40
Radius - Geometry (DP6):	R = 5

Tool probe 1 is in the machining area and is oriented

parallel to the axis of the workpiece coordinate

This radius must be taken into account in selecting the starting position for calibration of \_TPW[0,1], \_TPW[0,3] (increase distance from probe by  $2 \cdot R$ ).

Values of the settable ZO G54: Displacement: X=0, Z=60.000 mm, no rotation

Values of tool probe 1 in data block GUD6 which were determined manually to 5 mm accuracy beforehand (relative to the workpiece zero):

To obtain a minimum measurement path of 1 mm, the measurement path is programmed with FA = 1+5= 6 mm (max. total measurement path = 12 mm).

%_N_CALIBRATE_MTT_WCS_MPF	
N05 G54 G94 G90 DIAMOF	
N10 T7 D1	;Calibration tool
N15 G0 Z100 X120	;Starting position in minus X direction,
	;Traversal with ZO deactivated
N20 _TZL=0.001 _PRNUM=1 _VMS=0 _NMSP=1	;Parameters for calibration cycle
N21 _MVAR=10 _MA=2 _TSA=5 _FA=6	
N30 CYCLE982	;Calibration in minus X direction
N35 G0 Z80	;Approach new starting position
N38 _MA=1	;Select another measuring axis
N40 CYCLE982	;Calibration in minus Z direction
N45 G0 X10	;Approach new starting position
N48 _MA=2	



10.04

#### Measuring Cycles for Turning Machines 6.3 CYCLE982 tool: Measure turning and milling tools

N50 CYCLE982	;Calibration in plus X direction
N55 G0 Z-10	;Approach new starting position
N58 _MA=1	
N60 CYCLE982	;Calibration in plus Z direction
N65 G0 X10	;Approach start position in each axis
N70 Z100	
N80 X120	
N100 M2	;End of program

# =?

#### Explanation

The calibration tool moves out of the starting position of N5 (X120, Z100) in Z to the center of the probe with its "tool tip". An offset is applied to compensate for the calibration tool radius. This places the radius center point in the center of the probe. The tool tip position is shown: Z30 ((\_TPW[0,0] + (\_TPW[0,1]) / 2 -R = (50+20) / 2 -5=30)). This is followed by traversal in measuring axis X (\_MA=2, G18) to position X76 (\_TPW[0,2] + \_FA = 70 + 6 = 76). This is where actual calibration (like measurement) starts in the minus X direction. At the end the calibration tool is again at position X76.

The new trigger values in minus X are stored in the data of tool probe 1 (\_PRNUM=1) \_TP[0,2] if they deviate by more than 0.001 mm (\_TZL=0.001) from the old values. Deviations of up to 5 mm (\_TSA=5) are permissible.

After that, the sides in the minus Z direction, plus X direction, and plus Z direction are approached, calibrated, and the values entered in array \_TPW[0,...].

#### 6.3.3 Measuring the tool



#### Function

This cycle and its various measuring variants are for measuring:

\_MVAR=1: Turning tools (machine-related) This variant is described in detail in Subsection 6.2.4. \_MVAR=11: Turning tools (workpiece-related) \_MVAR=xxx01: Milling tools, drills (machine-related) \_MVAR=xxx11: Milling tools, drills (workpiece-related) Workpiece-related or machine-related measurement require an appropriately calibrated tool probe (see Subsection 6.3.2 or 6.2.2).

These measuring variants can only determine offset values that are in the measurement axis \_MA.

The cycle determines the new tool length (L1 or L2), for milling tools the radius too, and checks whether the difference from the old tool length can be corrected within a defined tolerance range:

Upper limits: Safe area \_TSA and Dimensional difference check \_TDIF,

Lower limit: Zero offset range \_TZL. If this range is not violated, the new tool length is accepted, otherwise an alarm is output. Violation of the lower limit is not corrected.



#### **Compensation strategy**

The tool measuring cycle is provided for various applications:

- Initial measurement of a tool (\_CHBIT[3]=0): The tool offset values in geometry and wear are replaced. The offset is applied in the geometry component of the length. The wear component is deleted.
- **Re-measurement of a tool** (\_CHBIT[3]=1): The resulting difference is calculated into the wear component (radius or length).

Empirical values may optionally be included. The mean value is not calculated.

If \_CHBIT[20]=1, positioning of the milling spindle at the value of \_STA1 can be suppressed. That is possible with the following milling cutter measuring variants: MVAR=xxx001 (with x: 0 or 1, no other values).

## Precondition

The tool probe must be calibrated. The approximate tool dimensions must be entered in the tool offset data: Tool type, cutting edge position on turning tools, radius, length 1, length 2. The tool to be measured must be active with its tool offset values when the cycle is called.

For a **milling cutter**, the setting data SD 42950: TOOL\_LENGTH\_TYPE = 2 must be set (length calculation as for turning tool). For milling tools, the tool spindle must be declared the master spindle.

The **drill** is also SD 42950: TOOL\_LENGTH\_TYPE = 0 possible (see Subsection 6.3.8).

### Parameters

MVAR	1 or xxx01	Measure tool (machine-related)
	11 or xxx11	Measure tool (workpiece-related)
		(> meas. cycle SW 6.3)
		More precise parameterization for milling tools
		is entered in the 3rd to 5th digits of _MVAR.
_MA	1, 2	Measuring axis
_STA1	REAL	For milling tools: Start angle
CORA	REAL	For milling tools:
		Offset angle position after reversal
		(for measurement with reversal only
		_MVAR=xx1x1)

The following additional parameters are also valid \_VMS, \_TZL, \_TDIF, \_TSA, \_FA, \_PRNUM, \_EVNUM, and \_NMSP. See Sections 2.2 and 2.3.

#### Sequence

10.04

#### Position before measuring cycle call

Before the cycle is called, the tool must be moved to the starting position, as shown in the diagram for turning tools. The measuring cycle then calculates the approach position automatically. This position determines the direction of measurement in the measuring axis \_MA.

For milling tools, the measuring point on the tool is determined by entered lengths 1 and 2 (please note: SD 42950: TOOL\_LENGTH\_TYPE).

If the radius value is not equal to zero, this is also a determining factor. The measuring point is then located on the side which the measuring probe faces (+R or – R). The axial or radial position of the tool must be specified (\_MVAR). This starting position must ensure collision-free approach.

In the case of milling tools, length and radius can be selected as an alternative to length only to determine the cutter radius.

For length and radius, two measuring points are required. These are approached from different sides of the measuring probe. First the measuring point facing the measuring probe at the starting point is approached. Then, after travel round the probe (in the direction of the starting point), the 2nd measuring point is measured in the opposite direction. If the spindle is stationary (M5) and measurement without reversal is selected, the 2nd measurement is performed with a spindle rotation of 180 degrees. The same cutting edge used for the 1st measurement is now used.

The L1 or L2 offset values and the cutter radius are calculated from these two measurements.

#### **Measurement with reversal** can be selected separately with \_MVAR: First the measuring point is measured in the

selected axis and in a milling spindle position according to starting angle \_STA1. Then the tool (spindle) is turned 180 degrees and measured again. The average value is the measured value. Measuring turning tool with different tool tip directions and starting positions suitable for both axes.



#### Measuring Cycles for Turning Machines 6.3 CYCLE982 tool: Measure turning and milling tools

Measurement with reversal causes a second measurement at each measuring point P with a spindle rotation through 180 degrees from the starting angle. The offset angle entered in \_CORA is summated to these 180 degrees. That enables selection of a certain 2nd milling cutting edge that is offset from the 1st cutting edge by precisely 180 degrees. Measurement with reversal permits measurement of two cutting edges of one tool. The mean value is the offset value.

If \_CHBIT[20]=1, selected measuring variants are possible for a milling cutter without taking the starting angle \_STA1 into account (see Subsection 6.3.7).

#### Note on measuring with rotating spindle

If selection of a certain miller cutting edge is not possible, it is possible to measure with a rotating spindle. The user must then program the direction of rotation, speed, and feedrate very carefully before calling up CYCLE982 to prevent damage to the probe. A low speed and feedrate must be selected.

#### Position after end of measuring cycle

When the cycle is complete, the tool nose is \_FA from the measuring surface and facing it.



Measuring variant	Given	Offset applied in	Milling tools, drills
	geometry		
Example:	L1=	L2	
Axial position,	L2=		Drill (tool type: 2xy)
R=0,	R=0		
Measuring without			X1 A L2
reversal,			F F F
calculate length only			
			Measuring
_MVAR=1			
_MA=1			
_			
			MT Z1



Measuring variant	Given geometry	Offset applied in	Milling tools, drills
Example: <b>Radial position,</b> R=0, Measuring without reversal, calculate length only _MVAR=10001 _MA=2	L1= L2= R=0	L1	Drill (tool type: 2xy)
Example: Axial position, R ≠ 0, Measuring without reversal, calculate length only _MVAR=1 _MA=1	L1= L2= R=	L2	Mill (tool type: 1xy) X1 Measuring point $\alpha$ Z1

Measuring variant	Given geometry	Offset applied in	Milling tools, drills
Example:	L1=	L1	
Radial position,	L2=		
. , R≠0.	R=		F
Measuring without			
reversal			
calculate length only			
calculate length only			
M\/AP=10001			······
_ΜΔ=2			Measuring
			point
			M Z1
Example:	L1=	R	
Axial position.	L2=		
R≠0	R=	R=ABS(P – L1)	
measuring with reversal			×1 <b>†</b>
calculate radius only			
calculate radius only			
MVAR=1101			point 2
_ΜΔ=2			
_ma-z			point 1 $\alpha$
ET mast be known			
			P P
			→ 71
			IVI Z1
Example:	L1=	L2	
Radial position,	L2=		
R ≠ 0,	R=	L2=(P - R)	F
measuring with reversal.			
calculate length only		Or other	
0,		measuring	
MVAR=10101		direction:	
 MA=1		L2=(P + R)	
– R must be known			point 1 point 2
			M Z1



Measuring variant	Given geometry	Offset applied in	Milling tools, drills
Example:	L1=	L2	
Radial position,	L2=	R	
R ≠ 0,	R=		F
Measuring without reversal,		L2=(P1 + P2)/2	
calculate length and radius,		R= ABS(P1-P2)/2	Measuring
2 measuring points			
necessary			FA FA Starting position of tool at beginning of cycle
_MVAR=12001			
			P2 P1
Notes:			
On starting, the			
measuring point must be			
outside the measurement	t		
cube coordinates in both			
coordinates.			
On the opposite side of			
the measuring cube (P2)			
measurement is			
performed with a rotated			
spindle (by 180 degrees).			
The same cutting edge is			
then measured. This only	,		
happens if the spindle is			

In this example, L1 refers to the upper cutting edge. If L1 is to be calculated in another measurement, the starting position must be below the measuring cube.

stationary and without

reversal.

Measuring variant	Given geometry	Offset applied in	Milling tools, drills
Example:	L1=	L1	
Axial position,	L2=	R	
R ≠ 0,	R=		
Measuring without		L1=(P1 + P2)/2	X1 L2
reversal,			
calculate length and		R=ABS(P1-P2)/2	
radius.			
2 measuring points			point 🗠
necessarv			
·····,			
MVAR=2001			
 MA=2			P 2 <b>1</b>
-			M <sup>T</sup> Z1
P2 is measured with a			
rotated spindle (by 180			
degrees), if			
measurement is			
performed with a			
stationary spindle.			
Example:	L1=	L2	
Radial position,	L2=	R	
R ≠ 0,	R=		F
Measurement with		L2=(P1 + P2)/2	
reversal at each			
measuring point,		R=ABS(P1-P2)/2	
calculate length and			Measuring Measuring
radius,			
2 measuring points			
necessary			
(4 measurements)			
_IVIVAR=12101 MA=1			Μ Ζ1

SINUMERIK 840D/840Di/810D User's Guide Measuring Cycles (BNM) - 10.04 Edition



# Measure milling tool in the radial position (machine-related)

For the end miller T3, D1 should be determined in the radial position when first measuring length L2 and radius R. Measurements will be without reversal. The cutting edge to be measured is in the milling spindle position 15 degrees.

The lengths and radius of tool T3 to be measured are roughly known and entered in offset field D1:

Tool type (DP1):	120
Cutting edge position (DP2)	
Length 1 - Geometry (DP3):	L1 = 60
Length 2 - Geometry (DP4):	L2 = 10
Radius - Geometry (DP6):	R = 14



Deviations from this value of less then 2.5 mm are expected.

The probe to be used is tool probe 1. This probe has already been completely calibrated (machinerelated). The precise values are entered in array \_TP[0,...] and are approximately:

\_TP[0,0]=220, \_TP[0,1]=200 \_TP[0,2]=400, \_TP[0,3]=380.

%_N_I3_MEA5_FR_MPF	
N1 G0 G18 G90 G94 DIAMOF	
N100 T3 D1	;Selection of the tool to be measured
N110 G0 SUPA Z285 X450	;Starting position for measurement
	;Traversal without ZO
N120 _TZL=0.001 _TSA=3 _FA=3 _PRNUM=1 _VMS=0	;Change of parameter definition for
_NMSP=1	;measurement, otherwise calibration
N121 _MA=1 _TDIF=2.5 _MVAR=12001 _STA1=15	
N130 _CHBIT[3]=0	;Offset in the geometry
N131 _CHBIT[20]=0	;Do not suppress _STA1
N140 CYCLE982	;Tool measurement L2, R
N180 G0 SUPA X450	;Retraction axis by axis
N190 SUPA Z285	
N200 M2	;End of program

10.04

#### Explanation

The spindle is positioned at 15 degrees with SPOS. Measuring point P1 is approached first. The measuring process is initiated in the negative Z direction (\_MA=1, starting position) with measuring velocity 300 mm/min (\_VMS=0, \_FA>1). The switching signal is expected by the probe 1 (\_PRNUM=1) within a distance of  $2 \cdot _FA=2$  mm. Otherwise, an alarm will be triggered. Measurement is performed once (\_NMSP=1). After successful measurement, tool T3 is \_FA=3 mm + tool radius in front of the probe.

The probe is then traveled around as shown in the figure.

On the opposite side of the probe (P2) measurement is performed with a rotated spindle (by 180 degrees). The same cutting edge is then measured.

Measurement is performed with the spindle stopped and without reversal.

After successful measurement, tool T3 is \_FA=3 mm + tool radius in front of the probe. The spindle remains in this position.

The radius and length L2 are determined precisely and tool parameters DP6 and DP4 of T3, D1 are entered. The values in result parameter array OVR[] are also entered.

In block N180, N190, the tool returns to the starting position and the program then ends.

10.04

#### 6.3.4 Automatic tool measurement



#### Function

This cycle and its various measuring variants are for measuring tools **automatically**:

_MVAR=2:	Turning tools (machine-related)
_MVAR=12:	Turning tools (workpiece-related)
_MVAR=xxx02:	Milling tools, drills (machine-related)
_MVAR=xxx12:	Milling tools, drills (workpiece-related)

Workpiece-related or machine-related measurement require a corresponding calibrated tool probe (see Subsection 6.3.2 or 6.2.2).

With milling tools/drills, the measurement is further specified with the 3rd to 5th decimal places of parameter \_MVAR. Here, SD 42950: TOOL\_LENGTH\_TYPE must be = 2

The function is as for non-automatic measurement. In automatic measurement, all offsets are determined. These are then defined with the tool type:

- Turning tool: Both lengths (2 measurements), for cutting edge positions SL = 5, 6, 7, and 8, only one length.
- Drill: Length according to axial or radial position (1 measurement)
- Milling tool: Both lengths and radius (4 meas.),

if the radius is R=0, both lengths only are calculated (2 measurements).

The calculated offsets are entered in the active D number of the active tool.

The offset strategy is defined via \_CHBIT[3] as for measurement.

The measuring cycle generates the approach blocks to the probe and the traverse movements for measuring length 1. Length 2 and for the milling cutter, the radius itself. The condition is a correctly selected starting position.

#### Precondition

As for non-automatic tool measurement





#### Parameters

_MVAR	2 or xxx02	Measure tool automatically (machine-related)		
	12 or xxx12	Measure tool automatically (workpiece-related)		
		(> meas. cycle SW 6.3)		
		More precise parameterization for milling tools is entered in		
		the 3rd to 5th digits of _MVAR.		
MA	1, 2	Measuring axis		
_STA1	REAL	For milling tools: Start angle		
CORA	REAL	For milling tools:		
		Offset angle position after reversal		
		(for measurement with reversal only MVAR=xx1x1)		

\_VMS, \_TZL, \_TDIF, \_TSA, \_FA, \_PRNUM, \_EVNUM,

and \_NMSP.

See Sections 2.2 and 2.3.

#### Sequence

#### Position before measuring cycle call

Before the cycle is called, the tool must be moved to the starting position, as shown in the diagram for turning tools. The measuring cycle then calculates the approach position automatically.

First the length (P1) in the abscissa (Z axis for G18) and then (P2) in the ordinate (X axis for G18) is measured. For turning tools, the measuring probe travels round the measuring cube at distance \_FA.

For milling tools, the measuring points on the tool are determined by entered lengths 1 and 2 (please note: SD 42950). If the radius value is not equal to zero, this is also a determining factor. The axial or radial position of the tool must be specified in \_MVAR, and the starting position approached accordingly. First, the values in the abscissa (Z axis for G18) are measured.

Measurement with reversal can be selected separately with \_MVAR.

The probe travels round the measuring cube at distance \_FA or according to the starting point coordinate (see figs.).



#### Position after end of measuring cycle

When the cycle is complete, the tool nose is again located at the starting point. A movement to this point is automatically generated in the cycle.



	0.			
Measuring variant	Given	Override	Milling tools	
	geometry	applied in		
Example:	L1=	L1		
Radial position,	L2=	L2		
R ≠ 0	R=	R	F	
measuring with reversal				
8 measurements		1 = (P3x + P4x)/2		
necessary (P1 to P/			- <u>R</u>	
necessary (F 1 to F4		l 2=(P1z + P2z)/2	Measuring	
each with reversal)			Measuring point 2	
		R=ABS(P17-P27)/2		
_MVAR=10102				
			P4	
			M <sup>®</sup> Z1	
Example:	L1=	L1	Measurements P1 to P4	
Axial position,	L2=	L2	measurements i i to i 4	
R ≠ 0,	R=	R	L2	
Measuring without				
reversal		L1=(P3x + P4x)/2		
4 measurements		· · · · · · · · · · · · · · · · · · ·		
		l 2=(P1z + P2z)/2		
necessary		(* ·_ · ·/ =		
		R = ABS(P3x - P4x)/2		
_WVAR=3002			P1, P2	
The probe travels around			L	
the measuring cube			a - distance from starting point	
opposite the starting			M Z1	
position side.				
Note:				
Length measurements				
for L2 (P1, P2) are				
performed here at the				
same measuring point 1				
without rotation the				
- without rotating the				
spindle by 180 degrees.				
The same cutting edge is				
always measured				
(starting angle _STA1).				



Measuring variant	Given	Override	Milling tools
	geometry	applied in	
Example:	L1=	L1	
Radial position,	L2=	L2	
R ≠ 0	R=	R	Starting position of tool at beginning of cycle
Measuring without			X1
reversal,		L1=(P3x + P4x)/2	F
4 measurements			
necessary		L2=(P1z + P2z)/2	
M\/A D-12002		R=	
_WIVAR=13002		ABS(P1z-P2z)/2	
The probe travels around			a - distance from starting position
the measuring cube			M <sup>M</sup> Z1
opposite the starting			
position side.			
Note:			
Length measurements			
for L1 (P3, P4) are			
performed here at the			
same measuring point 1			
<ul> <li>without rotating the</li> </ul>			
spindle by 180 degrees.			
The same cutting edge is			
always measured			
(starting angle _STA1).			

For measuring variants \_MVAR=0x3xx2 and \_MVAR=0x4xx2 (measure upper cutting edge automatically), the cutting mill must have suitable geometric dimension (end mill/cutter radius) for approaching the center of the probe with the cutting edge without collision. No check is made for suitability of the milling cutter

for this measuring variant. It is up to the user to ensure this.

 $\mathbf{O}$ 

Measuring variant	Given	Override	Milling tools
	aeometrv	applied in	
Example:	1 1=	11	
Axial position	12=	12	right of measuring cube
	R=	P	
$R \neq 0$ ,	IX		
weasuring without		1.1 - (D2x + D4x)/2	
reversal,		$L = (P_{3}X + P_{4}X)/2$	
4 measurements		$1.2 = (D1_7 + D2_7)/2$	· · · · · · · · · · · · · · · · · · ·
necessary			
		R = ABS(P3x-P4x)/2	
_MVAR=4002			
			P1 a
Direction of			P2 P2 P3
measurement for			
determining length L2			M <sup>4</sup> Measurements P1 to P4 Z1
opposite to traversing			
direction, measuring			Starting position of tool at beginning
procedure as for			of cycle - left of measuring cube
_MVAR=x3002 but with			
different traversing			
motion			
Notes:			
Length measurements			
for I 2 (P1 P2) are			point
nerformed here at the			<sup>™</sup> P4 FA
same measuring point –			a - distance from starting
without rotating the			position
spindle by 180 degrees			M Measurements P1 to P4 Z1
The same cutting edge is			Measurements 1 1 to 1 4
The same culling edge is			
(starting angle STA1)			
(starting angle _STAT).			cycle and various axial positions
The width of the milling			
the width of the mining			X1
tool must be considered			
when selecting the			
starting position or			
dimension a!			
			M Z1



7	-	1
	A	
		1
١.	•	4





#### 6.3.5 Incremental calibration



## Function

A tool probe can be calibrated with measuring variant

\_MVAR=100000 (machine related) or

\_MVAR=100010 (workpiece-related)

incrementally with a calibration tool.

The switching positions of the probe are not known. The values entered in the array of the probe are not evaluated. The calibration tool must have been positioned in front of the probe manually (in JOG mode) before the cycle is called.

The cycle uses the calibration tool to ascertain the current distance dimensions between the zero and the probe trigger point and automatically loads them into the appropriate data area in data block GUD6 (field \_TP [] or \_TPW[]).

Values are corrected without empirical and mean values.

#### Precondition

The lateral surfaces of the tool probe must be aligned parallel to the relevant axes (machine or workpiece coordinate system in abscissa and ordinate).

Length 1 and 2 and the radius of the calibration tool must be known exactly and stored in a tool offset data block. This tool offset must be active when the cycle is called. A turning tool must be specified as the tool type (type 5xy). The cutting edge position SL must be 3.

The calibration tool (tool tip) must, before CYCLE982 is started, have a position that causes the probe to switch in the specified direction MD for the measuring axis MA within path  $2 \cdot FA$ . Careful when positioning manually! Damage to the probe must be prevented.

#### Parameters

	_MVAR 100000		Calibrate tool probes incrementally (machine-related)		
10		100010	Calibrate tool probe incrementally (workpiece-related)		
	_MA	1, 2	Measuring axis		
	_MD	0, 1	Measuring direction: 0 - positive, 1 - negative		
The following additional parameters are also valid					
	_VMS, _FA, _PRNUM, and _NMSP.				
	Refer to Sections 2.2 and 2.3				

to Sections 2.2 and 2.3

#### Measuring Cycles for Turning Machines 6.3 CYCLE982 tool: Measure turning and milling tools

#### Sequence

10.04

#### Position before measuring cycle call

The calibration tool must be prepositioned as shown in the figure:

The "tip" of the calibration tool in the **measuring axis** \_MA is within distance  $2 \cdot FA$  **in front of** the measuring surface.

The center of the calibration tool tip in the **other axis** (offset axis) must be in the center of the probe. The measuring cycle starts measuring in the specified axis (\_MA) and measuring direction (\_MD) immediately on starting.

#### Position after end of measuring cycle

When the calibration procedure is completed the calibration tool is positioned on the starting position again.

#### Remark

A special tool is used as the calibration tool and is entered as a turning tool (5xy) with cutting edge 3. It is usually shaped (bent) such that it is also possible to approach point P4 for calibration (\_MA=1, \_MD=0). Calibration tool: See Subsection 6.2.2

However, it is not necessary to calibrate all 4 sides for **incremental** measurement. The side that is used for incremental measurement is sufficient.

That does not apply to automatic measurement. Here all 4 points must be calibrated or values entered for automatic central positioning of the tool to be measured.







6



#### Programming example

#### Calibrate tool probe incrementally

Tool probe 1 is in the machining area and is oriented parallel to the axis of the machine. Calibration is to be performed in the minus X direction and incrementally. The calibration tool is inserted in the turret as tool

T7. Values of the calibration tool T7 D1:

Tool type (DP1):	500
Cutting edge position (DP2)	3
Length 1 - Geometry (DP3):	L1 = 10
Length 2 - Geometry (DP4):	L2 = 40
Radius - Geometry (DP6):	R = 5

Values of tool probe 1 in data block GUD6 before calibration:

\_TP[0,0] = ?, \_TP[0,1] = ? \_TP[0,2] = ?, \_TP[0,3] = ?





# %\_N\_INCR\_CALIBRATE\_MPF N10 T7 D1 G94 ;Calibration tool is active, ;Starting position reached N20 \_MVAR=100000 \_MA=2 \_MD=1 \_FA=20 ;Parameters for calibration cycle \_PRNUM=1 \_VMS=0 \_NMSP=1 ;Calibration in minus X direction N30 CYCLE982 ;Calibration in minus X direction N99 M2 ;Calibration in minus X direction

# 2 Explanation

Before the program is started, the "tip" of the calibration tool T7 is in measuring axis X in a range 2 · FA=40 (dimension with reference to radius) in front of the probe. In axis Z, the probe tip center is centered with respect to the probe. When CYCLE982 is started, measurement starts in the negative X direction (MA=2, MD=1) with measuring velocity 300 mm/min (\_VMS=0, \_FA>1). The switching signal is expected by the probe 1 (PRNUM=1) within a distance of 2 · FA =40 mm. Otherwise, an alarm will be triggered. Measurement is performed once (\_NMSP=1). After successful measurement, the "tip" of T7 is in the starting position again. The calculated probe value is entered in TP[0,2]. Calibration with the measuring process has been completed in minus X.

#### 6.3.6 Incremental measurement



#### Function

This cycle and its various measuring variants are for measuring tools **incrementally**:

\_MVAR=100001: Turning tools (machine-related)

\_MVAR=100011: Turning tools (workpiece-related)

\_MVAR=1xxx01: Milling tools, drills (machine-related)

\_MVAR=1xxx11: Milling tools, drills (workpiece-related).

Workpiece-related or machine-related measurement require an appropriately calibrated tool probe (refer to Subsection 6.3.5). With milling tools/drills, the measurement is further specified in the 3rd to 5th digits of parameter \_MVAR.

It is possible to measure single tool lengths, or alternatively for milling tools the cutter radius.

The calculated offsets are entered in the active D number. The offset is entered in the **geometry data** and the wear data are reset (irrespective of \_CHBIT[3]).

Only the offset value that is in the measuring axis \_MA and measuring direction \_MD can be determined in a measurement.

If \_CHBIT[20]=1, positioning of the milling spindle at the value of \_STA1 can be suppressed (see Subsection 6.3.7). This is possible for milling cutter measuring variants: \_MVAR= xxx001 (where x : 0 or 1, no other values).

#### Preconditions

For incremental measurement, the tool probe must be calibrated in the measuring axis and direction in which measuring will be performed.

The tool T to be measured must be called with tool offset (D number).

The tool type is entered in the offset data.

For a milling cutter, setting data SD 42950:

TOOL\_LENGTH\_TYPE = 2 must be set (length calculation as for turning tool).

For milling tools, the tool spindle must be declared the master spindle. The **drill** is also SD 42950: TOOL\_LENGTH\_TYPE = 0 possible (see Subsection 6.3.8).



	- 1	
 _		

#### **Parameters** MVAR 1xxx01 Measure a tool incrementally - machine-related Measure a tool incrementally - workpiece-related 1xxx11 (> meas. cycle SW 6.3) More precise parameterization for milling tools/drills is entered in the 3rd to 5th digits of \_MVAR. MA 1, 2 Measuring axis MD 0, 1 Measuring direction: 0 - positive, 1 - negative STA1 Only for milling tools and if CHBIT[20]=0: Starting angle of the milling spindle \_CORA Only for milling tools and measurement with reversal: Offset angle position of the milling spindle after reversal

The following additional parameters are also valid

VMS, \_FA, PRNUM and \_NMSP.

Refer to Sections 2.2 and 2.3.

#### Sequence

#### Position before measuring cycle call

Before the cycle is called, the tool must be moved to the starting position, as is shown in the diagram for **turning tools**, e.g.: with traversal in JOG: The "tip" of the tool in the **measuring axis** \_MA is within distance  $2 \cdot _FA$  **in front of** the measuring surface. The center of the cutting edge radius on the turning tool in the **other axis** is in the center of the probe. If the cutting edge radius =0, it is the tool nose. For **milling tools**, the axial or radial position of the tool must be specified in \_MVAR;

#### as must measurement with reversal:

First the measuring point is measured in the selected axis and in a milling spindle position according to starting angle \_STA1. The tool (milling spindle) is then rotated through 180 degrees plus the value in \_CORA and measured again. The average value is the measured value.

If the milling spindle is activated when the cycle is started, measurement will be performed with a **rotating spindle**. In that case, the user must exercise special care when selecting the speed, direction of rotation, and feedrate!

If \_CHBIT[20]=1, selected measuring variants are possible for a milling cutter without taking the starting angle \_STA1 into account (see Subsection 6.3.7).

#### Position after end of measuring cycle

When the cycle is complete, the tool nose is again located at the starting position.





Measuring variant	Specified	Override	Milling tools, drills
	geometry	applied in	
Example:	L1=		
Axial position,	L2=	L2	
Drill, R=0,	R=0		
incremental			×1 <b>†</b>
measurement without			~1
reversal,			
calculation of the			L2 = ?
length in Z			
_MVAR=100001			
_MA=1			
Always position the			•
drill tip in the center of			M' Z1
the probe!			
Example:	I 1=		
Radial position	12=	11	
Drill R=0	R=0	<b>L</b> 1	F )
Measuring without	10		
rovorsal			
reversal,			
			¥
M\/AP=110001			
_MΔ=2			
2			
	1.1-		
Example:	L I =	L2	
Axial position,	L2=		
Miller, $R \neq 0$ ,	R=		
Measuring without			X1 <b>†</b>
reversal,			
calculation of the			L2 = ?
length in Z			
_MVAR=100001			
_MA=1			
			<b>↓</b>
			M <sup>r</sup> Z1

Measuring variant	Specified	Override	Milling tools, drills
	geometry	applied in	
Example:	L1=	L1	
Radial position,	L2=		
Miller, R ≠ 0,	R=		F
Measuring without			x1 t L2 ()
reversal,			
calculation of the			
length in X			
MVAR=110001			
_ _MA=2			
			M <sup>P</sup> Z1
Example:	1 1=		
	L 1= L 2=	R	
Millor $\mathbf{P} \neq 0$	L2 R=	IX IX	
model, $IX \neq 0$ ,	· · · · ·		
calculate radius			X1 F
calculate radius			
MVAR=101101			
In this case, L1 must			
be known.			
			M Z1
Example:	L1=	L2	
Radial position,	L2=		
Miller, $R \neq 0$ ,	R=		
measuring with reversal,			X1
calculation of the			F
length in Z			
_MVAR=110101			
_MA=1			
In this case, R must be			
KHUWH.			M Z1


Measuring variant	Specified	Override	Milling tools, drills
	geometry	applied in	
Example:	L1=	R	
Radial position,	L2=		
Miller, R ≠ 0,	R=		
measuring with reversal,			x1 <b>†</b>
calculate radius			F )
_MVAR=111101			
_MA=1			
In this case, L2 must			
be known.			
			M Z1

6



### Programming example

With the turning tool T3, D1 with SL=3, length L1 is to be determined **incrementally** and machine-related.

The probe to be used is tool probe 1. This probe is already calibrated in the minus X direction (machine-related). The precise values are entered in probe array \_TP[0,2,...].



%_N_INCR_MEAS_MPF	
N10 T3 D1 G94	;Turning tool T3 is active,
	;Starting position reached
N20 _MVAR=100001 _MA=2 _FA=20 _MD=1	;Parameters for the measuring cycle
_PRNUM=1 _VMS=0 _NMSP=1	
N30 CYCLE982	;Measurement in minus X direction
N99 M2	

# =?

#### Explanation

The probe has been calibrated in minus X. Before the program is started, the "tip" of the tool T3 is in measuring axis X in a range  $2 \cdot FA = 40$  mm (dimension with reference to radius) in front of the probe. In axis Z, the center of the cutting edge is centered with respect to the probe. If the cutting edge radius =0, it is the tool nose.

When CYCLE982 is started, measurement starts in the negative X direction ( $_MA=2$ , MD=1) with measuring velocity 300 mm/min ( $_VMS=0$ ,  $_FA>1$ ). The switching signal is expected by the probe 1 ( $_PRNUM=1$ ) within a distance of 2 ·  $_FA = 40$  mm. Otherwise, an alarm will be triggered. Measurement is performed once ( $_NMSP=1$ ).

After successful measurement, the "tip" of T3 is in the starting position again.

The calculated length L1 (tool type 5xy, \_MA=2,

\_MVAR=xx0xxx) is entered in D1 of T3 in the

geometry. The associated wear component is reset.

#### 6.3.7 Milling tool: Suppression of start angle positioning \_STA1



#### Function

To apply the angular position of the milling spindle (cutting edge of the miller contacting the probe) unchanged in the cycle and thus suppress the starting angle positioning with the value in \_STA1, you can set

#### \_CHBIT[20]=1

will be reached.

However, this only permits simple milling cutter measuring variants that do not have to use the starting angle in \_STA1, e.g.: no 2nd measurement or no re-positioning after measurement with reversal. Otherwise milling cutter measuring variants are possible that are also permitted for incremental measurement.

If the machine does not feature an SPOS-capable milling spindle, it is also possible to measure milling cutters with these measuring variants and \_CHBIT[20]=1.

Permissible measuring variants with milling cutter and \_CHBIT[20]=1:

xxx0x1 (where x: 0 or 1, no other values)

Other measuring variants with a miller will be rejected with an alarm message.

For measurement with rotating spindle and \_CHBIT[20]=1, only these measuring variants are permitted. Measurement with reversal is not permitted.

#### 6.3.8 Measuring drills – special applications (from measuring cycles SW 6.3)



#### Function

If **drills** are used on lathes with a length correction (offset) as for milling machines (SD 42950: TOOL\_LENGTH\_TYPE=0), then a drill can also be measured (gauged) in this application. **Length L1** is always calculated in the applicate (tool offset axis) of the current plane G17 to G19. This also characterizes the position of the tool. The usual position definition in \_MVAR (5th digit) is no longer relevant and is ignored. **G17**: L1 in Z axis (corresponds to axial position)

G18: L1 in Y axis (corresponds to axial position) G19: L1 in X axis (corresponds to radial position)

Length L1 is determined if

- the active tool of type 2xyx (drill)
- SD 42950: TOOL\_LENGTH\_TYPE=0
- G17 and G19 are active and
- a measuring variant is set: \_MVAR=1: Measuring (relative to the machine)
  - \_MVAR=11: Measuring (relative to the workpiece)
  - \_MVAR=1000x1: Incremental measurement, (referred to the machine or workpiece)

A measuring axis specified in \_MA is ignored. The 3rd axis (applicate) is used within the cycle.

Otherwise the description of the measuring variant applies.

#### **Requirement:**

The tool probe has been calibrated with **G18** active as is usual for turning tools.







#### 6.4 CYCLE973 calibrate workpiece probe

#### 6.4.1 Function overview



Programming



#### Function

CYCLE973

This cycle can calibrate a workpiece probe with various cutting edge positions in a

- reference groove or on a
- surface.

The surface for calibration on a "surface" is workpiece-related. It is only possible to calibrate in the selected axis and direction (perpendicular in front of the surface).

Calibration in a "reference groove" is relative to the machine. In this case, calibration is possible in one cycle call in both axis directions.

It is additionally possible select determining the skew of the probe in the idle position or determining the active probe ball diameter. The workpiece probe can only have cutting edge positions SL=7 or SL=8.

#### **Calibration principle**

The switching position of the workpiece probe in an axis is calculated into the measuring probe length. The trigger values calculated in this way (relative to ball center) is then entered in the corresponding array \_WP[] of array GUD6.DEF for the associated probe **\_PRNUM** (\_WP[\_PRNUM-1,...]).

For a complete description of the array \_WP[] of a workpiece probe, refer to Subsection 9.2.3.







#### **Measuring variants**

Measuring cycle CYCLE973 permits the following calibration variants that are specified with

parameter \_MVAR.

Digit			Meaning		
5	4	3	2	1	
-	-	-	-	0	Calibration on surface (workpiece-related)
			1	3	Calibrate in groove (machine-related)
0			1	3	Without determining skew of probe
1			1	3	Determining skew of probe
	1		1	3	1 axis direction (specify meas. axis _MA and axis direction _MD )
	2		1	3	2 axis directions (specify measuring axis _MA)
		0	1	3	Without determining diameter of probe ball
		1	1	3	Determining diameter of probe ball

#### Note:

When \_MVAR=x1x13 calibration is only performed in one direction. It is not possible to determine skew or calculate probe ball.

#### **Result parameters**

Measuring cycle CYCLE973 returns the following

values in block GUD5 for calibration:				
_OVR [4]	REAL	Actual value probe ball diameter		
_OVR [5]	REAL	Difference probe ball diameter		
_OVR [8]	REAL	Trigger point in minus direction, actual value, abscissa		
_OVR [10]	REAL	Trigger point in plus direction, actual value, abscissa		
_OVR [12]	REAL	Trigger point in minus direction, actual value, ordinate		
_OVR [14]	REAL	Trigger point in plus direction, actual value, ordinate		
_OVR [9]	REAL	Trigger point in minus direction, difference, abscissa		
_OVR [11]	REAL	Trigger point in plus direction, difference, abscissa		
_OVR [13]	REAL	Trigger point in minus direction, difference, ordinate		
_OVR [15]	REAL	Trigger point in plus direction, difference, ordinate		
_OVR [20]	REAL	Positional deviation abscissa		
_OVR [21]	REAL	Positional deviation ordinate		
_OVR [27]	REAL	Zero offset area		
_OVR [28]	REAL	Safe area		
_OVI [2]	INTEGER	Measuring cycle number		
_OVI [5]	INTEGER	Probe number		
_OVI [9]	INTEGER	Alarm number		

#### 6.4.2 Calibrating in the reference groove



#### Function

With this measuring cycle and measuring variants \_\_MVAR=xxx13

it is possible to calibrate a workpiece probe with cutting edge position SL=7 or SL=8 in a **reference groove** machine-related in the axes of the plane (abscissa, ordinate).

Calibration is possible in one direction (\_MVAR=x1x13) or in both directions of an axis (\_MVAR=x2x13).

It is also possible to calibrate the skew of the probe and the active diameter of the probe ball in both directions.

The workpiece probe calibrated is selected with **\_PRNUM.** The associated array \_WP[] in data block GUD6.DEF is \_WP[\_PRNUM-1,...] (for a detailed description of the data array, refer to Subsection 9.2.3).

The reference groove is selected with **\_CALNUM**. The associated array \_KB[] in data block GUD6.DEF is \_KB[\_CALNUM-1,...].

#### Precondition

The dimensions of the reference groove must already be stored in array \_KB[] of data block GUD6.DEF for the groove selected via \_CALNUM.

The workpiece probe must be called as a tool with a tool offset.









#### Parameters

_MV/	AR					Calibration variant			
-				1	3	Calibrate in groove (machine-related)			
	0			1	3	Without determining skew of probe			
-	1			1	3	Determining skew of probe			
-	1			1	3	1 axis direction (specify meas. axis _MA and axis direction _MD )			
-	2	2		1	3	2 axis directions (specify measuring axis _MA)			
-			0	1	3	Without determining diameter of probe ball			
-			1	1	3	Determining diameter of probe ball			
_MA			1, 2			Measuring axis			
_MD			0   1	positi nega	ve ax tive a	tis direction Measuring direction (for _MVAR=x1x13 only) xis direction			
_CAL	NUM		INT	EGEF	२	Number of reference groove (calibration			
						groove)			
_PRM	MUM		INT	EGEF	र	Probe number			

### The following additional parameters are also valid

\_VMS, \_TZL, \_TSA, \_FA, and \_NMSP. Refer to Sections 2.2 and 2.3.

#### Note:

When \_MVAR=x1x13 calibration is only performed in one direction. It is not possible to determine skew or calculate probe ball.



#### Attention!

The first time calibration is performed the default setting in the array of the probe is still "0". For that reason \_TSA> radius probe ball must be programmed to avoid alarm "Safe area violated".



#### Sequence

#### Position before measuring cycle call

The starting point must be selected such that the selected workpiece probe can be positioned in the cycle into the reference groove selected via \_CALNUM by the shortest path with paraxial collision-free movements in accordance with the active cutting edge position.

#### **Position after end of measuring cycle** On completion of calibration, the probe is positioned facing the calibration surface at distance \_FA.

10.04

### Programming example

#### Calibrate workpiece probe in reference groove

The workpiece probe 1 with cutting edge position SL=7 is to be calibrated in reference groove 1 in both axes and in both directions in X. The probe is inserted as tool T8, D1.

Probe lengths L1 and L2 always refer to the probe ball center and must be entered in the tool offset memory before the measuring cycle is called, T8, D1:

Tool type (DP1):	580
Cutting edge position (DP2)	7
Length 1 - Geometry (DP3):	L1 = 40.123
Length 2 - Geometry (DP4):	L2 = 100.456
Radius - Geometry DP6):	3.000



The data for reference groove 1 have already been

entered: \_KB[0,0] = 60.123, \_KB[0,1] = 50.054, \_KB[0,2] = 15.021

#### %\_N\_CALIBRATE\_IN\_GROOVE\_MPF

N10 T8 D1	;Tool offset of the probe
N20 G0 SUPA G90 DIAMOF Z125 X95	;Position in front of cycle call (starting
	;position), position without ZO
N30 _TZL=0 _TSA=1 _VMS=0 _NMSP=1 _FA=3	;Set parameters for calibration,
_PRNUM=1	;minus Z direction
N31 _MVAR=13 _MA=1 _MD=1 _CALNUM=1	
N40 CYCLE973	;Cycle call
N50 _MVAR=02013 _MA=2	;In X axis, both directions
N60 CYCLE973	;Cycle call
N70 G0 SUPA Z125	;Retraction in Z
N80 SUPA X95	;Retraction in X
N100 M2	:End of program



#### Explanation

The cycle automatically approaches reference groove 1 from the starting position and performs calibration in both axes and in the X axis in a double cycle call. The new trigger values are stored in the data of the workpiece probe 1 \_\_\_\_\_WP[0,1], \_\_WP[0,3], \_\_WP[0,4]. At the end, result array \_\_OVR[ ] contains the values of the 2nd cycle call.



#### 6.4.3 Calibration at a surface



#### Function

With this measuring cycle and measuring variant \_MVAR=0

it is possible to calibrate a workpiece probe with cutting edge positions SL=5 to 8 on a surface (workpiecerelated) and therefore determine the probe trigger points.

The position of the surfaces is defined in workpiece coordinates in **\_SETVAL**.

The workpiece probe calibrated is selected with **\_PRNUM.** The associated array \_WP[] in data block GUD6.DEF is \_WP[\_PRNUM-1,...].

#### Precondition

The surface must be parallel to an axis of the workpiece coordinate system and have low surface roughness.

The workpiece probe is called as a tool with tool offset and positioned opposite the calibration surface. 5xy should be entered as the tool type.



#### Parameters

_MVAR	0	Calibration on surface (workpiece-related)
_SETVAL	REAL	Setpoint referred to the workpiece zero, for
		facing axis in the diameter (DIAMON)
_MA	1, 2, 3 <sup>1)</sup>	Measuring axis
_MD	<ol> <li>positive axis direction</li> <li>negative axis direction</li> </ol>	Measuring direction
_PRNUM	INT	Probe number

1) It is also possible to calibrate in the 3rd axis (Y in G18), provided that this axis exists.

The following additional parameters are also valid

\_VMS, \_TZL, \_TSA, \_FA, and \_NMSP.

Refer to Sections 2.2 and 2.3.

#### Attention!

The first time calibration is performed the default setting in the array of the probe is still "0". For that reason \_TSA> probe ball radius must be programmed to avoid alarm "Safe area violated". 10.04



#### Sequence

10.04

#### Position before measuring cycle call

The starting point must be a position facing the calibration surface.

#### Position after end of measuring cycle

On completion of calibration, the probe is positioned facing the calibration surface at distance \_FA.



#### Programming example

#### Calibration of probe 1 on a surface

Workpiece probe 1 with cutting edge position SL=7 is to be calibrated on surface Z=-18 mm in direction minus Z.

The probe is inserted as tool T9, D1.

Probe lengths L1 and L2 always refer to the probe ball center and must be entered in the tool offset memory before the measuring cycle is called, T9, D1:

Tool type (DP1):	580
Cutting edge position (DP2)	7
Length 1 - Geometry (DP3):	L1 = 40.123
Length 2 - Geometry (DP4):	L2 = 100.456
Radius - Geometry DP6):	3.000

Zero offset, with settable ZO G54: ZOz

#### % N CALIBRATE IN Z MPF

Calibrat	e workpiece p	probe on surface	
Example:	SL=7, in minus 2	Z direction	. –
			F
	1		
<b>▲</b> ⊻	رم <sup>۲</sup>		
	`' <u>+</u>		-
		ų ų	
		L2	
	$\sim$		~
	//////5		X
(D 66)		///	(D 66)
	$\mathbb{N}$		1/
		<u>//X//////////////////////////////////</u>	
M	NVz	-18	Z1/Z
-			1

N10 G54 G90 G0 X66 Z90 T9 D1 DIAMON	;Activate ZO, select the tool offset	
	;of the probe	
	;Position before cycle call	
N20 _MVAR=0 _SETVAL=-18 _MA=1 _MD=1 _TZL=0	;Set parameters for calibration in minus Z	
_TSA=1 _PRNUM=1 _VMS=0 _NMSP=1 _FA=3	;direction, _SETVAL is negative!	
N30 CYCLE973	;Cycle call	
N40 G0 Z90	;Retraction in Z	
N50 X146	;Retraction in X	
N100 M2	;End of program	



#### Explanation

The surface with position Z=-18 is approached in the Z axis in the minus direction (\_SETVAL=-18, \_MA=1, \_MD=1).



#### Measuring Cycles for Turning Machines 6.4 CYCLE973 calibrate workpiece probe

Actual calibration starts \_FA=3 mm in front of the surface. The workpiece probe is then calibrated and ends up facing the surface again at distance \_FA from it.

The new trigger value in minus Z is entered in the data of probe 1 \_WP[0,1] and in the result field \_OVR[].

The original position is approached in block N40, N50.



#### 6.5 CYCLE974 workpiece: 1-point measurement

#### 6.5.1 Function overview



Programming

CYCLE974



#### Function

This measuring cycle can be used in various measurement variants to determine workpiece dimensions in a 1-point measurement. It is also possible to determine a zero offset (ZO) or an automatic tool offset.

- 1-point measurement and ZO determination
- 1-point measurement and tool offset
- 1-point measurement with reversal and tool
   offset

The measuring cycle determines the actual value of the workpiece with respect to the workpiece zero in the selected measuring axis \_MA and calculates the difference from a defined setpoint (setpoint-actual value).

An empirical value stored in data block GUD5 can be included. For variants "with tool offset" it is also possible to calculate a mean value over several parts. The cycle checks that a set tolerance range for the measured deviation is not violated and automatically corrects the ZO memory or tool offset memory selected in \_KNUM.

If KNUM=0, there is no offset.

#### Precondition

The probe must be calibrated in the measuring direction and as a tool with tool offset. The tool type is 5xy.

The cutting edge position can be 5 to 8 and must be suitable for the measurement task.







#### **Measuring variants**

Measuring cycle CYCLE974 permits the following measuring variants that are specified with parameter \_MVAR.

Value	Meaning	
0	1-point measurement and tool offset	
100	1-point measurement and ZO determination	
1000	1-point measurement with reversal and tool offset	

#### **Result parameters**

Depending on the measuring variant, measuring			
cycle CYCLE974 returns the following result values			
in the GUD5	block:		
_OVR [0]	REAL	Setpoint value for measuring axis	
_OVR [1]	REAL	Setpoint for abscissa	
_OVR [2]	REAL	Setpoint for ordinate	
_OVR [3]	REAL	Setpoint for applicate	
_OVR [4]	REAL	Actual value for measuring axis	
_OVR [8] <sup>1)</sup>	REAL	Upper tolerance limit for measuring axis	
_OVR [12] <sup>1)</sup>	REAL	Lower tolerance limit for measuring axis	
_OVR [16]	REAL	Difference for measuring axis	
_OVR [20] <sup>1)3)</sup>	REAL	Offset value	
_OVR [27] <sup>1)</sup>	REAL	Zero offset area	
_OVR [28]	REAL	Safe area	
_OVR [29] <sup>1)</sup>	REAL	Dimensional difference	
_OVR [30]	REAL	Empirical value	
_OVR [31] <sup>1)</sup>	REAL	Mean value	
_OVI [0]	INTEGER	D number or ZO number	
_OVI [2]	INTEGER	Measuring cycle number	
_OVI [4] <sup>1)</sup>	INTEGER	Weighting factor	
_OVI [5]	INTEGER	Probe number	
_OVI [6] <sup>1)</sup>	INTEGER	Mean value memory number	
_OVI [7]	INTEGER	Empirical value memory number	
_OVI [8]	INTEGER	Tool number	
_OVI [9]	INTEGER	Alarm number	
_OVI [11] <sup>2)</sup>	INTEGER	Status offset request	
1) For workpiece measurement with tool offset only			

2) For ZO correction only

3) From meas. cycles - SW 6.3: For "tool offset" the offset value always appears in \_OVR[20] as radius dimension.



#### Starting positions for various measuring tasks

The starting positions before cycle call depend on the measuring task, the value of the setpoint \_SETVAL, the measuring axes, and the cutting edge position (SL) of the workpiece probe.

The probe must be positioned facing the point to be measured and is reached by traversing measuring axis \_MA in the setpoint direction in the measuring cycle. The setpoint (position of the point) is defined by parameter \_SETVAL.

Measurement is possible parallel to and in the direction of the axes permitted by the "cutting edge position" of the workpiece probe inserted.







#### Function

With this measuring cycle and measuring variant \_MVAR=100

the actual value of a workpiece is determined with reference to the workpiece zero in the selected measuring axis \_MA.

An empirical value from data block GUD5 can be included with the correct sign.

The zero offset (ZO) is applied in such a way that the actual value adopts the required setpoint (\_SETVAL) in the workpiece coordinate system when the corrected ZO (frame) is used. Mirroring can be active in a frame of the frame sequence. Dimension factors must never be active.

The ZO to be corrected is specified in coded form with variable **\_KNUM** >0.

The ZO can be specified and corrected by various methods, e.g. in various settable frames, in various basic frames, system frames, fine offset, or coarse offset, etc.

For detailed information on specifying \_KNUM for the zero offset: Refer to Section 2.3.

#### Precondition

If necessary, the workpiece must be positioned in the correct angular spindle position with SPOS before the cycle is called.



6

# Parameters

_MVAR	100	1-point measurement and ZO determination
_SETVAL	REAL <sup>1)</sup>	Setpoint, with reference to the workpiece zero
_MA	1, 2, 3 <sup>1)</sup>	Measuring axis
_KNUM	0, >0	0: No automatic ZO correction
		>0: With automatic ZO correction
		(individual values: See Section 2.3, parameter _KNUM)

Measurement in the 3rd axis of the plane (with G18 in Y) is also possible, provided this axis exists. Moreover, for measurement in the 3rd axis of the plane with active G18 (measurement in the Y axis), the same setpoint parameterization can be used as for measurement in the X axis (facing axis), if \_CHBIT[19]=1 is set in block GUD6. The offset is then stored in the X component of the selected ZO memory.



## The following additional parameters are also valid

# \_VMS, \_TSA, \_FA, \_PRNUM, \_EVNUM, and \_NMSP.

Refer to Sections 2.2 and 2.3.

#### Sequence

#### Position before measuring cycle call

The probe must be positioned opposite the surface to be measured.

#### Position after end of measuring cycle

On completion of measurement, the probe is positioned facing the measuring surface at distance \_FA.

#### Attention!

Precise measurement is only possible with a probe calibrated under the measurement conditions, i.e. working plane and measuring velocity are the same for both measurement and calibration. If the probe is used in the spindle for a powered tool, the spindle orientation must also be considered. Deviations can cause additional measuring errors.



#### Programming example

#### ZO calculation at a workpiece

The intention is to determine the zero offset in the Z axis on a clamped workpiece with workpiece probe 1, inserted as tool T8, D1. The position determined should retain the value 60 mm in the new workpiece coordinate system for G54.

Measurement is also performed with G54.

The probe is already calibrated and the tool data are entered in T8, D1:

Tool type (DP1):	580
Cutting edge position (DP2)	7
Length 1 - Geometry (DP3):	L1 = 40.123
Length 2 - Geometry (DP4):	L2 = 100.456
Radius - Geometry DP6):	3.000

Zero offset, with settable ZO G54:

ZOz

#### %\_N\_ZO\_DETERMINING\_1\_MPF



N10 G54 G90 G18 DIAMON T8 D1	;Call ZO, tool = probe
N20 G0 X36 Z100	;Starting position before cycle call
N30_MVAR=100_SETVAL=60_MA=1_TSA=1_KNUM=1	;Parameters for cycle call
N40 CYCLE974	;Measurement in the Z direction
N50 G0 Z100	;Retraction in Z
N60 X114	;Retraction in X
N100 M2	;End of program

#### Note

If parameter \_VMS has value 0, the default value of the measuring cycle is used for the variable measuring velocity:

if \_FA=1: 150 mm/min. if \_FA>1: 300 mm/min. (see Section 2.3)

#### 6.5.3 1-point measurement and tool offset



#### Function

With this measuring cycle and measuring variant \_MVAR=0

the actual value of a workpiece is determined with reference to the workpiece zero in the selected measuring axis.

An offset can also be determined for a tool depending on that. This tool is specified in \_TNUM and \_TNAME. The D number and type of offset are specified in coded form in variable \_KNUM.

From measuring cycles SW 6.3, extended tool offset is available. With this function a tool from a particular stored tool environment \_TENV, and additive, setup offsets can be corrected by specifying the DL number in \_DLNUM.

Detailed information on the parameters: Refer to Section 2.3.

#### Empirical values and mean values

An empirical value stored in data block GUD5 in array \_EV[] can be included in calculation of the result after measurement is completed. Optionally, averaging is performed over a number of parts (array \_MV[]) and the tolerance bands are checked.

Both are activated in \_EVNUM (see Section 2.3).

#### Precondition

If necessary, the workpiece must be positioned in the correct angular spindle position with SPOS before the cycle is called.



_MVAR	0	1-point measurement and tool offset
_SETVAL	REAL <sup>2)</sup>	Setpoint (according to drawing)
		(in the case of facing axis (X) and diameter programming, this
		is a diameter dimension)
_MA	1, 2, 3 <sup>1)</sup>	Measuring axis
_KNUM	0, >0	0: Without automatic tool offset
		>0: With automatic tool offset
		(individual values: See Section 2.3, parameter _KNUM)
_TNUM	INTEGER, ≥0	Tool number for automatic tool offset
_TNAME	STRING[32]	Tool name for automatic tool offset
		(alternative to _TNUM with tool management active)
_DLNUM	INTEGER, ≥0	DL number for additive and setup offset
		(> meas. cycle SW 6.3)
_TENV	STRING[32]	Name of tool environment for automatic tool compensation
		(from measuring cycles SW 6.)
1) Measure	ement can also be ma	de in the 3rd axis of the plane
(for G18	in Y) if this is availab	le

2) By setting \_CHBIT[19] to 1 in the GUD6 block, the same parameterization can be used regarding the setpoint when measuring in the Y axis (3rd axis of the plane) for active G18 as when measuring in the X axis (transverse axis). In this case, the tool offset is also applied to L1 (effective length in X) if nothing different is specified by \_KNUM.

The following additional parameters are also valid

\_VMS, \_TZL, \_TMV, \_TUL, TLL, \_TDIF, \_TSA, \_FA, \_PRNUM, \_EVNUM, \_NMSP, and \_K.

Refer to Sections 2.2 and 2.3.

# Sequence

#### Position before measuring cycle call

The probe must be positioned opposite the surface to be measured.

#### Position after end of measuring cycle

On completion of measurement, the probe is positioned facing the measuring surface at distance \_FA.

#### Attention!

Precise measurement is only possible with a probe calibrated under the measurement conditions, i.e. working plane and measuring velocity are the same for both measurement and calibration. If the probe is used in the spindle for a powered tool, the spindle orientation must also be considered. Deviations can cause additional measuring errors.



#### Programming example

# 1-point measurements at outside and inside diameters with tool offsets

An outside diameter with tool T7, D1 and an inside diameter with tool T8, D1 has been machines on a workpiece. The set diameters have the dimensions shown in the figure.

If the absolute value of the difference determined is >0.002 mm, the length (in measuring axis \_MA) of the tool is to be automatically offset in the wear.

The maximum permissible deviation is taken as max. 0.5 mm. Max. 0.04 mm is permissible. To obtain a minimum measuring path of 0.5 mm, the measuring path is programmed as \_FA=0.5+0.5=1 mm (max. total measuring path = 2 mm).

The offset must take the empirical value in memory \_EV[12] into consideration for T 7, or \_EV[13] for T 8. Mean value calculation \_MV[12] or \_MV[13] and inclusion in calculation are also to be used.

This tool offset will therefore affect the production of the next workpieces or possible remachining.

Clamping for workpiece: Zero offset, with settable ZO G54: ZOz

Workpiece probe 1, inserted as tool **T9**, **D1**, is to be used as probe.

The probe is already calibrated. Arrays for workpiece probe 1: \_WP[0, ...]

The following is entered under T9, D1 in the tool offset memory:

Tool type (DP1):	580
Cutting edge position (DP2)	7
Length 1 - Geometry (DP3):	L1 = 40.123
Length 2 - Geometry (DP4):	L2 = 100.456
Radius - Geometry DP6):	3.000





10.04

%_N_ONE_POINT_MEAS_MPF	
N10 G54 G18 G90 T9 D1 DIAMON	;Call ZO, tool = probe
N20 G0 Z30 X90	;Preposition probe
N25 _CHBIT[4]=1	;With mean value calculation
N30 _TZL=0.002 _TMV=0.005 _TDIF=0.04 _TSA=0.5 _PRNUM=1 _VMS=0 _NMSP=1 _FA=1	;Parameters for cycle call
N31 _MVAR=0 _SETVAL=45 _TUL=0 _TLL=-0.01 _TNUM=7 _KNUM=1 _EVNUM=13 _K=2 _MA=2	
N40 CYCLE974	;Measurement on the outside diameter
N50 G0 Z60	;Place probe facing P2
N60 X0	
N70 Z40	
N80 _SETVAL=35 _TUL=0.015 _TLL=0 _TNUM=8 _EVNUM=14	
N90 CYCLE974	;Measurement on the inside diameter
N100 G0 Z110	;Retraction in Z
N110 X90	;Retraction in X
N200 M2	;End of program

# =?

#### Explanation

#### Measurement of outside diameter and offset T7

The difference calculated from the actual value and setpoint is compensated for by the empirical value in the empirical value memory \_EV[12] and compared with the tolerance parameter:

- If it is more than 0.5 mm (\_TSA), alarm "Safe area violated" is output and program execution cannot be continued.
- If it is more than 0.04 mm (\_TDIF), no compensation is performed and alarm "Permissible dimensional difference exceeded" is output and the program continues.
- If values \_TUL= -0.01, \_TLL=0 are violated upward or downward, the length of T7 D1 is compensated 100% by this difference. Alarm "Oversize" or "Undersize" is displayed and the program is continued.
- If 0.005 mm (\_TMV) is exceeded, the length in T7 D1 is compensated 100% by this difference.
   If it is less than 0.005 mm (\_TMV), the mean value is calculated (only if \_CHBIT[4]=1! with mean value memory) with the mean value in mean value memory \_MV[12] and by including weighting factor (\_K=2).



- If the mean value obtained is >0.002 (\_TZL), the reduced compensation of the length 1 for T7 D1 is the mean value/2 and the mean value is deleted in \_MV[12].
- If the mean value is <0.002 (\_TZL), there is no offset but it is stored in the mean value memory \_MV[12] if mean value storage (\_CHBIT[4]=1) is active.

The results are entered in result array \_OVR[ ]. The wear of the length 1 of T7, D1 is included if a change is necessary.

### Measurement of inside diameter and offset T8

Procedure as described for "Measurement output diameter".

Offset of T8 with appropriate modified values \_EV[13], \_MV[13] (EVNUM=14), \_TUL, \_TLL, SETVAL.

#### Note

The values of the workpiece tolerance parameters \_TUL, \_TLL were selected asymmetrically in the example. The result is then made symmetrical (see Subsection 2.3.11).



#### 6.5.4 1-point measurement with reversal and tool offset



#### Function

With this measuring cycle and measuring variant \_MVAR=1000

the workpiece actual value is ascertained with reference to the workpiece zero in the measuring axis by acquiring two opposite points on the diameter. Before taking the first measurement, the workpiece is positioned at the angular position programmed in parameter **\_STA1** with SPOS and the 180° reversal is automatically generated by the cycle before the second measurement.

The mean value is calculated from both measured values.

Otherwise, the same conditions and tool offset options apply as for measuring variant \_MVAR=10 "1-point measurement and tool offset" (see Subsection 6.5.4).

#### Parameters

_MVAR	1000	1-point measurement with reversal and tool offset	
_SETVAL	REAL <sup>2)</sup>	Setpoint (according to drawing)	
		(in the case of facing axis (X) and DIAMON , this is a diameter	
		dimension)	
_MA	1, 2, 3 <sup>1)</sup>	Measuring axis	
_STA1	REAL, >=0	Starting angle (spindle position)	
_KNUM	0, >0	0: Without automatic tool offset	
		>0: With automatic tool offset	
		(individual values: See Section 2.3, parameter _KNUM)	
_TNUM	INTEGER, ≥0	Tool number for automatic tool offset	
_TNAME	STRING[32]	Tool name for automatic tool offset	
		(alternative to _TNUM with tool management active)	
_DLNUM	INTEGER, ≥0	DL number for additive and setup offset	
		(from meas. cycle SW 6.3)	
_TENV	STRING[32]	Name of tool environment for automatic tool compensation	
		(from measuring cycles SW 6.)	

- 1) Measurement can also be made in the 3rd axis of the plane (for G18 in Y) if this is available.
- 2) By setting \_CHBIT[19] to 1 in the GUD6 block, the same parameterization regarding the setpoint can be used when measuring in the Y axis (3rd axis of the plane) for active G18 as when measuring in the X axis (transverse axis).



In this case, the tool offset is also applied to L1 (effective length in X) if nothing different is specified by \_KNUM. The following additional parameters are also valid \_VMS, \_TZL, \_TMV, \_TUL, TLL, \_TDIF, \_TSA, \_FA, \_PRNUM, \_EVNUM, \_NMSP, and \_K.

Refer to Sections 2.2 and 2.3.

# **\$**

### Programming example

# 1-point measurement at outside diameter, measuring with reversal

An outside diameter with tool T7, D1 has been machined on a workpiece. The set diameter has the dimension shown in the figure.

This outside diameter is to be measured with reversal. The spindle is SPOS-capable.

If the absolute value of the difference determined is >0.002, the length (in measuring axis \_MA) of the tool is to be automatically offset in the wear.

The maximum permissible deviation is taken as max. 1 mm. Max. 0.4 mm is permissible. To obtain a minimum measuring path of 1 mm, the measuring path is programmed as FA=1+1=2 mm (max. total measuring path = 4 mm).

The offset is not to consider an empirical value and no mean value is calculated or used.

Clamping for workpiece:

Zero offset, with settable ZO G54:

ZOz

Workpiece probe 1, inserted as tool **T9**, **D1**, is to be used as probe. The probe is already calibrated. Arrays for workpiece probe 1: WP[0, ...]

The following is entered under T9, D1 in the tool offset memory:

Tool type (DP1):	580
Cutting edge position (DP2)	7
Length 1 - Geometry (DP3):	L1 = 40.123
Length 2 - Geometry (DP4):	L2 = 100.456
Radius - Geometry DP6):	3.000



%_N_REVERSALMEAS_MPF	
N10 G54 G90 G18 T9 D1 DIAMON	;Call ZO, tool = probe
N20 G0 Z30 X90	;Preposition probe
N30 _MVAR=1000 _SETVAL=45 _TUL=0 _TLL=-0.01 _MA=2 _STA1=0 _KNUM=1 _TNUM=7 _EVNUM=0 _TZL=0.002 _TDIF=0.4 _TSA=1 _PRNUM=1 _VMS=0 _NMSP=1 _FA=2	;Parameters for cycle call
N40 CYCLE974	;Measuring cycle call
N50 G0 Z110	;Retraction in Z
N60 X90	;Retraction in X
N100 M2	;End of program



#### Note

The values of the workpiece tolerance parameters \_TUL, \_TLL were selected asymmetrically in the example. The result is then made symmetrical (see Subsection 2.3.11).



#### 6.6 CYCLE994 workpiece: 2-point measurement



#### Programming

CYCLE994



#### Function

This measuring cycle can be used to determine workpiece dimensions in 2-point measurements with various measuring variants.

Automatic tool offset is also possible.

The measuring cycle determines the actual value of the workpiece with respect to the workpiece zero in the selected measuring axis \_MA and calculates the difference from a defined setpoint (setpoint-actual value).

An empirical value stored in data block GUD5 can be included. It is also possible to calculate a mean value over several parts.

The cycle checks that a set tolerance range for the measured deviation is not violated and automatically corrects the tool offset memory selected in \_KNUM. If KNUM=0, there is no offset.

Two opposite measuring points in measuring axis \_MA are approached symmetrically to the workpiece zero at the distance of the setpoint \_SETVAL.

Sequence of operations: 1. measuring point positive, 2nd measuring point negative.

A safety zone is programmed in parameters \_SZA and \_SZO. This is considered in traversal with the measuring variant. The probe ball radius must also be considered by the user.

#### Precondition

The probe must be calibrated in the measuring direction (if  $\_CHBIT[7] = 0$ ) and called as a tool with tool offset.

The tool type is 5xy. The cutting edge position can be 5 to 8 and must be suitable for the measurement task.





The measuring cycle can be used for measurement **without** previous **calibration**.

Instead of the trigger values in \_WP[ ], the probe ball diameter entered in the probe array \_PRNUM ( WP[ PRNUM-1.0]) is used in the calculation.

The function is controlled via bit:

\_CHBIT[7]=1: Probe not calibrated (without use of trigger values). Use of ball probe diameter \_WP[\_PRNUM-1,0]) \_CHBIT[7]=0: Probe calibrated, use of

trigger values in \_WP[\_PRNUM-1,...])



#### **Tool offset**

An offset can be applied for the tool that machined the workpiece. This tool is specified in \_TNUM and \_TNAME. The D number and type of offset are specified in coded form in variable \_KNUM.

From measuring cycles SW 6.3, extended tool offset is available. With this function a tool from a particular stored tool environment **\_TENV**, and additive, setup offsets can be corrected by specifying the DL number in **\_DLNUM**.

Detailed information on the parameters: Refer to Section 2.3.

#### Empirical values and mean values

An empirical value stored in data block GUD5 in array \_EV[] can be included in calculation of the result after measurement is completed.

Optionally, averaging is performed over a number of parts (array \_MV[ ]) and the tolerance bands are checked.

Both are activated in \_EVNUM (see Section 2.3).



#### **Measuring variants**

Measuring cycle CYCLE994 permits the following measuring variants that are specified in parameter \_MVAR.

Value	Meaning
1	2-point measurement with programmed safety zone
	(This measuring variant is only for inside measurement!)
2	2-point measurement with programmed safety zone
	(For inside measurement without safety zone in this measuring variant!)

#### **Result parameters**

The measuring cycle CYCLE994 supplies the following values as results in the GUD5 block:

-			
_OVR [0]	REAL	Setpoint diameter or setpoint as radius dimension (note _MA)	
_OVR [1]	REAL	Setpoint diameter/radius in abscissa	$\rightarrow$ with _MA=1 only
_OVR [2]	REAL	Setpoint diameter/radius in ordinate	$\rightarrow$ with _MA=2 only
_OVR [3]	REAL	Setpoint diameter/radius in applicate	$\rightarrow$ with _MA=3 only
_OVR [4]	REAL	Actual value for diameter/radius	
_OVR [5]	REAL	Actual value diameter/radius in abscissa	$\rightarrow$ with _MA=1 only
_OVR [6]	REAL	Actual value diameter/radius in ordinate	$\rightarrow$ with _MA=2 only
_OVR [7]	REAL	Actual value diameter/radius in applicate	$\rightarrow$ with _MA=3 only
_OVR [8]	REAL	Upper Tolerance limit for diameter/radius	
_OVR [12]	REAL	Lower tolerance limit for diameter/radius	
_OVR [16]	REAL	Difference for diameter	
_OVR [17]	REAL	Difference diameter/radius in abscissa	$\rightarrow$ with _MA=1 only
_OVR [18]	REAL	Difference diameter/radius in ordinate	$\rightarrow$ with _MA=2 only
_OVR [19]	REAL	Difference diameter/radius in applicate	$\rightarrow$ with _MA=3 only
_OVR [20] <sup>1)</sup>	REAL	Offset value	
_OVR [27]	REAL	Zero offset area	
_OVR [28]	REAL	Safe area	
_OVR [29]	REAL	Dimensional difference	
_OVR [30]	REAL	Empirical value	
_OVR [31]	REAL	Mean value	
_OVI [0]	INTEGER	D number	
_OVI [2]	INTEGER	Measuring cycle number	
_OVI [4]	INTEGER	Weighting factor	
_OVI [5]	INTEGER	Probe number	
_OVI [6]	INTEGER	Mean value memory number	
_OVI [7]	INTEGER	Empirical value memory number	
_OVI [8]	INTEGER	Tool number	
_OVI [9]	INTEGER	Alarm number	
Whon mean	uring in the tr	overse axis and for diameter programming (DIA	MON) all of the

F

When measuring in the traverse axis and for diameter programming (DIAMON), all of the

dimensioned parameters are diameter dimensions, otherwise radius dimensions.

1) from meas. cycles - SW 6.3: For "tool offset" the offset value always appears in \_OVR[20] as radius dimension, independent of DIAMON or DIAMOF



#### Parameters

_MVAR	1 or 2	1: Inside measurement, 2-point measurement with safety zone
		2: 2-point measurement, safety zone only for outside measurement
_SETVAL	REAL <sup>2)</sup> Setpoint	
		If measurement is made in the transverse axis and diameter
		programming (DIAMON) is active, then _SETVAL is a diameter
		dimension, otherwise a radius dimension around the workpiece zero.
_MA	1, 2, 3 <sup>1)</sup>	Measuring axis
_SZA	REAL	Protection zone at the workpiece in the abscissa <sup>2)</sup>
		If the abscissa is a transverse axis and diameter programming
		(DIAMON) is active, then _SZA is a diameter dimension, otherwise
		a radius dimension around the workpiece zero.
_SZO	REAL	Protection zone at the workpiece in the ordinate <sup>2)</sup>
		If the ordinate is a transverse axis and diameter programming
		(DIAMON) is active, then _SZO is a diameter dimension, otherwise
		a radius dimension around the workpiece zero.
_KNUM	0, > 0	0: Without automatic tool offset
		>0: with automatic tool offset
		(individual values: See Section 2.3, parameter _KNUM)
_TNUM	INTEGER, ≥0	Tool number for automatic tool offset
_TNAME	STRING[32]	Tool name for automatic tool offset
		(alternative to _TNUM with tool management active)
_DLNUM	INTEGER, ≥0	DL number for additive and setup offset (from meas. cycle SW 6.3)
_TENV	STRING[32]	Name of tool environment for automatic tool compensation
		(from measuring cycles SW 6.)

 Measurement in the 3rd axis of the plane is also possible, provided this axis exists (\_MA=3: if G18 is in the Y axis).

For measurement in the 3rd axis (in G18 in Y\_SZO applies in this axis. \_SZA still applies in the 1st axis in the plane (Z axis in G18). Travel around is performed in the 1st axis of the plane (Z axis in G18). Setting \_CHBIT[19]=1 in block GUD6 enables the same setpoint and safety zone parameterization to be used for measurement in the 3rd axis (measurement in the Y axis)

measurement in the 3rd axis (measurement in the Y axis) with active G18 as for measurement in the X axis (facing axis).

The tool offset is then also in L1 if not specified differently in \_KNUM.

The following additional parameters are also valid \_VMS, \_TZL, \_TMV, \_TUL \_TLL, \_TDIF, \_TSA, \_FA, \_PRNUM, \_EVNUM, \_NMSP, and \_K. Refer to Sections 2.2 and 2.3.



#### Sequence

10.04

#### Position before measuring cycle call

The probe must be positioned opposite the **positive** measuring point.

#### Position after end of measuring cycle

After the end of measurement, the probe is facing the **negative** measuring point at distance \_FA.



#### Attention!

Precise measurement is only possible with a probe calibrated under the measurement conditions, i.e. working plane and measuring velocity are the same for both measurement and calibration. If the probe is used in the spindle for a powered tool, the spindle orientation must also be considered.

Deviations can cause additional measuring errors.



# Procedure for outside measurement with \_MVAR=2, \_MA=2:

(safety zone \_SZA, \_SZO active)

- 1: Approach path outside diameter (user)
- 2 to 7: Traverse paths generated by the cycle for Measurement at the outer diameter taking into account the protection zone \_SZA, \_SZO (4 to 6)
- 8 to 9: Retraction to the original point (user)





# Procedure for inside measurement with MVAR=2, MA=2:

(no safety zone active)

- 1, 2: Approach paths inside diameter (user)
- 3 to 5: Traverse paths generated by the cycle for Measurement on the inside diameter
- 6: Retraction paths to the original point (user)





#### Programming example

**Two-point measurement, outside and inside** An outside diameter with tool T8, D1 and an inside diameter with tool T9, D1 has been machines on a workpiece. The set diameters have the dimensions shown in the figure.

If the absolute value of the difference determined is >0.002 mm, the length (in measuring axis \_MA) of the tool is to be automatically offset in the wear.

The maximum permissible deviation is taken as max. 0.5 mm. Max. 0.04 mm is permissible. To obtain a minimum measuring path of 0.5 mm, the measuring path is programmed as FA= 0.5+0.5=1 mm (max. total measuring path = 2 mm).

The offset must take the empirical value in memory \_EV[2] into consideration for T 8, or \_EV[3] for T 9. Mean value calculation \_MV[2] or \_MV[3] and inclusion in calculation are also to be used.

This tool offsets will therefore affect the production of the next workpieces or possible remachining.

Clamping for workpiece: Zero offset, with settable ZO G54: ZOz The workpiece probe 1, used as tool **T1, D1**, should be used as probe.





The probe is already calibrated. Arrays for	
workpiece probe 1: _WP[0,]	
The following is entered under T1, D1 in the tool	
offset memory:	
Tool type (DP1): 580	
Cutting edge position (DP2) 7	
Length 1 - Geometry (DP3): $L1 = 40.123$	
Length 2 - Geometry (DP4): L2 = 100.456	
Radius - Geometry DP6): 3.000	
%_N_TWO_POINT_MEAS_MPF	
N10 T1 D1 DIAMON	;Call tool = probe (MT)
N20 G0 G54 Z30 X60	;ZO selection, Position probe facing P1
N25_CHBIT[4]=1_CHBIT[7]=0	;With mean value calc., calibrated MT
N30 _TLL=-0.01 _MA=2 _SZA=55 _SZO=55 _KNUM=1	;Parameter assignment for 1st cycle call
_K=3 _TZL=0.002 _TMV=0.005 _TDIF=0.04 _TSA=0.5	;(outside measurement)
_VMS=0 _NMSP=1 _FA=1 _MVAR=2	
N31 _SETVAL=45 _TUL=0 _TNUM=8 _EVNUM=3	
N40 CYCLE994	;2-point measurement outside
	;with safety zone (P1)
N50 G0 Z55	;Position probe facing P2
N60 X20	
N70 Z30	
N80 _SETVAL=35 _TUL=0.015 _TNUM=9 _EVNUM=4	;Parameter change for 2nd cycle call
	;(inside measurement)
N90 CYCLE994	;2-point measurement inside
	;without safety zone (P2)
N100 G0 Z110	;Retraction in Z
N110 X60	;Retraction in X
N200 M2	;End of program

#### Explanation

#### Measurement of outside diameter and offset T8

The difference calculated from the actual value and setpoint is compensated for by the empirical value in the empirical value memory \_EV[2] and compared with the tolerance parameter:

- If it is more than 0.5 mm (\_TSA), alarm "Safe area violated" is output and the program can no longer be executed.
- If it is more than 0.04 mm (\_TDIF), no compensation is performed and alarm "Permissible dimensional difference exceeded" is output and the program continues.
- If values \_TUL= -0.01, \_TLL=0 are violated upward or downward, the length of T8 D1 is compensated 100% by this difference. Alarm "Oversize" or "Undersize" is displayed and the program is continued.

- If 0.005 mm (\_TMV) is exceeded, the length in T8 D1 is compensated 100% by this difference.
- If it is less than 0.005 mm (\_TMV), the mean value is calculated (only if \_CHBIT[4]=1! with mean value memory) with the mean value in mean value memory \_MV[2] and by including weighting factor (\_K=3).
  - If the mean value obtained is >0.002 (\_TZL), the reduced offset of length 1 for T8 D1 is the mean value/2 and the mean value is deleted in \_MV[2].
  - If the mean value is <0.002 (\_TZL), there is no offset but it is stored in the mean value memory \_MV[2] if mean value storage (\_CHBIT[4]=1) is active.

The results are entered in result field \_OVR[]. The wear of the length L1 (KNUM=1, \_MA=2) of T8, D1 is included if a change is necessary.

#### Measurement of inside diameter and offset T9

Procedure as described for "Measurement output diameter".

Offset of T8 with appropriate modified values \_EV[3], \_MV[3] (EVNUM=4), \_TUL, \_SETVAL.

#### Note

The values of the workpiece tolerance parameters \_TUL, \_TLL were selected asymmetrically in the example. The result is then made symmetrical (see Subsection 2.3.11).



#### 6.7 Complex example for workpiece measurement

#### Explanation

The workpiece shown in the figure is to be measured with workpiece probe 1 with cutting edge position 7, inserted as tool T8, D1, in **CYCLE974**. This tool master is previously calibrated with **CYCLE973** in reference groove 1 in both axes in the negative direction. Measuring points P1 to P4 were machined with different tools T1, D1 to T4, D1. These tools are to be automatically corrected in the length (according to measuring axis \_MA) without empirical and mean values.







### Programming example

Calibration with workpiece probe CYCLE973, measurement of workpiece with CYCLE974

N10 T8 D1 DIAMON	;Select tool = probe
N20 SUPA G0 X300 Z150	;Approach starting position in X and Z, from ;which it is possible to approach the ;reference groove for calibration without ;collision
N30 _MVAR=13 _MA=1 _MD=1 _CALNUM=1 _TZL=0	Parameters for calibration in ref. groove
	·
N40 CYCLE973	;Calibrate probe in the minus Z direction
N50 _MA=2	;Another measuring axis
N60 CYCLE973	;Calibrate probe in the minus X direction
N70 G54 G0 Z40	;Select zero offset, traverse to measuring ;point in the Z axis
N80 X220	;Position probe facing P 1
N100 _TUL=0 _TLL=-0.01 _TZL=0.002 _EVNUM=0 _TDIF=0.2 _TSA=0.3 _PRNUM=1 _MVAR=0 _SETVAL=200 _MA=2 _TNUM=1 _KNUM=1	;Define parameters for measurement
N110 CYCLE974	;Measure P1
N120 G0 Z70	;Position probe facing P2
N130 X175	
N140 _MA=1 _SETVAL=50 _TUL=0.01 _TNUM=2 _KNUM=1	;Define parameters for measurement in ;another axis
N150 CYCLE974	;Measure P2
N160 G0 Z180	;Position probe opposite P3
N170 _MA=2 _SETVAL=150 _TUL=0.005 _TLL=-0.003 _TNUM=3 _KNUM=1	;Change parameters for measurement
N180 CYCLE974	;Measure P3
N190 G0 Z150	;Position probe opposite P4
N200 X50	
N210_MA=1_SETVAL=100_TUL=0.01_TLL=-0.01	;Change parameters for measurement
_TNUM=4 _KNUM=1	
_TNUM=4 _KNUM=1 N220 CYCLE974	;Measure P4
_TNUM=4 _KNUM=1 N220 CYCLE974 N230 G0 SUPA Z250	;Measure P4 ;Retraction in Z
_TNUM=4 _KNUM=1 N220 CYCLE974 N230 G0 SUPA Z250 N240 SUPA X280	;Measure P4 ;Retraction in Z ;Retraction in X
### **Miscellaneous Functions**

7.1	Logging the measurement results	
7.1.1	Overview of the logging cycles	
7.1.2	Log file	
7.1.3	Handling the log cycles	
7.1.4	Variables when logging	
7.1.5	Selecting log content	
7.1.6	Log format	
7.1.7	Log header	
7.1.8	Example: Creating a measurement result log	7-371
7.2	Measuring cycle support in the program editor (up to measuring cycles	
	SW 5.4)	7-374
7.2.1	Files of the measuring cycle support	
7.2.2	Loading the measuring cycle support	
7.2.3	Assigning calls and measuring cycles	7-376
7.2.4	Description of the parameterizing cycles	7-377
7.3	Measuring cycle support in the program editor (from measuring cycles	
	SW 6.2)	7-387
7.3.1	Menus, explanation of the cycles	
7.3.2	Pre-setting of the measuring cycle support	
7.4	Measuring result screens	



### 7.1 Logging the measurement results

The measuring cycles support measuring cycle logging into a file on the control. There are no special hardware requirements for logging measurement results. It is executed solely by the software.

### 7.1.1 Overview of the logging cycles

CYCLE100	Log ON
CYCLE101	Log OFF
CYCLE105	Generate log contents
CYCLE106	Log sequential controller
CYCLE113	Internal subroutine: Log
CYCLE118	Format real values

### 7.1.2 Log file

# Fu

### Function

The log file is stored in the directory where the calling program is located. You can specify the file name for the log file. The restrictions that apply to program names also apply here. Only letters, numbers and underscores are permitted, and the name must commence with two letters or a letter followed by an underscore. The file always has the extension "MPF".

The maximum length of the log file is limited in MD 11420: LEN\_PROTOCOL\_FILE. If the system detects during writing that a data record is too long, another log file is automatically created. Underscore and a digit are added to the name specified in \_PROTNAME[1] and the following message is output:

#### "New log file has been created".

In this way, up to 9 subsequent logs can be stored in the control. After the 10th log operation is halted and the following message is output:

### "Please specify new log name".

After restart, operation is continued. If a log file with the same name already exists before logging is started, then it is deleted before writing is started.



### 7.1.3 Handling the log cycles



### Function

- The log is enabled and disabled via the program (CYCLE100/CYCLE101). This requires a cycle call without setting any parameters.
- After disabling the log function, the log files can be exported from the part program memory to diskette (MMC102/103 and HMI Advanced only) or via RS-232-C.



### Sequence

When used with the measuring cycles, it is sufficient to activate and deactivate the logging with CYCLE100 and CYCLE101 respectively. Logging is carried out with the parameters described in Subsections 7.1.5 to 7.1.7. Logging is executed by CYCLE105, CYCLE106, CYCLE113, and CYCLE118. These cycles are called up in conjunction with measuring cycles. The log cycles may be used independently of the measuring cycles. CYCLE100 and CYCLE101, and CYCLE105 and CYCLE106 are called explicitly in this context. CYCLE113 and CYCLE118 are called internally. You can also be called separately, if necessary.



### CYCLE100 Log ON

After the log is enabled, an existing file with the specified name is automatically deleted in the control. All follow-up logs with \_PROTNAME[1]\_digit are only deleted when the preceding logs overflow. The log is reopened and the header is entered. The internal status variables are set.

### CYCLE101 Log OFF

Disables the logging function and resets the internal flag.

### CYCLE105(PAR1) Generate log contents

 This cycle generates up to 4 lines of log contents (lines of values) according to the entries in the variables of the data block (GUD6). It allows you to generate only value lines or only the log header depending on the setting for PAR1.





Formatting one number:

- The input and output values are specified directly in the first 3 parameters passed on cycle call, i.e. the number to be formatted (PAR1), the number of digits (PAR2), and the name of a string variable for the formatted string that is returned.
- For the return value, up to 12 characters are provided, i.e. the return variable is of type STRING[12].



#### Formatting more than one number:

- Up to 10 numbers can be formatted; the actual number of numbers to be formatted is passed to the cycle in PAR4.
- The cycle takes the numbers from consecutive R parameters, .e.g. R11 to R20.
- It is possible to start with any R parameter. It is passed to the cycle on calling (PAR5).
- The number of decimal places is specified in the same way as for formatting one number (PAR2).
- The return values are in the predefined variables \_TXT[0] up to \_TXT[9] (variable field data block GUD6).



### Parameters

PAR1	REAL		Number to be formatted
PAR2	INTEGER	≥0	Number of decimal places
PAR3	STRING[12]		Formatted return as STRING
PAR4	INTEGER	≥1	Number of numbers to be formatted
PAR5	INTEGER	≥0	Number of the first R parameter only relevant if
			PAR4>1



### Programming example

#### Example 1: Formatting a single number

The value of **one** variable of type REAL is to be formatted for 2 decimal places and displayed:

DEF STRING[12] TEXTVAR	
DEF REAL VAR1	
VAR1=100/\$PI	;\$PI - circle constant PI = 3.1415927
CYCLE118 (VAR1,2,TEXTVAR)	
MSG("VAR1:"< <var1<<" textvar='31.83"&lt;/td'></var1<<">	
M30	

#### Example 2: Formatting three numbers

The intention is to format the values of R parameters **R11 to R13** for the decimal places in \_DIGIT and display them. DIGIT is a variable in data block GUD6 for setting the

decimal places in the measuring cycles. The value in the default setting is \_DIGIT=3. PI = 3.1415927. The results in STRING format are

supplied in \_TXT[0] to \_TXT[2]; the value for the last parameter (R13) also in TEXTVAR.



DEF STRING[12] TEXTVAR
R11=0.1/\$PI R12=1/\$PI R13=10/\$PI
CYCLE118(0,_DIGIT,TEXTVAR,3,11)
MSG("_TXT[0]="<<_TXT[0]<<" _TXT[1]="_TXT[1]<<" _TXT[2]="< <txt[2])< td=""></txt[2])<>
M0 ;Stop – view values:
;"_TXT[0]=0.032_TXT[1]=0.318_TXT[2]=3.183
M30

### 7.1.4 Variables when logging

### Function

With these parameters you can

- select the content of the log,
- format the log,
- determine the content of the log header.

In the measuring cycle, data logging is controlled via

the following data bit:

the following data	on.						
_CBIT[6]=	0 \	With measuring cycle name and measuring variant					
	1 Without measuring cycle name and measuring variant						
_CBIT[11]=	] = 0 Standard log header						
	1 l	Jser-defined lo	g header				
The following varia	ables (data bl	lock GUD6) de	scribe the contents of the measuring log:				
Variable Type Default value Table of Contents							
_PROTNAME [2]	STRING[32]	Blank string	_PROTNAME [0] = Name of the main program from				
			which the log is written				
			_PROTNAME [1] = Name of the log file				
HEADLINE[10]	STRING[80]	Blank string	_HEADLINE[0]HEADLINE[9]				
			The user can enter customized texts in these STRINGs;				
			they are included in the log				
Variable	Туре	Default value	Table of Contents				
_PROTFORM[6]	INTEGER	60	_PROTFORM[0] = Number of lines per page				
		80	_PROTFORM[1] = Number of characters per line				
		1	_PROTFORM[2] = First page number				
		3	_PROTFORM[3] = Number of customized header lines				
		1	_PROTFORM[4] = Number of value lines in the log				
		12	_PROTFORM[5] = Number of characters per column				
_PROTSYM[2]	CHAR	" " "	_PROTSYM[0] = Separators between the values in				
			the log				
		"#"	_PROTSYM[1] = Special characters for identification				
			when tolerance limits are exceeded				
_PROTVAL[13]	STRING[80]		_PROTVAL[0] = Contents of the header line (line 9)				
			_PROTVAL[1] = Contents of the header line (line 10)				
			PROTVAL[2][5] = Specification of the values to				
			be logged in successive lines				
_TXT[100]	STRING[12]		Field for formatted STRINGs				

### 7.1.5 Selecting log content



### Function

The measurement result log contains parts that are fixed and some that can be set. It always contains:

- Measuring cycle (cycle name)
- Measuring variant (value of \_MVAR)

Output of the measuring cycle and measuring variant can be suppressed by setting \_CBIT[6].

The following additional data can be included in a log:

- Time (specification \_TIME)
- Axis of the associated measuring axes
  - Specification \_AXIS:

The axis name is entered automatically according to the measuring axis entered in MA.

- or specification \_AXIS1... 3:

\_AXIS1: Axis name of the abscissa in the selected plane

- \_AXIS2: Axis name of the ordinate in the selected plane
- \_AXIS3: Axis name of the applicate in in the selected plane
- All result data provided by the measuring cycle in the \_OVR[ ] array.
- R parameters
- Comment texts and
- STRINGs stored in \_TXT[] (GUD6)

The logging values to be selected must correspond to the measuring cycle and the selected measuring variant. This makes for versatile adaptation of the contents of the log to meet your requirements.

Specification of the log contents is conducted via the variable \_PROTVAL[ ].

The strings stored in \_PROTVAL[0] and \_PROTVAL[1] are used as header lines for the log (see example in Subsection 7.1.9, Lines 8 - 10). \_PROTVAL[2]...[5] specify the line contents of the individual log lines.



If you change the measuring cycle or the measuring variant, you may have to adapt \_PROTVAL[2]...[5] (see example in Subsection 7.1.9).

The comma is used as the separator between variables.

### Example

_PROTVAL[2]="R27,_OVR[0],_OVR[4],_OVR[8],_OVR[12],_OVR[16],_TIME"
_PROTVAL[3]="_AXIS,_OVR[1],_OVR[5],_OVR[9],_OVR[13],_OVR[17], INCH"
_PROTVAL[4]="_AXIS,_OVR[2],_OVR[6],_OVR[10],_OVR[14],_OVR[18], Metr"

R27 is just an example of an R parameter. The texts "INCH" and "Metro" at the end of the second and third line are examples for comment texts. This makes it easy, for example, to append dimensions after the measurement results.

### 7.1.6 Log format



### Programming

The following values can be specified for the

log format:

_PROTFORM[1]       INTEGER       Number of characters per line         _PROTFORM[2]       INTEGER       First page number         _PROTFORM[3]       INTEGER       Number of customized log header lines         _PROTFORM[4]       INTEGER       Number of value lines in the log         _PROTFORM[5]       INTEGER       Column width/variable column width         _PROTSYM[0]       CHAR       Separators between the values in the log         _PROTSYM[1]       CHAR       Special characters for identification when tolerance limits are exceeded         _DIGIT       INTEGER       Number of decimal places	_PROTFORM[0]	INTEGER	Number of lines per page including log header
_PROTFORM[2]       INTEGER       First page number         _PROTFORM[3]       INTEGER       Number of customized log header lines         _PROTFORM[4]       INTEGER       Number of value lines in the log         _PROTFORM[5]       INTEGER       Column width/variable column width         _PROTSYM[0]       CHAR       Separators between the values in the log         _PROTSYM[1]       CHAR       Special characters for identification when tolerance limits are exceeded         _DIGIT       INTEGER       Number of decimal places	_PROTFORM[1]	INTEGER	Number of characters per line
_PROTFORM[3]       INTEGER       Number of customized log header lines         _PROTFORM[4]       INTEGER       Number of value lines in the log         _PROTFORM[5]       INTEGER       Column width/variable column width         _PROTSYM[0]       CHAR       Separators between the values in the log         _PROTSYM[1]       CHAR       Special characters for identification when tolerance limits are exceeded         _DIGIT       INTEGER       Number of decimal places	_PROTFORM[2]	INTEGER	First page number
PROTFORM[4]       INTEGER       Number of value lines in the log         PROTFORM[5]       INTEGER       Column width/variable column width         PROTSYM[0]       CHAR       Separators between the values in the log         PROTSYM[1]       CHAR       Special characters for identification when tolerance limits are exceeded         DIGIT       INTEGER       Number of decimal places	_PROTFORM[3]	INTEGER	Number of customized log header lines
PROTFORM[5]       INTEGER       Column width/variable column width         PROTSYM[0]       CHAR       Separators between the values in the log         PROTSYM[1]       CHAR       Special characters for identification when tolerance limits are exceeded         DIGIT       INTEGER       Number of decimal places	_PROTFORM[4]	INTEGER	Number of value lines in the log
PROTSYM[0]       CHAR       Separators between the values in the log         PROTSYM[1]       CHAR       Special characters for identification when tolerance limits are exceeded         DIGIT       INTEGER       Number of decimal places	_PROTFORM[5]	INTEGER	Column width/variable column width
PROTSYM[1]       CHAR       Special characters for identification when tolerance limits are exceeded         _DIGIT       INTEGER       Number of decimal places	_PROTSYM[0]	CHAR	Separators between the values in the log
Imits are exceeded       DIGIT     INTEGER       Number of decimal places	_PROTSYM[1]	CHAR	Special characters for identification when tolerance
DIGIT INTEGER Number of decimal places			limits are exceeded
	_DIGIT	INTEGER	Number of decimal places



### Explanation

The value set in parameter \_PROTFORM[0] determines when a log header with title lines is output again. If this parameter is set to zero, the log only contains a header at the beginning.

The value of parameter \_PROTFORM[5] determines the column width of the log. If the parameter=0, the column width of each column is derived from the string lengths (number of characters between the commas) of the 1st header line (\_PROTVAL[0]). This makes it possible to individually define the width of each column. If the value >0, each column is formatted to this value if the string length allows it. Logging of variables always has priority, i.e. when specified format limits are exceeded they are modified and an alarm without terminating execution is generated.

The number of decimal places can be set via the variable \_DIGIT in GUD6 data block (display precision).





### Function

The log header can be customized by the operator or a log header prepared by the standard measuring cycles can be used.

The selection is made via measuring cycle data bit \_\_CBIT[11].

\_CBIT[11]=0: Default log header

\_CBIT[11]=1: User-defined log header

The contents of the header are stored in an array of string variables \_HEADLINE [10], which are automatically output when logging (CYCLE100) is enabled. The maximum number of header lines can be changed during measuring cycle start-up (\_PROTFORM[3]).

Each field element contains a line for the log header.

## =?

### Explanation

### User-defined log header

The content of the string array \_HEADLINE [ ] is entered in line 1 ff. The number of header lines can be defined by the user (according to the length of the HEADLINE array).

### Default log header

The default log header contains fixed and variable components.

All variable components are in bold formatting, that

is:

Line 1 Number of pages,

### Line 3 Program name,

Line 5, 6, 7 (\_HEADLINE [0-2]) ff. and

Line 9 (\_PROTVAL[0])

l ine 10	<i>.</i>		г 1	1	١
		PROTVAL	L⊥	Ы.	)

Line 1 is fixed and predefined.

10.04

10.04

# Miscellaneous Functions 7.1 Logging the measurement results

Line 2 Line 3 Program: MEASPROGRAM_1 Line 4 Line 5 Part number: 123456789 Line 6 Job number: 6878 Line 7 Processed by: Müller Phone: 1234 Line 8	Line 1	Date:	98/09/15		Time:	10:05:30	Page: <b>1</b>
Line 3 Program: MEASPROGRAM_1 Line 4 Line 5 Part number: 123456789 Line 6 Job number: 6878 Line 7 Processed by: Müller Phone: 1234 Line 8	Line 2						
Line 4 Line 5 Part number: 123456789 Line 6 Job number: 6878 Line 7 Processed by: Müller Phone: 1234 Line 8	Line 3	Program:	MEASPROGR	AM_1			
Line 5 Part number: 123456789 Line 6 Job number: 6878 Line 7 Processed by: Müller Phone: 1234 Line 8	Line 4						
Line 6 Job number: 6878 Line 7 Processed by: Müller Phone: 1234 Line 8	Line 5	Part num	ber: 1234567	789			
Line 7 Processed by: Müller Phone: 1234 Line 8	Line 6	Job num	oer: 6878				
Line 8	Line 7	Processe	d by: Müller	Phone: 123	34		
Line 9       Meas.       Axis       Set       Actual , Difference       Time         Line 10       point       value       value       value         Line 11	Line 8						
Line 10 point value value Line 11	Line 9	Meas.	, Axis	, Set , Act	ual , Differe	ence , Time	
Line 11	Line 10	point		value val	ue		
<pre>When filling in the standard log header shown above the following program lines must be inserted in the main program before the measuring cycle is called: DEF INT PARTNUM, JOBNUMCBIT[11]=0 ;Log with default header PARTNUM=123456789 JOBNUM=6878 _PROTNAME[0]="MEASPROGRAM_1" PROTNAME[1] ="MY_LOG1"HEADLINE[0]="Part number: "&lt;<partnum <="" <<partnum="" _headline[1]="Supervisor: Müller Phone: 1234" protval[0]="Meas. , Axis , Set , Act. value, Difference , Time" td=""><td>Line 11</td><td></td><td></td><td></td><td></td><td></td><td></td></partnum></pre>	Line 11						
the following program lines must be inserted in the main program before the measuring cycle is called: DEF INT PARTNUM, JOBNUM CBIT[11]=0 ;Log with default header PARTNUM=123456789 JOBNUM=6878 _PROTNAME[0]="MEASPROGRAM_1" PROTNAME[1] ="MY_LOG1" HEADLINE[0]="Part number: "< <partnum _HEADLINE[1]="Job number: "&lt;<partnum _HEADLINE[1]="Job number: "&lt;<jobnum _HEADLINE[2]="Supervisor: Müller Phone: 1234" _PROTVAL[0]="Meas. , Axis , Set , Act. value, Difference , Time" PROTVAL[1]="point value</jobnum </partnum </partnum 	When fil	ling in the st	andard log he	eader shown abo	ve		
<pre>main program before the measuring cycle is called: DEF INT PARTNUM, JOBNUM CBIT[11]=0 ;Log with default header PARTNUM=123456789 JOBNUM=6878 _PROTNAME[0]="MEASPROGRAM_1" PROTNAME[1] ="MY_LOG1" HEADLINE[0]="Part number: "&lt;<partnum HEADLINE[1]="Job number: "&lt;<partnum HEADLINE[1]="Job number: "&lt;<jobnum HEADLINE[2]="Supervisor: Müller Phone: 1234" PROTVAL[0]="Meas. , Axis , Set , Act. value, Difference , Time" PROTVAL[1]="point value</jobnum </partnum </partnum </pre>	the follow	wing program	m lines must l	be inserted in the	;		
DEF INT PARTNUM, JOBNUM         _CBIT[11]=0       ;Log with default header         PARTNUM=123456789       JOBNUM=6878       _PROTNAME[0]="MEASPROGRAM_1"         _PROTNAME[1] ="MY_LOG1"	main pro	ogram before	e the measuri	ng cycle is called	1:		
_CBIT[11]=0       ;Log with default header         PARTNUM=123456789       JOBNUM=6878       PROTNAME[0]="MEASPROGRAM_1"         _PROTNAME[1] ="MY_LOG1"	DEF INT	PARTNUM, .	JOBNUM				
PARTNUM=123456789       JOBNUM=6878 _PROTNAME[0]="MEASPROGRAM_1"         _PROTNAME[1] ="MY_LOG1"         _HEADLINE[0]="Part number: "< <partnum< td="">         _HEADLINE[1]="Job number: "&lt;<jobnum< td="">         _HEADLINE[2]="Supervisor: Müller       Phone: 1234"         _PROTVAL[0]="Meas.       , Axis       , Set       , Act. value, Difference       , Time"         _PROTVAL[1]="point       walue</jobnum<></partnum<>	_CBIT[	11]=0			;Log w	ith default header	
PROTNAME[1] ="MY_LOG1" HEADLINE[0]="Part number: "< <partnum HEADLINE[1]="Job number:"&lt;<jobnum HEADLINE[2]="Supervisor: Müller Phone: 1234" PROTVAL[0]="Meas. , Axis , Set , Act. value, Difference , Time" PROTVAL[1]="point value</jobnum </partnum 	PARTNU	M=12345678	JOBNUM=	6878 PROTNAM	IE [0]="MEA	ASPROGRAM_1"	
<pre></pre>	PROTN	AME[1] ="	MY LOG1"				
	- HEADL	INE[0]="P	 art number	: "< <partn< td=""><td>UM</td><td></td><td></td></partn<>	UM		
	- HEADL	INE[1]="J	ob number:	"< <jobnum< td=""><td></td><td></td><td></td></jobnum<>			
PROTVAL[0]="Meas. , Axis , Set , Act. value, Difference , Time"	- HEADL	INE[2]="S	upervisor:	Müller		Phone: 1234"	
	PROTV	AT.[0]="Me	as. Ax	is . Set	. Act	value. Differen	ce . Time"
		$\frac{1}{\Delta T [1] = "no$	int	, see	, 1100.		,

### 7.1.8 Example: Creating a measurement result log

Log vie	w						
%N_LOG	G_1_MPF						
Line 1	Date:	96/11/15			Time:	10:05:30	Pa
Line 2							
Line 3	Program:	MEASPROGR	AM_1				
Line 4							
Line 5	Part num	ber: 123456	789				
Line 6	Job num	ber: 6878					
Line 7	Processe	ed by: Müller	Pho	ne: 1234			
Line 8							-
Line 9	Meas.	, Axis	, Set	, Actual	, Differen	ce,Time	
Line 10	point		value	value			
Line 11	 CYCL E97	 '8 MVAR	100				
Line 12	1	.Z	, 80.000	. 79.987	0.013	. 09:35:12	
Line 14	•	, _	,	,	,	,	
Line 16	CYCLE97	7. MVAR	. 102				
Line 17	2	, <u> </u>	64 000	64 009	0 009	09.36.45	
	2		, 04.000	, 07.003	, 0.005	,	



÷

### Programming

The log with the default log header shown above is created u	ising
the following program. The example shows the user how to h	landle
ine log.	
%_N_MEASPROGRAM_1_MPF	
;\$PATH=/_N_MPF_DIR	
Measure shaft with measuring log	
DEF INT PARTNUM, JOBNUM, MP_COUNTER	
, Set parameters for log	
_CBIT[11]=0	;Log with default header
Log header	
PARTNUM=123456789 JOBNUM=6878	:Name of calling program
PROTNAME [0] = "MEASPROGRAM 1"	,
PROTNAME [1] = "LOG 1"	;Name of log file
HEADLINE[0]="Part number: "< <partnum< td=""><td></td></partnum<>	
HEADLINE[1]="Job number:"< <jobnum< td=""><td></td></jobnum<>	
HEADLINE[2]="Supervisor: Müller Phone: 1234"	
	Formats: Default values from
PROTSYM[0]="," PROTSYM[1]="*"	;Define separators and special
PROTFORM[0]=60	;60 lines per page
 PROTFORM[1]=80	:80 characters per line
PROTFORM[2]=1	:Start of page 1
PROTFORM[3]=3	Three customized log header
PROTFORM [ 4 ] = 1	:One value line
PROTFORM[5]=12	:12 characters per line
	,
: Header lines	
PROTVAL[0]="Meas. , Axis , Set , Act.	value, Difference , Time"
PROTVAL[1]="point , , value"	
Other value assignments	
MP COUNTER=1 TXT[0]=< <mp counter<="" td=""><td>;Assign counter for</td></mp>	;Assign counter for
	measurement log
Perform measurements with log	
N100 G0 G17 G90 T3 D1 Z100 F1000	;Approach start position for
N110 X70 Y90	
_MVAR=100 _SETVAL=80 _MA=3 _TSA=2 _FA=2	;Set measuring cycle
	;Measuring variant: Measure
Contents of the value lines	
_PROTVAL[2]="_TXT[0],_AXIS,_OVR[0],_OVR[4],_O	VR[16],_TIME"
N150 CYCLE100	;Activate log
N160 CYCLE978	;Measure surface
N170 Z200	;Retraction in Z
N180 X64 Y38	;Position above shaft center
N185 Z130	;Lower in Z
	0.1
M/AR=102 SET/AI=70 EA=2 TSA=2 ID=-20	Set measuring cycle
_MVAR=102 _SETVAL=70 _FA=2 _TSA=2 _ID=-20	;Set measuring cycle
	;Set measuring cycle ;Measuring variant: Measure sha
	;Set measuring cycle ;Measuring variant: Measure sha ;two value lines

MP_COUNTER=MP_COUNTER+1_TXT[0]=< <mp_counter< th=""><th>;Increase user-def. counter for measurements</th></mp_counter<>	;Increase user-def. counter for measurements
N190 CYCLE977	;Measure shaft
N210 CYCLE101	;Deactivate log
N220 Z200	;Retraction in Z
N290 M2	







### Function

In SW 4.3 and higher, cycle support for measuring cycles in the ASCII editor is provided as for the standard cycles.

With this support function, the parameters described as mandatory parameters are input for each measuring cycle. For the additional parameters the last values input are retained. Furthermore, it is possible to change the additional parameters.

The measuring cycles are selected in the editor by using the vertical softkeys. The softkey menu is divided up according to measuring tasks, e.g. "Calibration" and then "Calibration in hole" or "Tool probe". This results in a one-to-one assignment between softkeys and measuring cycles.

As from SW 5 of the MMC100/100.2, MMC102/103 and HMI Advanced/Embedded, measuring cycle support will be accessible via softkeys

>		Support		New		Measuring
	$\rightarrow$		$\rightarrow$	cycle	$\rightarrow$	cycles

from the extended menu of the editor.

In the edited program there are calls with parameter list, e.g. CYCLE\_PARA(...) for additional parameter assignment CYCLE\_976(...) for calibrating in a hole, CYCLE\_CAL\_TOOLSETTER(...) for calibrating the tool probe.





### Function

Measuring cycle support requires the following files:

- COV.COM
   Configuring the softkeys for cycle selection
- SC.COM
   Configuring the input screens for the individual
   parameters
- Auxiliary cycle\*.spf Additional cycles with parameter list that transfer the input parameters to the data blocks (GUD) of the measuring cycles and call the measuring cycles.

These files are combined in the following two archives on the measuring cycle diskette:

- MCSUPP\_1.COM
- MCSUPP\_2.COM

### 7.2.2 Loading the measuring cycle support



### Function

The files mcsupp\_1.com and mcsupp\_2.com are loaded from diskette or via RS-232-C (V24) with "Data in" into the "Services" menu.

With the MMC 102/103 the auxiliary cycle programs (see list Subsection 7.2.3) must be transferred to the NCU with "Load".

The Power ON is executed.



### 7.2.3 Assigning calls and measuring cycles



### Function

The following table provides an overview of:

- Measuring task,
- Measuring cycle,
- Call.

Measuring task, function	Measuring cycle	Call in the program
Auxiliary parameters	-	CYCLE_PARA()
Calibrating tool probes	CYCLE971,	CYCLE_CAL_TOOLSETTER()
	CYCLE972,	
	CYCLE982	
Calibrate workpiece probe on surface	CYCLE973,	CYCLE_CAL_PROBE()
	CYCLE976	
Calibrate workpiece probe in reference groove	CYCLE973	CYCLE_973()
Calibrate workpiece probe in hole	CYCLE976	CYCLE_976()
Measure milling tool on milling machines	CYCLE971	CYCLE_971()
Measure turning tool	CYCLE972	CYCLE_972()
Measure hole/shaft parallel to axis/at an angle	CYCLE977,	CYCLE_977_979A()
	CYCLE979	
Measure groove/web parallel to axis/at an	CYCLE977,	CYCLE_977_979B()
angle	CYCLE979	
Measure rectangle inside/outside parallel to	CYCLE977	CYCLE_977_979C()
axis		
Single-point measurement milling machine	CYCLE978	CYCLE_978()
Angle measurement	CYCLE998	CYCLE_998()
Measure corner, specifying angles	CYCLE961	CYCLE_961_W
Measure corner, specifying points	CYCLE961	CYCLE_961_P
Single-point measurement turning	CYCLE974	CYCLE_974()
Two-point measurement	CYCLE994	CYCLE_994()

### 7.2.4 Description of the parameterizing cycles



### Function

The individual parameterization cycles of the measuring cycles together with their input parameters are described below.

The parameter names in the table directly refer to the defining parameters of the measuring cycle in question in the GUD variables. If no parameter is given, it is a selection field in the input screen form for particular functions.



### Setting additional parameters – CYCLE\_PARA

With CYCLE\_PARA, you can program all generally valid parameters of the measuring cycles. These parameters do not depend on the individual measuring variant. If an entered parameter has the value 0, the target parameter is not overwritten, i.e. its previous value is retained.

	_	
Γ.	=	=
	_	=
	_	- 1

_FA	REAL	>0	Measurement path in mm
_VMS	REAL	≥0	Variable measuring speed
_NMSP	INTEGER	>0	Number of measurements at the same location
_RF	REAL	>0	CYCLE979 only:
			Feedrate in circular-path programming
_PRNUM	INTEGER	>0	Probe number
_CORA	REAL		Only if mono probe is used:
			Offset angle position
_TZL	REAL	≥0	Tolerance range for zero offset
_TMV	REAL	≥0	Select range for offset with mean value calculation, greater than _TZL
_TUL	REAL		Upper tolerance range workpiece, oversize acc. to drawing
_TLL	REAL		Lower tolerance range workpiece, undersize acc. to drawing
_TSA	REAL	>0	Safe area for measuring result
_EVNUM	INTEGER	≥0	Number of empirical value memory that is calculated
_K	INTEGER	≥1	Weighting factor for mean value calculation
_TDIF	REAL	>0	Tolerance range for dimensional difference check



10.04



### Calibrating a tool probe – CYCLE\_CAL\_TOOLSETTER With CYCLE\_CAL\_TOOLSETTER With CYCLE\_CAL\_TOOLSETTER softkeys Measurem. or Calibrat. (CYCLE971)

softkeys turning  $\rightarrow$  TL probe (CYCLE972) measuring cycles CYCLE971 and CYCLE972 can be

parameterized to calibrate a tool measuring probe.

### Parameters

	INTEGER		Selection: Cycle number
			971 for CYCLE971 (milling machine), 972 for CYCLE972 (turning machine)
_MA	INTEGER >0		Number of measuring axis and for CYCLE972 also the offset axis
_PRNUM	INTEGER >0		Probe number
	INTEGER		For CYCLE971 only
			Selection: Measuring variant
			0absolute calibration/1incremental calibration
_FA	REAL	>0	Measurement path



### Calibration on surface – CYCLE\_CAL\_PROBE



	INTEGER	Selection: Cycle number
		976 for CYCLE976 (milling machine),
		973 for CYCLE973 (turning machine)
_SETVAL	REAL	Calibration setpoint with respect to workpiece zero
_MA	INTEGER 13	Number of the measuring axis
_MD	INTEGER 0, 1	Measuring direction
_PRNUM	INTEGER >0	Probe number
_MVAR	INTEGER ≥0	Selection: Measuring variant (for CYCLE976 only)
		0: Calibration on any surface
		10000: Calibrating in the 3rd axis with calculation of probe length



### Calibration in groove – CYCLE\_973

With CYCLE\_973

softkeys  $\overset{-}{\underset{\text{turning}}{\text{Measurem.}}} \rightarrow \overset{-}{\underset{\text{calibrat.}}{\text{Probe}}}$ 

CYCLE973 can be parameterized to calibrate a reference groove.



### Parameters

_SETVAL	REAL	Setpoint
	INTEGER	Selection: Positional deviation
		0without/1with specification of positional deviation
	INTEGER	Selection: Number of axes
		Number of axes to be calibrated, 1, 2
	INTEGER	Selection: Ball calculation
		0without/1with calculation of probe ball diameter
_MA	INTEGER >0	Number of measuring axis 1, 2
_MD	INTEGER ≥0	Determines measuring direction
		0in positive direction / 1in negative direction
_CALNUM	INTEGER ≥0	Selection of calibration groove with number
_PRNUM	INTEGER >0	Probe number



### Calibration in hole – CYCLE\_976

With CYCLE\_976

softkeys Measurem.  $\rightarrow$  Probe calibrat.

CYCLE976 can be parameterized to calibrate a hole.

### Parameters

#### \_SETVAL REAL Setpoint INTEGER Selection: Angular position 0...Paraxial calibration/1...calibration at an angle **INTEGER** Selection: Positional deviation 0...without/1...with specification of positional deviation **INTEGER** Selection: Number of axes Number of axes to be calibrated, 1, 2 or 4 INTEGER Selection: Ball calculation 0...without/1...with calculation of probe ball diameter MA **INTEGER >0** Number of measuring axis 1, 2 \_MD INTEGER ≥0 Determines measuring direction 0...in positive direction / 1...in negative direction \_STA1 REAL Angle (for calibration at an angle only) PRNUM **INTEGER >0** Probe number Selection: Hole type 0...hole center known / 1...unknown





### Measuring milling tools – CYCLE\_971

### With CYCLE\_971

softkeys  $\overset{-}{\underset{\text{milling}}{\text{Measurem.}}} \rightarrow \overset{-}{\underset{\text{measurem.}}{\text{Tool}}}$ 

CYCLE971 can be parameterized for tool measurement.

### Parameters

_MVAR	INTEGEF	२ >०	Measuring variant
_MA	INTEGEF	२ >०	Number of the measuring axis
_ID	REAL	>0	Offset
_PRNUM	INTEGEF	२ >०	Probe number
_MFS[0]	REAL	>0	Feed 1st probing (only with _CBIT[12]=1)
_MFS[1]	REAL	>0	Speed 1st probing (only with _CBIT[12]=1)
_MFS[2]	REAL	≥0	Feed 1st probing (only with _CBIT[12]=1)
_MFS[3]	REAL	≥0	Speed 1st probing (only with _CBIT[12]=1)
_MFS[4]	REAL	≥0	Feed 1st probing (only with _CBIT[12]=1)
_MFS[5]	REAL	≥0	Speed 1st probing (only with _CBIT[12]=1)



### Measuring turning tools – CYCLE\_972

CYCLE\_972 can be used to parameterize CYCLE976 to gauge tools.



_MA	INTEGER >0	Number of the measuring axis
_PRNUM	INTEGER >0	Probe number



#### With CYCLE\_977\_979A

	Measurem.		Workpiece		Hole	
softkeys	milling	$\rightarrow$	measure	$\rightarrow$	Shaft	

measuring variants xxx1 and xxx2 of measuring cycles CYCLE977 and CYCLE979 can be parameterized.

Parameters

	INTEGER	Selection: Angular position
		977paraxial measurement / 979measurement at an angle
_MVAR	INTEGER >0	Measuring variant
_SETVAL	REAL	Setpoint
_ID	REAL	Infeed path
_SZA	REAL	Safety zone (for paraxial measurement only)
_TNUM	INTEGER ≥0	Tool number for automatic offset
_TNAME	STRING	Tool name with active tool management
_KNUM	INTEGER ≥0	Offset number
		D number for measurement/ZO number for calculating zero offset
_CPA	REAL	Center point hole, shaft 1st axis
_CPO	REAL	Center point hole, shaft 2nd axis
_STA1	REAL	Start angle
_INCA	REAL	Indexing angle
_PRNUM	INTEGER >0	Measuring probe number
		CYCLE979 only:
		The number of measuring points is assigned from the thousands digit;
		03 measuring points, 14 measuring points



### Measure groove, web – CYCLE\_977\_979B

With CYCLE\_977\_979B

 $\begin{array}{c} \text{Measurem.} \\ \text{softkeys} \end{array} \xrightarrow[]{\text{Measurem.}} \rightarrow \end{array} \xrightarrow[]{\text{Workpiece}} \\ \xrightarrow[]{\text{web}} \xrightarrow[]{\text{web}} \\ \xrightarrow[]{\text{web}} \\ \xrightarrow[]{\text{web}} \\ \xrightarrow[]{\text{Workpiece}} \\ \xrightarrow[]{\text{web}} \\ \xrightarrow[]{\text{web}} \\ \xrightarrow[]{\text{Workpiece}} \\ \xrightarrow[]{\text{$ 

measuring variants xxx3 and xxx4 of measuring cycles CYCLE977 and CYCLE979 can be parameterized.



### Parameters

	INTEGER	Selection: Angular position	
		977paraxial measurement / 979measurement at an angle	
_MVAR	INTEGER >0	Measuring variant	
_SETVAL	REAL	Setpoint	
_ID	REAL	Infeed path	
_MA	INTEGER >0	Number of measuring axis 1, 2	
_TNUM	INTEGER ≥0	Tool number for automatic offset	

 $\odot$  Siemens AG, 2004. All rights reserved SINUMERIK 840D/840Di/810D User's Guide Measuring Cycles (BNM) – 10.04 Edition



#### Miscellaneous Functions 7.2 Measuring cycle support in the program editor

_TNAME	STRING	Tool name with active tool management		
_KNUM	INTEGER ≥0	Offset number		
		D number for measurement/ZO number for calculating zero offset		
_CPA	REAL	Center groove, web 1st axis		
_CPO	REAL	Center groove, web 2nd axis		
_STA1	REAL	Start angle		
_SZA	REAL	Safe area (for paraxial measurement only)		
_PRNUM	INTEGER >0	Probe number		

### Measure rectangle – CYCLE\_977\_979C

With CYCLE\_977\_979C



measuring variants xxx5 and xxx6 of measuring cycle CYCLE977 can be parameterized.

### Parameters

Ê

_MVAR	INTEGER >0	Measuring variant
_SETV[0]	REAL	Setpoint length
_SETV[1]	REAL	Setpoint width
_ID	REAL	Infeed path
_SZA	REAL	Protection zone length
_SZO	REAL	Protection zone width
_TNUM	INTEGER ≥0	Tool number for automatic offset
_TNAME	STRING	Tool name with active tool management
_KNUM	INTEGER ≥0	Offset number
		D number for measurement/ZO number for calculating zero offset



10.04



### Single-point measurement – CYCLE\_978

With CYCLE_978					
softkeys	Measurem. milling	$\rightarrow$	Workpiece measure	$\rightarrow$	Plane

CYCLE978 can be parameterized.

## 

### Parameters

_MVAR	INTEGER ≥0	Measuring variant
_SETVAL	REAL	Setpoint
_MA	INTEGER >0	Measuring axis 1, 2, 3
_TNUM	INTEGER ≥0	Tool number for automatic offset
_TNAME	STRING	Tool name with active tool management
_KNUM	INTEGER ≥0	Offset number
		D number for measurement/ZO number for calculating zero offset



### Angle measurement – CYCLE\_998

With CYCLE\_998

	—					
	Measurem.		Workpiece		Angle	
softkevs	milling	$\rightarrow$	measure	$\rightarrow$		
· · · · <b>,</b> ·						

CYCLE998 can be parameterized.

### 

MVAR	INTEGER >0	Measuring variant
_SETVAL	REAL	Setpoint
_ID	REAL	To be controlled
_RA	INTEGER ≥0	Number of rotary axis
_MA	INTEGER >0	Number of measuring axis and offset axis 102302
_KNUM	INTEGER ≥0	ZO number
_STA1	REAL	Angle
_PRNUM	INTEGER >0	Probe number





With CYCLE_961_W						
softkeys	Measurem. milling	$\rightarrow$	Workpiece measure	$\rightarrow$	Corner	
measuring variants 105 to 108 for CYCLE961 can be						

parameterized.

### 

	INTEGER		Selection: Outside or inside corner
	INTEGER		UInside corner / 1outside corner
SETV[0]			Distance between starting point and measuring point 2 without sign
SETV[1]		>0	Distance between starting point and measuring point 2, without sign
		>0	Distance between starting point and measuring point 4, without sign
_10	REAL	>0	without sign
_STA1	REAL		Enter approx. angle between 1st axis (abscissa) and 1st edge, in clockwise direction with negative sign
_INCA	REAL	<>0	Angle between 1st and 2nd edge of workpiece,
			in clockwise direction with negative sign
_KNUM	INTEGEF	R ≥0	ZO number
_SETV[4]	REAL		For 3 measuring points only:
			Selection: Override
			Selection: Override 1measured corner entered as zero point
			Selection: Override 1measured corner entered as zero point 2measured corner is entered in 1st axis offset by the value in
			Selection: Override 1measured corner entered as zero point 2measured corner is entered in 1st axis offset by the value in _SETV[3] and as a zero point
			<ul> <li>Selection: Override</li> <li>1measured corner entered as zero point</li> <li>2measured corner is entered in 1st axis offset by the value in _SETV[3] and as a zero point</li> <li>3measured corner is offset in both axes and entered as zero point</li> </ul>
			<ul> <li>Selection: Override</li> <li>1measured corner entered as zero point</li> <li>2measured corner is entered in 1st axis offset by the value in _SETV[3] and as a zero point</li> <li>3measured corner is offset in both axes and entered as zero point</li> <li>4measured corner is offset in 2nd axis by the value</li> </ul>
			<ul> <li>Selection: Override</li> <li>1measured corner entered as zero point</li> <li>2measured corner is entered in 1st axis offset by the value in _SETV[3] and as a zero point</li> <li>3measured corner is offset in both axes and entered as zero point</li> <li>4measured corner is offset in 2nd axis by the value from _SETV[3] and entered as zero point.</li> </ul>
_SETV[2]	REAL		<ul> <li>Selection: Override</li> <li>1measured corner entered as zero point</li> <li>2measured corner is entered in 1st axis offset by the value in _SETV[3] and as a zero point</li> <li>3measured corner is offset in both axes and entered as zero point</li> <li>4measured corner is offset in 2nd axis by the value from _SETV[3] and entered as zero point.</li> </ul> For 3 measuring points only:
_SETV[2]	REAL		<ul> <li>Selection: Override</li> <li>1measured corner entered as zero point</li> <li>2measured corner is entered in 1st axis offset by the value in _SETV[3] and as a zero point</li> <li>3measured corner is offset in both axes and entered as zero point</li> <li>4measured corner is offset in 2nd axis by the value from _SETV[3] and entered as zero point.</li> <li>For 3 measuring points only:</li> <li>Offset of coordinate origin in 1st axis (abscissa)</li> </ul>
_SETV[2]	REAL		<ul> <li>Selection: Override</li> <li>1measured corner entered as zero point</li> <li>2measured corner is entered in 1st axis offset by the value in _SETV[3] and as a zero point</li> <li>3measured corner is offset in both axes and entered as zero point</li> <li>4measured corner is offset in 2nd axis by the value from _SETV[3] and entered as zero point.</li> <li>For 3 measuring points only:</li> <li>Offset of coordinate origin in 1st axis (abscissa)</li> <li>For 3 measuring points only:</li> </ul>
_SETV[2] _SETV[3]	REAL		<ul> <li>Selection: Override</li> <li>1measured corner entered as zero point</li> <li>2measured corner is entered in 1st axis offset by the value in _SETV[3] and as a zero point</li> <li>3measured corner is offset in both axes and entered as zero point</li> <li>4measured corner is offset in 2nd axis by the value from _SETV[3] and entered as zero point.</li> <li>For 3 measuring points only:</li> <li>Offset of coordinate origin in 1st axis (abscissa)</li> <li>For 3 measuring points only:</li> <li>Offset of coordinate origin in 2nd axis (ordinate)</li> </ul>





With CYCLE\_961\_P softkeys  $Measurem. \rightarrow Morkpiece \rightarrow Corner$ milling  $\rightarrow Morkpiece \rightarrow Corner$ 

measuring variants 117 to 118 for CYCLE961 can be parameterized.

### Parameters

	INTEGER	Selection: Outside or inside corner
		0inside corner / 1outside corner
_ID	REAL >0	Infeed path of measuring probe to measuring height, without sign
_SETV[0]	REAL	Starting position for measuring the 1st point in the 1st axis (abscissa)
_SETV[1]	REAL	Starting position for measuring the 1st point in the 2nd axis (ordinate)
_SETV[2]	REAL	Starting position for measuring the 2nd point in the 1st axis (abscissa)
_SETV[3]	REAL	Starting position for measuring the 2nd point in the 2nd axis (ordinate)
_SETV[4]	REAL	Starting position for measuring the 3rd point in the 1st axis (abscissa)
_SETV[5]	REAL	Starting position for measuring the 3rd point in the 2nd axis (ordinate)
_SETV[6]	REAL	Starting position for measuring the 4th point in the 1st axis (abscissa)
_SETV[7]	REAL	Starting position for measuring the 4th point in the 2nd axis (ordinate)
_KNUM	INTEGER ≥0	ZO number
_PRNUM	INTEGER >0	Probe number



### Single-point measurement – CYCLE\_974

With CYCLE\_974



CYCLE974 can be parameterized.

_MVAR	INTEGER ≥0	Measuring variant
_SETVAL	REAL	Setpoint
_MA	INTEGER >0	Number of measuring axis 1, 2, 3
_TNUM	INTEGER ≥0	Tool number for automatic offset
_TNAME	STRING	Tool name with active tool management
_KNUM	INTEGER >0	Offset number
		D number for measurement/ZO number for calculating zero offset
_PRNUM	INTEGER >0	Probe number
_STA1	REAL	Start angle





### Two-point measurement – CYCLE\_994

### With CYCLE\_994

	_				
	Measurem.		Workpiece		Two-point
softkeys	turning	$\rightarrow$	measure	$\rightarrow$	measurem.
,					

CYCLE994 can be parameterized.

_MVAR	INTEGER 1, 2	Measuring variant
_SETVAL	REAL	Setpoint
_MA	INTEGER >0	Number of measuring axis 1, 2, 3
_TNUM	INTEGER ≥0	Tool number for automatic offset
_TNAME	STRING	Tool name with active tool management
_KNUM	INTEGER ≥0	Offset number
		D number for measurement/ZO number for calculating zero offset
_SZA	REAL	Protection zone on workpiece, 1st axis (abscissa)
_SZO	REAL	Protection zone on workpiece, 2nd axis (ordinate)
_PRNUM	INTEGER >0	Probe number

# 7.3 Measuring cycle support in the program editor (from measuring cycles SW 6.2)



From measuring cycles SW 6.2, the program editor provides extended measuring cycle support for inserting measuring cycle calls into the program.

### **Requirement:**

HMI Advanced/Embedded as from SW 6.2 required.



### Function

This measuring cycle support provides the following functionality:

- Measuring cycle selection via softkeys
- Input screen forms for parameter assignment with help displays
- Retranslatable code is generated from the individual screen forms.



### 7.3.1 Menus, explanation of the cycles



### Explanation

The input screens for the measuring cycles are selected depending on the technology being used via horizontal softkeys.





Input screen forms for measuring cycles for turning technology.

Measurem. milling

Input screens for measuring cycles for milling technology.







.

<<

Back to selection menu turning.

measurement.

















### Measuring a hole parallel to the axis with

### protection zone

### (generated with measuring cycle support)

N100 G17 G0 G90 Z20 F2000 S500 M3	;Main block
N110 T7 M6	;Insert probe
N120 G17 G0 G90 X50 Y50	;Position probe in X/Y plane
	;on hole center point
N130 Z20 D1	;Position Z axis in hole
; from the measuring cycle support using the	
; NC code _MZ_MASK[0]=1 generated using the	
; input screen	
N130 _MVAR=1001 _SETVAL=100.000 _PRNUM=101	;Parameter passing to
_KNUM=2002 _FA=2 _TSA=0.23	;measuring cycle
_VMS=0 _NMSP=1 _ID=-20.000 _SZA=50.000	
_CORA=0.03 _TZL=0.01 _TDIF=0.2 _TUL=0.065	
_TLL=-0.065 _CHBIT[4]=0 _K=1 _EVNUM=2 -TNUM=1	
CYCLE977	;Call measuring cycle
;* end of NC code generated by measuring cycle ;	
; support	

### N200 M30

...

#### Input screen for

measuring a hole, parallel with the axis, with protection zone (CYCLE977).

Program	CHAN1	Auto	MPFO				
// Channel reset		Program a	aborted				
				ROV SI	BL1		1
Meas. hole	CYCLE977			Measu	ure paraxial or u	<mark>Inder angle</mark>	Alter- native
						<u> </u>	
		Angle	pos.	P	araxial		Hole
Y≱.	_SETVAL	Prot. :	zone		N	0	
		Offset			Lengl	h	Shaft
		Offset			norm	al	
P3 P1	P3 P1	Offsel			N	0	
	\P2 / /	Tool r	10.	_TNUM	1.00	0	
		Tool e	edge no.		2.00	0	
		Setpt.	value	_SETVA	L 100.00	0	
		Meas.	path fac	_FA	2.00	0 -	Abort
¥	×	Area		_TSA	0.23	0	ADOIL
		Probe	number	_PRNUM	4 1.00	0	
		Meas.	feed	_VMS	0.00	0	ОК
<u>^</u>							U.N.

;End of program

### 7.3.2 Pre-setting of the measuring cycle support



### Explanation

A field \_MZ\_MASK is declared in the data block (GUD6) in which the screen forms can be adapted:

- technological measuring conditions
- to measuring variants

The settings in the data block for the \_MZ\_MASK field can be changed in a screen form in "Setup" operating area.

nbetrieb nahme	CHAN1	AUTO	MPF0	
🥢 Kanal R	ESET		Programm abgebrochen	
			ROV	
				Alter-
Messzykle	n		Auswahlmöglichkeiten für NV- u. WZ-Korrektur	nativ
_MZ_MAS	K(O) Zyklusaufr	uf	direkt	
_MZ_MAS	K[1] NV- und W	Z-Korrektur	Standard	
_MZ_MAS	K[2] Messverfa	hren	Standard	
_MZ_MAS	K[3] Erfahrung:	swerte	ohne	
_MZ_MAS	K[4] Mittelwert	oildung	ohne	
_MZ_MAS	K[5] Messtaste	rtyp	Multitaster	
_MZ_MAS	K[7] Vorgabe F	S beim WZ Mess	en ohne	
Lingesch	rankte Korrekturm	oglichkeiten für _K	NUM in den Masken	
				zurück

Variable	Value	Default	Meaning		
_MZ_MASK[0]	0	-	An indirect measuring cycle call is inserted in the NC code.		
			Example: CYCLE977/drill-hole		
			CYCLE_PARA()		
			CYCLE_977_979A(977,)		
	1	1	A direct measuring cycle call is inserted in the NC code.		
			Example: CYCLE977/drill-hole		
			_MVAR=1 _KNUM=1 _PRNUM=1		
	_		CYCLE977		
_MZ_MASK[1]	0	0	The workpiece screen forms contain the following selection options		
			for zero offset and tool offset:		
			Zero offset - default:		
			Settable zero offsets		
			Last channel-specific basic frame		
			Tool offset – default:		
			Milling: Tool radius is corrected		
			Turning: Length offset in the measuring axis		
	1	-	The workpiece screen forms contain the following selection options		
			for zero offset and tool offset:		
			Zero offset – extended:		
			Settable zero offsets		
			Last channel-specific basic frame		
			Offset in system frame		
			Offset in active frame		
			Offset in any basic frame (global or channel-specific)		
			Tool offset – extended:		
			• Offset radius, length, or length selection (L1, L2, or L3)		
			<ul> <li>Calculation of measurement results normal or inverted</li> </ul>		
			Offset in setun/additive offset		



Variable	Value	Default	Meaning
_MZ_MASK[2]	0	0	Forms without input fields for parameters:
			_VMS: Measuring velocity
			• _NMSP: Number of measurements at the same location.
			The following default values are entered in the NC code for the
			parameters:
			_VMS=0 corresponds to 150 mm/min or 5.9055 inch/min
			<ul> <li>_NMSP=1 number of measurements = 1</li> </ul>
	1	-	Forms with input fields for parameters:
			_VMS: Measuring velocity
			_NMSP: Number of measurements at the same location
_MZ_MASK[3]	0	0	Screen forms for workpiece measurement with automatic
			tool offset and tool measurement do not contain an input
			field for the following parameters:
			<ul> <li>_EVNUM: Number of empirical value memory</li> </ul>
			The following default value is entered in the NC code:
			• _EVNUM=0
			No empirical value memory is taken into account.
	1	-	Screen forms for workpiece measurement with automatic tool
			offset and tool measurement contain an input field for the following
			parameters:
			<ul> <li>_EVNUM: Number of empirical value memory</li> </ul>
_MZ_MASK[4]	0	0	Screen forms without input fields for the following parameters for
			mean value calculation with automatic tool offset:
			_TMV: Select range for offset with mean value
			calculation
			K: Weighting factor for mean value calculation
			EVNUM: mean value memory number
			The following default values are entered in the NC code:
			<ul> <li>_TMV=ABS(_TULTLL)/3</li> </ul>
			• _K=1
			• _EVNUM=0
			• _CHBIT[4]=0
	1	-	Screen forms with input fields for the following parameters for
			mean value calculation with automatic tool offset:
			_IMV: Select range for offset with mean value
			Calculation
			• _K. Weighting factor for mean value calculation
			• Lowing is also optored in the NC code:
MZ MASK[5]		0	CODIT[4]=1  Probe type for worknigge measurement is a multi-probe
_MZ_MASK[5]	0	0	Probe type for workpiece measurement is a multi probe
	I		The relevant screen forms show an input field for the offset
			andle CORA.
MZ MASK[6]			Reserved



Variable	Value	Default	Meaning
_MZ_MASK[7]	0	0	Screen form for CYCLE971 – tool measurement/milling does not contain input fields for feedrate and spindle speed. F and S are calculated within the cycle. The following is also entered in the NC code: • _CBIT[12]=0
	1	-	Screen form for CYCLE971 – tool measurement/milling contains input fields for feedrate and spindle speed. The following is also entered in the NC code: • _CBIT[12]=1

References:

/BEM/, HMI Embedded Operator's Guide

/IAM/, Installation and Start-Up Guide HMI/MMC IM2 "Installation and Start-up of HMI Embedded"



### Retranslation

Recompilation of programs allows you to change existing programs using the cycle support. When recompiling measuring cycle calls, please note that a field of defaults for programming is active (\_MZ\_MASK) in addition to the screen forms. If there has been a change in this settings between program creation and recompilation, the changes will also be included in the program.

Programs with measuring cycle calls cannot be recompiled after a change in the type of tool programming, i.e. change in the machine data setting

- MD 18102: MM\_TYPE\_OF\_CUTTING\_EDGE
- MD 18080: MM\_TOOL\_MANAGEMENT\_MASK.



### 7.4 Measuring result screens

### Function

Measurement result displays will be shown automatically during measuring cycle runtime if \_CHBIT[10]=1. If \_CHBIT[10]=0 (default setting), the measurement result displays are now shown.

Depending on the setting in \_CHBIT[11] and \_CHBIT[18]

- the measurement result displays are automatically deselected at the end of measuring cycle (\_CHBIT[11]=0, \_CHBIT[18]=0) or
- the measurement result displays must be acknowledged with the NC start key (\_CHBIT[11]=1, \_CHBIT[18]=0). In this case, the measuring cycle outputs the message: "Please acknowledge meas. result display with NC start"

or

 The measurement result displays are retained until the next measuring cycle call (\_CHBIT[11]=0, \_CHBIT[18]=1).



### Explanation

The measuring cycles can display different measuring result screens depending on the measuring variant:

- Calibrating tool probes
- Tool measurement
- Calibrating workpiece probes
- Workpiece measurement

The result displays contain the following data:

### **Calibrating tool probes**

- Measuring cycle and measuring variant
- Trigger values of axis directions and differences
- Probe number
- Safe area




- Measuring cycle and measuring variant
- Actual values and differences for tool offsets
- Safe area and permissible dimensional difference
- T-, D-number

#### Calibrating workpiece probes

- Measuring cycle and measuring variant
- Trigger values of axis directions and differences
- Positional deviation during calibration in the plane
- Probe number
- Safe area

#### Workpiece measurement

- Measuring cycle and measuring variant
- Setpoints, actual values and their differences
- Tolerance upper and lower limits (for tool offset)
- Offset value
- Probe number
- Safe area and permissible dimensional difference
- T number, D number, and DL number or ZO memory number for automatic offset

#### Example of measurement result display

Zyklen CHAN1 AUTO				WKS.DIR\MI TEST_102.M	esse IPF	EN_SCA_TEST.V	VPD	
💮 Kana	d unterbrochen							
🐴 Halt:	M0/M1 aktiv				ROV			
	NV-EF	MITTLUNG AN	WELLE					
Messerg	Jebnis CYCLE9	177						1
NV-Ern	nittlung Welle				M	lessvariante:	102	
		Sollwert		Istwert		Differenz		
X		100.000000		100.373567	10	0.373567	mm	
Y		55.000000		54.773794		-0.226206	mm	
We	lle	50.000000		41.013243		-8.986757	mm	
NV	-Speicher-Nr.	1						
Me	sstasternr.	102	Vertrau	ensbereich		10.000000	mm	
Kein	e Zugriffsrecht							
Kon			T					1
		-	-			-		



Notes



8

# Part 2 Description of Functions

# Hardware, Software and Installation

8.1	Hardware prerequisites	
8.1.1	General hardware prerequisites	
8.1.2	Measuring probe connection	
8.1.3	Measuring in JOG	
8.2	Software prerequisites	
8.2.1	Software versions NC and MMC/HMI	
8.2.2	Measuring in JOG	
	5	



### 8.1 Hardware prerequisites

#### 8.1.1 General hardware prerequisites

#### Axis assignment

For correct execution of the measuring cycles the machine axes must be assigned according to DIN 66217.



#### Probes that can be used

The measuring cycles require connection of switching probes; see the descriptions in Section 1.5.

#### 8.1.2 Measuring probe connection

On the SINUMERIK 840D, 840Di and 810D, the probe is connected via the I/O interface X121 which is located on the front panel of the NCU/CCU module.

#### Example: Probe connection to X121 on CCU module





8

Interfaces, control and display elements of NCU module



# **Interface**

• I/O interface

37-pin subminiature D connector (X121), **maximum 2** measuring probes can be connected;

The 24 V external power supply for the binary inputs is also located on this connector.

Excerpt from PIN assignment table for front panel connector X121:

PIN		Description
		External power supply
1	M24EXT	External ground
2	M24EXT	External ground
		Connection of probe 1
9	MEPUS 0	Measuring pulse signal input
10	MEPUC 0	Measuring pulse common input
		External power supply
20	P24EXT	External P24 V
21	P24EXT	External P24 V
		Connection of probe 2
28	MEPUS 1	Measuring pulse signal input
29	MEPUC 1	Measuring pulse common input

The interfaces (e.g. pin assignment) are described in detail in

References: /PHD/, Hardware Configuring Guide

# 8.1.3 Measuring in JOG



## Explanation

Measuring in JOG is only possible with

- SINUMERIK 840D
- SINUMERIK 810D
- HMI Advanced (PCU 50) or MMC 103



#### 8.2 Software prerequisites

#### How the measuring cycles are supplied

The measuring cycle software is supplied on diskette.

#### 8.2.1 Software versions NC and MMC/HMI

#### NC software version

The measuring cycles can only be used with NCK SW 6.3 and higher.

#### MMC SW

The measuring cycles require an MMC/HMI software version as from SW 4.4; see compatibility list below.

#### **PLC** program

No adaptation of the PLC user program is required to run the measuring cycles.

# Compatibility of measuring cycle versions – NCU/CCU, HMI/MMC

The following table provides an overview of the software versions of the NCU/CCU and HMI/MMC components and their compatibility with the measuring cycles. The table indicates whether the basic functionality of the measuring cycle version or the subfunction listed can be executed with the corresponding other system component.

Some cycles or functions may have additional software requirements. These are described in the description of the cycle or function.

	Meas. cycle	Meas. cycle	Meas. in JOG	Meas. in	Meas. cycle
	SW 5.3/5.4	SW 6.2	SW 5.3/5.4	JOG SW 6.2	support SW 6.2
NCU			-	·	
SW 4.4	x	n	х	n	No requirements
SW 5.3	x	n	x	n	on NCU, pure
SW 6.2	x	n	х	n	HMI function
SW 6.3	x	x	х	x	
SW 6.4	x	x	х	x	
CCU		•	•		•
SW 2.4	x	n	х	n	No requirements
SW 3.3	x	n	х	n	on NCU, pure
SW 4.2	x	n	х	n	HMI function
SW 6.3	x	x	х	x	
SW 6.4	x	x	х	x	
HMI Embedded		•	•		•
SW 6.1	x	х	Measuring i	n JOG not	n
SW 6.2	x	x	possible with H	MI Embedded	х
SW 6.3	x	x			Х
MMC 100.2		•	•		•
SW 4.4	x	х	Measuring i	n JOG not	Not possible
SW 5.1	х	х	possible with	MMC100.2	with MMC100.2
SW 5.2	x	х			
SW 5.3	x	х			
HMI Advanced					•
SW 6.1	X	х	Х	x	n
SW 6.2	x	х	х	x	Х
SW 6.3	х	x	X	x	X
MMC 103					•
SW 4.4	x	x	n	n	Not possible
SW 5.1	x	x	n	n	with MMC103
SW 5.2	x	x	n	n	
SW 5.3	x	x	x	x	

x executable n not executable

# 8.2.2 Measuring in JOG



#### Options

"Measurement in JOG" can only be used if the "Inter-modal actions" option (ASUPs and synchronized actions in all modes) is active.





Ħ

# Function

#### Measure command

The control has the command MEAS for generating a measuring block.

The measuring input number is set in the command parameters.

#### Saving measurement results

The results of the measurement command are stored in the system data of the NCK and can be accessed from the program.

They are:

\$AC_MEA[ <no.>]</no.>	Software switching signal for the probe			
	No. stands for measuring input number			
\$AA_MW[ <axis>]</axis>	Measured value of the axis in workpiece coordinates			
	Axis stands for the name of the measuring axis			
\$AA_MM[ <axis>]</axis>	Measured value of the axis in machine coordinates			

References: /PG/, Programming Guide

#### PLC service display

The functional test for the probe is conducted via an NC program.

The measuring signal can be controlled via the diagnostics menu "PLC status".

Status display for measurement signal

Probe 1 deflected	DB10	DB X107.0
Probe 2 deflected	DB10	DB X107.1



# Example of functional check

%\_N\_TEST\_PROBE\_MPF

H=/_N_MPF_DIR	
g program probe connection	
DEF INT MTSIGNAL	;Flag for trigger status
DEF INT ME_NR=1	;Measurement input number
G17 T1 D1	;Select tool offset for probe
G0 G90 X0 F150	;Starting position and meas. velocity
MEAS=ME_NR G1 X100	;Measurement at measuring input 1
	;in the X axis
STOPRE	;Preprocessing stop
MTSIGNAL=\$AC_MEA[ME_NR]	;Read software switching signal
	;at 1st measurement input
IF MTSIGNAL == 0 GOTOF _ERROR1	;Evaluation of signal
R1=\$AA_MW[X]	;Read measured value in workpiece
	;coordinates
MO	;View measured value in R1
M2	
_ERROR1: MSG ("Probe not switching!")	;Display "Probe not switching"
MO	
M2	
	H=/_N_MPF_DIR g program probe connection DEF INT MTSIGNAL DEF INT ME_NR=1 G17 T1 D1 G0 G90 X0 F150 MEAS=ME_NR G1 X100 STOPRE MTSIGNAL=\$AC_MEA[ME_NR] IF MTSIGNAL == 0 GOTOF _ERROR1 R1=\$AA_MW[X] M0 M2 _ERROR1: MSG ("Probe not switching!") M0 M2

9

# **Data Description**

9.1 9 1 1	Machine data for executing measuring cycles	9-408 9-408
9.1.2	Additional machine data	9-400
9.2	Cycle data	9-413
9.2.1	Data blocks for measuring cycles: GUD5.DEF and GUD6.DEF	9-413
9.2.2	Data adjustment to a specific machine	9-418
9.2.3	Central values	9-420
9.2.4	Central bits	9-426
9.2.5	Central strings	9-429
9.2.6	Channel-oriented values	9-430
9.2.7	Channel-oriented bits	9-433
9.3	Data for measuring in JOG	9-440
9.3.1	Machine data for ensuring ability to function	9-440
9.3.2	Modifying the GUD7 data block	9-442
9.3.3	Settings in data block GUD6	9-446
9.3.4	Loading files for measuring in JOG	9-447



#### 9.1.1 Machine data that configure the memory

# Function

The measuring cycles use their own **GUD** and **LUD variables** (**G**lobal **U**ser **D**ata and Local **U**ser **D**ata). The memory provided is configured via machine data in the control. The values must be increased.

The memory requirement for GUD variables arises on loading data blocks GUD5 and GUD6; and if the "**Measurement in JOG**" function is installed, also on loading GUD7 (from SW 6.3: GUD7\_MC).

For special data for "measuring in JOG", refer to Section 9.3

LUD variables are created during program and cycle runtime. So the memory space is not used until execution. A sufficient reserve must be available because different amounts of memory are required at different times.

**System frames** are used in the cycles. The memory space is provided via special machine data settings in the control and set by default.

#### Notice

The following information in the machine data **only** covers use of SIEMENS measuring cycles.

When using **ShopMill** or **ShopTurn**, comply with the information for these products.

#### Note:

Future versions measuring cycle versions may require more memory. For current data, see file SIEMENSD.TXT or SIEMENSE.TXT on supplied diskette 1. 10.04

# 9.1 Machine data for executing measuring cycles



#### Notice

10.04

Machine data whose name starts "\_MM" are memory configuring data. Changing these data reallocates the memory. Data it contains will be lost. If this memory is in the non-volatile memory range of the control (**SRAM**), the **user data** (loaded programs and definitions, R-parameters, tool offsets, zero offsets, etc.) it contains will be lost. **Therefore:** When performing initial start-up, set these

machine data at the beginning of installation.

If data is already contained in the user memory, all data of the user memory must be saved before changing the machine data. After a successful change followed by a new control start-up, this data can be downloaded again.

#### Memory configuring machine data, SRAM

<b>18118</b> MD number	MM_NUM_GUD_MODULES Number of data blocks					
Default setting: 7 When using measuring cy	put limit: 1		Maximum input limit: 9			
Changes effective after PO	Protection level: 2/7			Unit: -		
Data type: DWORD	Applies from SW version: SW 2					
Meaning: Number of GUD files in the active file system (SRAM)						

<b>18120</b> MD number	MM_NUM_GUD_NAMES_NCK Number of GUD variable names in the control						
Default setting: 10 Minimur When using measuring cycles: 30		Minimum inp	um input limit: 0		Maximum input limit: plus		
Changes effective after POWER ON			Protection level: 2/7			Unit: -	
Data type: DWORD				Applies from	SW version:	SW 1	
Meaning:	Memory reservation for GUD variable names, NCK (SRAM) range Note: The number of GUDs may also be limited by MD 18150.						

<b>18130</b> MD number	MM_NUM_GUD_NAMES_CHAN Number of GUD variable names per channel						
Default setting: 40 When using measuring cycles: 100 with measurement in JOG: 130		Minimum input limit: 0		Maximum input limit: plus			
Changes effective after POV	VER ON	Protection level: 2/7		/el: 2/7		Unit: -	
Data type: DWORD			Applies from SW version: SW 1				
Meaning:	Vemory reservation for GUD variable names in each channel, CHAN (SRAM) range Note: The number of GUDs may also be limited by MD 18150.						

<b>18150</b> MD number	MM_GUD_VALUES_MEM Memory for values of the GUD variables					
Default setting: 12 / 16 <sup>1)</sup> When using measuring cycles: 28 / 32 <sup>1)</sup> (see Table below)		Minimum input limit: 0		Maximum input limit: plus		
Changes effective after POWER ON			Protection level: 2/7			Unit: KB
Data type: DWORD		Applies from SW version: SV		SW 1		
Meaning:	Memory for user variables (SRAM): Provides indication of the number of GUD variables that can be created. The FRAME data type requires almost 1 Kbyte. It is the largest data type. Data type BOOLEAN requires only 1 byte; type REAL 8 bytes. An array of variables requires only one name (MD 18120, MD 18130) – but memory space for a elements. Here, the total memory requirement is entered for all GUD variables, for NCK area, and for all channels.					nber of GUD variables that can ne largest data type. Data type 30) – but memory space for all es, for NCK area, and for all MD 18130.

1) NCU-specific



#### Memory requirement of GUD variables for meas. cycles

The machine data settings recommended above provide overall values that enable work with measuring cycles. Deviating settings may be required for particular applications. The memory requirement of each measuring cycle is therefore specified here (approximate values). It may be higher in future measuring cycle versions. The resulting memory requirement must be added to other

user memory requirements.

Range	Number of names	Memory in Kbytes
		MD 18150
NCK	MD 18120: <b>20</b>	7
CHAN	MD 18130: <b>60</b>	Two channels: 8
(GUD5, GUD6 only)		(number of channels * 4)
CHAN	MD 18130: <b>30</b>	Two channels: 1
(from GUD7, also		(number of channels * 0.5)
for "measuring in JOG")		
		Example: Measuring cycles with 2
		channels and "measuring in JOG": 16

<b>28082</b> MD number	MM_SYSTER Configuration	M_SYSTEM_FRAME_MASK onfiguration screen form for channel-specific system frames						
Default setting: 21hex (bit 0	Minimum input limit: 0 =1, bit 5=1)				Maximum input limit: 7Fhex			
Changes effective after POWER ON			Protection lev	vel: 2/7		Unit: -		
Data type: INT				(valid from m	neas. cycles S	SW 6.2)		
Meaning:	Bit 0: 1 syste Bit 5: 1 syste	t 0: 1 system frames for preset mode and scratching t 5: 1 system frames for cycles (must always be available for cycles)						

#### Memory configuring machine data, DRAM

<b>18170</b> MD number	MM_NUM_N Number of na	IM_NUM_MAX_FUNC_NAMES Index of miscellaneous functions - cycles						
Default setting: 40 <b>When using measuring cy</b>	Minimum inp uring cycles: 80			put limit: 0		Maximum input limit: plus		
Changes effective after POWER ON			Protection le	vel: 2/7		Unit: -		
Data type: DWORD				Applies from	SW version:	SW 1		
Meaning:	Number of cy	Number of cycles with input parameters (DRAM)						

<b>18180</b> MD number	MM_NUM_N Number of pa	IM_NUM_MAX_FUNC_PARAM Jumber of parameters in the miscellaneous functions - cycles						
Default setting: 300 Min		Minimum input limit: 0		Maximum input limit: plus				
Changes effective after POWER ON		Protection level: 2/7			Unit: -			
Data type: DWORD				Applies from	SW version:	SW 1		
Meaning:	Number of pa	umber of parameters for cycles acc. to MD 18170 (DRAM)						

<b>28020</b> MD number	MM_NUM_LU Number of LU	MM_NUM_LUD_NAMES_TOTAL Number of LUD variable names (in total in all program levels)						
Default setting: 200 When using measuring cy	vcles: 200	/linimum inp	ut limit: 0	Maximum input limit:		out limit: plus		
Changes effective after POWER ON			Protection level: 2/7 Unit: -					
Data type: DWORD				Applies from	SW version:	SW 3.2		
Meaning:	Number of local user variables – for each channel (DRAM) Note: The number of LUDs may also be limited by MD 28040.							

<b>28040</b> MD number	MM_NUM_L Memory for v	IM_NUM_LUD_VALUES_MEM /lemory for values of the LUD variables (DRAM)					
Default setting: 12 / 25 <sup>1)</sup> <b>When using measuring cy</b> 29 <sup>1)</sup>	Minimum input limit: 0				Maximum inp	put limit: plus	
Changes effective after POWER ON			Protection lev	/el: 2/7		Unit: KB	
Data type: DWORD			Applies from SW version: SW 3.2				
Meaning:	Provides indication of the number of LUD variables that can be created. The FRAME data type requires almost 1 Kbyte. It is the largest data type. The BOOLEAN data type, however, requires only 1 byte. An array of variables requires only one name (MD 18120, MD 18130) – but memory space for all elements. <b>Note:</b> The number of LUDs may also be limited by MD 28020.						

1) NCU-specific



# 9.1.2 Additional machine data



# Machine data for adapting the probe

<b>13200</b> MD number	MEAS_PROB Switching cha	IEAS_PROBE_LOW_ACTIVE[n] witching characteristics of the measuring probe (n=0: Measuring input 1, n=1: Measuring input 2					
Default setting: FALSE Minin When using measurement cycles: FALSE		Minimum input limit: FALSE		Maximum inp	out limit: TRUE		
Changes effective after POV	VER ON		Protection le	vel: 2/7		Unit: -	
Data type: BOOLEAN			L	Applies from	SW version:	2.2	
Meaning:	Value 0 (FAL Value 1 (TRU <b>Note:</b> When I to the voltage	.SE): non-deflected state deflected state JE) non-deflected state deflected state programming MEAS, the c e level. The voltage level is		0 24 24 0 lata always re s set with this	V (low) IV (high) I V (high) V (low) efers to "not d MD.	eflected", "deflected" – but not	

#### Machine data for adapting MMC commands in cycles

<b>10132</b> MD number	MMC_CMD_ Monitoring ti	MC_CMD_TIMEOUT onitoring time for MMC command in part program						
Default setting: 1.0 When using measuring cy	cles: 3.0	Minimum input limit: 1.0 es: 3.0			Maximum inp	out limit: 100.0		
Changes effective after POWER ON			Protection level: 2/7			Unit: s		
Data type: DOUBLE				Applies from	SW version:	SW 3.2		
Meaning:	Monitors the	onitors the time until the MMC acknowledges a command from the part program.						

#### Machine data for logging

<b>11420</b> MD number	LEN_PROTOCOL_F File size for log files	EN_PROTOCOL_FILE						
Default setting: 1 When using measuring cy	Minimum input limit: 1				Maximum input limit: 1000 000			
Changes effective after POWER ON		Protection level: 2/7 Unit: Ki		Unit: KB				
Data type: DWORD				Applies from	SW version:	SW4.3		
Meaning:	Maximum size of protocol files that are created with WRITE command.			mand.				





### 9.2 Cycle data

### 9.2.1 Data blocks for measuring cycles: GUD5.DEF and GUD6.DEF



#### Function

The measuring cycle data are stored in two separate definition blocks:

- GUD5.DEF Data block for measuring cycle users
- GUD6.DEF Data module for machine manufacturers



#### Data block GUD5.DEF

The input and output parameters for measuring cycles are stored in the data block GUD5.DEF; their status flags and arrays for the empirical and mean values are also defined here.

The sizes of the fields for the empirical and mean values must also be configured by the machine manufacturer at measuring cycle start-up. The preset values, however, are defined by the measuring cycle operator.

In the as-delivered state, for example, the following settings are active:

_EV[20]	REAL	Number of empirical values
_MV[20]	REAL	Number of mean values



# Data block GUD6.DEF

The general, global, and channel-specific measuring cycle data are configured in the GUD6.DEF data block.

This block is supplied with the measuring cycles in its standard configuration and must be adapted to the specific requirements of the machine by the machine manufacturer.

In the as-delivered state, the following settings are active:

#### **Global data**

_CVAL[4]=(3,3,3,0)	INTEGER	3 arrays each for
		<ul> <li>tool probe, machine-related (_TP[3,10])</li> </ul>
		<ul> <li>workpiece probe (_WP[3,11])</li> </ul>
		<ul> <li>gauging block (_KB[3,7])</li> </ul>
		<ul> <li>tool probe, workpiece-related (_TPW[3,10])</li> </ul>
_TP[3,10]	REAL	3 arrays for tool probes, machine-related
_WP[3,11]	REAL	3 arrays for workpiece probes
_KB[3,7]	REAL	3 arrays for gauging blocks
_TPW[3,10]	REAL	3 arrays for tool probes, workpiece-related (from measuring
		cycles SW 6.3)
_CM[8]=(100,1000,1,	REAL	Only active if _CBIT[12]=0
0.005,20,4,10,0),		Monitoring data for tool measurement with rotating spindle
		and cyclic calculation:
		<ul> <li>max. peripheral speed 100 m/min</li> </ul>
		<ul> <li>max. speed 1000 rev/min</li> </ul>
		• F <sub>min</sub> =1mm/min
		<ul> <li>measuring accuracy 0.005 mm</li> </ul>
		• F <sub>max</sub> for probing 20 mm/min
		direction of rotation M4
		<ul> <li>probing twice with feed factor 10 on first probing</li> </ul>
_MFS[6]	REAL	Only active if _CBIT[12]=1
		Speed and feed set by user during tool measurement with
		rotating spindle
_CBIT[16]=(0,0,0,1,0,	BOOL	Central bits
0,0,0,1,0,0,0,0,0,0,0)		[0]: 0: A measurement is not repeated after a limit has
		been violated dimensional difference and safe
		area
		[1]: 0: No M0 on measurement repeat
		[2]: 0: No M0 on "oversize", "undersize", "dim. difference"
		[3]: 1: Metric basic system
		[4]: 0: Internal data item

		[5]: 0:	Tool measurement and calibration with CYCLE982
			performed in basic coordinate system (machine
			coordinate system with switched-off
			kinematic transformation)
		[6]: 0:	Logging with information about the measuring
		L · J ·	cycle measuring variant
		[7]: 0:	Internal data item
		[8]: 1:	Offset for mono probe setting with CORA.
		[9]: 0:	Logging off
		[10]: 0:	Internal data item
		[11]: 0:	Use of standard log header
		[12]: 0:	Cycle internal calculation of feedrate and
			feedrate for tool measurement with rotating
			spindle
		[13]: 0:	Without deleting fields TP[1, TPW[1, WP[1,
			KB[], EV[] and MV[]
		[14]: 0:	Length of the workpiece measuring probe, referred
			to the center of the probe sphere (ball)
		[15]: 0:	Internal data item
SI[3]=("","6","")	STRING[8]	Central	strings (system information)
		• Inter	nal data item
		<ul> <li>Soft</li> </ul>	ware version of the control
		<ul> <li>Inter</li> </ul>	nal data item
PROTNAME[2]	STRING[32]	Nar	he of main program the log is from
	011110[02]	<ul> <li>Nam</li> </ul>	ne of log file
	STRING[80]	Strings f	for log header
			natting
_1 3 1 12)	, INTEGER	• 60 li	nes per page
1,0,1,12),			heractors por lino
		• OUC	
			page number is 1
		• Nurr	iber of neader lines is 3
		• Nurr	iber of value lines in the log is 1
		• Num	ber of characters per column is 12
_PROTSYM[2]=(";","#")	CHAR	<ul> <li>Separation</li> </ul>	arator in the log is ";"
		<ul> <li>Spe</li> </ul>	cial character for identification of measuring
		cycle	es is "#"
_PROTVAL[13]	STRING[100]	Title line	in log;
		Specifica	ation of the values to be logged
		• 01	: two title lines
		• 25	: up to 4 value lines
		• 61	2: internal fields
_PMI[4]	INTEGER	Field for	internal flags for logging
_SP_B[20]	INTEGER	Variable	column width
_TXT[100]	STRING[12]	Field for	formatted strings
DIGIT=3	INTEGER	Number	of decimal places is 3

9

_M2_MASK[20] =(1,0,0,0,0,0,0,0)	INTEGER	<ul> <li>Setting data for measuring cycle support</li> <li>generation of a direct measuring cycle call</li> <li>without extended ZO/tool offset</li> <li>without input field for measuring speed and measuring feed</li> <li>without empirical values</li> <li>without mean value calculation</li> <li>workpiece probe is a multidirectional probe,</li> <li>internal data item</li> <li>cycle-internal calculation of feedrate and spindle rotation for tool measurement with rotating spindle (CYCLE971)</li> </ul>
Channel-specific data		
_EVMVNUM[2]=(20,20)	INTEGER	<ul> <li>Number of empirical values and mean values</li> <li>20 memories for empirical values</li> <li>20 memories for mean values</li> </ul>
_SPEED[4] =(50,1000,1000,900),	REAL	<ul> <li>Traversing velocities for intermediate positioning</li> <li>50% rapid traverse velocity</li> <li>positioning feedrate in the plane 1000 mm/min</li> <li>positioning feedrate in infeed axis 1000 mm/min</li> <li>fast measurement feedrate 900 mm/min</li> </ul>
_CHBIT[30] =(0,1,1,0,0,0,0,0,0,0,0,0,0,0,0,0,0,0,0,0	BOOL	<ul> <li>[0]: 0: Measuring input 1 for connecting a workpiece probe</li> <li>[1]: 1: Measuring input 2 for connecting a tool probe</li> <li>[2]: 1: Collision monitoring active for traversing blocks generated by measuring cycles</li> <li>[3]: 0: Tool data are entered into the geometry memory when the tool is measured</li> <li>[4]: 0: No mean value storage</li> <li>[5]: 0: The empirical value is subtracted from the measured actual value</li> <li>[6]: 0: When measuring workpieces with automatic tool offset, an additive offset/correction is made in the wear memory</li> <li>[7]: 0: CYCLE994 uses the trigger values for offset</li> <li>[8]: 0: When measuring workpieces with automatic tool offset, an additive offset is made in the total offset</li> <li>[9]: 0: Internal data item</li> <li>[10]: 0: No measurement result display</li> <li>[11]: 0: Measurement result display is deselected at the end of cycle</li> <li>[12]: 0: Internal data item</li> <li>[13]: 0: No coupling of the spindle position with coordinate rotation in the plane</li> <li>[14]: 0: Spindle positioning acc. to default</li> <li>[15]: 0: Up to 5 measurement attempts</li> <li>[16]: 0: Retraction from the measuring location with the same velocity as for the intermediate positioning</li> </ul>

# Data Description 9.2 Cycle data



		<ul> <li>[17]: 0: Measurement feed only defined by _VMS</li> <li>[18]: 0: Automatic de-selection of the measuring result screen with end of cycle</li> <li>[19]: 0: Normal handling of the Y axis for rotating meas. cycles,</li> <li>[20]: 0: Spindle positioning with measurement with CYCLE982</li> <li>[21]: 0: Internal data item</li> </ul>
		[22]: 0: Last measurement with reduced speed for tool measurement with rotating spindle (CYCLE971)
		<ul><li>[23]: 0: Internal data item</li><li>[24]: 1: Metric basic system</li><li>[2529]: 0: Internal data item</li></ul>
_IP_CF=0	INTEGER	No tool probe manufacturer specified (type)
_MT_COMP=0	INTEGER	No additional offset of the measurement result display on tool measurement with rotating spindle (CYCLE971)
_MT_EC_R[6,5]=(0,,0)	REAL	User-defined array for offsetting the measurement result on tool radius measurement and rotating spindle (CYCLE971)
_MT_EC_L[6,5]=(0,,0	REAL	User-defined array for offsetting the measurement result on tool length measurement and rotating spindle (CYCLE971)
_JM_I[5]=(0,1,1,17,0)	INTEGER	<ul> <li>INT value field for JOG measurement</li> <li>no set array number for probes like in ShopMill</li> <li>number of array for workpiece probe is 1</li> <li>number of array for tool probe is 1</li> <li>working plane for measurement in JOG is G17</li> <li>active ZO number on measurement in JOG is 0 (G500)</li> </ul>
_JM_B[7]= (0,1,0,0,0,0,0)	BOOL	<ul> <li>BOOL value field for JOG measurement</li> <li>offset in geometry on tool measurement</li> <li>1 measurement attempt</li> <li>retraction from meas. point at same velocity as intermediate positioning</li> <li>no fast measurement feed</li> </ul>
_MC_MTL[3]= (33.3,33.3,33.3)	INTEGER	Measuring probe offset (correction) for sphere measurement (only relevant in CYCLE997 for the measurement variant with "determining the sphere diameter") Ratio between the tracer length/pin-sphere (ball) radius (\$TC_DP6) Pre-assignment (default) for 3 measuring probes: 100/3 If this variable is not available, then the trigger values are not corrected. [Array index]: _PRNUM-1 (from meas. cycles SW 6.3)
_MC_SIMSIM=1	INTEGER	0: Skip measuring cycles during simulation 1: Run measuring cycles during simulation (from meas. cycles SW 6.3)
_MC_SIMDIFF=0	REAL	Value for simulated measuring difference (from meas. cycles SW 6.3)



# 9.2.2 Data adjustment to a specific machine



# Function

There are two main steps for adapting the data to a specific machine:

- 1. **Adapting the data definitions** in the GUD blocks and loading them in the PLC.
- 2. **Subsequent adjustment** of certain values in part program.



## 1. Adapting the data definitions

The following example shows how to adapt the data blocks GUD5.DEF and GUD6.DEF to a machine with SINUMERIK 840D with the characteristics described below:

- and 2 data fields (data arrays) for using tool probes
- and 2 data fields (data arrays) for using workpiece probes
- without calibration groove pair,
- only 10 empirical and mean values

The following example only shows the lines to be

#### changed!

#### Example:

%\_N\_GUD6\_DEF

;\$PATH=/\_N\_DEF\_DIR

;(date) adaptation to a machine\_1

...

N10 DEF NCK INT \_CVAL[4]=(**2**,**2**,**0**,0)<sup>1)</sup>

N11 DEF NCK REAL \_TP[**2**,10]=(0,0,0,0,0,0,0,133,0,2)<sup>1)</sup>

N12 DEF NCK REAL \_WP[2,11]<sup>1)</sup>

;N13 DEF NCK REAL \_KB[3,7]<sup>1)</sup>

N40 DEF CHAN INT \_EVMVNUM[2]=(10,10)<sup>1)</sup>

... M17

 Characters and digits in bold (highlighted) have been changed with respect to the status when first supplied





#### 9.2.3 Central values

#### Data block GUD6.DEF

	_CVAL[] Number of	elements, arrays	
		Minimum input limit: -	faximum input limit: -
Changes valid after va	alue assignment	Protection level: -	Unit: -
Data type: INT		Applies from S	W version: 3.2 / 6.3
Meaning:	CVAL[0]	Number of arrays for tool probes TP[	Preset default
		(machine-related)	3
	_CVAL[1]	Number of arrays for workpiece probe	s_WP[] 3
	_CVAL[2] CVAL[3]	Number of arrays for gauging blocks _ Number of arrays for tool probes TP\	KB[] 3 W[]
	· · · -[•]	(relative to the workpiece, from meas.	cycles SW 6.3) 3

1	TD/ 1				
	_IP[] Array for tool	probes (machine-related)			
	-	Minimum input limit: -		Maximum input limit: -	
Changes valid after value ass	ignment	Protection lev	el: -	Unit: -	
Data type: REAL			Applies from	n SW version: 3.2	
Meaning:					Preset default
ارا م	ndex "k" stan Assignment	ds for the number of the cu on milling Trigger point in minus di	urrent array	(_PRNUM-1)	0
-	 TP[k,1]	Trigger point in plus dire	ction X	(1st geometry axis)	0
	_TP[k,2]	Trigger point in minus di	rection Y	(2nd geometry axis)	0
	_TP[k,3]	Trigger point in plus dire	ction Y	(2nd geometry axis)	0
	_TP[k,4]	Trigger point in minus di	rection Z	(3rd geometry axis)	0
	_TP[k,5]	Trigger point in plus dire	ction Z	(3rd geometry axis)	0
	_TP[k,6]	Edge length/disk diamet	er		0
	_TP[k,7]	Axes and directions for ' (from meas. cycles SW)	'automatic ( 6.3)	calibration"	133
	_TP[k,8)	Probe type 0: cube	, - XX		0
		101: disk i 201: disk ir	n XY I ZX		
		301: disk ir	YZ		
-	TP[k,9]	Distance between upper (= calibration depth, mea	edge of too	bl probe and lower edge of depth for cutter radius)	tool 2
م	ssignment	for turning (probe type: pr	imarily cub	e)	
	_TP[k,0]	Trigger point in minus di	rection of a	bscissa	0
	_TP[k,1]	Trigger point in plus dire	ction of abs	cissa	0
	_TP[k,2]	Trigger point in minus di	rection of o	rdinate	0
-	_TP[k,3]	Trigger point in plus dire	ction of ord	inate	0
	_TP[k,4]	irrelevant			0
	_TP[k,9]	irrelevant			0
Tool probe on milling	machine				
Example: Probe type di	sk in XY (	TP[k,8]=101)		Z1 Trigger points in the	
- F	- (_			geometry axes	
				_TP[k,6]	
				-Z	Д
			_'' [K,4]		_TP[k,9]
					Calibration depth,
			_TP[k,5]	+Z	measurement dept
Fool probe on turning	machine				v
Example: G18 plane. v	alues ma	chine-related		+X	~
p				MT _TP[k,1] TP[	[k,0] X1
X1/X 🕇 Triana	r nointe			Y1	
(ordinate) in abs	cissa (G18)				
			TPIk 21	-Y	
_'' ['∠]					
	∔ - ⊨				
	1				
_TP[k,3] +X			_TP[k,3]	+Y	
+Z	-Z	(abscissa)			v
M		71/7		+X -	^
_TP[k,1]	_TP[k,0]			MTP[k,1]TP[l	k,0] X1
Tool probe mac	hine related.				
cube type	,		т	ool probe machine related	
				lisk-type in XY ( TPIk 81=10	)1)
					.,

	Array for tool	probes (workpiece-related)			
		Vinimum input limit: -		Maximum input limit: -	
Changes valid after va	alue assignment	Protection level:	-	Unit: -	
Data type: REAL		Apr	olies from	SW version: 6.3	
Meaning:					Preset defaul
	Index "k" stan	ds for the number of the curre	ent array	(_PRNUM-1)	
	Assignment	on milling			
	_TPW[k,0]	Trigger point in minus direc	tion X	(1st geometry axis)	0
	_TPW[k,1]	Trigger point in plus direction	on X	(1st geometry axis)	0
	_TPW[k,2]	Trigger point in minus direc	tion Y	(2nd geometry axis)	0
	_TPW[k,3]	Trigger point in plus direction	on Y	(2nd geometry axis)	0
	_TPW[k,4]	Trigger point in minus direc	tion <b>Z</b>	(3rd geometry axis)	0
	_TPW[k,5]	Trigger point in plus direction	on <b>Z</b>	(3rd geometry axis)	0
	_TPW[k,6]	Edge length/disk diameter			0
	_TPW[k,7]	Axes and directions for "au	tomatic c	alibration"	133
	_TPW[k,8)	Probe type 0: cube			0
		101: disk in	XY		
		201: disk in	ZX		
		301: disk in	ΥZ		
	_TPW[k,9]	Distance between upper ec	dge of too	I probe and lower edge of too	2
		(= calibration depth, measu	irement d	epth for cutter radius)	
	Assianment	for turning			
	TPW[k,0]	Trigger point in minus direc	tion of at	oscissa	0
	TPW[k,1]	Trigger point in plus direction	on of abs	cissa	0
	TPW[k,2]	Trigger point in minus direc	tion of or	dinate	0
		Trigger point in plus direction	on of ordi	nate	0
	TPW[k,4]	irrelevant			0
	to				
	TPW[k,9]	irrelevant			0

For illustration, see analogous explanation of \_TP[]

**Tool probe types** 

Ē





		Minimum input limit: -	Maximum	input limit: -
Changes valid after va	alue assignment	Protection level: -		Unit: -
Data type: REAL		Applies	from SW version	n: 3.2
Meaning:				Preset default
	Index "k" sta	inds for the number of the current a	array (_CALNUN	1-1)
	Reference s	lot for calibration of a workpiece pr	obe with cutting	edge position SL=7 (tool type:
	5xy)			
	KB[k,0]	Groove edge in plus direction,	Groove edge in plus direction, ordinate	
	KB[k,1]	Groove edge in minus direction	n, ordinate	0
	_KB[k,2]	Groove base in abscissa		0
	Reference g	roove for calibration of a workpiece	e probe with cutt	ing edge position SL=8 (tool type:
	5xy)			
	_KB[k,3]	Groove edge in plus direction,	abscissa	0
	_KB[k,4]	Groove edge in minus directior	n, abscissa	0
	_KB[k,5]	Upper edge groove in ordinate		0
	_KB[k,6]	Groove base in ordinate		0
	Notes:			
	The values of	of the grooves are always machine	-related and rad	ius dimensions.
	Both groove	s have the same depth.		

Overview of pair of reference grooves for calibration (for turning only) The representation refers to the working plane

defined by G18.



#### For tool measurement with CYCLE971 only

	_ <b>CM[ ]</b> Monitoring	functions for t	ool measurement w	vith rota	ting spindle, only activ	e if _CBIT[12]=0
		Minimum inp	out limit: -		Maximum input limit: ·	-
Changes valid after value as	ssignment		Protection level: -		Unit: -	
Data type: REAL			Applie	es from	SW version: 4.3	
Meaning:	_CM[0] _CM[1] _CM[2] _CM[3] _CM[4] _CM[5] _CM[6] _CM[7]	Max. perm Max. perm Minimum Required Max. perm Direction Feed factor Feed factor	nissible peripheral s nissible speed [rpm feedrate for 1st pro measuring accurac nissible feedrate for of spindle rotation or 1	speed [r ] bing [mi y [mm] · probing	n/min]/[feet/min] m/min]	Preset default 100 1000 1 0.005 20 4 10 0

14			
	r	-	-

#### For tool measurement with CYCLE971 only

	_MFS[] Feeds and	speeds for to	ol measuren	nent with rot	tating spindle, o	only active if	_CBIT[12]=1
		Minimum inp	out limit: -		Maximum ir	nput limit: -	
Changes valid after value as	signment	•	Protection I	evel: -	•	Unit: -	
Data type: REAL				Applies fr	om SW version	: 4.3	
Meaning:	_MFS[0] _MFS[1] _MFS[2] _MFS[3] _MFS[4] _MFS[5]	Speed 1st Feed 1st   Speed 2n Feed 2nd Speed 3rd Feed 3rd	t probing orobing d probing probing d probing probing				Preset default 0 0 0 0 0 0 0



9)

#### 9.2.4 Central bits

#### Data block GUD6.DEF

	_CBIT[]			
	Central bits		<b>h</b> a	
		Minimum input limit: 0	Maximum input limit: 1	
Changes valid after value as	ssignment	Protection level: -	Unit: -	
Data type: BOOLEAN		Applies from	SW version: 3.2	
Comment:				Preset default
	_CBIT[0]	Measurement repetition after violation	on of dimensional difference an	ıd
		Sate area 0: no measurement repeat		0
		1: measurement repeat, up to 4		
	_CBIT[1]	Alarm and M0 for measurement repo	eat with _CBIT[0]=1	0
		0: no alarm, no M0 generated	ofour cost usuant	
	CRITI21	1: MU and an alarm are generated b	etore each repeat	
	_CBIT[2]	"Permissible dimensional difference	exceeded"	0
		0: no generation of M0 for the above	alarms	·
		1: generation of M0 for the above ala	arms	
	_CBIT[3]	Central marker for basic dimension	system of the control	1
		1. Basic system setting is metric		
	_CBIT[4]	currently not assigned		0
	_CBIT[5]	Tool measurement and calibration ir	the WCS in CYCLE982	0
		0: machine-related measurement an	id calibration	
	Noto:	1: workpiece-related measurement a	and calibration	
	Nole.	From measuring cycles SW 6.3. fun	ction switchover via MVAR is	available.
	_CBIT[6]	Logging without output of the measu	Iring cycle name and	
		measuring variant		0
		0: Measuring cycle name and measure 1: Those outputs will be suppressed	uring variant will be output.	
	_CBIT[7]	currently not assigned		0
	CBIT[8]	Offset of mono probe position		1
	_02.1[0]	0: no offset		
		1: offset of spindle by angle _CORA	,	
	_CBIT[9]	Assigned internally		0
	_CBIT[10]	currently not assigned		0
	_CBIT[11]	Selection of log header for logging		0
		U: Standard		
	CBIT[12]	Feed and speed in CYCL F971		0
		0: calculation by measuring cycle its	elf	-
		1: set by user in array _MFS[]		
	_CBIT[13]	Deletion of the measuring cycle arra	ys in the GUD6	0
		1: delete TP[] TPW[] WP[] k	(BII EVII MVII CBITII3)	1
	CBIT[14]	Length reference of the workpiece p	robe in milling measuring cycle	es 0
		0: length relative to probe ball center	r	
		1: length relative to end		
	_CBIT[15]	I ranster of workpiece probe data int	to the tool offset in CYCLE976	0
		1: result of probe ball calculation on	calibration will be entered in t	he
	1	geometry memory of the workpiec	e probe (radius)	

Ϊ

E

	Central bits	– detailed description					
?	Measurement repetition after violation of dimensional difference and safe area						
	_CBIT[0]=0:	When the parameters for the dimensional difference (_TDIF) and safe area (_TSA) are exceeded, the measurement is not repeated. An appropriate alarm is displayed.					
	_CBIT[0]=1:	When the parameters for the dimensional difference (_TDIF) and safe area (_TSA) are exceeded, a measurement is repeated. An alarm is also displayed in the repeat measurements if <b>_CBIT[1]</b> is set.					
2	Alarm and M	I0 for measurement repeat with _CBIT[0]=1					
	_CBIT[1]=0: _CBIT[1]=1:	No alarm, no M0 in measurement repeats If the limits of the parameters for the dimensional difference and safe area are exceeded, M0 is generated and the repeat measurement must be started with NC-START. An alarm is displayed in the alarm line that does not have to be exceeded.					
2	M0 for tolera	ince alarms "oversize", "undersize", or					
	"permissible	e dimensional difference exceeded"					
	_CBIT[2]=0:	When the alarms "oversize", "undersize", or "permissible dimensional difference exceeded" occur, no M0 is generated.					
	_CBIT[2]=1:	M0 is generated when these alarms occur.					
2	Central mark	ker for basic dimension system of the control					
•	_CBIT[3]=0: _CBIT[3]=1:	Basic system is based on inches The basic system is metric					
	When starting up the measuring cycles, this bit has to be set according to the basic settings of the PLC (MD 10240). MD 10240: SCALING_SYSTEM_IS_METRIC must be set.						
	If modifying the longer matchin be converted t modification ar	e basic settings of the control results in _CBIT[3] no ng MD 10240, arrays _TP[], _WP[], _KB[] and _EV[] will he first time a measuring cycle is called after the nd a message will be output.					
	The data for to are also conve	ol measurement with rotating spindle (_CM[], _MFS[]) erted as are the data in the _SPEED[ ] field.					

2	Tool measur CYCLE982	rement and calibration in the WCS in
	_CBIT[5]=0:	The tool is measured and the tool measuring probe is calibrated in the machine coordinate system. The tool probe data are stored in the TP[1 field
	_CBIT[5]=1:	The tool can be measured and the tool probe calibrated in the active WCS. In this case, when calibrating and measuring, the same prerequisites must exist regarding the actual WCS. This means that tools with active transformation can also be measured, e.g. TRAANG. Notice: When calibrating and measuring, the _TP[] field is also used here.

#### Note:

From measuring cycles SW 6.3, function switchover via \_MVAR is available. A separate tool probe array is used there, the \_TPW[] field, for calibration/measurement in the WCS.

# Logging without output of the measuring cycle name and measuring variant

_CBIT[6]=0:	When logging, the measuring cycle name and the
	measuring variant are written into the log.
_CBIT[6]=1:	When logging, the measuring cycle name and
	the measuring variant are not output into the log.



# Offset of the mono probe setting

\_CBIT[8]=0: No compensation

\_CBIT[8]=1: If the workpiece measuring probe is a mono probe, its position (spindle position) is corrected by the angular value in \_CORA.



\_CBIT[11]=0: The standard log header is used. \_CBIT[11]=1 A user-defined log header is used.



# Feed and speed in CYCLE971

\_CBIT[12]=0: When measuring milling tools with the spindle rotating, the **measuring cycle** calculates the feed and speed.
\_CBIT[12]=1: The **user** enters the feed and speed in data field

(data array) \_MFS[ ].

-2	Deletion of t	he measuring cycle arrays in the GUD6
	_CBIT[13]=0: _CBIT[13]=1:	No deletion For the following measuring cycle call, the data fields (data arrays) _TP[], _TPW[], _WP[], _KB[], _EV[], _MV[] and _CBIT[13] are set to zero.
?	Length refer measuring c	ence of the workpiece probe in milling ycles
	_CBIT[14]=0:	The length 1 of the measuring probe referred to the <b>center</b> of the probe sphere (ball) should be entered into the tool offset.
	_CBIT[14]=1:	The length 1 of the measuring probe referred to the <b>end</b> of the probe sphere (ball) should be entered into the tool offset.
-?	Transfer of v CYCLE976	vorkpiece probe data into the tool offset in

\_CBIT[15]=0: No transfer

\_CBIT[15]=1: With measuring variant "Calibration with probe ball calculation", the determined "effective probe sphere diameter" (\_WP[k,0]), converted as radius value is entered into the radius geometry memory of the active workpiece measuring probe as tool offset.

## 9.2.5 Central strings

# Data block GUD6.DEF

	_ <b>SI[ ]</b> Central str	ings (only up t	o measuring c	cycles SW 6	5.2)
	·	Minimum inp	out limit: -		Maximum input limit: -
Changes valid after value assignment		Protection level: -		vel: -	Unit: -
Data type: STRING			1	Applies fro	m SW version:
Meaning:		ourropthy	not oppigned		Preset default
	_SI[0] _SI[1] _SI[2]	Software version currently not assigned			"6"



# Software version

Here you have to enter the first digit of the version of the NCU software on the control, e.g. for SW 05.xx.xx, enter 5.



#### 9.2.6 Channel-oriented values

#### Data block GUD5.DEF

	_ <b>EV[ ]</b> Empirical value	es			
	М	nimum input limit: -		Maximum in	put limit: -
Changes valid after value assignment		Protection level: -			Unit: -
Data type: REAL			Applies fro	om SW version:	3.2
Meaning:	Index "k" stand	Is for the number of th	e current arr	ay -1	Preset default
_EV[k] Number of empirical values					0
	NA) /F 1				

Mean values							
		Minimum inp	ut limit: -		Maximum inp	out limit:	-
Changes valid after value as		Protection level: -			Unit: -		
Data type: REAL				Applies from	SW version:	3.2	
Meaning:							Preset default
	Index "k" sta _MV[k]	ands for the n Mean valu	umber of the	current array			0

#### Data block GUD6.DEF

_EVMVNUM[] Number of empirical values and mean values							
Minimum input limit: 0					Maximum inp	out limit: -	
Changes valid after value assignment			Protection level: -			Unit: -	
Data type: INT				Applies from	SW version:	3.2	
Meaning:						Preset defa	ault
	_EVMVNUM	1[0] Ni	umber of emp	irical values		20	
	_EVMVNUM	1[1] Nu	umber of mea	n values		20	

	_SPEED[] Traversing ve	elocities for i	ntermediate p	ositioning			
	N	/linimum inp	ut limit: 0		Maximum inp	out limit: 10	0
Changes valid after value as	ssignment		Protection lev	/el: -		Unit: -	
Data type: REAL				Applies from	n SW version:	3.2	
Meaning:	_SPEED[0] Intermediate positioning in the measuring cycle with rapid traverse in % non-active collision monitoring (values between 1 and 100)					ith J plane	Preset default 50
	_SPEED[2]	with active collision monitoring Intermediate positioning in the measuring cycle with position					1000 ng
	Fast meas	ed axis with co suring feed	oiiision moni	toring active		1000 900	
Note: Adjust value to probe and machine used, if necessary! Too high values can damage the probe!							

# Traverse velocities for intermediate positioning \_SPEED[0] to [2]

Intermediate positions before the actual measuring block are calculated in the measuring cycles. This positions can

- be approached **with collision monitoring** (\_CHBIT[2]=1, default) or
- without collision monitoring (\_CHBIT[2]=0).

The appropriate velocities are used for approach as specified in these settings.

### With collision monitoring (\_CHBIT[2]=1):

With **\_SPEED[1]** the feedrate is applied to traversing in the plane and with **\_SPEED[2]** to traversing in the infeed axis (applicate). If the probe switches while approaching these intermediate positions, the movement is interrupted and the alarm "probe collision" is output.

#### Without collision monitoring (\_CHBIT[2]=0):

The intermediate positions are approached with the **percentage maximum axis velocity** (rapid traverse) specified in **\_SPEED[0]**. With \_SPEED[0]=0 and \_SPEED[0]=100 the maximum axis velocity applies. **Caution!** The user must ensure that collisions are ruled out.

# Measuring feed \_VMS, fast measuring feed \_SPEED[3]

Measurement is performed with the measuring feed of \_VMS. If \_VMS=0 and \_FA=1: 150 mm/min.

If \_VMS=0 and \_FA>1: 300 mm/min (see Section 2.3) If \_CHBIT[17]=1 and \_FA>1 probing is performed twice. The fast measuring feed \_SPEED[3] is used for the first probing. After the probe has switched, it is retracted by 2 mm. This is followed by actual measurement with the feedrate programmed in \_VMS.

# Measurement retraction velocity

Retraction from the measuring point is usually performed with the same speed (\_SPEED[1], [2]) or percentage of the rapid traverse as approach to the intermediate position (see above). However, while **collision monitoring** (\_CHBIT[2]=1) is active, it is possible with **CHBIT[16]=1** to switch to the percentage of rapid traverse in \_**SPEED[0**].



_TP_CF Tool probe type (manufacturer)						
	Mir	Minimum input limit: 0 Maximum input limi				
Changes valid after value assignment		Protection level: -		U	nit: -	
Data type: INTEGER Applies from S				m SW version: 6.	2	
Meaning:	Applies to tool r 0: r 1:	Applies to tool measurement with rotating spindle (CYCLE971 only)         0:       No specified validity         1:       TT130 (Heidenhain)         2:       TS27R (Renishaw)				

_MT_COMP Measurement result offset for tool measurement with rotating spindle (CYCLE971 only)						
		Minimum inp	out limit: 0	iput limit: 2		
Changes valid after value a	ssignment	Protection level: -		Unit: -		
Data type: INTEGER		Applies from SW version			6.2	
Meaning:	0: 1: 2:	no offset cycle-inter offset via	rnal offset (only active if user-defined offset table	Preset default 0		

	Mir	nimum input limit: -		Maximum in	put limit: -		
Changes valid after value as	Protection le	vel: -	•	Unit: mm			
Data type: REAL			Applies from	NSW version:	6.2		
Meaning:	Preset default Measurement result offset for tool measurement with rotating spindle (CYCLE971 only) _MT_EC_R[0,1]MT_EC_R[0,4] 4 tool radii from small to large _MT_EC_R[1,0]MT_EC_R[5,0] 5 peripheral velocities from low to high are specified _MT_EC_R[i,k] where i=15, k=14 20 offset values If _MT_COMP=2, then actual radius = measured radiusMT_EC_R[i,k] where i=15 next lowest table value for peripheral velocity and K=14 next lowest table value for tool radius						

	_MT_EC_L Offset table (CYCLE971	[6,5] for measurement res only)	ult offset for too	ol length measurement with rotating spindle		
		Minimum input limit:	-	Maximum input limit: -		
Changes valid after value as	ssignment	Protecti	on level: -	Unit: mm		
Data type: REALR		Applies fro	om SW version: 6.2			
Meaning:	Preset default Measurement result offset for tool measurement with rotating spindle (CYCLE971 only) _MT_EC_L[0,1]MT_EC_L[0,4] 4 tool lengths from small to large _MT_EC_L[1,0]MT_EC_L[5,0] 5 peripheral velocities from low to high are specified _MT_EC_L[i,k] where i=15, k=14 20 offset values If _MT_COMP=2, then actual length = measured lengthMT_EC_L[i,k] where i=15 next lowest table value for peripheral velocity and K=1. 4 next lowest table value for tool radius					


#### 9.2.7 Channel-oriented bits

#### Data block GUD6.DEF

	_CHBIT Channel bits		
	Mi	inimum input limit: - Maximum input limit: -	
Changes valid after value as	ssignment	Protection level: - Unit: -	
Data type: BOOLEAN		Applies from SW version: 3.2	
Meaning:		Pres	set default
	_CHBIT[0]	Measurement input for workpiece measurement 0: measurement input 1 1: measurement input 2	0
	_CHBIT[1]	Measurement input for tool measurement: 0: measurement input 1 1: measurement input 2	1
	_CHBIT[2]	Collision monitoring for intermediate positioning 0: OFF 1: ON	1
	_CHBIT[3]	Tool offset mode with tool measurement 0: first-time measurement (determining geometry) 1: remeasuring (determining wear)	0
	_CHBIT[4]	Mean value for workpiece measurement with automatic tool offset (_EVNUM>0) 0: no mean value derivation over several parts 1: with mean value formation and calculation	0
	_CHBIT[5]	Inclusion of empirical value (_EVNUM>0) 0: subtraction of actual value 1: addition to actual value	0
	_CHBIT[6]	Tool offset mode with workpiece measurement with automatic tool offset 0: Offset in wear 1: offset in geometry, delete wear	0
from meas. cycles - SW 6.3		For additive and setup offset and _CHBIT[8]=0: 0: offset in additive offset 1: offset in set-up offset, delete additive offset	
	_CHBIT[7]	Measured value offset in CYCLE994 0: use of trigger values of the probe _WP[k,1] 1: use of the active ball diameter of the probe _WP[k,0]	0
from meas. cycles - SW 6.3	_CHBIT[8]	Offset mode for workpiece measurement with automatic tool offset 0: additive, setup offset according to _CHBIT[6] 1: offset additive in set-up offset, independent of _CHBIT[6]	0
	_CHBIT[9]	currently not assigned	0
	_CHBIT[10]	Measuring result display 0: OFF 1: ON	0
	_CHBIT[11]	Acknowledgment measurement result screen with NC start 0: OFF (If _CHBIT[18]=0, the display is automatically deselected at end of cycle.) 1: ON (M0 is generated in the cycle.)	0
	_CHBIT[12]	currently not assigned	
	_CHBIT[13]	Coupling spindle position with coordinate rotation in active plane for workpiece measurement with multi probe 0: OFF 1: ON	0
	_CHBIT[14]	Adapt spindle positioning, if _CHBIT[13]=1 0: according to default 1: adapted angle values	0
	_CHBIT[15]	Number of measurements on failure to switch 0: up to 5 measurements 1: only 1 measurement	0



_CHBIT[16]       Retraction velocity from the measuring point       0         0: velocity as for intermediate positioning       0: velocity as for intermediate positioning ON: _CHBIT[2]=1)         _CHBIT[17]       Feed during measurement on intermediate positioning ON: _CHBIT[2]=1)         0: with feed in _VMS       0         0: with feed in _VMS       0         0: with feed in _CHBIT[11]       0         0: or ist measurement feed in _SPEED[3]       0         0: offect as set in _CHBIT[11].       0         1: only active if _CHBIT[11].       0         1: only active if CHBIT[11].       0         1: only active if CHBIT[11].       0         0: only active if CHBIT[11].       0         1: only active if CYCLE974 or CYCLE994!       0         0: on special treatment       0         0: on special treatment of V axis with G18       0         0: on special treatment       0         0: on special treatment of V axis active in the ordinate (X axis)         The tool offset is in the length that is active in the ordinate (X axis)         The tool offset is in the length batis active in the ordinate (X axis).         0       0         0: onspecial treatment the ordinate component (X axis).         0       0         0: offset additive in FINE       0				
CHBIT[17]       Feed during measurement 0: with feed in _VMS         1: on 1st measurement feed in _SPEED[3] on 2nd measurement feed in _VMS        CHBIT[18]       Static measurement result display         0: effect as set in _CHBIT[11].         1: only active if _CHBIT[11].         1: only active if _CHBIT[11].         1: only active if _CHBIT[11].         1: only active for CYCLE974 or CYCLE994!        CHBIT[19]         Special treatment         0: on special treatment         1: setpoint setting and parameterization (_SETVAL, _TUL, TLL, SZO) for the Y axis (applicate) as for the parameterization of the ordinate (X axis)         The tool offset is in the length that is active in the ordinate (X axis). (L1 usually) – as long as viaKNUM no other length settings are made. The ZO correction is applied in the specified ZO memory in the ordinate component (X axis).         Only active for CYCLE982I _CHBIT[20]       Suppression OFF 1: suppression OFF 1: suppression OFF         1: suppression OF       0         0: offset additive in FINE       0         0: offset additive in FINE       0         0: last measurement with reduced speed at _CBIT[12]=0         1: no speed reduction for tool measurement 0: last measurement with reduced speed at _CBIT[12]=0         1: no speed reduction       0         0: last measurement with reduced speed at _CBIT[12]=0         1: no recoding		_CHBIT[16]	Retraction velocity from the measuring point 0: velocity as for intermediate positioning 1: with percentage of rapid traverse velocity (_SPEED[0]) (only active with collision monitoring ON: CHBIT[2]=1)	0
_CHBIT[18]       Static measurement result display       0         0: effect as set in _CHBIT[11].       1: only active if _CHBIT[11]=0: Measuring result display remains until next measuring cycle is called       0         Only active for CYCLE974 or CYCLE994!       0       0         _CHBIT[19]       Special treatment of Y axis with G18       0         0: no special treatment       0       0: no special treatment       0         1: setpoint setting and parameterization (_SETVAL, _TUL, TLL, SZO) for the Y axis (applicate) as for the parameterization of the ordinate (X axis)       0         The tool offset is in the length that is active in the ordinate (X axis) (L1 usually) - as long as via _KNUM no other length settings are made. The ZO correction is applied in the specified ZO memory in the ordinate component (X axis).       0         Only active for CYCLE982!       _CHBIT[20]       Suppression of the starting angle positioning _STA1       0         0: suppression ON       0: suppression ON       0       0       0       0         _CHBIT[21]       ZO mode 0: offset additive in FINE       0       0       0       0         1: offset in COARSE, delete FINE       0       0       0       0       0       0         _CHBIT[21]       ZO mode 0: offset additive in FINE       0       0       0       0       0       0       0       0 <td< td=""><td></td><td>_CHBIT[17]</td><td>Feed during measurement 0: with feed in _VMS 1: on 1st measurement feed in _SPEED[3] on 2nd measurement feed in _VMS</td><td>0</td></td<>		_CHBIT[17]	Feed during measurement 0: with feed in _VMS 1: on 1st measurement feed in _SPEED[3] on 2nd measurement feed in _VMS	0
0: effect as set in _CHBIT[11].         1: only active if _CHBIT[11]=0:         Measuring result display remains until next         measuring cycle is called         Only active for CYCLE974 or CYCLE994!         _CHBIT[19]         Special treatment of Y axis with G18         0: no special treatment         1: setpoint setting and parameterization (_SETVAL, _TUL, _TLL, SZO) for the Y axis (applicate) as for the parameterization of the ordinate (X axis)         The tool offset is in the length that is active in the ordinate (X axis)         The tool offset is in the length that is active in the ordinate (X axis).         The ZO correction is applied in the specified ZO memory in the ordinate component (X axis).         Only active for CYCLE982!         _CHBIT[20]       Suppression of the starting angle positioning _STA1         0       0: suppression ON         Only active for CYCLE974, CYCLE977, CYCLE978, CYCLE979, CYCLE9997!         _CHBIT[21]       ZO mode         0       0: offset additive in FINE         1: offset in COARSE, delete FINE       0         0.101 active for CYCLE971!       0         _CHBIT[22]       Speed reduction for tool measurement         0       0         0       0         0.101 active for CYCLE971!       0         _CHBIT[22]       Speed reduction for		_CHBIT[18]	Static measurement result display	0
Only active for CYCLE974 or CYCLE994!         _CHBIT[19]       Special treatment of Y axis with G18       0         0       0: no special treatment       1: setpoint setting and parameterization (_SETVAL, _TUL, _TLL, SZO) for the Y axis (applicate) as for the parameterization of the ordinate (X axis)       The tool offset is in the length that is active in the ordinate (X axis) (X axis) (L1 usually) - as long as via _KNUM no other length settings are made. The ZO correction is applied in the specified ZO memory in the ordinate component (X axis).         Only active for CYCLE9821       _CHBIT[20]       Suppression of the starting angle positioning _STA1       0         0: offset additive in FINE       0       0: offset in COARSE, delete FINE       0         0. Only active for CYCLE974, CYCLE977, CYCLE978, CYCLE9997!       _CHBIT[21]       ZO mode       0         0: offset additive in FINE       1: offset in COARSE, delete FINE       0       0         0. Only active for CYCLE971!       _CHBIT[22]       Speed reduction for tool measurement       0         0: last measurement with reduced speed at _CBIT[12]=0       1: no speed reduction       0       1: ast measurement with reduced speed at _CBIT[12]=0         from meas. cycles - SW 6.3       Only active for CYCLE982!			<ul> <li>0: effect as set in _CHBIT[11].</li> <li>1: only active if _CHBIT[11]=0: Measuring result display remains until next measuring cycle is called</li> </ul>	
_CHBIT[19]       Special treatment of Y axis with G18       0         0: no special treatment       0: no special treatment       0: no special treatment         1: setpoint setting and parameterization (_SETVAL, _TUL, _TLL, SZO) for the Y axis (applicate) as for the parameterization of the ordinate (X axis)       The tool offset is in the length that is active in the ordinate (X axis) (L1 usually) – as long as via _KNUM no other length settings are made. The ZO correction is applied in the specified ZO memory in the ordinate component (X axis).         Only active for CYCLE982!       _CHBIT[20]       Suppression of the starting angle positioning _STA1       0         0: suppression OFF       1: suppression OFF       0       0       0       0         1: suppression ON       0: offset additive in FINE       0       0       0       0       0         0: offset additive in GARSE, delete FINE       0       0: offset in COARSE, delete FINE       0		Only active	for CYCLE974 or CYCLE994!	
0: No Special treatment         1: setpoint setting and parameterization (_SETVAL, _TUL, _TLL, SZO) for the Y axis (applicate) as for the parameterization of the ordinate (X axis) The tool offset is in the length that is active in the ordinate (X axis) (L1 usually) – as long as via _KNUM no other length settings are made. The ZO correction is applied in the specified ZO memory in the ordinate component (X axis).         Only active for CYCLE982! _CHBIT[20]       Only active for CYCLE982! _CHBIT[21]       0         0: suppression OFF 1: suppression OFF 1: suppression ON       0         0: Only active for CYCLE974, CYCLE977, CYCLE978, CYCLE979, CYCLE9997! _CHBIT[21]       0         0: Offset additive in FINE 1: offset in COARSE, delete FINE       0         0: last measurement with reduced speed at _CBIT[12]=0 1: no speed reduction 0: last measurement with reduced speed at _CBIT[12]=0 1: no speed reduction       0         from meas. cycles - SW 6.3       Only active for CYCLE982! _CHBIT[23]       Recoding of tool point direction during tool measurement 0: last measurement with reduced speed at _CBIT[12]=0 1: no speed reduction       0         from meas. cycles - SW 6.3       Only active for CYCLE982! _CHBIT[24]       Recoding of tool point direction during tool measurement 0: no recoding 1: internal recording (tool point direction mirroring about X)       0		_CHBIT[19]	Special treatment of Y axis with G18	0
Intervision of the tool offset is in the length that is active in the ordinate (X axis) (L1 usually) – as long as via _KNUM no other length settings are made. The ZO correction is applied in the specified ZO memory in the ordinate component (X axis).         Only active for CYCLE982! _CHBIT[20]       Only pression of the starting angle positioning _STA1       0         0: suppression OFF 1: suppression ON       0       0         Only active for CYCLE974, CYCLE977, CYCLE978, CYCLE979, CYCLE9997!       0         _CHBIT[21]       ZO mode       0         0: offset additive in FINE 1: offset in COARSE, delete FINE       0         Only active for CYCLE971! _CHBIT[22]       Speed reduction for tool measurement 0: last measurement with reduced speed at _CBIT[12]=0         1: no speed reduction       1: no speed reduction       0         from meas. cycles - SW 6.3       Only active for CYCLE982! _CHBIT[23]       Recoding of tool point direction during tool measurement 0: last measurement with reduced speed at _CBIT[12]=0         1: internal recording 1: internal recording (tool point direction mirroring about X)       0			1: setpoint setting and parameterization (_SETVAL, _TUL, _TLL, SZO) for the Y axis (applicate) as for the parameterization of the ordinate (X axis)	
ivia_KNUM no other length settings are made. The ZO correction is applied in the specified ZO memory in the ordinate component (X axis).         Only active for CYCLE982!        CHBIT[20]       Suppression of the starting angle positioning _STA1         0       0: suppression OFF         1: suppression ON         Only active for CYCLE974, CYCLE977, CYCLE978, CYCLE979, CYCLE9997!        CHBIT[21]       ZO mode         0: offset additive in FINE         1: offset in COARSE, delete FINE         Only active for CYCLE971!         _CHBIT[22]       Speed reduction for tool measurement         0: last measurement with reduced speed at _CBIT[12]=0         1: no speed reduction         from meas. cycles - SW 6.3       Only active for CYCLE982!         CHBIT[23]       Recoding of tool point direction during tool measurement       0         0: no recoding       1: internal recording (tool point direction mirroring about X)         CHBIT[24]       Channel-oriented marker for basic dimension system       1			I he tool offset is in the length that is active in the ordinate	
The ZO correction is applied in the specified ZO memory in the ordinate component (X axis).         Only active for CYCLE982! _CHBIT[20]       Suppression of the starting angle positioning _STA1       0         0: suppression OFF       1: suppression OFF       0         1: suppression ON       Only active for CYCLE974, CYCLE977, CYCLE978, CYCLE979, CYCLE9997!       0         CHBIT[21]       ZO mode       0       0         0: offset additive in FINE       0       0       0         1: offset in COARSE, delete FINE       0       0       0         Only active for CYCLE971!       CHBIT[22]       Speed reduction for tool measurement       0         0: last measurement with reduced speed at _CBIT[12]=0       1: no speed reduction       0       0: no recoding         from meas. cycles - SW 6.3       Only active for CYCLE982!       CHBIT[23]       Recoding of tool point direction during tool measurement       0         0: no recoding       1: internal recording (tool point direction mirroring about X)       1			via KNUM no other length settings are made	
in the ordinate component (X axis).         Only active for CYCLE982!         _CHBIT[20]       Suppression of the starting angle positioning _STA1         0:       suppression OFF         1:       suppression OF         Only active for CYCLE974, CYCLE977, CYCLE978, CYCLE979, CYCLE9997!         _CHBIT[21]       ZO mode         0:       offset additive in FINE         1:       offset in COARSE, delete FINE         Only active for CYCLE971!       CHBIT[22]         CHBIT[22]       Speed reduction for tool measurement         0:       last measurement with reduced speed at _CBIT[12]=0         1:       no speed reduction         from meas. cycles - SW 6.3       Only active for CYCLE982!         CHBIT[23]       Recoding of tool point direction during tool measurement         0:       no recoding         1:       internal recording (tool point direction mirroring about X)			The ZO correction is applied in the specified ZO memory	
Only active for CYCLE982! _CHBIT[20]       Suppression of the starting angle positioning _STA1       0         0: suppression OFF       1: suppression OFF       0         1: suppression ON       Only active for CYCLE974, CYCLE977, CYCLE978, CYCLE979, CYCLE9997!         _CHBIT[21]       ZO mode       0         0: offset additive in FINE       0         1: offset in COARSE, delete FINE       0         Only active for CYCLE971!       CHBIT[22]         CHBIT[22]       Speed reduction for tool measurement         0: last measurement with reduced speed at _CBIT[12]=0       1: no speed reduction         1: no speed reduction       0         from meas. cycles - SW 6.3       Only active for CYCLE982!         CHBIT[23]       Recoding of tool point direction during tool measurement         0: no recoding       0: no recoding         1: internal recording (tool point direction mirroring about X)       CHBIT[24]			in the ordinate component (X axis).	
CHBIT[20]       Suppression of the starting angle positioning _STA1       0         0: suppression OFF       0       0: suppression OFF         1: suppression ON       Only active for CYCLE974, CYCLE977, CYCLE978, CYCLE979, CYCLE99997!        CHBIT[21]       ZO mode       0         0: offset additive in FINE       0         1: offset in COARSE, delete FINE       0         Only active for CYCLE971!      CHBIT[22]         Speed reduction for tool measurement       0         with rotating spindle and multiple measurement       0         0: last measurement with reduced speed at _CBIT[12]=0       1: no speed reduction         1: no speed reduction       0         from meas. cycles - SW 6.3       Only active for CYCLE982!         CHBIT[23]       Recoding of tool point direction during tool measurement       0         0: no recoding       1: internal recording (tool point direction mirroring about X)       1		Only active	for CYCLE982!	•
Only active for CYCLE974, CYCLE977, CYCLE978, CYCLE979, CYCLE99997!           _CHBIT[21]         ZO mode         0           0: offset additive in FINE         0           1: offset in COARSE, delete FINE         0           Only active for CYCLE971!         _CHBIT[22]           Speed reduction for tool measurement         0           with rotating spindle and multiple measurement         0           0: last measurement with reduced speed at _CBIT[12]=0         1: no speed reduction           1: no speed reduction         0           from meas. cycles - SW 6.3         Only active for CYCLE982!           CHBIT[23]         Recoding of tool point direction during tool measurement           0: no recoding         1: internal recording (tool point direction mirroring about X)           CHBIT[24]         Channel-oriented marker for basic dimension system		_CHBIT[20]	0: suppression OFF 1: suppression ON	0
_CHBIT[21]       ZO mode       0         0: offset additive in FINE       1: offset in COARSE, delete FINE       0         Only active for CYCLE971!       _CHBIT[22]       Speed reduction for tool measurement       0         with rotating spindle and multiple measurement       0       0       0         0: last measurement with reduced speed at _CBIT[12]=0       0       0       0         1: no speed reduction       0       0       0       0         from meas. cycles - SW 6.3       Only active for CYCLE982!       0       0       0         CHBIT[23]       Recoding of tool point direction during tool measurement       0		Only activ	e for CYCLE974, CYCLE977, CYCLE978, CYCLE979, CYCL	E9997!
from meas. cycles - SW 6.3 CHBIT[23] CHBIT[23] Recoding of tool point direction during tool measurement 0 CHBIT[24] Channel-oriented marker for basic dimension system 1		_CHBIT[21]	ZO mode	0
Only active for CYCLE971!       _CHBIT[22]       Speed reduction for tool measurement       0         with rotating spindle and multiple measurement       0       0         with rotating spindle and multiple measurement       0         0: last measurement with reduced speed at _CBIT[12]=0       1: no speed reduction         from meas. cycles - SW 6.3       Only active for CYCLE982!         _CHBIT[23]       Recoding of tool point direction during tool measurement       0         0: no recoding       0: no recoding       0: no recoding         1: internal recording (tool point direction mirroring about X)       CHBIT[24]       Channel-oriented marker for basic dimension system			0: offset additive in FINE 1: offset in COARSE, delete FINE	
_CHBIT[22]       Speed reduction for tool measurement       0         with rotating spindle and multiple measurement       0         0: last measurement with reduced speed at _CBIT[12]=0       1: no speed reduction         from meas. cycles - SW 6.3       Only active for CYCLE982!         _CHBIT[23]       Recoding of tool point direction during tool measurement       0         0: no recoding       0: no recoding       0: no recoding         1: internal recording (tool point direction mirroring about X)       1		Only active	for CYCLE971!	•
from meas. cycles - SW 6.3 CHBIT[23] Recoding of tool point direction during tool measurement 0 0: no recoding 1: internal recording (tool point direction mirroring about X) CHBIT[24] Channel-oriented marker for basic dimension system 1		_CHBIT[22]	with rotating spindle and multiple measurement 0: last measurement with reduced speed at _CBIT[12]=0	0
from meas. cycles - SW 6.3 Only active for CYCLE982! CHBIT[23] Recoding of tool point direction during tool measurement 0 0: no recoding 1: internal recording (tool point direction mirroring about X) CHBIT[24] Channel-oriented marker for basic dimension system 1			1: no speed reduction	
CHBIT[23] Recoaling or tool point direction during tool measurement 0 0: no recoding 1: internal recording (tool point direction mirroring about X) CHBIT[24] Channel-oriented marker for basic dimension system 1	from meas. cycles - SW 6.3	Only active	for CYCLE982!	0
CHBIT[24] Channel-oriented marker for basic dimension system 1		_CHBI1[23]	0: no recoding 1: internal recording 1: internal recording (tool point direction mirroring about X)	0
		CHBIT[24]	Channel-oriented marker for basic dimension system	1
(For deviations from MD 10240, cycle parameters are internally converted and this bit is again set.)		_ · · · · · · · · · · · · · · · · · · ·	(For deviations from MD 10240, cycle parameters are interna converted and this bit is again set.)	lly
U: The basic system is inch 1: The basic system is metric			<ul><li>U: I ne basic system is inch</li><li>1: The basic system is metric</li></ul>	

10.04



=?	Tool offset m automatic to	node for workpiece measurement with ol offset
	_CHBIT[6]=0:	The determined offset value is added in the <b>wear</b> memory (length and radius) of the specified tool and is incorporated in the D number specified using KNUM.
	_CHBIT[6]=1:	The length and the radius of the specified tool is corrected by the determined offset value and entered into the appropriate <b>geometry</b> memory. The corresponding wear memory is calculated and then set to zero.
	From measurin	ng cycles SW 6.3, an offset in the set-up/additive
	offset can also	be programmed. The type of calculation is also
	defined by _CH	IBIT[6] and _CHBIT[8]:
	_CHBIT[6]=0	The calculated offset value is included additively in
	_CHBIT[8]=0	the appropriate additive offset value memory.
	_CHBIT[6]=1	The determined offset value is incorporated, taking into account
	_CHBIT[8]=0	the appropriate additive offset value, into the setting-up offset memory and the additive offset memory is deleted.
	_CHBIT[8]=1	Independent of _CHBIT[6], the determined offset value is additively incorporated into the appropriate setting-up offset memory.
- 7	Measured va	lue offset in CYCLE994
	_CHBIT[7]=0:	In order to determine the actual value, the <b>trigger values</b> of the measuring probe, saved in the _WP[_PRNUM-1,14) are used.
	_CHBIT[7]=1:	In order to determine the actual value, the effective <b>diameter</b> of the measuring probe, saved in the _WP[_PRNUM-1,0] is used.



## Offset mode for workpiece measurement with automatic tool offset

\_CHBIT[8]: Explanation --> see \_CHBIT[6]

#### Measuring result display

\_CHBIT[10]=0: OFF \_CHBIT[10]=1: ON – After measuring or calibrating, a measurement result screen is automatically displayed.

=?	Acknowledgr start	nent measurement result screen with NC
	_CHBIT[11]=0:	The measurement result screen is automatically de-selected at the end of the cycle. Also _CHBIT[18] must be =0, otherwise the effect as described for <b>_CHBIT[18]</b> =1 is obtained.
	_CHBIT[11]=1:	After the measurement result screen is displayed, the cycle generates M0. The measurement cycle is continued and the screen is de-selected after the NC start.
-	Static measu	rement result display
= {	_CHBIT[18]=0: _CHBIT[18]=1:	Effect is defined by <b>_CHBIT[11]</b> . The measurement result screen display is kept until the next measuring cycle is called. The NC program processing is not interrupted. <b>_CHBIT[10]</b> must be set, <b>_CHBIT[11]</b> must be 0!
=?	Coupling spin active plane f	ndle position with coordinate rotation in for workpiece measurement with multi probe
	_CHBIT[13]=0:	OFF
	CHBIT[13]=1.	There is no coupling between the spindle position and active coordinate rotation in the plane.
	Notice:	When multi-probes are being used, the spindle is positioned depending on the active coordinate rotation in the plane (rotation around the applicate (feed axis)) so that the same positions of the probe sphere are probed when calibrating and measuring. Note: Pay attention to <b>_CHBIT[14]</b> ! If additional rotations are active in the other planes/ axes, then this function is not effective!
_?	Adapt spindle	e positioning, if _CHBIT[13]=1
	_CHBIT[14]=0:	Spindle positioning is performed acc. to default. Angle of coordinate rotation in the plane 0°: Spindle positioning 0° Angle of coordinate rotation in the plane 90°: Spindle positioning 270°
	_CHBIT[14]=1:	Spindle positioning 270 Spindle pos. is performed in opposite direction. Angle of coordinate rotation in the plane 0°: Spindle positioning 0° Angle of coordinate rotation in the plane 90°: Spindle positioning 90°
8	<ul><li>A coordinate ro</li><li>one rotation</li><li>one rotation</li><li>one rotation</li></ul>	tation in the active plane is: a around the Z axis with G17, a around the Y axis with G18 or a around the X axis with G19,



Number of measurements on failure to switch CHBIT[15]=0: A maximum of 5 measuring attempts are made before alarm "Probe not switching" is generated. CHBIT[15]=1: After one unsuccessful measurement attempt, the fault "measuring sensor does not switch" is generated. Retraction velocity from the measuring point CHBIT[16]=0: The retraction from the measuring point is realized with the same velocity as for an intermediate positioning operation. CHBIT[16]=1: The retraction velocity is realized with the percentage rapid traverse velocity, defined in SPEED[0] and is only effective when collision monitoring is active (\_CHBIT[2]=1). Feed during measurement CHBIT[17]=0: Measurement is performed with the feed programmed in VMS. Pay attention to consequences if VMS=0! \_CHBIT[17]=1: Initially, the axis traverses with the measuring feed \_SPEED[3], after switching, there is a retraction of 2 mm from the measuring position and the actual measurement starts with the feed from VMS. Measurement with the feed from SPEED[3] is only realized for a measurement distance/ travel > 2 mm. Static measurement result display CHBIT[18]: Explanation -> see CHBIT[10], CHBIT[11]



## Special treatment of Y axis with G18 in CYCLE974 or CYCLE994

\_CHBIT[19]=0: No special treatment for Y axis (applicate) \_CHBIT[19]=1: Setpoint setting and parameterization (\_SETVAL, \_TUL, \_TLL, SZO) for the Y axis (applicate) are performed as for parameterization of the ordinate (X axis). The tool offset is applied to the length that is active in the ordinate (X axis) (usually L1), as long as other lengths are not specified by \_KNUM. The ZO correction is applied in the specified ZO memory in the ordinate component (X axis). 10.04

=?	Suppression CYCLE982	of the starting angle positioning _STA1 in
	_CHBIT[20]=0:	For certain measuring variants, the milling spindle is positioned with STA1.
	_CHBIT[20]=1:	When measuring milling tools, for basic measuring variants, it is possible to suppress positioning of the milling spindle to the value of the starting angle _STA1. This is possible for the following measuring variants to measure milling tools: _MVAR=xxx001 (where x: 0 or 1, no other values)
=?	ZO mode in C CYCLE997	CYCLE974, CYCLE977, CYCLE978, CYCLE979,
	_CHBIT[21]=0:	The offset is applied additively in FINE, if MD 18600: MM_FRAME_FINE_TRANS=1, otherwise in COARSE.
	_CHBIT[21]=1:	The offset is applied in COARSE. FINE is taken into account and is then subsequently deleted.
_?	Speed reduct	tion in tool measurement in CYCLE971
	_CHBIT[22]=0:	When measuring tools with rotating spindle and when the speed ( <b>_CBIT[12]=0</b> ) is calculated in the cycle, for multiple measurements, the last measurement is carried-out at a reduced speed.
	_CHBI1[22]=1:	For multiple measurements with the spindle rotating and calculation in the cycle, the speed remains constant.
_7	Re-coding of	the cutting edge position for tool
	measuremen SW 6.3)	t in CYCLE982 (from measuring cycles
	_CHBIT[23]=0: _CHBIT[23]=1:	Standard setting, no recoding Internal recording, tool point direction mirroring about X axis (tool turret -180 degrees, Z not mirrored)
_7	Channel-orie	nted marker for basic dimension system
= {	_CHBIT[24]=0: _CHBIT[24]=1:	Basic system is based on inches The basic system is metric
	When starting u according to the (MD 10240: SC If modifying the longer matching values are conv	up the measuring cycles, this bit has to be set e basic settings of the PLC (MD 10240). ALING_SYSTEM_IS_METRIC) must be set. basic settings of the control results in _CBIT[24] no g MD 10240, the channel-specific dimensioned verted in each channel the first time a measuring

cycle is called.



#### 9.3 Data for measuring in JOG

#### 9.3.1 Machine data for ensuring ability to function

11602	ASUP_START_M	ASUP_START_MASK			
MD number	Ignore stop cond	Ignore stop conditions for ASUB			
Default setting: 0 for measuring in JOG: (Bit0=1)	1, 3	Minimu	m input limit: 0	Maximum input limit: 3	
Changes effective after F	POWER ON		Protection level: 2/7	Unit: -	
Data type: DWORD			Applies from	m SW version: SW 4.1	
Meaning:	Bit 0: 1 ASUB sta	art possik	ble in JOG		
11604	ASUP_START_F	PRIO_LE	EVEL		
MD number	Priorities for ASU	IP_STAF	RT_MASK effective		
Default setting: 0 for measuring in JOG:	1 - 64H	Minimu	m input limit: 0	Maximum input limit: 64H	
Changes effective after F	POWER ON		Protection level: 2/7	Unit: HEX	
Data type: INT			Applies from	m SW version: SW 4.1	
Meaning:	"ASUP_START_	MASK" i	ncluded from ASUB priori	ity "64H" to ASUB priority 1.	
20050	AXCONF_GEOA	X_ASSI	GN_TAB[i]		
MD number	Assignment of ge [i = Geometry at	eometry a xis index	axes to channel axes ]: 0 2		
Default setting: 1, 2, 3 for measuring in JOG:		Minimu	m input limit: 0	Maximum input limit: 15	
Change applies after PO	IST DE present. IWER ON:		Protection level: 2/7	Unit:	
Data type BYTE			Applies from	m SW version: SW 1	
Meaning:	The MD specifies are mapped. Inde	The MD specifies channel axes onto which the axes of the Cartesian coordinate system (WCS) are mapped. Index i assumes values 0, 1, 2 and refers to the 1st to 3rd geometry axis.			
20110		MACK			
MD number	Define control de	fault sett	ting after power-up and R	ESET	
Default setting: 0 for measuring in JOG: (Bit0=1. Bit2=1. Bit6=1.	Default setting: 0 for measuring in JOG: min. 4045H (9:40-1 Bit2-1 Bit2-1 Bit2-1 Bit2-1)				
Changes effective after F	RESET		Protection level: 2/7	Unit: HEX	
Data type: DWORD		Applies from SW version: SW 2			
Meaning:	Bit 0: 1 tool lengt Bit 2: 1 Bit 6: 1 Bit 6: 1 The t	Bit 0: 1 tool length offset active Bit 2: 1 Bit 6: 1 Bit 6: 1 Bit 14: 1 The current setting of the basic frame is retained.			
20112 MD number	Define control de	START_MODE_MASK Define control default setting after part program start			
Default setting: 400H for measuring in JOG 4 (Bit6=0)	100H	Minimu	m input limit: 0	Maximum input limit: 07FFFH	
Changes effective after F	RESET		Protection level: 2/7	Unit: HEX	
Data type DWORD			Applies from	m SW version: SW 3.2	
Meaning:	Bit 6: 0 active too	l length	correction is retained		

10.04

	·					
20310	TOOL_MANAGEMENT_MASK					
MD number	Channel-specific activation of tool management					
	Note: MD only relevant if tool management is active (Bit0=1)					
Default setting: 0		Minimur	n input limit: (	)	Maximum input limit:	0FFFFhex
if Bit0=1						
for measuring in JOG: min	n. <b>4001H</b>					
Changes effective after PO	WER ON		Protection le	vel: 2/7	Unit: HE	X
Data type DWORD				Applies from	SW version: SW 2	
Meaning:	Bit 0: 1 Tool management active Bit 14: 1 Automatic tool change on RESET and START according to MD 20110: RESET_MODE_MASK					
24006	CHSFRAME_RE	CHSFRAME RESET MASK				
MD number	Reset behavior of the channel-specific system frame, actual value setting and scratching (basis reference)					
Default setting: 0 for measuring in JOG: min (Bit0=1)	g: 0 Minimum g in JOG: min. 1			)	Maximum input limit:	07FFFH
Changes effective after RES	SET		Protection le	vel: 2/7	Unit: HE	X
Data type: DWORD	Data type: DWORD Applies from SW version: SW 2					
Meaning:	Bit 0: 1 system fra	ame for I	PRESET and	scratching (b	asis reference) is acti	ve after reset.
	1					
24007	CHSFRAME_RE	SET_CL	EAR_MASK			
MD number	Clear system frames on reset Note: MD only relevant if frames are configured (MD 28082 SYSTEM_ERAME_MASK)					
Default setting: 0 for measuring in JOG: Bit 0=0				)	Maximum input limit:	07FFFH
Changes effective after RES	SET		Protection level: 2/7		Unit: HE	X
Data type: DWORD				Applies from	SW version: SW 2	
Meaning:	Bit 0:0 system frame for PRESET and scratching (basis reference) is not deleted at reset.					



#### 9.3.2 Modifying the GUD7 data block

Function

Measurement in JOG requires data definitions in data block GUD7.DEF.

Different procedures are followed depending on the configuration of the overall system.

- If the "ShopMill" programming interface is installed on the control, then all of the required GUD data are already active. No reloading or additional modification of definition blocks is required.
- 2. From SW 6.02.17 of the measuring cycle package, the definition files GUD7.DEF and SMAC.DEF are split-up. Now a differentiation is made between basis definition files and application-specific definition files. All applications, measuring cycles, technology cycles, ShopMill etc. only incorporate their own application-specific definitions. For this purpose, new cycle files GUD7\_xxx.DEF and SMAC\_xxx.DEF have been introduced, and are located in the HMI data management, in the definition directory DEF.DIR. This means, when measuring in JOG, from measuring cycles SW 6.02.17, the data block GUD7\_MC.DEF supplied must be used. A secondary condition/limitation is that file GUD7.DEF must be used also for measuring cycles from SW 6.02.17. For installation, refer to Section 10.1.

### Commissioning "measuring in JOG" for the first time, up to measuring cycles SW 6.2.16

3. In this case, definition block GUD7.DEF must be modified. Select definition file GUD7.DEF in menu "Services" in directory "Definitions" with the arrow keys and unload it by pressing the softkey "Unload". Before modifying data in GUD7, the "old" data should be backed-up (archived). Then open file GUD7.DEF by pressing the Enter key. In the section "Measure", remove the semicolons at the beginning of each definition line with the DEL key. This affects the definition lines as from the comment "Definitions for measurement in JOG and ShipMill" (starting with data item E\_MESS\_IS\_METRIC).

DEF CHAN INT	E_MESS_IS_METRIC=1
DEF CHAN INT	E_MESS_IS_METRIC_SPEZ_VAR=1
DEF CHAN BOOL	E_MESS_MS_IN=0
DEF CHAN BOOL	E_MESS_MT_IN=1
DEF CHAN REAL	E MESS D=5
DEF CHAN REAL	E MESS D M=50
DEF CHAN REAL	E MESS D L=2
DEF CHAN REAL	E MESS D R=1
DEF CHAN REAL	E_MESS_FM=300
DEF CHAN REAL	E MESS F=2000
DEF CHAN REAL	E_MESS_FZ=2000
DEF CHAN REAL	E_MESS_MAX_V=100
DEF CHAN REAL	E_MESS_MAX_S=1000
DEF CHAN REAL	E_MESS_MAX_F=20
DEF CHAN REAL	E_MESS_MIN_F=1
DEF CHAN REAL	E_MESS_F_FAK1=10
DEF CHAN REAL	E_MESS_F_FAK2
DEF CHAN REAL	E_MESS_MIN_D=0.01
DEF CHAN INT	E_MESS_MT_TYP[3]=SET(0,0,0)
DEF CHAN INT	E_MESS_MT_AX[3]=SET(133,133,133)
DEF CHAN REAL	E_MESS_MT_DL[3]
DEF CHAN REAL	E_MESS_MT_DR[3]
DEF CHAN REAL	E_MESS_MT_DZ[3]=SET(2,2,2)
DEF CHAN INT	E_MESS_MT_DIR[3]=SET(-1,-1,-1)
DEF CHAN REAL	E_MESS_MT_D=10
DEF CHAN REAL	E_MESS_MT_FM=100
DEF CHAN INT	E_MESS_MT_CF
DEF CHAN INT	E_MESS_MT_COMP
DEF CHAN REAL	E_MESS[3]
DEF CHAN REAL	E_MEAS
DEF CHAN BOOL	E MESS SETT[10]

#### Possibility of minimizing the memory requirements

4. The number of available data fields (data arrays) regarding the measuring probes that can be connected, can be adapted by the machinery manufacturer to the specific relationships and situations. When supplied, 3 data fields (data arrays) are available for tool measuring probes, workpiece measuring probes and calibration gauging blocks. If, for example, in the specific application case, only one data field (data array) is required for the workpiece and the tool measuring probe, and there is no gauging block available, then the appropriate parameters can be adapted in data block GUD6 and in GUD7\_DEF and GUD7\_MC.DEF. The details can be taken from the following example.

#### Example:

There is only one tool and workpiece measuring probe on a specific milling machine. The tool is to be only gauged in the interpolation plane G17. In order to minimize the NC memory requirement in the SINUMERIK, the following changes are required:

#### - GUD7.DEF or GUD7 MC.DEF

DEF CHAN INT	E_MESS_MT_TYP[1]=SET(0)
DEF CHAN INT	E_MESS_MT_AX[1]=SET(133)
DEF CHAN REAL	E_MESS_MT_DL[1]
DEF CHAN REAL	E_MESS_MT_DR[1]
DEF CHAN REAL	E_MESS_MT_DZ[1]=SET(2)
DEF CHAN INT	E_MESS_MT_DIR[1]=SET(-1)
DEF CHAN INT	E_MESS_CALL_D[1]=SET(0) from measuring cycles SW 6.3
DEF CHAN INT	E_MESS_CALL_L[1]=SET(0) from measuring cycles SW 6.3

- GUD6.DEF N10 DEF NCK INT \_CVAL[4]=(**1**,**1**,0,0) ; 1x tool and 1x workpiece measuring probe data set N11 DEF NCK REAL \_TP[**1**,10]=(0,0,0,0,0,0,133,0,2) N12 DEF NCK REAL \_WP[**1**,11] ;N13 DEF NCK REAL \_KB[3,7] ;N111 DEF NCK REAL \_TPW[3,10]

The definition blocks should be opened in the HMI area Services\Definitions using the operator panel "INPUT key" and after editing, re-closed using the vertical softkey "Close editor"; i.e. to save and to activate.

When supplied, the following basic settings are active:

E_MESS_IS_METRIC=1	INT	All dimensioned data are metric
E_MESS_IS_METRIC_SPEZ_VAR=1		
E_MESS_MS_IN=0	BOOL	Work piece probe connected to measuring
		input 1
E_MESS_MT_IN=1	BOOL	Tool probe connected to measuring input 2
E_MESS_D=5	REAL	Internal data for measuring in JOG not
		relevant
E_MESS_D_M=50	REAL	Measuring path for manual measuring [mm]
		(in front of and behind the measuring point)
E_MESS_D_L=2	REAL	Measuring path for length measurement [mm]
		for tool measurement
		(in front of and behind the measuring point)
E_MESS_D_R=1	REAL	Measuring path for radius measurement [mm]
		for tool measurement
		(in front of and behind the measuring point)
E_MESS_FM=300	REAL	Measuring feedrate [mm/min] for workpiece
		measurement and calibration
E_MESS_F=2000	REAL	Plane feedrate for collision monitoring
		[mm/min]
E_MESS_FZ=2000	REAL	Infeed feedrate for collision monitoring
		[mm/min]
E_MESS_CAL_D=[3]	REAL	Diameter, calibration ring [mm] 0
(from measuring cycles SW 6.3)		
E_MESS_CAL_L=[3]	REAL	Calibration height in the feed axis
(from measuring cycles SW 6.3)		(referred to the WCS) [mm] 0
E_MESS_MAX_V=100	REAL	Max. peripheral speed for measuring with
		rotating spindle [m/min]
E_MESS_MAX_S=1000	REAL	Max. spindle speed for measuring with
		rotating spindle [rpm]
E_MESS_MAX_F=20	REAL	Max. feedrate for measuring with rotating
		spindle [mm/min]
E_MESS_MIN_F=1	REAL	Min. feedrate for measuring with rotating
		spindle on 1st probing [mm/min]
E_MESS_F_FAK1=10	REAL	On tool measurement with rotating spindle,
		traversal with 10 times measuring feed is
		performed in the 1st probing (limitation by
		E_MESS_MAX_F)



10.04

Data Description

E_MESS_F_FAK2=0	REAL	For tool measurement with rotating spindle, 2nd probing is performed with measuring feed. There is no 3rd probing.
E_MESS_MIN_D=0.01	REAL	Measuring accuracy for measuring with rotating spindle [mm/min]
E_MESS_MT_TYP[3]=SET(0,0,0)	INT	3 arrays for tool measuring probe; tool measuring probe type; cube
E_MESS_MT_AX[3]=SET(133,133,133)	INT	Permissible axis directions for tool probe in X and Y in plus and minus direction, in Z in minus direction only
E_MESS_MT_DL[3]	REAL	Active diameter of tool measuring probe for length measurement 0
E_MESS_MT_DR[3]	REAL	Active diameter of tool measuring probe for radius measurement 0
E_MESS_MT_DZ[3]=SET(2,2,2)	REAL	Distance between upper edge of tool probe and lower edge of tool probe [mm] when measuring the tool radius 2
E_MESS_MT_DIR[3]=SET(-1,-1,-1)	INT	Approach direction in plane of tool measuring probe for tool measurement (minus direction in 1st plane axis) -1
E_MESS_MT_D=10	REAL	Calibrate measurement path for tool probe and tool measurement with motionless spindle (before and after expected switching position)
E_MESS_MT_FM=100	REAL	Calibrate measuring feed for tool probe and tool measurement with motionless spindle
E_MESS_MT_CF=0	INT	No tool probe make (manufacturer) specified
E_MESS_MT_COMP=0	INT	No offset of the measurement result on tool measurement with rotating spindle
E_MESS[3]	REAL	Internal values
E_MEAS	REAL	Internal values
E_MESS_RETT	FRAME	Internal values
E_MESS_SETT[10]	BOOL	Field for settings
E MESS AM	BOOL	Internal values

#### Attention!

Arrays E\_MESS\_MT\_DL[] and E\_MESS\_DR[] (active diameter, width of tool measuring probe for length/radius measurement) must be assigned.

	E_MESS_S Field for set	ETT[] tings					
		Minimum inp	out limit: 0		Maximum inp	out limit: 1	
Changes valid after value as	ssignment		Protection lev	el: -		Unit: -	
Data type: BOOL				Applies from	SW version:	6.2	
Meaning:	E_MESS_SE E_MESS_SE	ETT[0] 0: S 1: S ETT[19] ir	pindle coupling pindle position nternal date	g with coord on cycle ca	inate rotation Il is initial pos	ition	Preset default 0



#### 9.3.3 Settings in data block GUD6

### 

### Function

The channel specific arrays \_JM\_I[ ], and \_JM\_B[ ] in data block GUD6 are used for adaptation to the requirements of the machine

N92 DEF CHAN INT \_JM\_I[10]=SET(0,1,1,17,100,0,0,0,0,0)

	_JM_I[] INT value fiel	d for JOG measurement		
	ľ	/linimum input limit: -	Maximum input limit: -	
Changes valid after value as	ssignment	Protection level: -	Unit: -	
Data type: INT		Applie	s from SW version: 5.3	
Meaning:				Preset default
	_JM_I[0]	Setting for the workpiece p 0: set by _JM_I[1] 1: set as with ShopMill G17 plane Workpiece prot G18 plane Workpiece prot G19 plane Workpiece prot	probe number pe number 1 pe number 2 pe number 3	0
	_JM_I[1]	Probe number and probe to Only active if JM I[0]=0	type for workpiece measurement	nt 1
	_JM_I[2]	Probe number for tool mea	asurement	1
	_JM_[[3]	plane 17: measurement in G17 p 18: measurement in G18 p 19: measurement in G19 p Each other value: Measure	blane blane blane ement made in the plane define	17 ed by the machine
	_JM_I[4]	data Definition of the active ZO 0: measurement made wit 199: Measurement is m zero offset G54G57 or G505G599 1: G544: G57 599: G505G599 100: measurement made w	on measurement h G500 hade with the defined, settable 9 where with ZO defined in machine dat	0 a
As from measuring cycle version SW 6.3	_JM_I[5]	0: Standard offset (only me settable ZO 1: Extended offset (additio channel-specific basis fi	easurement, basis reference, nal global and rame)	0
	_JM_I[6] _JM_I[7] _JM_I[8] _JM_I[9]	Internal copy of _JM_I[4] Reserved Reserved Reserved		

10.04

	_JM_B[] BOOL value f	ield for JOG measurement	Maximum input limit	
	IV			
Changes valid after value	assignment	Protection level: -	Unit: -	
Data type: BOOLEAN		Applies fro	m SW version: 5.3	
Meaning:				Preset default
	_JM_B[0]	Tool offset mode for tool meas 0: offset in geometry on tool m 1: Offset in wear	surement neasurement	0
	_JM_B[1]	Number of measurement atter 0: 5 measurement attempts 1: 1 measurement attempt	npts	1
	_JM_B[2]	Retraction from the measuring 0: retraction as for intermediat 1: retraction with rapid travers	location e positioning e	0
	_JM_B[3]	Fast measuring feed 0: measuring with measuring f 1:1. measuring with feed in _S 2. measuring with measuring	eed SPEED[3] g feed	0
	_JM_B[5] _JM_B[6]	Not assigned Internal data item		0 0
Up to measuring cycle version SW 6.2	_JM_B[4]	0: Selection "basic frame" in th last channel-specific basic f 1: Selection of "basis" in scree system frame "set zero" (SE	ne screen means offset frame en form means offset in SETERAME	t in O

#### 9.3.4 Loading files for measuring in JOG

### Function

The files stored in the JOG\_MEAS\CYCLES\SPFFILES directory in the software supplied

CYC_JM.SPF	Auxiliary cycle for measuring
CYC_JMA.SPF	Aux. cycle for activation (as from SW 6.3)
CYC_JMC.SPF	Aux. cycle for calculation
E_MS_CAL.SPF	For calibrating a workpiece measuring probe
E_MS_CAN.SPF	To measure a corner
E_MS_HOL.SPF	To measure a hole
E_MS_PIN.SPF	To measure a spigot/shaft
E_MS_POC.SPF	For measurement of a rectangular pocket
	(from measuring cycles SW 6.3)
E_MS_SPI.SPF	For measurement of a rectangular spigot
	(from measuring cycles SW 6.3)
E_MT_CAL.SPF	For calibrating a tool measuring probe
E_MT_LEN.SPF	For length measurement of a tool
E_MT_RAD.SPF	For radius measurement of a tool
E_SP_NPV.SPF	Auxiliary cycle for activating the ZO that is
	active during measurement

#### Data Description 9.3 Data for measuring in JOG

are transferred to the control into directory "Standard cycles" from the diskette in menu "Services" after selection of softkey "Data in", "Diskette" and selection of the file in question and then pressing the "Start" softkey. They must then be loaded into the NC memory by pressing the softkey "Load". After the next Power on, they are known to the control.

When performing a software upgrade, make sure the variable values in the existing DEF files are kept. Back them up first, if necessary!

The other files

JOG_MEAS.COM (up to meas. cycles SW 6.2)	Configuring file for measuring in JOG user
COM_FILE.COM (from meas. cycles SW 6.3)	interface
MA_JOG.COM (up to meas. cycles SW 6.2)	Configuring file for the softkeys for
COM_FILE.COM (from meas. cycles SW 6.3)	measuring in JOG in the JOG basic display
BMP_FILE.ARC (up to measuring cycles SW 6.2)	Help displays for measuring in JOG
BMP_FILE.EXE	
MC_CST0.ARC (from measuring cycles SW 6.3)	All screens/displays
or	
MC_CST1.ARC	

must also be transferred to the control.



- BMP FILE.EXE contains:
- BMPJ640.EXE
- BMPJ800.EXE .
- BMPJ1024.EXE •

Note.

Up-to-date information is provided in the file SIEMENSD.TXT or SIEMENSE.TXT of the software being supplied.

Aux. displays, resolution 640 · 480

Aux. displays, resolution 800 · 600

Aux. displays, resolution 1024 · 768



### Start-Up

10.1	Commissioning measuring cycles for the first time	10-450
10.2	Upgrading measuring cycles	10-456
10.3	Sequence for probe installation	10-457
10.4	Example of calculating the repeat accuracy	10-459

#### 10 Start-Up 10.1 Commissioning measuring cycles for the first time

#### 10.1 Commissioning measuring cycles for the first time

#### Preconditions

- The hardware and software requirements of the measuring cycle version have been fulfilled (see Chapter "Hardware, software and installation").
- The probe is functional (Functional check probe connection → see Chapter "Hardware, software and installation").
- Be sure to archive the existing state before beginning installation; export series start-up archive now, if necessary.

#### NOTICE!

Please heed the information on measuring cycle startup in connection with ShopMill / ShopTurn in file SIEMENSD.TXT / SIEMENSE.TXT!



## Step 1 – set up memory configuring machine data

- Set memory configuring machine data for measuring cycles (→ for data see Section 9.1 "Machine data for executing measuring cycles" and the up-to-date information in SIEMENSD.TXT / SIEMENSE.TXT included with the delivered software).
- The following machine data must be observed:
   Machine data for GUDs MD18118: MM\_NUM\_GUD\_MODULES MD18120: MM\_NUM\_GUD\_NAMES\_NCK MD18130: MM\_NUM\_GUD\_NAMES\_CHAN MD18150: MM\_GUD\_VALUES\_MEM
  - Machine data for file systems MD18280: MM\_NUM\_FILES\_PER\_DIR MD18320: MM\_NUM\_FILES\_IN\_FILESYSTEM
  - Further machine data MD28082: MM\_SYSTEM\_FRAME\_MASK
- Export and reimport NC series start-up.

<sup>10.04</sup> **10** 





#### Step 2 - set up other machine data

- Set other machine data for measuring cycles (→ for data see Section 9.1 "Machine data for executing measuring cycles" and the up-to-date information in SIEMENSD.TXT / SIEMENSE.TXT included with the delivered software).
- The following machine data must be observed:
   Machine data for cycle interface MD18170: MM\_NUM\_MAX\_FUNC\_NAMES MD18180: MM\_NUM\_MAX\_FUNC\_PARAM MD28020: MM\_NUM\_LUD\_NAMES\_TOTOL MD28040: MM\_NUM\_LUD\_VALUES\_MEM MD10132: MMC\_CMD\_TIMEOUT MD11420: LEN\_PROTOCOL\_FILE MD13200: MEAS\_PROBE\_LOW\_ACTIV
- Perform an NC reset

#### Step 3 – load definition files

- Load GUD5.DEF and GUD6.DEF via "Services", "Data in" from diskette or via RS232
- Select blocks singly and press softkey "Activate"
- Go to step 4 without measuring in JOG

#### Only for measuring in JOG up to measuring cycles SW 6.02.16:

- Adjust GUD7.DEF (→ refer to Subsection 9.3.2 "Modifying the GUD7 data block")
- If a GUD7.DEF block was already active in the NCU, save the current values of existing GUD variables from the NCU (with "Services", "Data out", "NC-active data" NCK/channel, select block) and then reload.

# Only when measuring in JOG from measuring cycles SW 6.3 (also from patch measuring cycles SW 6.02.17) with HMI Advance from SW 6.3:

- If GUD7.DEF is already active in the NCU, unload it (but first save current values of existing GUD variables)
- Import file GUD7\_MC.DEF from diskette and activate it
- Re-activate GUD7.DEF (from Patch measuring cycles SW 6.02.17) or if previously not available, download and activate from "Archive\cycles archive".
- GUD7.DEF and GUD7\_MC.DEF must always have the same version.

#### Start-Up 10.1 Commissioning measuring cycles for the first time



### 10.04



#### Step 4 – load cycle programs

- Import cycle archive files (or single cycles) from diskette or via RS232.
- Load new cycles into the NCU (in the case of HMI Advanced: if now archive file was used or cycles were added).
- NC reset to update the cycle interface

#### Step 5 – set measuring cycle data

- Check default values of the GUD variables and set other values, if necessary.
- That is done by selecting and changing the variables in "Parameters", "User data" ... or using a program (see Section 9.3).
- Data description (see → Section 9.2 "Cycle data", Section 9.3 "Data for measuring in JOG" and upto-date information in SIEMENSD.TXT / SIEMENSE.TXT included with the software delivered).

#### Step 6 – load measuring cycle texts

When using measuring cycles with HMI up to SW 6.2, the measuring cycle text files have to be loaded afterward. If you require languages other than the standard languages, please order the appropriate HMI language versions.



#### Step 6 for HMI Advanced

- Load the archive file HMI\_ADV\MC\_TEXT.COM from diskette or via RS232.
- It contains the measuring cycle text files in the five standard languages.



#### Step 6 for HMI Embedded

- The text files for HMI Embedded are supplied singly (→ see diskette structure SIEMENSD.TXT / SIEMENSE.TXT).
- Out of the language-specific files ALMC.TXT and ALZ.XT, select the files of the required language path.
- The text files have to be linked in the HMI Software
   (→ see Upgrade Instructions for HMI Embedded).





Step 7 – load files for measurement result display



### Step 7 for HMI Advanced

- Load the files for HMI Advanced - HMI\_ADV\MCRESULT.COM, - HMI\_ADV\BMP\_RESU.ARC via RS232 or from floppy disk (from measuring cycles SW 6.3: Pictures/images are contained in file MC\_CSTx.ARC).
- After loading the files for the measurement result displays, check and adjust the entry for the text format in system file HMI MMC2\MBDDE.INI Change the entry to display the measurement result displays from

   MEASURE\_CYCLES\_RESULT=DOS to

```
- MEASURE_CYCLES_RESULT=
```

```
(delete "DOS"!).
```

### Step 7 for HMI Embedded

- Link the files for HMI Embedded
  - HMI\_EMB\MCRESULT\MZBILD01.COM
  - HMI\_EMB\MCRESULT\MZBILD02.COM
  - HMI\_EMB\MCRESULT\MZBILD03.COM
  - HMI\_EMB\MCRESULT\MZBILD04.COM
  - into the HMI software via the application.
- Integrate the measurement result displays into the display file for programming a flash card:

#### - Requirement:

- The system and application diskette has already been installed on your PC.
- The MZBILDnn.COM files have been packed as MZBILDnn.CO\_.

Commands for packing: arj a MZBILD01.CO\_ MZBILD01.COM arj a MZBILD02.CO\_ MZBILD02.COM arj a MZBILD03.CO\_ MZBILD03.COM arj a MZBILD04.CO\_ MZBILD04.COM

- Sequence:
  - Change to directory "INSTUTIL" in your application path and start "APP\_INST.EXE". The selection menu for software installation is displayed.
  - Select menu item < 2 > "Modify configuration".

#### Start-Up 10.1 Commissioning measuring cycles for the first time



A further selection menu appears. Here, select item < 6 > "Add \*.\* Files ...". As the file name enter your graphics files path and file name in the input screen form.

e.g.: A: \HMI\_EMB\MCRESULT\MZBILD01.CO\_

- Press the Return key to confirm your input.
- The "Esc" key will take you back to the basic menu and you can now create your software as "PCU\_20.ABB". That programs the flash card. Now proceed according to the start-up instructions for HMI Embedded.



## Step 8 – load and activate measuring cycle support



#### Step 8 for HMI Advanced

- From the software supplied located in subdirectory hmi\_emb, the following files are required from directory HMI\_ADV:
  - HMI\_ADV\BMP\_xxx.EXE (from measuring cycles SW 6.3: Pictures/images are contained in file MC\_CSTx.ARC) xxx stands for the screen resolution
  - HMI\_ADV\MCSUPP.COM HMI\_ADV\AEDITOR.COM HMI\_ADV\STARTUP.COM

#### Handling the EXE file:

 This file is not loaded into the control, but initially has to be loaded onto a PC. Starting this file automatically unpacks it, creating the unpacked archives:
 MZ BMP01.ARC ... MZ BMPxx.ARC

- These must be copied onto floppy disk and downloaded into the HMI or via V.24 (binary format). These archives contain the bitmaps for measuring cycles support.
- Load MCSUPP.COM, AEDITOR.COM, and STARTUP.COM from diskette or via RS232, then restart HMI.



#### Note:

Up-to-date information is provided in the file SIEMENSD.TXT or SIEMENSE.TXT of the software being supplied.





#### Step 8 for HMI Embedded

- To pack the COM files, a tool is provided in MAKE\_COM.BAT.
- Sequence: Set-up an empty directory and copy the following files there:
  - HMI\_EMB\MCSUPP\\*.COM
  - HMI\_EMB\TOOLS\ARJ.EXE
  - HMI\_EMB\TOOLS\MAKE\_COM.BAT
- Start MAKE\_COM.BAT
- The procedure to follow with the \*.CO\_ files is the same as that for linking the measuring result displays.
- Also add the archive file of bitmaps supplied CST.ARJ or CST\_10.ARJ (depending on screen resolution) via menu item < 6 > "Add \*.\* Files ...".
- Adapt file common.com and then load the following file into the NCU via RS232
  - HMI\_EMB\MCSUPP\COMMON.COM
  - Adaptation: Remove semicolon in front of SC617, SC326, and SC327

#### Step 9 – configure measuring cycle support

The measuring cycle screen forms can be set via GUD field \_MZ\_MASK. It is of type integer and is located in the GUD6 under NCK global data. In the "Start-up" operating area under softkey "Meas. cycles", "Manage data", a start-up screen form appears allowing you to make the settings.



## Step 10 – load and activate measurement in JOG (with HMI Advanced only)

- Set machine data for measurement in JOG
   (→ for data see Section 9.3 "Data for measurement in JOG")
- The following machine data must be observed: MD11602: ASUP\_START\_MASK MD11604: ASUP\_START\_PRIO\_LEVEL MD20110: RESET\_MODE\_MASK MD20112: START\_MODE\_MASK
- Load the files for measurement in JOG (→ see Subsection 9.3.4 "Loading files for measurement in JOG")
- Activate softkeys for measurement in JOG in file MA\_JOG.COM (by deleting comments)
- Restart the HMI

#### 10 Start-Up 10.2 Upgrading measuring cycles

#### 10.2 Upgrading measuring cycles

### =?

#### Explanation

Upgrading measuring cycles is basically performed in the same order as initial installation. Special notes:

- Only use files with the same measuring cycle software version. Do not mix files with different versions!
- Always heed the up-to-date information in SIEMENSD.TXT / SIEMENSE.TXT.
- Whether you need to increase the memory configuring machine data (step 1) depends on the state of the machine and the previous measuring cycle version.
- When performing a software upgrade, make sure the variable values in the existing DEF files are preserved. If necessary, save them first (step 3).
- Adaptation of the measuring cycle data usually only necessary when upgrading an original version 5.x (or 4.x); when upgrading within SW 6 it is enough to reconstruct the set values by previously archiving of the NC-active data (except where other settings are explicitly prescribed in SIEMENSD.TXT / SIEMENSE.TXT).



#### 10.3 Sequence for probe installation













#### 10.4 Example of calculating the repeat accuracy

### Function

#### Test program

This program allows the measuring scatter (repeat accuracy) of the entire measuring system (machine-probe-signal transmission to NC) to be calculated.

In the example, ten measurements are taken in the X axis and the measured value recorded in the workpiece coordinates. It is therefore possible to determine the random

dimensional deviations, i.e. those that are not subject to any trend.

#### Example:

%_N_	CHECK_ACCURATE_MPF	
;\$PAT	H=/_N_MPF_DIR	
N05	DEF INT SIGNAL, KK	;Variable definition
N10	DEF REAL MEASVALUE_IN_X[10]	
N15	G17 T1 D1	;Start conditions, preselect tool offset for ;probe
N20	ANF: G0 X0 F150	;Prepositioning in the measured axis
N25	MEAS=+1 G1 X50	;Measurement at 1st measurement ;input with switching signal from "not ;deflected" to "deflected" in the X axis. ;Probe switching expected between X0 and ;X50.
N30	STOPRE	;Stop decoding for subsequent ;evaluation of result (pre-processing stop)
N35	SIGNAL= \$AC_MEA[1]	;Read software switching signal at ;1st measurement input
N37	IF SIGNAL == 0 GOTO _ERROR1	;Check switching signal
N40	MEASVALUE_IN_X[kk]=\$AA_MW[X]	;Read measured value in workpiece ;coordinates
N50	KK=KK+1	
N60	IF KK<10 GOTOB ANF	;Repeat 10 times
N65	MO	
N66	STOPRE	;Preprocessing stop
N70	M2	
N80	_ERROR1: MSG ("Probe is not switching")	
N90	MO	
N95	M2	



## 0 Start-Up 10.4 Example of calculating the repeat accuracy



#### Explanation

After the parameter display (user-defined variables) have been selected, the measurement results can be read in field MEASVALUE\_IN\_X[10] provided that the program is still being processed.

#### Alarms



#### **General information**

If faulty states are detected in the measuring cycles, an alarm is generated and execution of the measuring cycle is aborted. In addition, the measuring cycles issue messages in the dialog line of the PLC. These messages do not interrupt execution.



#### Error handling in the measuring cycles

Alarms with numbers between 61000 and 62999 are generated in the measuring cycles. This range is again subdivided according to alarm responses and acknowledgment criteria.

The text displayed with the alarm number provides an explanation of the cause of the error.

Alarm number	Delete criterion	Alarm responses	
61000 61999	NC_RESET	Block preprocessing in the NC is aborted	
62000 62999	Backspace key	Program execution is not interrupted;	
		display only.	



#### Overview of measuring cycle alarms

The following table displays the errors which occur in the measuring cycles, together with error location and tips for remedying the errors.

Alarm number	Alarm text	Source	Explanation, remedy
61016	"System frame for cycles missing"	All	MD 28082: MM_SYSTEM_FRAME_MASK, set bit 5=1
61230	"Tool probe diameter too small"	E_MT_CAL E_MT_RAD E_MT_LEN	Correct in the data block GUD7 variable E_MESS_MT_DR[n] or E_MESS_MT_DL[n] for probe n+1 (measuring in JOG)
61301	"Probe not switching"	All	<ul> <li>Check measuring input</li> <li>Check measurement path</li> <li>Probe defective</li> </ul>
61302	"Probe - collision"	All	There is an obstacle in the probe's traversing path.
61303	"Safe area violated"	All	<ul><li>Check setpoint</li><li>Increase parameter _TSA</li></ul>
61306	"Permissible dimensional difference exceeded"	CYCLE971 CYCLE972 CYCLE974 CYCLE977 CYCLE978 CYCLE979 CYCLE982 CYCLE994	<ul> <li>Check setpoint</li> <li>Increase parameter _TDIF</li> </ul>
61307	"Incorrect measuring variant"	All	Parameter _MVAR has an illegal value.
61308	"Check measurement path 2a"	All	Parameter $_FA$ is $\leq 0$ .
61309	"Check probe type"	All, except CYCLE971 CYCLE972 CYCLE982	Tool type of workpiece probe in TO memory is not allowed
		CYCLE971	No permissible tool probe type entered in _TP[x,8], or in the case of tool probe type "disk" check the permissible working plane G17G19.
61310	"Scale factor is active"	All	Measurements are not possible when the scale factor is active.
61311	"No D number is active"	All	There is no tool offset selected for the probe (with workpiece measuring) or no tool offset selected for the active tool (with tool measuring).
61312	"Check measuring cycle number"	All	Measuring cycle called not permissible.
61313	"Check probe number"	All	The probe number is illegal (_PRNUM). Remedy: Correct _PRNUM or set up array _TP[] or _WP[] for additional tool or workpiece probes and adapt _CVAL[0]/_CVAL[1] accordingly.



Alarm number	Alarm text	Source	Explanation, remedy
61314	"Check selected tool type"	CYCLE971	Tool probe not permitted for tool
		CYCLE972	measurement/tool probe calibration.
		CYCLE982	
61315	"Check tool edge position"	CYCLE972	Check cutting edge position of tool
		CYCLE973	(measuring probe) in TO memory
		CYCLE974	
		CYCLE982	
		CYCLE994	
61316	"Center point and radius	CYCLE979	It is not possible to calculate a
	cannot be determined"		circle from the measured points.
61317	"Check parameter	CYCLE979	Parameterization faulty; needs 3
	CYCLE116"		or 4 points for calculating the
			center point
61318	"Check weighting factor _K"	CYCLE974	Parameter _K is 0.
		CYCLE977	
		CYCLE978	
		CYCLE979	
		CYCLE994	
		CYCLE998	
61319	"Check call parameter	As 61318	Internal fault measuring cycles.
	CYCLE114"		
61320	"Check tool number"	All	If tool management is active,
			parameter _TNUM=0, and parameter
			_TNAME is not assigned or the
			specified tool name for tool
			management is not known.
61321	"Check ZO memory	CYCLE974	The ZO with the number specified
	number"	CYCLE977	in _KNUM does not exist.
		CYCLE978	
		CYCLE979	
		CYCLE994	
		CYCLE998	
61322	"Check 4th digit in _KNUM"	CYCLE974	The specified digit of _KNUM
61323	"Check 5th digit in _KNUM"		contains invalid values.
61324	"Check 6th digit in _KNUM"	CYCLE970	AISO CHECK _IVIVAR!
		CVCLE979	
		CYCLE994	
		CYCLE114	
61325	"Check measuring	All, except	Parameter for the measuring axis
	axis/offset axis"	CYCLE979	MA has an incorrect value.
61326	"Check measuring direction"	CYCLE973	Parameter for the measuring
5.020		CYCLE976	direction MD has an incorrect
			value
61327	"Program reset necessary"	Allevent	NC reset necessary
01021	r rogram reset necessal y	CYCI F973	No reset necessary
		CYCLE976	
		0101010	



Alarm number	Alarm text	Source	Explanation, remedy
61328	"Check D number"	All	The D number in parameter
			_KNUM is 0.
61329	"Check rotary axis"	CYCLE998	The axis number specified in
	,		parameter RA is not assigned to a
			name (MD 20080) or the axis is not
			configured as a rotary axis
61000	"Coordinate rotation active"		(MD 50500).
01330	Coordinate rotation active	CYCLE972	measurements are not possible in
		CYCLE973	a rotated coordinate system.
		CYCLE974	
		CYCLE994	
61331	"Angle too large, change	CYCLE998	Parameter _STA1 is too large for
	measuring axis"		the specified measuring axis;
			select another measuring axis.
61332	"Change position of tool tip"	CYCLE971	Position of tool is not correct;
		CYCLE972	change starting point of
		CYCLE982	measurement.
		E_MI_CAL	
		E_MI_LEN	
61333	"Check calibration block		Parameter CALNUM is too large:
01000	number"	OTOLLOTO	1 Reduce CALINI M to a
	number		
61004	"Check protection popo"		
01334	Check protection zone	CICLE9//	Parameter _SZA/_SZO too large
01000		A 11	
61336	"Geometry axes not	All	No geometry axes are configured;
	available		change machine data in
			MD 20060.
61338	"Positioning speed is zero"	All	Parameter _SPEED[1],
			_SPEED[2] in GUD6 is 0
61339	"Offset factor rapid traverse	All	Check parameter _SPEED[0] in
	< 0"		GUD6.
61340	"Incorrect alarm number"	All	Internal fault measuring cycles.
61341	"Probe in active plane	CYCLE974	Calibrate probe before cycle call.
	not calibrated"	CYCLE977	
		CYCLE978	
		CYCLE979	
61342	"Software version entry	All	Up to meas. cycles SW 6.2
	in GUD6 incomplete or		SI[1] in GUD6 has either no value
	wrong format"		or a value < 3:
			from measuring cycles SW 6.3
			Lingrade NCK-SW release
			opyrade NON-OW TELEASE



Alarm number

61343

Alarm text

"Tool for specified tool

identifier does not exist"

Source	Explanation, remedy
All	Check name of tool identifier
All	Remove tool from other spindle.
All	Reduce D number in _KNUM, che
	software or shallow D number MD
CYCLE961	Parameters _SETV[0] or _SETV[7 are not assigned or are less than

61344	"Several tools are active"	All	Remove tool from other spindle.
61345	"Parameterized D number	All	Reduce D number in _KNUM, check
	(_KNUM) too large"		software or shallow D number MD
61346	"Distance between starting	CYCLE961	Parameters _SETV[0] or _SETV[1]
	point and measuring point		are not assigned or are less than 0
	$\_$ SETV[0] and $\_$ SETV[1] $\le$		
	0"		
61347	"Angle 1st edge - 2nd edge	CYCLE961	Parameter _INCA is 0.
	is 0"		
61349	"Distance between tool	CYCLE971	Parameter _TP[x,9] distance between
	probe top edge and		tool probe top edge and bottom edge
	measuring position for tool		is 0; relevant for radius measurement
	radius measurement is 0"		
61350	"Feedrate, speed for tool	CYCLE971	Measurement feed and/or spindle
	measurement with rotating		speed for tool measurement with
	spindle not programmed in		rotating spindle not specified in GUD
	_MFS"		variable _MFS[2].
61351	"Tool length or radius is 0"	CYCLE971	The length or radius for the active tool
			is zero.
61352	"Illegal path for	CYCLE106	The path specification for the log file
	log file"		is incorrect.
61353	"Path for log file does not	CYCLE106	The specified directory does not exist
	exist"		or the path indicated is incorrect.
61354	"Log file not found"	CYCLE106	No name was specified for the log file.
61355	"Incorrect file type for	CYCLE106	The file extension for the log file is
	log file"		incorrect.
61356	"Log file already in use"	CYCLE106	The log file is already used by another
			NC program.
61357	"No resources available"	CYCLE106	Insufficient NC memory available,
			delete files.
61358	"Logging error"	CYCLE106	Internal error, contact hotline
61359	"Continue with RESET"	CYCLE106	Internal error, contact hotline
61360	"Undefined logging job -	CYCLE106	The cycle CYCLE106 was called with
	press RESET to continue"		an incorrect parameter.
61361	"Unable to log variable"	CYCLE105	The value specified in _PROTVAL[]
61362	"Too many values"		4 parameter for CVCL E118 is greater
01302	Too many values	GIGLEIIO	than 10.



Alarm number	Alarm text	Source	Explanation, remedy
61363	"Max. number of value lines exceeded"	CYCLE105	Reduce number of lines.
61364	"Check distance between measuring point 1 and measuring point 2.	CYCLE998	Parameter _ID is $\leq 0$ .
61365	"Check circular feed"	CYCLE979	Parameter _RF is $\leq 0$ .
61366	"Direction of rotation for tool measurement with rotating spindle in _CM[5] is not defined"	CYCLE971	Permissible values for array _CM[5] in the GUD6 module are 3 (corresponds to M3) and 4 (corresponds to M4)
61367	"The points P1 and P2, P3 and P4 are identical".	CYCLE961	Various positions specified for the different positions of _SETV[07].
61368	"The straight line defined by P1 and P2 or P3 and P4 do not produce an intersection"	CYCLE961	Various positions specified for the different positions of _SETV[07].
61369	"Unable to uniquely determine position of corner check parameter _SETV[07]"	CYCLE961 ,	Define P1 and P2, or P3 and P4 so that the intersection of the straight lines through these points lies outside the section defined by P1 and P2 or P3 and P4.
61370	"_PROTVAL[0] - _PROTVAL[5] do not contain entries"	CYCLE105	Assign values to _PROTVAL[05].
61371	"The log produced by the column width and number columns exceed 200 characters per line"	CYCLE105	Reduce the column width or number of columns.
61372	"Selected measuring variant requires an SPOS-capable spindle"	t All	Change measuring variant or check machine equipment.
61373	"Mono probe requires an SPOS-capable spindle"	All	Check machine equipment



Alarm number	Alarm text	Source	Explanation, remedy
61401	"Probe is not responding,	CYCLE961	Unable to reach setpoint position
	end position"	CYCLE971	end position exceeded
		CYCLE977	
		CYCLE978	
		CYCLE998	
61402	"Probe collision, travel	CYCLE977	The position path in the plane has
	limitation through software		been limited by the software end
	end position"		position for measuring variants
			shaft/web. Infeed in the infeed axis
044001)		A.11	caused the sensor response.
61403''	"Internal cycle error in frame calculation"	All	Call SIEMENS notline
61404 <sup>1)</sup>	"Internal cycle error on tool offset"	All	Check dependent tool data
61405 <sup>1)</sup>	"Tool environment does not	All	Correct name or create this
	exist in _TENV"		environment
61406 <sup>1)</sup>	"Check DL_NUMMER in	All	
- 1)	_DLNUM"		
61407''	"Check the 7th digit and higher of KNUM"	All	
61408 <sup>1)</sup>	"Sum offset not available"	All	MD 18080, set bit 8=1
61409 <sup>1)</sup>	"Setup offsets not available"	All	MD 18112, set bit 4=1
61410 <sup>1)</sup>	"Option or offset size not	All	The magnitude to be corrected
	available"		requires an option or increased MD values.
61411 <sup>1)</sup>	"Frame calculation not	CYCLE997	Check setpoints, actual values
	possible, check values"	CYCLE119	
61412 <sup>1)</sup>	"Channel basic frame not	CYCLE997	Set MD 28081>0,
	available"	CYCLE119	\$P_CHBFRMASK>0
61413 <sup>1)</sup>	"Check setpoint ball diameter, _SETVAL <=0"	CYCLE997	Check setpoint for ball diameter
61414 <sup>1)</sup>	"Distortion of the triangle	CYCLE997	Check setpoints, actual values
	over limit"	CYCLE119	
61415"	"Check probe/machining	CYCLE971	Use probe permitted for machining
	plane"		plane (_IP[x,8], _IPW[x,8]) or
C144C <sup>1)</sup>	"A divet errow size	A 11	change machining plane
01410	TPI 1/ CV/ALIO11" or	All	Adjust _CVAL entry for humber of
	 "Adjust array size		probe of gauging block data arrays
	WP[1/ CVAL[1]!" or		
	"Adjust array size		
	_KP[]/_CVAL[2]!" or		
	"Adjust array size		
	_TWP[ ]/_CVAL[3]!"		

1) from measuring cycles SW 6.3



62303       "Safe area violated"       All <ul> <li>Check setpoint</li> <li>Increase parameter_TSA</li> </ul> 62304       "Allowance"       CYCLE974       Actual/setpoint difference is greater         CYCLE979       CYCLE979       (parameter_TUL)         CYCLE979       CYCLE974       Actual/setpoint difference is less than         62305       "Undersize"       CYCLE974       Actual/setpoint difference is less than         CYCLE979       CYCLE974       Actual/setpoint difference is less than         CYCLE979       TLL)       CYCLE978         CYCLE979       CYCLE979       TLL)         CYCLE979       CYCLE979       Parameter_TDIF, tool data are not         CYCLE978       CYCLE978       cyrcLe978         difference exceeded"       CYCLE978       cyrcLe979         CYCLE978       CYCLE978       cyrcLe979         CYCLE978       CYCLE979       cyrcLe979         CYCLE979       CYCLE979       cyrcLe979         CYCLE979       CYCLE979       cyrcLe979         CYCLE979       CYCLE979       cyrcLe979         CYCLE979       CYCLE978       cyrcLe979         CYCLE979       CYCLE978       cyrcLe974         62307       "Max. num	Alarm number	Alarm text	Source	Explanation, remedy
<ul> <li>Increase parameter _TSA</li> <li>62304 "Allowance" CYCLE974 CYCLE977 CYCLE977 than the upper tolerance level CYCLE978 (parameter _TUL)</li> <li>62305 "Undersize" CYCLE974 Actual/setpoint difference is less than CYCLE979 CYCLE974 Actual/setpoint difference is less than CYCLE977 the lower tolerance level (parameter CYCLE978 _TLL)</li> <li>62306 "Permissible dimensional difference exceeded" CYCLE971 Actual/setpoint difference is greater than the tolerance parameter _TDIF, tool data are not CYCLE974 cYCLE979 CYCLE974</li> <li>62306 "Permissible dimensional difference exceeded" CYCLE971 CYCLE974 CYCLE974 CYCLE974 CYCLE974</li> <li>62307 "Max. number of characters CYCLE105 per line exceeded" CYCLE105</li> <li>Number of characters per line not sufficient</li> <li>Increase value in _PROTFILE[1]</li> <li>62308 "Variable column width not possible"</li> <li>CYCLE105 No variable column width of 12 characters is used.</li> <li>A fixed column width of 12 characters is used.</li> <li>Remedy: Complete header in _PROTVAL[]</li> <li>62309 "Column width not sufficient"</li> <li>CYCLE105 Value to be logged is greater than the column width.</li> </ul>	62303	"Safe area violated"	All	Check setpoint
62304       "Allowance"       CYCLE974 CYCLE977       Actual/setpoint difference is greater than the upper tolerance level (parameter _TUL)         62305       "Undersize"       CYCLE978 CYCLE979       (parameter _TUL)         62305       "Undersize"       CYCLE974 CYCLE977       Actual/setpoint difference is less than CYCLE979         62306       "Permissible dimensional difference exceeded"       CYCLE972 CYCLE974       Actual/setpoint difference is cYCLE978 CYCLE979         62306       "Permissible dimensional difference exceeded"       CYCLE972 CYCLE974       Actual/setpoint difference is cYCLE979         62307       "Max. number of characters per line exceeded"       CYCLE978 CYCLE978       -TLL)         62307       "Max. number of characters per line exceeded"       • Number of characters per line not sufficient       • Number of characters per line not sufficient         62308       "Variable column width not possible"       CYCLE105       • No variable column width of 12 characters is used.         62309       "Column width not sufficient"       CYCLE105       • Value to be logged is greater than the column width.				<ul> <li>Increase parameter _TSA</li> </ul>
CYCLE977 CYCLE978 CYCLE978 CYCLE979       than the upper tolerance level (parameter _TUL)         62305       "Undersize"       CYCLE974 CYCLE974       Actual/setpoint difference is less than CYCLE978 CYCLE978         62306       "Permissible dimensional difference exceeded"       CYCLE971 CYCLE974       Actual/setpoint difference is greater than the tolerance parameter _TDIF, tool data are not CYCLE978 CYCLE978         62307       "Max. number of characters per line exceeded"       CYCLE105 CYCLE978       • Number of characters per line not sufficient         62308       "Variable column width not possible"       CYCLE105       • Number of characters per line not sufficient         62308       "Variable column width not possible"       CYCLE105       • No variable column width of 12 characters is used.         62309       "Column width not sufficient"       CYCLE105       • Value to be logged is greater than the column width.	62304	"Allowance"	CYCLE974	Actual/setpoint difference is greater
CYCLE978 CYCLE994       (parameter _TUL)         62305       "Undersize"       CYCLE974 CYCLE974       Actual/setpoint difference is less than CYCLE977         62306       "Permissible dimensional difference exceeded"       CYCLE971 CYCLE979       Actual/setpoint difference is greater than the tolerance CYCLE974         62306       "Permissible dimensional difference exceeded"       CYCLE972 CYCLE974       greater than the tolerance parameter _TDIF, tool data are not CYCLE978 CYCLE978         62307       "Max. number of characters per line exceeded"       CYCLE105 CYCLE979       • Number of characters per line not sufficient         62308       "Variable column width not possible"       CYCLE105       • No variable column width of 12 characters is used.         62309       "Column width not sufficient"       CYCLE105       • Value to be logged is greater than the column width.			CYCLE977	than the upper tolerance level
CYCLE979 CYCLE994         62305       "Undersize"       CYCLE974 CYCLE977       Actual/setpoint difference is less than the lower tolerance level (parameter CYCLE978 _TLL)         62306       "Permissible dimensional difference exceeded"       CYCLE974 CYCLE974       Actual/setpoint difference is greater than the tolerance         62306       "Permissible dimensional difference exceeded"       CYCLE974 CYCLE974       Actual/setpoint difference is greater than the tolerance         62307       "Max. number of characters per line exceeded"       CYCLE978 CYCLE978 CYCLE979       • Number of characters per line not sufficient         62308       "Variable column width not possible"       CYCLE105       • Number of characters per line not sufficient         62308       "Variable column width not possible"       CYCLE105       • No variable column width of 12 characters is used.         62309       "Column width not sufficient"       CYCLE105       • Value to be logged is greater than the column width.			CYCLE978	(parameter _TUL)
62305       "Undersize"       CYCLE974 CYCLE977       Actual/setpoint difference is less than the lower tolerance level (parameter CYCLE978 TLL)         62306       "Permissible dimensional difference exceeded"       CYCLE971 CYCLE974       Actual/setpoint difference is greater than the tolerance parameter _TDIF, tool data are not CYCLE977         62306       "Permissible dimensional difference exceeded"       CYCLE972 CYCLE974       greater than the tolerance parameter _TDIF, tool data are not CYCLE978         62307       "Max. number of characters per line exceeded"       CYCLE105       • Number of characters per line not sufficient         62308       "Variable column width not possible"       CYCLE105       • Number of column width of 12 characters is used.         62309       "Column width not sufficient"       CYCLE105       • Value to be logged is greater than the column width.			CYCLE979	
62305       "Undersize"       CYCLE974 CYCLE977 CYCLE978 CYCLE979 CYCLE994       Actual/setpoint difference is less than the lower tolerance level (parameter CYCLE979 CYCLE994         62306       "Permissible dimensional difference exceeded"       CYCLE971 CYCLE974 CYCLE974       Actual/setpoint difference is greater than the tolerance parameter _TDIF, tool data are not CYCLE978 CYCLE978         62307       "Max. number of characters per line exceeded"       CYCLE105 CYCLE978       • Number of characters per line not sufficient         62308       "Variable column width not possible"       CYCLE105 not possible"       • No variable column width of 12 characters is used.         62309       "Column width not sufficient"       CYCLE105 CYCLE105       • Value to be logged is greater than the column width.			CYCLE994	
CYCLE977 CYCLE978 CYCLE979 CYCLE994       the lower tolerance level (parameter TLL)         62306       "Permissible dimensional difference exceeded"       CYCLE971 CYCLE972       Actual/setpoint difference is greater than the tolerance parameter _TDIF, tool data are not CYCLE977         62307       "Max. number of characters per line exceeded"       CYCLE105 CYCLE978       • Number of characters per line not sufficient         62307       "Max. number of characters per line exceeded"       • Number of characters per line not sufficient         62308       "Variable column width not possible"       CYCLE105         62309       "Column width not sufficient"       CYCLE105 CYCLE105	62305	"Undersize"	CYCLE974	Actual/setpoint difference is less than
CYCLE978 CYCLE994       _TLL)         62306       "Permissible dimensional difference exceeded"       CYCLE971 CYCLE972       Actual/setpoint difference is greater than the tolerance CYCLE974         62306       "Permissible dimensional difference exceeded"       CYCLE972 CYCLE974       greater than the tolerance parameter _TDIF, tool data are not CYCLE977         62307       "Max. number of characters per line exceeded"       CYCLE105 CYCLE982 CYCLE994       • Number of characters per line not sufficient         62308       "Variable column width not possible"       CYCLE105       • No variable column widths can be generated because no header exists.         • A fixed column width of 12 characters is used.       • A fixed column width of 12 characters is used.         • Remedy: Complete header in _PROTVAL[]       • Value to be logged is greater than the column width.			CYCLE977	the lower tolerance level (parameter
CYCLE979 CYCLE994         62306       "Permissible dimensional difference exceeded"       CYCLE971 CYCLE972       Actual/setpoint difference is greater than the tolerance parameter _TDIF, tool data are not CYCLE977         CYCLE978       CYCLE978       corrected.         CYCLE994       CYCLE998         62307       "Max. number of characters per line exceeded"       CYCLE105         62308       "Variable column width not possible"       CYCLE105         62308       "Variable column width not possible"       CYCLE105         62309       "Column width not sufficient"       CYCLE105         62309       "Column width not sufficient"       CYCLE105			CYCLE978	_TLL)
62306       "Permissible dimensional difference exceeded"       CYCLE971       Actual/setpoint difference is greater than the tolerance parameter _TDIF, tool data are not corrected.         CYCLE974       CYCLE977       corrected.         CYCLE978       CYCLE978       cycLE994         62307       "Max. number of characters CYCLE904       • Number of characters per line not sufficient         62308       "Variable column width not possible"       CYCLE105       • No variable column width of 12 characters is used.         62309       "Column width not sufficient"       CYCLE105       • Value to be logged is greater than the column width.			CYCLE979	
62306       "Permissible dimensional difference exceeded"       CYCLE971 CYCLE971 CYCLE972 CYCLE974 CYCLE974 CYCLE974 CYCLE974 CYCLE977 CYCLE978 CYCLE978 CYCLE979 CYCLE982 CYCLE994       Actual/setpoint difference is greater than the tolerance parameter _TDIF, tool data are not corrected.         62307       "Max. number of characters CYCLE105 per line exceeded"       • Number of characters per line not sufficient         62308       "Variable column width not possible"       CYCLE105       • No variable column width of 12 characters is used.         62309       "Column width not sufficient"       CYCLE105       • A fixed column width.			CYCLE994	
difference exceeded"       CYCLE972 CYCLE974       greater than the tolerance parameter _TDIF, tool data are not corrected.         CYCLE977       CYCLE978 CYCLE978       corrected.         62307       "Max. number of characters CYCLE105 per line exceeded"       • Number of characters per line not sufficient         62308       "Variable column width not possible"       CYCLE105 CYCLE105       • No variable column widths can be generated because no header exists.         62309       "Column width not sufficient"       CYCLE105 CYCLE105       • No variable column width of 12 characters is used.         62309       "Column width not sufficient"       CYCLE105       • Value to be logged is greater than the column width.	62306	"Permissible dimensional	CYCLE971	Actual/setpoint difference is
CYCLE974 CYCLE977 CYCLE978 CYCLE978 CYCLE979       parameter _TDIF, tool data are not corrected.         62307       "Max. number of characters per line exceeded"       CYCLE105 CYCLE994         62308       "Variable column width not possible"       CYCLE105 CYCLE105       • Number of characters per line not sufficient         62308       "Variable column width not possible"       CYCLE105       • No variable column widths can be generated because no header exists.         A fixed column width of 12 characters is used.       • A fixed column width of 12 characters is used.         62309       "Column width not sufficient"       CYCLE105 CYCLE105       • Value to be logged is greater than the column width.		difference exceeded"	CYCLE972	greater than the tolerance
CYCLE977       corrected.         CYCLE978       CYCLE979         CYCLE979       CYCLE982         CYCLE994       CYCLE982         62307       "Max. number of characters       CYCLE105         per line exceeded"       Increase value in _PROTFILE[1]         62308       "Variable column width not possible"       CYCLE105         not possible"       No variable column width of 12 characters is used.         A fixed column width of 12 characters is used.       Remedy: Complete header in _PROTVAL[]         62309       "Column width not sufficient"       CYCLE105			CYCLE974	parameter _TDIF, tool data are not
CYCLE978       CYCLE979         CYCLE982       CYCLE994         62307       "Max. number of characters       CYCLE105         per line exceeded"       • Number of characters per line not sufficient         62308       "Variable column width not possible"       CYCLE105         62308       "Variable column width not sufficient"       CYCLE105         62308       "Variable column width not sufficient"       CYCLE105         62309       "Column width not sufficient"       CYCLE105         62309       "Column width not sufficient"       CYCLE105			CYCLE977	corrected.
CYCLE979 CYCLE982 CYCLE994         62307       "Max. number of characters CYCLE105 per line exceeded"       • Number of characters per line not sufficient         62308       "Variable column width not possible"       CYCLE105 VCLE105       • No variable column widths can be generated because no header exists.         62309       "Column width not sufficient"       CYCLE105 CYCLE105       • No variable column width of 12 characters is used.         62309       "Column width not sufficient"       CYCLE105 CYCLE105       • Value to be logged is greater than the column width.			CYCLE978	
62307       "Max. number of characters       CYCLE994         62307       "Max. number of characters       CYCLE105         per line exceeded"       Increase value in _PROTFILE[1]         62308       "Variable column width not possible"       CYCLE105         62308       "Variable column width not possible"       CYCLE105         62309       "Column width not sufficient"       CYCLE105         62309       "Column width not sufficient"       CYCLE105			CYCLE979	
62307       "Max. number of characters       CYCLE105       • Number of characters per line not sufficient         62308       "Variable column width not possible"       • CYCLE105       • No variable column widths can be generated because no header exists.         62309       "Column width not sufficient"       • CYCLE105       • A fixed column width of 12 characters is used.         62309       "Column width not sufficient"       • CYCLE105       • Value to be logged is greater than the column width.			CYCLE982	
62307       "Max. number of characters CYCLE105 per line exceeded"       • Number of characters per line not sufficient         62308       "Variable column width not possible"       CYCLE105 of the column width of the column width.         62309       "Column width not sufficient"       CYCLE105 CYCLE105       • Value to be logged is greater than the column width.			CYCLE994	
per line exceeded"       sufficient         62308       "Variable column width not possible"       CYCLE105       No variable column widths can be generated because no header exists.         62309       "Column width not sufficient"       CYCLE105       No variable column width of 12 characters is used.         62309       "Column width not sufficient"       CYCLE105       Value to be logged is greater than the column width.	62307	"Max. number of characters	CYCLE105	Number of characters per line not
<ul> <li>Increase value in _PROTFILE[1]</li> <li>62308 "Variable column width not possible"</li> <li>62308 "Variable column width not possible"</li> <li>A fixed column width of 12 characters is used.</li> <li>Remedy: Complete header in _PROTVAL[]</li> <li>62309 "Column width not sufficient"</li> <li>CYCLE105</li> <li>Value to be logged is greater than the column width.</li> </ul>		per line exceeded"		sufficient
62308       "Variable column width not possible"       CYCLE105       • No variable column widths can be generated because no header exists.         • A fixed column width of 12 characters is used.       • A fixed column width of 12 characters is used.         • Remedy: Complete header in _PROTVAL[]         62309       "Column width not sufficient"       CYCLE105 CYCLE105       • Value to be logged is greater than the column width.				<ul> <li>Increase value in _PROTFILE[1]</li> </ul>
not possible" generated because no header exists. A fixed column width of 12 characters is used. Remedy: Complete header in _PROTVAL[] 62309 "Column width not sufficient" CYCLE105 Value to be logged is greater than the column width.	62308	"Variable column width	CYCLE105	• No variable column widths can be
exists. A fixed column width of 12 characters is used. Remedy: Complete header in _PROTVAL[] 62309 "Column width not sufficient" CYCLE105 Value to be logged is greater than the column width.		not possible"		generated because no header
<ul> <li>A fixed column width of 12 characters is used.</li> <li>Remedy: Complete header in _PROTVAL[]</li> <li>62309 "Column width not sufficient" CYCLE105</li> <li>Value to be logged is greater than the column width.</li> </ul>				exists.
62309       "Column width not sufficient"       CYCLE105 sufficient"       • Value to be logged is greater than the column width.				A fixed column width of 12
Remedy: Complete header inPROTVAL[]     Column width not CYCLE105 Value to be logged is greater than the column width.				characters is used.
62309       "Column width not sufficient"       CYCLE105       • Value to be logged is greater than the column width.				Remedy: Complete header in
62309 "Column width not CYCLE105 • Value to be logged is greater than sufficient" the column width.				_PROTVAL[]
sufficient" the column width.	62309	"Column width not	CYCLE105	• Value to be logged is greater than
		sufficient"		the column width.
Adapt PROTFORM[5] or change				Adapt PROTFORM[5] or change
header for variable column width.				header for variable column width.
62310 "The max. number of CYCLE105 The max. number of characters per	62310	"The max. number of	CYCLE105	The max. number of characters per
characters per line is limited line has been limited to 200.		characters per line is limited	d	line has been limited to 200.
to 200"		to 200"		
62311 "The max. number of CYCLE105 The max. number of characters per	62311	"The max. number of	CYCLE105	The max. number of characters per
adiusted"		adiusted"		




### Adaptation of the Measuring Cycles to Previous Software Versions

12.1	Adaptation of the measuring cycle version to NC software versions	2-470
12.2	CYCLE103: Cycle for parameter definition (only up to measuring cycles SW 4.5)	2-471
12.3	Measuring cycle sub-routines	<u>?</u> -472
12.4	Start-up of the measuring cycle user interface for MMC 102 (only up to measuring cycles SW 4.4)	2-474



10.04





### Function

Parameter \_SI[1] in data block GUD6 is used to adapt measuring cycle versions up to SW 5 to previous NC software versions. In the delivery status of the measuring cycles, the current software status of the control is entered in parameter \_SI[1], i. e. 5 for SW 5. This parameter must be changed to match the measuring cycles <SW 6 to older software releases. Example: When using measuring cycles status 5.x.x on a control with SW 4,  $\rightarrow$ \_SI[1] = 4 Requirement: To use the measuring cycles, the software status of the control must be  $\geq$  3.





# 12.2 CYCLE103: Cycle for parameter definition (only up to measuring cycles SW 4.5)



### Function

Values can be assigned to measuring cycle parameters with CYCLE103.



Activation of this function depends on the configuration of the measuring cycle interface in the MMC.

Ĩ

Never ignore information provided by the machine manufacturer!

As of measuring cycles SW 4.5, CYCLE103 is no longer supported or developed further. Instead, use the cycle support for measuring cycles to supply the parameter data. Please refer to Section 7.2 for a detailed description.



#### Explanation

When CYCLE103 is selected and started, an input dialog for setting parameters for the measuring cycles is opened.

During the course of this dialog, a series of input screen forms are opened one after the other on top of the current display. Once the values have been entered each display must be concluded by pressing the OK key in the vertical softkey bar. At the end of the dialog, the message

"Input dialog successfully completed" is displayed in the dialog line of the control and the display before dialog mode was activated is reconstructed.

It is immediately possible to select and start the last measuring cycle assigned parameters.







#### Function

These measuring cycle subroutines are called directly by the cycles. With the exception of CYCLE100, CYCLE101, and CYCLE116, these subroutines cannot be executed by a direct call.



#### Programming

Cycle	Function	≥ SW 4	$\geq$ SW 4.5 $\geq$ SW 6.2
CYCLE100	Activate logging	Х	
CYCLE101	Deactivate logging	Х	
CYCLE102	Measuring result display		
CYCLE103	Parameter setting in interactive mode		
CYCLE104	Internal subroutine: Measuring cycle interface		
CYCLE105	Internal subroutine: Log	Х	
CYCLE106	Internal subroutine: Log	Х	
CYCLE107	Output of measuring cycle messages	only up	to measuring cycles – SW 6.2
CYCLE108	Output of measuring cycle alarms	only up	to measuring cycles – SW 6.2
CYCLE109	Internal subroutine: Data transfer		Х
CYCLE110	Internal subroutine: Plausibility checks		
CYCLE111	Internal subroutine: Measuring functions		
CYCLE112	Internal subroutine: Measuring functions		
CYCLE113	Internal subroutine: Log	Х	
CYCLE114	Internal subroutine: Load ZO memory, load WCS wear		
	Internal subroutine: Load WCS wear		X
CYCLE115	Internal subroutine: Load ZO memory		Х
CYCLE116	Calculation of the center point and radius on a circle		
CYCLE117	Internal subroutine: Measuring functions		
CYCLE118	Internal subroutine: Log	Х	



If the operating area is switched over during the course of the input dialog, the dialog can be selected again at a later stage with "Cycles" softkey in the extended menu.



<u>ا</u> ب	Programming	
=:	CYCLE103	
	Programming example	
	Workpiece probe calibration	
	CALIBRATE_IN_X_Y	
	N10 G54 G17 G0 X100 Y80	Position probe at the center of the hole and select ZO
	N15 T9 D1 Z10	Select length compensation, position probe in hole
	N20 CYCLE103	The operator can assign the parameters for calibration cycle CYCLE976 in interactive mode
	N25 CYCLE976	Measuring cycle call for calibration in the X-Y plane
	N50 M30	End of program





# 12.4 Start-up of the measuring cycle user interface for MMC 102 (only up to measuring cycles SW 4.4)



#### Function

In SW 3.2 and higher, the measuring cycles offer the option of displaying the measurement result screens and setting the input parameters via a dialog (call CYCLE103). These functions require some adjustments in the MMC software on the control.



#### Explanation

#### MMC 102

In the "Start-up" operating area you can access the MMC file system via the softkeys "MMC" and "DOS-Shell".

In the file c:\mmc2\comic.nsk the comment has to be removed in the second line.

**REM** TOPIC(...  $\Rightarrow$  **TOPIC**(...

Then the MMC has to be started again.

#### Measuring cycle user interface test

The cycle CYCLE103 can be activated and run in automatic mode.

When functioning properly, a screen is displayed with an overview of the measuring cycles; the dialog box for setting the measuring parameter cycles can be opened from here.



## Appendix

А	Overview of measuring cycle parameters	A-476
В	Abbreviations	A-514
С	Terms	A-516
D	References	A-522
Е	Commands, identifiers	A-534
F	Index	I-537



### A Overview of measuring cycle parameters



#### Parameter definition



Parameters must be defined

Parameter is not used in the cycle

The definition of the parameter depends on the measuring variant, other parameters, or on the machine configuration

CYCLE 961			Workpiece	e measuren	nent			
Parameters GUD5	Туре	Com- parable parameter 840C		Automatic setup of inside and outside corner for G17: in XY plane for G18: in ZX plane				
			Speci	ifving dista	nces and a	nales	Specifvin	a 4 points
			Corner inside	Corner outside	Corner inside	Corner outside	Corner inside	Corner outside
CALNUM	INTEGER	R12	3 medsur		4 measur			
_CORA	REAL	R13						
_CPA	REAL	R20						
_CPO	REAL	R21						
_EVNUM	INTEGER	R11						
_FA	REAL >0	R28	Only in	cluded if cal interna	M culated larç il value	easuremen ger than	t path in mm	
_ID	REAL	R19		Retraction in infeed axis, incremental for overtravel of corner if _ID=0 travels around the corner		Retraction in infeed axis, incremental for overtravel of corner if _ID=0 travels around the corner	Infeed of posit measuring dep	ioning depth to th (incremental)
_INCA	REAL 179.5 179.5 degrees	R26	Angle from workpiece	i 1st edge to (clockwise r	o 2nd edge negative)	of the		
_ <sup>K</sup>	INTEGER	rz9						





CYCLE 961			Workpiec	e measurer	ment				
Parameters GUD5	Туре	Com- parable parameter 840C	Automatic setup of inside and outside corner for G17: in XY plane for G18: in ZX plane for G19: in YZ plane					er	
			Spec	ifying dista	nces and a	angles	Specifyin	g 4 points	
			Corner	Corner	Corner	Corner	Corner inside	Corner outside	
			inside	outside	inside	outside			
			3 measur	ing points	4 measur	ing points			
_KNUM	INTEGER	R10		V	vithout/with	automatic o	offset of the ZO memor	У	
	>=0			0 Without	t offset				
			1	99 automa	tic offset in	ZO G540	G57 G505G599		
			10	00 automa	tic offset in	basic frame	e G500		
			101110	26 automa	tic ZO corre	ection in 1st	to 16th basic frame		
		D20	20	00 automa	itic ZO in sy	stem frame	• •		
_MA	INTEGER	R30							
	INTEGER	R31							
_MVAR	>0	RZJ				Measurin	ig variant		
			105	106	107	108	117	118	
_NMSP	INTEGER	R27		N	lumber of n	neasureme	nts at the same location	n	
PRNUM	INTEGER	R22				Probe r	number		
	>0		(numbe	r of the data	i field assigi	ned to the w	vorkpiece probe GUD6	:_WP[_PRNUM-1])	
_RA	INTEGER	R31							
_RF	REAL	R31							
_SETVAL	REAL	R32							
_SETV[0]	REAL		Distance b	etween star	ting point a	nd	Coordinates of point F	P1 in the active	
			measuring	point 2 (pos	sitive only)		workpiece coordinate	system (abscissa)	
_SETV[1]	REAL		Distance b	etween star	ting point a	nd	Coordinates of point F	P1 in the active	
			measuring	point 4 (pos	sitive only)		workpiece coordinate	system (ordinate)	
_SETV[2]	REAL		Distance be	tween			Coordinates of point F	P2 in the active	
			corner point	in abscissa			workpiece coordinate	system (abscissa)	
			only active i	f					
			_SETV[4]>1	1					
_SETV[3]	REAL		Distance be	tween			Coordinates of point F	P2 in the active	
			corner point	ing required			workpiece coordinate	system (ordinate)	
			only active i	f					
			_SETV[4]>1	1					
_SETV[4]	REAL		1: Measured	d corner			Coordinates of point F	P3 in the active	
			∠: Uffset in a 3: Offset in a	abscissa abscissa			workpiece coordinate	system (abscissa)	
			and ordin	ate					
			4: Offset in	Offset in ordinate					



υ	.0	)4	

CYCLE 961			Workpiece measurement						
Parameters	Туре	Com-	Automatic setup of inside and outside corner						
GUD5		parameter				for G17: in	n XY plane		
		840C				for G18: ir	n ZX plane		
						for G19: ir	n YZ plane		
			Speci	ifying dista	nces and a	ingles	Specifyin	g 4 points	
			Corner	Corner	Corner	Corner	Corner inside	Corner outside	
			inside	outside	inside	outside			
			3 measur	ing points	4 measur	ing points			
_SETV[5]	REAL						Coordinates of point F	P3 in the active	
							workpiece coordinate	system (ordinate)	
_SETV[6]	REAL						Coordinates of point F	P4 in the active	
							workpiece coordinate system (abscissa)		
_SETV[7]	REAL						Coordinates of point P4 in the active		
							workpiece coordinate	system (ordinate)	
_STA1	REAL 0360	R24	Approx. an	igle of posit.	direction o	f the			
	degrees		abscissa w	ith respect	to 1st edge	of the			
			workpiece	(reference e	edge), clock	wise			
			negative		•	•			
_SZA	REAL	R19							
_SZO	REAL	R18							
_TDIF	REAL	R37							
_TMV	REAL	R34							
_TNAME	STRING[]								
_TNUM	INTEGER	R9							
_TUL	REAL	R40							
_TLL	REAL	R41							
_TSA	REAL	R36							
_TZL	REAL	R33							
_VMS	REAL	R25			Va	riable meas	suring velocity		
	>=0			(for _VM	S=0: 150 m	ım/min (if _l	FA=1); 300 mm/min (if	_FA>1))	



/ _	<u> </u>

CYCLE971			Tool measurement of milling tools on milling machines				
Parameters GUD5	Туре	Com- parable Para- meters 840C	Abscissa for G17: X=1 for G18: Z=1 for G19: Y=1 Calibrating	Possible axes       Possible axes       Abscissa (_MA=1) / Ordinate (_MA=2) / Applicate (_MA=3)       7:     X=1     Y=2     Z=3       3:     Z=1     X=2     Y=3       3:     Y=1     Z=2     X=3			
			Relative to machine	Relative to workpiece	Relative to machine	Relative to workpiece	
_CALNUM	INTEGER	R12		•		•	
_CORA	REAL	R13					
_CPA	REAL	R20					
_ _CPO	REAL	R21					
_EVNUM	INTEGER >=0	R11			Empirical value number of GUD5:_EV[	memory number <sup>:</sup> data field _ <b>EVNUM-1]</b>	
_FA	REAL	R28		Measuremer	nt path in mm		
	20		For incremental calibrate travel is specified by the	tion, the direction of e sign of _FA.			
_ID	REAL >=0	R19			Normally 0, on multiple between the highest po and the length for radiu radius for length measu	cutters the offset int of the cutting edge s measurement (or the irement).	
_INCA	REAL	R26			5	,	
_K	INTEGER	R29					
_KNUM	INTEGER	R10					
_MA	INTEGER >=1	R30	1: calibration in +/- direct 2: calibration in +/- direct 3: calibration in +/- direct Also possible for cali 102:a) calculation of ce b) calibration in 2 ( 201:a) calculation of ce b) calibration in 1 ( Not for increment	Measuring ion in 1 (abscissa) ion in 2 (ordinate) ion in 3 (applicate) ibration in plane: inter in 1 (abscissa) ordinate) inter in 2 (ordinate) abscissa) ial calibration!	g axis 13 1: meas. of radius in di (abscissa) 2: meas. of radius in di 3: meas. of length at cr tool probe 103: meas. of the length, (abscissa) 203: meas. of the length, (ordinate)	rection 1 rection 2 (ordinate) enter point of the offset about radius 1 offset about radius 2	
_MD	INTEGER	R31					
L_MVAK	>=0	1120	Calibration in measurem positioning on center 0	Measurin ent axis acc. to previous of measurement cube 10	y variant Measure of le with motion 1 Measurement witl direction of rotation befo	ngth of radius less spindle 11 n rotating spindle; re cycle call, if spindle is	
	INTEGER	P27	Incremental traverse motion in r 10000 from meas. cy Automatic 100000	I calibration, measuring axis only 10010 /cles - SW 6.3: calibration 100010	already For motionless spindl 2	rotating. e direction in <b>_CM[5]</b> 12	
_NMSP	>0	R2/		Number of measureme	nts at the same location		



CYCLE971			Tool measurement of milling tools on milling machines						
Parameters	Туре	Com-		Possib	le axes				
GUD5		parable	Abscissa	(_MA=1) / Ordinate (_	MA=2) / Applicate (_	MA=3)			
		Para- meters	for G17: X=1	Y=2	Z=3				
		840C	for G18: Z=1	X=2	Y=3				
			for G19: Y=1	Z=2	X=3				
			Calibrate	tool probe	Measu	re tool			
			Relative to machine	Relative to	Relative to machine	Relative to			
				workpiece		workpiece			
_PRNUM	INTEGER >0	R22		Tool prob	e number				
			(numb	er of the data field ass	igned to the workpiece	probe			
				GUE					
			_TP[_PRNUM-1,i])	_TPW[_PRNUM-1,i]	_TP[_PRNUM-1,i])	_TPW[_PRNUM-1,i]			
				(as from meas. cycle SW 6.3))		(as from meas. cycle SW 6.3))			
RA	INTEGER	R31							
_ _RF	REAL	R31							
_SETVAL	REAL	R32							
_SETV[8]	REAL								
_STA1	REAL	R24							
	degrees								
_SZA	REAL	R19							
_SZO	REAL	R18							
_TDIF	REAL	R37	]		Dimension dif	ference check			
TMV	REAL	R34							
_ _TNAME	STRING[32]								
_TNUM	INTEGER	R9							
_TUL	REAL	R40							
_TLL	REAL	R41							
_TSA	REAL	R36		Safe	area				
TZL	REAL	R33		Zero off	set area				
	>=0 REAI	R25		Variable maa					
	>=0		(for VM	Variable meas	EA=1): 300 mm/min (if	FA>1))			
CMI	REAL	REAL		<u> </u>	Cvcle-internal calculat				
GUD6					monitoring data in <b>CI</b>	ип ип			
data item									
			-						
_MFS[]	REAL	REAL			Specification of S, F b	y user in _MFS[]			
GUD6-					Only active if _CBIT[	12j=1			
data item									

10.04





CYCLE 972	-		Tool measurement of turning tools with cutting edge position 1 – 8				
			on turning machines (machine-related)				
Parameters	Туре	Com-	Possib	le axes			
GUD5		parable	Abscissa (_MA=1) / Ordinate (_	MA=2)			
		Para- meters	for G17: X=1 Y=2	-			
		840C	for G18: Z=1 X=2				
			for G19: Y=1 Z=2				
			Calibrate tool probe	Measure tool			
_CALNUM	INTEGER	R12					
_CORA	REAL	R13					
_CPA	REAL	R20					
_CPO	REAL	R21					
_EVNUM	INTEGER	R11		Empirical value memory number			
	>=0			number of data field			
				GUD5:_EV[_EVNUM-1]			
_FA	REAL >0	R28	Measurement path in mm				
_ID	REAL	R19					
_INCA	REAL	R26					
_K	INTEGER	R29					
_KNUM	INTEGER	R10					
_MA	INTEGER >0	R30	Measuring	g axis 12			
_MD	INTEGER	R31					
_MVAR	INTEGER	R23	Measurir	ng variant			
			0	1			
_NMSP	INTEGER >=1	R27	Number of measureme	nts at the same location			
_PRNUM	INTEGER	R22	Tool prob	e number			
	2-1		(number of the data field assigned to the	he tool probe GUD6:_TP[_PRNUM-1])			
_RA	INTEGER	R31					
_RF	REAL	R31					
_SETVAL	REAL	R32					
_SETV[8]	REAL						
_STA1	REAL	R24					
_SZA	REAL	R19					
_SZO	REAL	R18					
_TDIF	REAL	R37		Dimension difference check			
_TMV	REAL	R34					
_TNAME	STRING[]						
_TNUM	INTEGER	R9					
_TUL	REAL	R40					
_TLL	REAL	R41					
_TSA	REAL >0	R36	Safe	area			
_TZL	REAL >=0	R33	Zero off	set area			
_VMS	REAL	R25	Variable meas	suring velocity			
	>=∪		(for _VMS=0: 150 mm/min (if _	FA=1); 300 mm/min (if _FA>1))			



CYCLE 973			Workpiece measurement					
Parameters	Туре	Com-	Possib	le axes				
GUD5		parable Para-	Abscissa (_MA=1) / Ordinate (_	_MA=2) / Applicate (_MA=3)				
		meters	for G17: X=1 Y=2	Z=3				
		0400	TOF G18:         Z=1         X=2           for C10:         V=1         7=2	Y=3 Y-3				
			Calibrating wo	nkniece probes				
			Relative to machine	Relative to workpiece				
			Reference groove	Surface				
_CALNUM	INTEGER	R12	Number of the gauging block					
	20		(number of the data field assigned					
0054	DEAL	D12	GUD6: _KB[_CALNUM-1])					
	REAL	R13						
	DEAL	P21						
_EVNUM	DEAL	R11 D20						
_FA	>0	RZ0	Measuremer	nt path in mm				
_ID	REAL	R19						
_INCA	REAL	R26						
_K	INTEGER	R29						
_KNUM	INTEGER	R10						
_MA	INTEGER >0	R30	Measuring axis 1, 2	Measuring axis 13				
_MD	INTEGER	R31	Measuring direction ( 0 =	= positive / 1 = negative )				
_MVAR	INTEGER >=0	R23	Measurir	ng variant				
			xxx13	0				
			34321           13 Reference groove	Sunace				
			0       Without incl. probe ball in calc.					
			(for calibration in groove only)					
			Image:					
			meas. axis and axis direction)					
			I 2 Two axis directions					
			(specify measuring axis)					
			I U Without position calculation					
			Calibration in groove only)					
_NMSP	INTEGER	R27	Number of measureme	nts at the same location				
PRNUM	INTEGER	R22	Tool prob	be number				
	>0		(number of the data field assigned to the	he tool probe GUD6:_WP[_PRNUM-1])				
_RA	INTEGER	R31						
_RF	REAL	R31						
_SETVAL	REAL	R32 R42		Calibration setpoint				
_SETV[8]	REAL							
_SZA	REAL	R19						
_SZO	REAL	K18						



CYCLE 973

Parameters

Туре

Com-

Workpiece	e measurement		
		Possible a	xes
	Abscissa (_MA=1	) / Ordinate (_MA=	=2) / Applicate (_MA=3)
for G17:	X=1	Y=2	Z=3
for G18:	Z=1	X=2	Y=3
for G19:	Y=1	Z=2	X=3
	C	alibrating workpi	ece probes
	Relative to mach	ine	Relative to workpiece

	<b>7</b> 1									
GUD5		parable	A	bscissa (_MA=1	l) / Ordinate (_MA=	2) / Applicate (_MA=3)				
		Para- meters	for G17:	X=1	Y=2	Z=3				
		840C	for G18:	Z=1	X=2	Y=3				
			for G19:	Y=1	Z=2	X=3				
				(	Calibrating workpie	ce probes				
			F	Relative to mach	nine	Relative to workpiece				
			Reference groove			Surface				
_STA1	REAL	R24								
_TDIF	REAL	R37								
_TMV	REAL	R34								
_TNAME	STRING[32]									
_TNUM	INTEGER	R9								
_TUL	REAL	R40								
_TLL	REAL	R41								
_TSA	REAL >0	R36		Safe area						
_TZL	REAL >=0	R33		Zero offset area						
_VMS	REAL	R25			Variable measuring	y velocity				
	>=0			(for _VMS=0: 15	50 mm/min (if _FA=1	); 300 mm/min (if _FA>1))				



CYCLE 974			Workpiece measurement						
CYCLE 994									
Para-	Туре	Com-		Possible measuring axe	25				
meters		parable	Abscissa (MA	A=1) / Ordinate ( MA=2) / App	olicate ( MA=3)				
GUD5		Para-	for G17: X=1	Y=2	Z=3				
		meters	for G18: Z=1	X=2	Y=3				
		04UC	for G19: Y=1	7=2	X=3				
			Measuring with	Measuring with a	utomatic tool offset				
			automatic 70	incusuring with a					
			correction						
			d naint	d point d point with row					
		<b>P1</b> 2	i point		2 point on diameter				
_CALNUM	INTEGER	RI2							
_CORA	REAL	R13							
_CPA	REAL	R20							
_CPO	REAL	R21							
_EVNUM	INTEGER	R11		Empirical value memory nur	nber				
	>=0		nu	mber of data field GUD5:_EV[_l	EVNUM-1]				
				Mean value r	memory number				
				number of the data field	d GUD5:_MV[_EVNUM-1]				
				Only active if GUD6:_CHBIT[4]=1					
FA	REAL	R28		Measurement path in mn	n				
– ID	REAL	R19							
– INCA	REAL	R26							
e	INTEGER	R29		Weighting factor k for	r mean value calculation				
	>0 INTEGER	R10	Without/with automatic offset	Without / with automa	tic tool offset (D number)				
	>=0		of the ZO memory	0 wit	hout tool offset				
			0 Without offset						
			199	Normal D number structure	Flat D number structure				
			automatic offset in ZO	7654321	<u>987654321</u>				
			G54G57 G505G599	_ _ _D number	_ _ _  _ D number				
			1000	0/1	0/1				
			automatic offset in basic frame	measuring axis or	measuring axis or				
			6500	setup or additive	setup or additive				
			10111026	offset (from meas.	offset (from meas.				
			automatic ZO correction in	I         Cycles SW 6.3)       I         2 radius offset or	I       Cycles SW 0.3)     I       2 radius offset or				
			1st to 16th channel basic	setup or additive	setup or additive				
			frame	offset (from meas.	offset (from meas.				
			10511066	cycles SW 6.3)	cycles SW 6.3)				
			automatic ZO correction in	0 offset normal	0 offset normal				
			1st to 16th global basic frame	1 offset inverted	Image: 1     1     Onset inverted       Image: 1     0     offset according to				
			automatic ZO correction in	4. digit	6. digit				
			system frame	1 offset of L1	1 offset of L1				
			9999	2 offset of L2	2 offset of L2				
			automatic ZO correction in	3 Offset of L3     4 Radius compensation	3 Offset of L3     4 Radius compensation				
			active frame G54G57,	O offset in length /	<b>0</b> offset in length /				
			G505G599 or with active	radius	radius				
			G500 in the last active	(> meas. 1 offset in setup or	(> meas. 1 offset in setup or				
			channel-specific basic frame	cycle additive offset	cycle additive offset				
				SW 6.3) 2 offset in length or	radius acc. to TENV				
				3 offset in setup or	3 offset in setup/				
				additive offset	or additive offset				
				acc. to _TENV	acc. to _TENV				







CYCLE 974			Workpiec	e measuremen	t					
CYCLE 994	_	-				<del></del>				
Para-	Туре	Com-			Possik	ole measuri	ng axes			
meters		Para-	for 017.	Abscissa (_M/	A=1) / Ordina	ate (_MA=2) 	Applicate (_	_MA=3)		
GODS		meters		X=1 7-1	r V	=2	Z=3			
		840C	for G10:	Z=1 V=1	7	-2 -2	1=3 X=3	2		
			Moas		2	Mossuring	with automat	-J		
			auto	matic ZO		Measuring	with automat			
			co	rrection						
			CY	CLE974		CYCLE974		CYCLE994		
			1	point	1 point 1 point with reversal		ith reversal	2 point on diameter		
_MA	INTEGER >0	R30			Me	asuring axis	13			
_MD	INTEGER	R31								
_MVAR	INTEGER >0	R23			M	easuring var	iant			
				100	0	1(	000	1		
								2		
_NMSP	INTEGER >0	R27		Number of measurements at the same location						
_PRNUM	INTEGER >0	R22	(numb	er of the data fie	Workpiece probe number eld assigned to the workpiece probe GUD6:_WP[_PRNUM-1])					
_RA	INTEGER	R31								
_RF	REAL	R31								
_SETVAL	REAL	R42 R32	S	etpoint		Setpoin	t (according to	o drawing)		
_SETV[8]	REAL									
_STA1	REAL	R26				Start	tandle			
	degrees					Otan				
_SZA	REAL	R19						Safety zone abscissa		
_SZO	REAL	R18						Safety zone ordinate		
_TDIF	REAL >0	R37				Dimer	nsion differenc	e check		
_TNAME	STRING[]				Tool name	(alternative	for "_TNUM" if	tool management active)		
_TENV	STRING[]				Name of	tool environi (from m	ment for autom	natic tool compensation es SW 6.)		
_TNUM	INTEGER	R9				Tool numb	per for automat	tic tool offset		
_TMV	REAL >0	R34				Offset range	with mean va			
TUI	REAL	R40				oper toleran	ce limit (accord	ding to drawing)		
TLL	REAL	R41				ower toleran	ce limit (accord	ding to drawing)		
_TSA	REAL	R36				Safe area		<u> </u>		
_TZL	REAL	R33					Zero offset are	ea		
_VMS	REAL	R25			Variabl	e measuring	y velocity			
	>=0			(for _VMS=	•0: 150 mm/m	nin (if _FA=1	); 300 mm/min	ı (if _FA>1))		



CYCLE 976			Workpiece measurement		
Para-	Туре	Com-		Possible measuring axes	
meters		Parable Para-	Abscissa (_MA=1)	/ Ordinate (_MA=2) / Applic	ate (_MA=3)
GUD5		meters	for G17: X=1	Y=2	Z=3
		840C	for G18: Z=1	X=2 7-0	Y=3
			tor G19: Y=1	Z=Z	X=3
			Drill bolo	Drill bolo with	Surface
			with known center	unknown center	Sunace
_CALNUM	INTEGER	R12			
_CORA	REAL	R13		Offset angular position	
	0359.5			(only active if mono probe)	
_CPA	REAL	R20			
_CPO	REAL	R21			
_EVNUM	INTEGER	R11			
_FA	REAL >0	R28		Measurement path in mm	
_ID	REAL	R19			
_INCA	0360 degrees	R26			
_K	INTEGER	R29			
_KNUM	INTEGER	R10			
_MA	INTEGER >0	R30		Measuring axis	
_MD	INTEGER >0	R31	Measuring	g direction ( $0 = \text{positive} / 1 = 1$	negative)
_MVAR	INTEGER >0	R23		Measuring variant	
			xxxx01	xxxx08	x0000
			Calibration in plane		Calibration on surface
			<b>6 5 4 3 2 1 1</b> calibration in hole	with known center	_INVAR=0 calibration on surface
			I     I     I     I     I     I     I       I     I     I     I     I     I     I	with unknown center	MVAR=10000
					calibration on surface with
			···· · · · · · · · · · · · · · · · · ·		calculation of the probe
			1111		length only permissible with
			I       0 without including	probe ball in calculation	_MA=3!
			I     <b>1</b> including probe ba	all in calculation (for meas. in plar	e)
			1   0 4 axis directions	necify measuring axis and axis	
			l direction)	pecity measuring axis and axis	
			I I <b>2</b> 2 axis directions (	specify measuring axis)	
				/	
			I I 0 without position	a calculation	
			II <b>1</b> with position de	etermination	
			I 0 Paraxial calibra	tion (in the plane)	
	INTEGER	0.7	1 Calibration at a	ny angle (in the plane)	
_NMSP	INTEGER >0	R27	Number	of measurements at the same	location





CYCLE 976			Workpiece	measurement			
Para- meters	Туре	Com- parable		Possible bscissa (MA=1)	measuring axe	es abscissa/ ord	linate
GUD5		Para-	for G17:	X=1	Y=2	Z=:	3
		meters 840C	for G18:	Z=1	X=2	Y=	3
		0400	for G19:	Y=1	Z=2	X=	3
				C	alibrating work	piece probes	
			D	rill-hole	Drill-ho	ole with	Surface
			with k	nown center	unknow	n center	
_PRNUM	INTEGER >0	R22		Prot	oe type / workpie	ce probe numbe	r
				<u>321</u> 	2-d	igit number	
				1	1	Mono probe	
					0	Multi probe	
				(number of the	e data field assig	ned to the workp	piece probe
				Gl	JD6:_WP[_PRN	UM(2-digit)-1])	
_RA	INTEGER	R31					
_RF	REAL	R31					
_SETVAL	REAL	R32			Calibration	setpoint	
_SETV[8]	REAL						
_STA1	REAL	R24		Start	angle		
_SZA	REAL	R19					
_SZO	REAL	R18					
_TDIF	REAL	R37					
_TMV	REAL	R34					
_TNAME	STRING[]						
_TNUM	INTEGER	R9					
_TUL	REAL	R40					
_TLL	REAL	R41					
_TSA	REAL >0	R36			Safe a	rea	
_TZL	REAL >=0	R33			Zero offse	et area	
_VMS	REAL >=0	R25		(for _VMS=0: 15	Variable measu 0 mm/min (if _F/	ıring velocity A=1); 300 mm/m	in (if _FA>1))





CYCLE 977			Workpiece measurement							
Para-	Туре	Com-			Possible mea	asuring axes				
meters		parable	Abs	cissa (_MA=1)	/ Ordinate (_M/	A=2)				
GUD5		meters	for G17:	X=1	Y=2					
		840C	for G18:	Z=1	X=2					
			for G19:	Y=1	Z=2					
			Меа	suring with au	tomatic tool of	fset	Measu	ring wit	th autom	atic
						_	Z	O corr	ection	
			Hole	Shaft	Groove	Web	Hole	Shaft	Groove	Web
_CALNUM	INTEGER	R12								
_CORA	REAL 0359.5	R13			Offset angu	lar position				
	REAL	R20			(only active if	mono probe)				
	REAL	R21								
	INTEGER	R11						1	l	
	>=0		numbe	r of data field <b>G</b>	ID5 FVI FVN	IIIM_11				
			Tidifibe	Mean value m	emory number					
			number	number of the data field GUD5:_MV[_EVNUM-1]						
			C	Only active if GUD6:_CHBIT[4]=1						
_FA	REAL >0	R28			Measuremen	t path in mm				
_ID	REAL	R19			Infeed a	pplicate				
INCA	REAL	R26								
	0360 degrees									
_K	INTEGER	R29								
KNIIM	>=0 INTEGER	R10	Without / with aut	omatic tool offset	(D number)		Without/w	ith autor	l natic offse	et of
	>=0		0 without too	ol offset			the ZO m	emory		
			Normal D numbe	r structure	Flat D number s	tructure	0 withou	ut offset		
			<u>7654321</u>		987654321		199			
			D nur	nber <sup>1)</sup>		D number <sup>2)</sup>	automatic	offset in	ZO G54(	G57
				gth offset in the	lengt	th offset in the	G505G5	599		
			me	asuring axis or	mea	suring axis or	automatic	offset in	basic fram	e
			set	up or additive	setu	p or additive	G500			
			me	as. cycle SW 6.3)	cycle	es SW 6.3)	1011102	26		
			2 rad	ius offset or	2 radiu	is offset or	automatic	ZO co	rrection in	
				set (as from	offse	et (from meas.	105110	66	i Dasicira	ame
			me	as. cycle SW 6.3)	cycle	es SW 6.3)	automatic	ZO corre	ection in	
				set normal		tset normal	1st to 16th	n global b	asic frame	;
			<b>0</b> offs	set according to	0 offse	et according to	2000	70.1		
			4th	digit	6. dig	git at of L1		ZO in sy	stem trame	e
			<b>2</b> offs	set of L2	2 offse	et of L2	automatic	ZO corre	ection in ac	tive
			3 offs	et of L3	3 offse	et of L3	frame G54	4G57, (	G505G59	99 or
			<b>4</b> Rac	set in length	I 4 Radi	et in length / radius	with active	e G500 ir	the last ac	ctive
			(> meas. or r	adius	(> meas. 1 offse	et in setup or	channel-s	pecific ba	asic frame	
			cycle 1 offs SW 6 3) Sw	et in setup or	cycle addit	tive offset				
			2 offs	set in length or	radiu	us acc. to _TENV				
			rad	ius acc. to _TENV	3 offse	et in setup or				
			3 offs add	et in setup or litive offset	addit acc.	to TENV				
			acc	. to _TENV	2) if MD 18105 >	999 also valid				
			1) if MD 18105 ≤9	single-digit,	for normal D nu	umber structure				
		ļ	>9<1000, three	-aigit D number						



CYCLE 977			Workpie	ce measurement						
Para-	Туре	Com-			Poss	ible measurir	ng axes			
meters		parable Para-		Abscissa (_MA=	1) / Ordin	ate (_MA=2)				
GUD5		meters	for G17:	X=1	١	(=2				
		840C	for G18:	Z=1	)	(=2				
			for G19:	Y=1	2	2=2				
			Ме	asuring with autor	matic too	offset	Me	asuring v	vith autor	natic
								ZO co	rrection	
			Hole	Shaft	Groove	Web	Hole	Shaft	Groove	Web
_MA	INTEGER >0	R30			Measuri	ng axis 12			Measurir	ig axis 12
_MD	INTEGER	R31								
_MVAR	INTEGER	R23			N	leasuring vari	ant			
	20			1xxx measureme	nt traveling	g around or ta	king acco	unt of a s	afety zone	
			1	2	3	4	101	102	103	104
_NMSP	INTEGER >0	R27		Number of measurements at the same location						
_PRNUM	INTEGER	R22		P	Probe type	/ workpiece p	robe num	ber		
				<u>3 2 1</u>						
				_   2-digit number						
				I		1 Mc	no probe			
				(aurahan af	the data	U IVIL	liti probe			
				(number of the data field assigned to the workpiece probe						
D۸	INTEGER	R31			GOD0•		z-uigit)- i	])		
	REAL	R31								
	REAL	R42/		O a ta aliat (a a a a adia	1	)		0-	L	
	DEAL	R32		Setpoint (according	g to drawi	ng)		Se	tpoint	
_SETV[8]	REAL	DOG								
_STA1	REAL 0360	R20								
074	degrees	D10							<u> </u>	
_SZA	REAL	R 19		Protection	on zone in	abscissa (on	ly for _MV	AR=1xxx	)	
	REAL	R10		Protecti	on zone ir	ordinate (onl	y for _MV	AR=1xxx	) *	
	>0	K37		Dimension differe	ence chec	k				
_TMV	REAL >0	R34	Offs	et range with mean	value cal	culation				
_TNAME	STRING[]			Tool nar	me					
			(alterna	ative to "_TNUM" w	ith tool ma	anagement				
				active	)					
_TENV	STRING[]		Name	e of tool environmer	nt for autor	matic tool				
			compe	ensation (from meas	suring cyc	es SW 6.)				
_TNUM	INTEGER >=0	R9		Tool num	iber					
	DEAL	D40		for automatic tool c	compensation	tion				
	REAL	R40	Uppe	r tolerance limit (ac	cording to	drawing)				
	DEAL	P26	Lowe	r tolerance limit (ac	cording to	arawing)				
_ISA	×EAL	K30				Safe area				
_TZL	REAL >=0	R33		Zero offset	area					
_VMS	REAL	R25			Variat	le measuring	velocity		×	
	>=0			(for _VMS=0:	150 mm/ı	min (if _FA=1)	; 300 mm	/min (if _F	<sup>-</sup> A>1))	









CYCLE 978			Workpiece measurement					
Parameters GUD5	Туре	Com- parable Param- eters 840C	Possible me Abscissa (_MA=1) / Ordinate (_I for G17: X=1 Y=2 for G18: Z=1 X=2 for G19: Y=1 Z=2	asuring axes IA=2)				
			Measuring with sutematic	Macouring with outomatic				
			tool offeet					
			CYCL E978	CYCL E978				
MA	INTEGER	R30	Measurin	axis 13				
- MD	>0 INTEGER	R31	····•	,				
MVAR	INTEGER	R23	Measuring variant					
-	>=0		0	100				
			1000 <sup>1)</sup> 1100 <sup>1)</sup>					
_NMSP	INTEGER	R27	Number of measureme	nts at the same location				
_PRNUM	INTEGER	R22	Probe type / workp	iece probe number				
	>0		<u>3 2 1</u>					
			    _   2-digit number					
			<u> </u>	Mono probe				
			0	Multi probe				
				signed to the workpiece probe				
	RFAI	R32	Sotpoint (app. to the drawing)	NOM(2-digit)-1])				
	INTEGER	R31	Setpoint (acc. to the drawing)	Selpoint				
_RF	REAL	R31						
_'N STA1	REAL	R24						
_0	0360							
	dogroop							
SZA	degrees REAL	R19						
_SZA _SZO	degrees REAL REAL	R19 R18						
_SZA _SZO TDIF	degrees REAL REAL REAL	R19 R18 R37	Dimension difference check					
_SZA _SZO _TDIF	degrees REAL REAL REAL >0 REAL	R19 R18 R37 R34	Dimension difference check					
_SZA _SZO _TDIF _TMV	degrees REAL REAL >0 REAL >0	R19 R18 R37 R34	Dimension difference check Offset range with mean value calculation only active if <b>GUD6:_CHBIT[4]=1</b>					
_SZA _SZO _TDIF _TMV _TNAME	degrees REAL REAL >0 REAL >0 STRING[]	R19 R18 R37 R34	Dimension difference check Offset range with mean value calculation only active if <b>GUD6:_CHBIT[4]=1</b> Tool name					
_SZA _SZO _TDIF _TMV _TNAME	degrees REAL REAL >0 REAL >0 STRING[]	R19 R18 R37 R34	Dimension difference check Offset range with mean value calculation only active if <b>GUD6:_CHBIT[4]=1</b> Tool name (alternative for "_TNUM" if tool management					
_SZA _SZO _TDIF _TMV _TNAME	degrees REAL REAL >0 REAL >0 STRING[]	R19 R18 R37 R34	Dimension difference check Offset range with mean value calculation only active if <b>GUD6:_CHBIT[4]=1</b> Tool name (alternative for "_TNUM" if tool management active)					
_SZA _SZO _TDIF _TMV _TNAME _TENV	degrees REAL REAL >0 REAL >0 STRING[]	R19 R18 R37 R34	Dimension difference check Offset range with mean value calculation only active if <b>GUD6:_CHBIT[4]=1</b> Tool name (alternative for "_TNUM" if tool management active) Name of tool environment for automatic tool					
_SZA _SZO _TDIF _TMV _TNAME _TENV	degrees REAL REAL >0 REAL >0 STRING[]	R19 R18 R37 R34	Dimension difference check Offset range with mean value calculation only active if <b>GUD6:_CHBIT[4]=1</b> Tool name (alternative for "_TNUM" if tool management active) Name of tool environment for automatic tool compensation					
_SZA _SZO _TDIF _TMV _TNAME _TNAME	degrees REAL REAL >0 REAL >0 STRING[]	R19 R18 R37 R34	Dimension difference check Offset range with mean value calculation only active if <b>GUD6:_CHBIT[4]=1</b> Tool name (alternative for "_TNUM" if tool management active) Name of tool environment for automatic tool compensation (from measuring cycles SW 6.3)					
_SZA _SZO _TDIF _TMV _TNAME _TENV _TENV	degrees REAL REAL >0 REAL >0 STRING[] STRING[] INTEGER >=0	R19 R18 R37 R34 R34	Dimension difference check Offset range with mean value calculation only active if <b>GUD6:_CHBIT[4]=1</b> Tool name (alternative for "_TNUM" if tool management active) Name of tool environment for automatic tool compensation (from measuring cycles SW 6.3) Tool number for automatic tool offset					
_SZA _SZO _TDIF _TMV _TNAME _TENV _TENV _TNUM _TUL	degrees REAL REAL >0 REAL >0 STRING[] INTEGER >=0 REAL REAL	R19 R18 R37 R34 R34 R9 R40	Dimension difference check Offset range with mean value calculation only active if <b>GUD6:_CHBIT[4]=1</b> Tool name (alternative for "_TNUM" if tool management active) Name of tool environment for automatic tool compensation (from measuring cycles SW 6.3) Tool number for automatic tool offset Upper tolerance limit (according to drawing)					
_SZA _SZO _TDIF _TMV _TNAME _TNAME _TENV _TNUM _TUL _TUL _TLL	degrees REAL REAL >0 REAL >0 STRING[] INTEGER >=0 REAL REAL 	R19 R18 R37 R34 R34 R9 R40 R41	Dimension difference check Offset range with mean value calculation only active if <b>GUD6:_CHBIT[4]=1</b> Tool name (alternative for "_TNUM" if tool management active) Name of tool environment for automatic tool compensation (from measuring cycles SW 6.3) Tool number for automatic tool offset Upper tolerance limit (according to drawing) Lower tolerance limit (according to drawing)					
_SZA _SZO _TDIF _TMV _TNAME _TNAME _TENV _TENV _TUL _TUL _TLL _TSA	degrees REAL REAL >0 REAL >0 STRING[] INTEGER >=0 REAL REAL >0	R19 R18 R37 R34 R34 R34 R34 R40 R41 R36	Dimension difference check Offset range with mean value calculation only active if <b>GUD6:_CHBIT[4]=1</b> Tool name (alternative for "_TNUM" if tool management active) Name of tool environment for automatic tool compensation (from measuring cycles SW 6.3) Tool number for automatic tool offset Upper tolerance limit (according to drawing) Lower tolerance limit (according to drawing)	area				
_SZA _SZO _TDIF _TMV _TNAME _TNAME _TENV _TENV _TUL _TLL _TLL _TSA _TZL	degrees REAL REAL >0 REAL >0 STRING[] INTEGER >=0 REAL REAL >0 REAL >0	R19 R18 R37 R34 R34 R34 R9 R40 R41 R36 R33	Dimension difference check Offset range with mean value calculation only active if <b>GUD6:_CHBIT[4]=1</b> Tool name (alternative for "_TNUM" if tool management active) Name of tool environment for automatic tool compensation (from measuring cycles SW 6.3) Tool number for automatic tool offset Upper tolerance limit (according to drawing) Lower tolerance limit (according to drawing) Safe Zero offset area	area				
_SZA _SZO _TDIF _TMV _TNAME _TNAME _TENV _TENV _TUL _TUL _TLL _TSA _TZL _VMS	degrees REAL REAL >0 REAL >0 STRING[] STRING[] STRING[] REAL >0 REAL >0 REAL >0 REAL >0 REAL >0	R19 R18 R37 R34 R34 R34 R34 R40 R41 R36 R33 R25	Dimension difference check Offset range with mean value calculation only active if <b>GUD6:_CHBIT[4]=1</b> Tool name (alternative for "_TNUM" if tool management active) Name of tool environment for automatic tool compensation (from measuring cycles SW 6.3) Tool number for automatic tool offset Upper tolerance limit (according to drawing) Lower tolerance limit (according to drawing) Safe Zero offset area	area suring velocity				

1) Difference measurement (not with mono probe)



CYCLE 979	)		Workpiece measurement							
Para-	Туре	Compar		Possible mea	asurements					
meters		able Para-		G17: X -ነ	/ plane					
GUD5		meters		G18: Z ->	( plane					
		840C								
			Measuring with automatic				Measuring with automatic			
			tool offset				ZO correction			
	INTEGER	R12	Hole Shaft	Groove	Web	Hole	Shaft	Groove	Web	
	REAL	R13		Offerent energy						
_CORA	0359.5	i ti i		only active if r	ar position nono probe					
_CPA	REAL	R20	Center absc	Center abscissa (with reference to the workpiece zero)						
_CPO	REAL	R21	Center ordir	ate (with referer	nce to the workp	piece ze	ero)		_	
_EVNUM	INTEGER >=0	R11	Empirical value memory number number of data field GUD5:_EV[_EVNUM-1]							
			number of the data field GUD5: MVI EVNUM-11							
			Only active if <b>GU</b>	D6: CHBIT[4]=	1					
_FA	REAL >0	R28		Measurement	path in mm			1		
_ID	REAL	R19			Infeed				Infeed	
					applicate				appli- cate	
_INCA	REAL	R26	Indexing angle			Inde ang	xing gle			
_K	INTEGER >0	R29	Weighting factor k for	mean value calcula	ation					
_KNUM	INTEGER >=0	R10	without / with automatic tool offset <b>0</b> without tool offset Normal D number structure	(D number) Flat D numb	per structure	without/with automatic offset of the ZO memory <b>0</b> without offset				
			7 6 5 4 3 2 1                  D number <sup>1)</sup>        O/1              O/1              O/1              O/1              O/1              O/1              O/1              O/1              O/1         setup or additive            O/1            O/1         meas. cycle SW 6.3)              O/1            O/1         meas. cycle SW 6.3)            O/1            O/1         meas. cycle SW 6.3)            O/1         0 offset normal            O/1         0 offset normal            O/1         1 offset of L1          O/1         2 offset of L2           3 offset of L3           4 Radius compensation        Offset in length	9876543212           0           1           1           1           1           1           1           1           1           1           1           1           1	e number <sup>2)</sup> th offset in the suring axis or p or additive et (as from s. cycle SW 6.3) us offset or p or additive et (as from s. cycle SW 6.3) et normal et normal et normal et normal et according to git et of L1 et of L2 et of L3 ius compensation et in length addius et in setup or n offset et in length or us acc. to _TENV e9 also valid	199 automai G54G G5051 1000 automai G500 10111 automai in 1st to 10511 1st to 11 2000 automai frame G54G or with a in the la basic fra	tic offse 57 G599 tic offse tic ZO c 0 16th ba 1066 tic ZO c 6th glob tic ZO c m frame tic ZO c 557, G50 active G ast active ame	t in ZO t in basic f orrection asic frame orrection i al basic fr orrection i 05G599 500 e channel-	rame n ame specific	



CYC	CLE 979		Workpiece me	asurement							
Para-	Туре	Com-		Possible measurements							
meters		Parable Para-			G17: X -	Y plane					
GUD5		meters			G18: Z -)	K plane					
		840C		G19: Y -Z plane							
			Меа	asuring with au	tomatic tool of	fset	Meas	uring \	with auto	omatic	
				ZO correction							
			Hole	Hole Shaft Groove Web Hole Shaft Groove Web							
_MA	INTEGER	R30									
_MVAR	INTEGER >0	R23			Measuring	g variant					
	-		1	1 2 3 4 101 102 103 104							
_NMSP	INTEGER >0	R27		Number	of measuremen	ts at the same I	ocatior	1			
_PRNUM		R22	Nun	nber of > measu	ring points / pro	be type / workpi	iece pro	obe nui	mber		
	>0			<u>4 321</u>							
					2-digit	number					
					1 N	lono probe					
					0 N	/ulti probe					
					0 3	measuring poir	nts				
					1 4	measuring poir	nts				
				(number of the	e data field assi	gned to the worl	kpiece	probe			
				Gl	UD6:_WP[_PRM	NUM(2-digit) –1	]				
_RA	INTEGER	R31									
_RF	REAL	R31	Veloc	ty for			Veloc	ity for			
			circular in	terpolation			circ	ular			
							interp	olation			
_SETVAL	REAL	R32 R42		Setpoint (accore	ding to drawing)			Se	tpoint		
_SETV[8]	REAL										
_STA1	REAL	R24		•	Start a	ingle			•		
_SZA	REAL	R19									
_SZO	REAL	R18									
_TDIF	REAL >0	R37		Dimension dif	ference check						
_TMV	REAL	R34	Offs	et range with me	ean value calcul	ation					
			c	only active if <b>GU</b>	D6:_CHBIT[4]=	1					
_TNAME	STRING[]			Tool	name						
	OTDINOU		(alternative	to "_TNUM" wit	th tool managen	nent active)					
_TENV	STRING[]		Name of tool	environment for	automatic tool	compensation					
	INTEGER	PO		(from measuring	g cycles SW 6.)						
	>=0	179		for automat	umber						
ТШ	REAL	R40	Unnor		according to dr						
	REAL	R41			according to dra	awing) awing)					
TSA	REAL	R36			Safe 4	area					
– TZL	>0 REAL	R33		Zoro off							
	>=0 REAI	R25									
	>=0	1723		for VMS-0.45		uring velocity	min /if				
					o mm⊭mm (ir_F	A-1), 300 mm/	umi (n	_FA>1,	1)		



CYCLE982			Tool measurement of turning, drilling, and milling tools for turning machines					
Parameters	Туре	Com-			Possible meas	suring axes		
GUD5		parable	Abso	;issa (_MA=1) /	Ordinate (_MA	A=2)		
		meters	for G17:	X=1	Y=2			
		840C	for G18:	Z=1	X=2			
			for G19:	Y=1	Z=2			
			Calibrate t	ool probe	Measure tool		automatically	
			Relative to	Relative to	Relative to	Relative to	Relative to	Relative to
			machine	workpiece	machine	workpiece	machine	workpiece
_CORA	REAL 0359.5	R13			Offset angle a	after reversal wh	en measuring	milling tools
_СРА	REAL	R20						
_CPO	REAL	R21						
_EVNUM	INTEGER >=0	R11			E	mpirical value m number of o GUD5: EV[	nemory numbe data field EVNUM-1]	r
_FA	REAL >0	R28			Measurement	path in mm		
_ID	REAL	R19						
_INCA	REAL 0360	R26						
_к	INTEGER	R29						
_KNUM	INTEGER	R10						
_MA	INTEGER >0	R30		Measuring	axis 12			
_MD	INTEGER	R31						
_MVAR	INTEGER	R23			Measuring	variant		
			0         6 5 4 3 2 1   </th <th><ol> <li>Calibration</li> <li>Measureme measureme</li> <li>Automatic m</li> <li>Relative to m</li> <li>Relative to m</li> <li>Relative to m</li> <li>Relative to m</li> <li>Measuring w</li> <li>Measuring w</li> <li>Measureme</li> <li>Measureme</li> <li>Automatic m measuring w</li> <li>Automatic m measuring w</li> <li>Automatic m measuring w</li> <li>Automatic m measuring w</li> <li>Axial positic (radius in or</li> <li>Radial positic (radius in at 0 Measureme</li> </ol></th> <th>xxxx01 ant of turning (SI ant in _MA heasurement in machine the workpiece (f <b>s only (also aut</b> without reversal with reversal ant: only correct ant: only correct int: only correct int: only correct int: only correct ant: only</th> <th>xxxx11 - 1-8), milling, and abscissa and or rom meas. cycle </th> <th>xxxx02 nd drilling tools dinate es SW 6.3) ting data SD42 t digit=1 it=1 1st digit=1 d radius, if 1st tarting positior d radius, if 1st irection</th> <th>xxxx12</th>	<ol> <li>Calibration</li> <li>Measureme measureme</li> <li>Automatic m</li> <li>Relative to m</li> <li>Relative to m</li> <li>Relative to m</li> <li>Relative to m</li> <li>Measuring w</li> <li>Measuring w</li> <li>Measureme</li> <li>Measureme</li> <li>Automatic m measuring w</li> <li>Automatic m measuring w</li> <li>Automatic m measuring w</li> <li>Automatic m measuring w</li> <li>Axial positic (radius in or</li> <li>Radial positic (radius in at 0 Measureme</li> </ol>	xxxx01 ant of turning (SI ant in _MA heasurement in machine the workpiece (f <b>s only (also aut</b> without reversal with reversal ant: only correct ant: only correct int: only correct int: only correct int: only correct ant: only	xxxx11 - 1-8), milling, and abscissa and or rom meas. cycle 	xxxx02 nd drilling tools dinate es SW 6.3) ting data SD42 t digit=1 it=1 1st digit=1 d radius, if 1st tarting positior d radius, if 1st irection	xxxx12





CYCLE982			Tool measure	Tool measurement of turning, drilling, and milling tools for turning machines					
Para-	Туре	Com-			Possible mea	asuring axes			
meters		parable Para-	Abs	scissa (_MA=1)	/ Ordinate (_N	IA=2)			
GUD5		meters	for G17:	X=1	Y=2				
		840C	for G18:	Z=1	X=2				
			for G19:	Y=1	Z=2				
			Calibrate	tool probe	Measu	re tool	Measuring	g the tool	
							automa	atically	
			Relative to	Relative to	Relative to	Relative to	Relative to	Relative to	
			machine	workpiece	machine	workpiece	machine	workpiece	
_NMSP	INTEGER >0	R27		Number	of measuremer	nts at the same	location		
_PRNUM	INTEGER >0	R22			Tool prob	e number			
				(number of th	e data field assi	gned to the wo	rkpiece probe		
					GU	D6:			
			_TP[_PRNUM-1,i]	_TPW[_PRNUM-	_TP[_PRNUM-1,i]	TPW[_PRNUM-	_TP[_PRNUM-1,i]	TPW[PRNUM-	
				meas, cvcle		meas, cvcle		meas, cvcle	
				SW 6.3)		SW 6.3)		SW 6.3)	
_RA	INTEGER	R31							
_RF	REAL	R31							
_SETVAL	REAL	R32							
_SETV[8]	REAL								
_STA1	REAL	R24			Startin	a anale when n	neasuring milling	u tools	
	degrees					0 - 0	5 5		
_SZA	REAL	R19							
_SZO	REAL	R18							
_TDIF	REAL >0	R37				Dimension dif	ference check		
_TMV	REAL	R34							
_TNAME	STRING[]								
_TNUM	INTEGER	R9							
_TUL	REAL	R40							
_TLL	REAL	R41							
_TSA	REAL >0	R36	Safe area						
_TZL	REAL >=0	R33	Zero offset area						
_VMS	REAL	R25		Variable measuring velocity					
	~=0			(for _VMS=0: 1	50 mm/min (if _l	FA=1); 300 mm	/min (if _FA>1))		



CYCLE997			Workpiece measurement				
Para-	Туре	Com-		Р	ossible me	asure	ments
meters		parable Para-			G17: X ·	-Y plai	ne
GUD5		meters			G18: Z ·	-X plar	ne
		840C			G19: Y	-Z plar	ne
				Measuring	g with auto	omatic	ZO correction
				1 sphere			3 spheres
_FA	REAL >0	R28		N	leasuremer	nt path	in mm
_INCA	REAL	R26	Ir	cremental angle (for	_MVAR=xx	:1109 c	only, measuring at an angle)
_KNUM	INTEGER >=0	R10		without/with a	utomatic off	set of	the ZO memory
					0 witho	out offs	set
			199 automa	tic offset in ZO		199	automatic offset in ZO
			G505	G599			G505G599
			1000 automa	tic offset in basic frame		1000	automatic offset in basic frame
			G500	automatic 70 correction	h in	1011	G500 1026 automatic ZO correction in
			10111020	1. up to 16th channel, b	asic frame		1. up to 16th channel, basic frame
			10511066	automatic ZO correction	n in	2000	automatic ZO correction in
				1. up to 16th global bas	ic frame		system frame
			2000	automatic ZO correction	n in	9999	automatic ZO correction in
			0000 outo	system frame	ativo		active frame
			fram	e G54 G57 G505 G5	99		or with G500 active in last
			or w	th G500 active in last			active channel-specific
			activ	e channel-specific basic	frame		basic frame
_MVAR	INTEGER	R23			Measurir	ng varia	ant
	20			x0xxxx			x1xxxx
			<u>765432</u>	1			
				109 measui 119 measui	re sphere and re sphere and	d ZO de d ZO de	etermination without measurement repeat
				0 measu	rement parax	a Lo ut	axes of the WCS)
			iii '	1 measur	rement at an	angle (	(intermediate positioning on circular path)
				0 measur 1 measur	re 1 sphere re 3 spheres		
			 	01 3 meas 11 4 meas	a. points on c a. points for c	ircle de ircle de	termination, for measuring at an angle only termination, for measuring at an angle only
			I	0 without 1 with dia	diameter det meter detern	termina ninatior	tion (sphere diameter known) า
_NMSP	INTEGER >0	R27		Number of n	neasureme	nts at t	the same location
_PRNUM	INTEGER	R22		Workpiece p	probe numb	er (for	multi probe only)
			Values: 1 up to 99				
			(number of the data field assigned to the workpiece probe				
RF	REAL	R31	Velocity for intermediate paths on circular path (G2 or G3)				
-			(for MVAR=xx1109 only, measuring at an angle)				
_STA1	REAL	R26		Starting angle (for _N	/IVAR=xx11	109 on	ly, measuring at an angle)
_TNVL	REAL						Limit for triangle distortion
						(f	or _MVAR=x1x109 only, measure 3
							spheres and ZO correction)
_SETV[8]	REAL		Setpoints, center point of the spheres (balls)				
_TSA	REAL >0	R36			Safe	area	
_VMS	REAL	R25		Va	riable meas	suring	velocity
	>=()			(for _VMS=0: 150 m	nm/min (if _	FA=1)	; 300 mm/min (if _FA>1))





CYCLE 998			Workpiece measurement				
Parameters	Туре	Com-	Possible measuring axes				
GUD5		parable	Absci	ssa (_MA=1) / Ordinate (_	MA=2) / Applicate (_MA=3)		
		Para- meters	for G17:	X=1 Y=2	Z=3		
		840C	for G18:	Z=1 X=2	Y=3		
			for G19:	Y=1 Z=2	X=3		
				Measuring with aut	omatic ZO correction		
				1 angle	2 angle		
_CALNUM	INTEGER	R12					
_CORA	REAL	R13		Offset ang	ular position		
	0359.5			(only active	if mono probe)		
_CPA	REAL	R20					
_CPO	REAL	R21					
_EVNUM	INTEGER	R11					
_FA	>=0 REAL >0	R28		Measureme	nt path in mm		
_ID	REAL	R19	Distance betwee	n measuring points P1 and	Distance between measuring points P1 and		
	>0		P2	in offset axis	P2 in abscissa		
_INCA	IREAL	R26			Setpoint or angle in ordinate		
_K	INTEGER	R29					
KNIIM	>U INTEGER	R10		without/with automatic o	ffset of the ZO memory		
	>=0			0 wi	thout offset		
			199	automatic offset in 70			
				G54G57			
				G505G599			
			1000	automatic offset in basic fr	ame G500		
			10111026	automatic ZO correction ir	1 1st to 16th basic frame		
			2000	automatic ZO in system fra	ame		
			9999	automatic ZO correction ir	active frame		
				G54G57, G505G599			
				or with G500 active in last	active channel-specific basic frame		
_MA	INTEGER	R30	Offset axis/meas	uring axis			
	20		102301				
				Measuring axis			
			l	Offset axis			
_MD	INTEGER	R31	for _MVAR=1xx1	0x only	for _MVAR=1xx10x only		
_MVAR	INTEGER >=0	R23		Measur	ng variant		
	Ŭ			1xx10x paraxial positionin	g between measuring points		
				105	106		
				1105 <sup>1)</sup>			
_NMSP	INTEGER	R27	Number of measurements at the same location				
	Differen	ce meas	surement (not with	mono probe)			



CYCLE 998			Workpiece measurement						
Parameters	Туре	Com-			Possible me	asuring axes			
GUD5		parable Para-		Abscissa (_MA=1	) / Ordinate (_I	MA=2) / Applicate (_MA=3)			
		meters	for G17:	X=1	Y=2	Z=3			
		840C	for G18:	Z=1	X=2	Y=3			
			for G19:	Y=1	Z=2	X=3			
				Meas	uring with auto	omatic ZO correction			
				1 angle		2 angle			
_PRNUM	INTEGER >0	R22		Pro	be type / workp	iece probe number			
	-			<u>3 2 1</u>					
					2-d	ligit number			
					1	Mono probe			
				<i>,</i> , , , , , , , , , , , , , , , , , ,	0	Multi probe			
				(number of th	e data field ass	signed to the workpiece probe			
	DEAL	Daa		G	UD6:_WP[_PR	NUM(2-digit)-1])			
_SETVAL	REAL	RJZ	Setpoint i	in measuring point 1	in meas. axis"	Setpoint in measuring point P1 in applicate?			
_SETV[0] <sup>2)</sup>	REAL >0					Distance between measuring points P1 and			
		D21				P3 in ordinate			
_RA	INTEGER	R31	_RA=0: 0	coordinate system is	rotated				
			_RA>0: r	Number of the rotary	axis in which				
	DEAL	D21	t	the offset is applied					
_RF	REAL	R31	ļ			F			
_STA1	REAL 0360	R24		Setpoint angle		Setpoint for angle about the abscissa			
	degrees	D10							
_SZA	REAL	R19 D10							
_SZO	REAL	R10							
_TDIF	REAL >0	R37							
TMV	REAL	R34							
TNAME	STRING[]								
	INTEGER	R9							
	>=0	<b>D</b> 40							
_IUL	REAL	R40							
_TLL	REAL	R41							
_TSA	REAL >0	R36			Safe are	ea angle			
TZL	REAL	R33							
	>=0 REAL	R25	+		Variable mee				
_vivis	>=0	1.20		(for VMS-0.1		Sum y velocity $EA = 1 + 300 \text{ mm/min} \text{ (if } EA > 1 + 1)$			
	not for		100100			FA-1), 300 MM//MM (II_FA>1))			
			IXXIUX						







Result par	ameters ca	libration								
							CYCLE971	CYCLE972 CYCLE982	CYCLE973	CYCLE976
GUD5	Data type	Meaning					Ť			
_OVR [0]	REAL									
_OVR [1]	REAL									
_OVR [2]	REAL									
_OVR [3]	REAL									
_OVR [4]	REAL	Actual point	Probe b	oall diameter						
_OVR [5]	REAL	Difference	Probe b	oall diameter						
_OVR [6]	REAL	Center of hole				Abscissa				
_OVR [7]	REAL	Center of hole				Ordinate				
_OVR [8]	REAL	Trigger point	Minus	direction	Actual point	Abscissa				
_OVR [9]	REAL	Trigger point	Minus	direction	Difference	Abscissa				
_OVR [10]	REAL	Trigger point	Plus	direction	Actual point	Abscissa				
_OVR [11]	REAL	Trigger point	Plus	direction	Difference	Abscissa				
_OVR [12]	REAL	Trigger point	Minus	direction	Actual point	Ordinate				
_OVR [13]	REAL	Trigger point	Minus	direction	Difference	Ordinate				
_OVR [14]	REAL	Trigger point	Plus	direction	Actual point	Ordinate				
_OVR [15]	REAL	Trigger point	Plus	direction	Difference	Ordinate				
_OVR [16]	REAL	Trigger point	Minus	direction	Actual point	Applicate				
_OVR [17]	REAL	Trigger point	Minus	direction	Difference	Applicate				
_OVR [18]	REAL	Trigger point	Plus	direction	Actual point	Applicate				
_OVR [19]	REAL	Trigger point	Plus	direction	Difference	Applicate				
_OVR [20]	REAL	Positional deviation				Abscissa				
_OVR [21]	REAL	Positional deviation				Ordinate				
_OVR [22]	REAL									
_OVR [23]	REAL									
_OVR [24]	REAL	Angle at which the trigger	points were	determined						
_OVR [25]	REAL									
_OVR [26]	REAL									
_OVR [27]	REAL	Zero offset area								
_OVR [28]	REAL	Safe area								
_OVR [29]	REAL									
_OVI [0]	INT									
_OVI [1]	INT									
_OVI [2]	INT	Measuring cycle number								
_OVI [3]	INT	Measuring variant								
_OVI [4]	INT									
_OVI [5]	INT	Probe number								
_OVI [6]	INT									
_OVI [7]	INT									
_OVI [8]	INT									
_OVI [9]	INT	Alarm number								



Result par	ameters me	easurement (turning machin	ies)		
GUD5	Data type	Meaning	CYCLE974	CYCLE994	CYCLE972 CYCLE982
_OVR [0]	REAL	Setpoint	Measuring axis	Diameter/radius	
_OVR [1]	REAL	Setpoint	Abscissa	Abscissa	
OVR [2]	REAL	Setpoint	Ordinate	Ordinate	
OVR [3]	REAL	Setpoint	Applicate	Applicate	
 OVR [4]	REAL	Actual point	Measuring axis	Diameter/radius	
OVR [5]	REAL	Actual point	Abscissa	Abscissa	-
OVR [6]	REAL	Actual point	Ordinate	Ordinate	
OVR [7]	REAL	Actual point	Applicate	Applicate	
	DEAL	Tolerance Upper limit <sup>1)</sup>	Measuring axis	Diameter/radius	
_OVR [8]	REAL	Actual point	-		Length L1
_OVR [9]	REAL	Difference			Length L1
OVR [10]	REAL	Actual point			Length L2
OVR [11]	REAL	Difference			Length L2
		Tolerance Lower limit <sup>1)</sup>	Measuring axis	Diameter/radius	<u> </u>
_OVR [12]	REAL		<u> </u>		Radius only
		Actual point			CYCLE982
	DEAL	Difference			Radius only
_OVR [13]	REAL	Difference			CYCLE982
_OVR [14]	REAL				
_OVR [15]	REAL				
_OVR [16]	REAL	Difference	Measuring axis	Diameter/radius	
OVR [17]	REAL	Difference	Abscissa	Abscissa	
OVR [18]	REAL	Difference	Ordinate	Ordinate	
_OVR [19]	REAL	Difference	Applicate	Applicate	
OVR [20]	REAL	Offset value			
OVR [21]	REAL				
_OVR [22]	REAL				
OVR [23]	REAL				
_OVR [24]	REAL				
_OVR [25]	REAL				
OVR [26]	REAL				
 OVR [27]	REAL	Zero offset range <sup>1)</sup>			
OVR [28]	REAL	Safe area			
_OVR [29]	REAL	Permissible dimension difference <sup>1)</sup>			
OVR [30]	REAL	Empirical value			
 OVR [31]	REAL	Mean value <sup>1)</sup>			
 OVI [0]	INT	D number / ZO number			
OVI [1]	INT				
OVI [2]	INT	Measuring cycle number			
OVI [3]	INT	Measuring variant			
OVI [4]	INT	Weighting factor <sup>1)</sup>			
OVI [5]	INT	Probe number			
OVI [6]	INT	Mean value memory number <sup>1)</sup>			
_OVI [7]	INT	Empirical value memory			
	INT	Tool number			
		Status offset request <sup>2)</sup>			
		Internal error number			
[12]					

for automatic tool offset only;
 for automatic ZO correction only



Result para	meters r	neasurement (milli	ing and machining centers	5)		
GUD5	Data	Meaning	CYCLE961	CYCLE997	C	YCLE998
	type				1 angle	2 angle
_OVR [0]	REAL	Setpoint		Sphere diameter 1st sphere	Angle	Angle about abscissa
_OVR [1]	REAL	Setpoint		Center point coordinate abscissa 1st sphere		Angle about ordinate
_OVR [2]	REAL	Setpoint		Center point coordinate ordinate 1st sphere		
_OVR [3]	REAL	Setpoint		Center point coordinate applicate 1st sphere		
_OVR [4]	REAL	Actual point	Angle with abscissa axis (WCS)	Sphere diameter 1st sphere	Angle	Angle about abscissa
_OVR [5]	REAL	Actual point	Corner point in abscissa (WCS)	Center point coordinate abscissa 1st sphere		Angle about ordinate
_OVR [6]	REAL	Actual point	Corner point in ordinate (WCS)	Center point coordinate ordinate 1st sphere		
_OVR [7]	REAL	Actual point		Center point coordinate applicate 1st sphere		
_OVR [8]	REAL	Difference		Sphere diameter 1st sphere		
_OVR [9]	REAL	Difference		Center point coordinate abscissa 1st sphere		
_OVR [10]	REAL	Difference		Center point coordinate ordinate 1st sphere		
_OVR [11]	REAL	Difference		Center point coordinate applicate 1st sphere		
_OVR [12]	REAL	Actual point		Sphere diameter 2nd sphere <sup>1)</sup>		
_OVR [13]	REAL	Actual point		Center point coordinate abscissa 2nd sphere <sup>1)</sup>		
_OVR [14]	REAL	Actual point		Center point coordinate ordinate 2nd sphere <sup>1)</sup>		
_OVR [15]	REAL	Actual point		Center point coordinate applicate 2nd sphere <sup>1)</sup>		
_OVR [16]	REAL	Difference		Sphere diameter 2nd sphere <sup>1)</sup>	Angle	Angle about abscissa
_OVR [17]	REAL	Difference		Center point coordinate abscissa 2nd sphere <sup>1)</sup>		Angle about ordinate
_OVR [18]	REAL	Difference		Center point coordinate ordinate 2nd sphere <sup>1)</sup>		
_OVR [19]	REAL	Difference		Center point coordinate applicate 2nd sphere <sup>1)</sup>		
_OVR [20]	REAL	Actual point	Angle with abscissa axis (MCS)	Sphere diameter 3rd sphere <sup>1)</sup>		
		Offset value			Angle	
_OVR [21]	REAL	Actual point	Corner point in abscissa (MCS)	Center point coordinate abscissa 3rd sphere <sup>1)</sup>		
		Offset value				Angle about abscissa
_OVR [22]	REAL	Actual point	Corner point in ordinate (MCS)	Center point coordinate ordinate 3rd sphere <sup>1)</sup>		
		Offset value				Angle about ordinate
OVR [23]	REAL	Actual point		Center point coordinate applicate 3rd sphere <sup>1)</sup>		
_0,[20]		Offset value				Angle about applicate



1) for measuring variants \_MVAR=x1x109 only, measure 3 spheres



GUD5	Data	Meaning	CYCLE961	CYCLE997	CYCLE998	
	type				1 angle	2 angle
_OVR [24]	REAL	Difference		Sphere diameter 3rd sphere <sup>1)</sup>		
_OVR [25]	REAL	Difference		Center point coordinate abscissa 3rd sphere <sup>1)</sup>		
_OVR [26]	REAL	Difference		Center point coordinate ordinate 3rd sphere <sup>1)</sup>		
_OVR [27]	REAL	Difference		Center point coordinate applicate 3rd sphere <sup>1)</sup>		
_OVR [28]	REAL	Safe area				
_OVR [30]	REAL	Empirical value				
_OVR [31]	REAL	Mean value				
_OVI [0]	INT	ZO number				*
_OVI [1]	INT					
_OVI [2]	INT	Measuring cycle number				
_OVI [3]	INT	Measuring variant				
_OVI [4]	INT	Weighting factor				
_OVI [5]	INT	Probe number				
_OVI [6]	INT	Mean value memory number				
_OVI [7]	INT	Empirical value, memory number				
_OVI [8]	INT	Tool number				
_OVI [9]	INT	Alarm number				
_OVI[10						
_OVI [11]	INT	Status offset request				
_OVI12]	INT	Internal error number		Internal measurement evaluation		



1) for measuring variants \_MVAR=x1x109 only, measure 3 spheres



10.04



Result parameters measurement (milling and machining centers)										
GUD5	Data	Meaning	CYCL	.E977	CYCLE978	CYCLE979				
	type		_MVAR=xxx1	_MVAR=xxx5						
			to	_MVAR=xxx6						
			_MVAR=xxx4							
_OVR [0]	REAL	Setpoint	Hole		Measuring axis	Hole				
			Shaft			Shaft				
			Groove			Groove				
			Web			Web				
		Setpoint rectangle length		Abscissa						
_OVR [1]	REAL	Setpoint center point/center	Abscissa		Abscissa	Abscissa				
		Setpoint rectangle length		Ordinate						
_OVR [2]	REAL	Setpoint center point/center	Ordinate		Ordinate	Ordinate				
		Setpoint for rectangle center point		Abscissa						
_OVR [3]	REAL	Setpoint			Applicate					
		Setpoint for rectangle center point		Ordinate						
_OVR [4]	REAL	Actual value for diameter/width	Hole		Measuring axis	Hole				
			Shaft			Shaft				
			Groove			Groove				
			Web			Web				
		Actual value rectangle length		Abscissa						
_OVR [5]	REAL	Actual value center point/center	Abscissa			Abscissa				
		Actual value rectangle length		Ordinate						
_OVR [6]	REAL	Actual point	Ordinate			Ordinate				
		Actual value rectangle center point		Abscissa						
_OVR [7]	REAL	Actual point								
		Actual value rectangle center point		Ordinate						
_OVR [8] <sup>1)</sup>	REAL	Upper tolerance limit	Hole		Measuring axis	Hole				
		diameter/width	Shaft			Shaft				
			Groove			Groove				
			Web			Web				
		Upper tolerance limit rectangle length		Abscissa						
_OVR [9]''	REAL	Upper tolerance limit rectangle length		Ordinate						
_OVR [10]	REAL									
_OVR [11]	REAL									
_OVR [12]' <sup>/</sup>	REAL	Lower tolerance limit	Hole		Measuring axis	Hole				
		diameter/width	Shaft			Shaft				
			Groove			Groove				
			Web			Web				
		Lower tolerance limit rectangle length		Abscissa						
_OVR [13] <sup>1/</sup>	REAL	Lower tolerance limit rectangle length		Ordinate						
_OVR [14]	REAL									
_OVR [15]	REAL									
_OVR [16]	REAL	Difference diameter/width	Hole		Measuring axis	Hole				
			Shaft			Shaft				
			Groove			Groove				
			Web			Web				
		Difference rectangle length		Abscissa						

Α



1) for workpiece measurement with tool offset only



Result para	ameters	measurement (milling and machinin	g centers) 1)			
GUD5	Data	Meaning	CYCI	_E977	CYCLE978	CYCLE979
	type		_MVAR=xxx1 to _MVAR=xxx4	_MVAR=xxx5 _MVAR=xxx6		
_OVR [17]	REAL	Difference center point/center	Abscissa			Abscissa
		Difference rectangle length		Ordinate		
_OVR [18]	REAL	Difference center point/center	Ordinate			Ordinate
		Difference of rectangle center point		Abscissa		
_OVR [19]	REAL	Difference of rectangle center point		Ordinate		
_OVR [20]	REAL	Offset value	1)	1)	1)	1)
_OVR [21]	REAL					
_OVR [22]	REAL					
_OVR [23]	REAL					
_OVR [24]	REAL					
_OVR [25]	REAL					
_OVR [26]	REAL					
_OVR [27]	REAL	Zero offset area	1)	1)	1)	1)
_OVR [28]	REAL	Safe area				1)
_OVR [29]	REAL	Permissible dimension difference	1)	1)	1)	1)
_OVR [30]	REAL	Empirical value	1)	1)		1)
_OVR [31]	REAL	Mean value	1)	1)	1)	1)
_OVI [0]	INT	D number / ZO number				
_OVI [1]	INT					
_OVI [2]	INT	Measuring cycle number				
_OVI [3]	INT	Measuring variant				
_OVI [4]	INT	Weighting factor	1)	1)	1)	1)
_OVI [5]	INT	Probe number				
_OVI [6]	INT	Mean value memory number	1)	1)	1)	1)
_OVI [7]	INT	Empirical value memory number	1)	1)		1)
_OVI [8]	INT	Tool number				
_OVI [9]	INT	Alarm number				
_OVI [11]	INT	Status offset request (for ZO correction only)				
_OVI12]	INT					
_OVI13]	INT	DL number (from meas. cycles SW 6.3)	1)	1)	1)	1)



1) for workpiece measurement with tool offset only




No machine	uala				
MD number	Name of identifier	Description	Max. input value	Default value	Value for meas. cycles
10132	MMC-CMD-TIMEOUT	Monitoring time for MMC command in part program	100	1	3
11420	LEN_PROTOCOL_FILE	File size for log files	100	1	5
13200	MEAS_PROBE_LOW_ACTIV	Switching characteristics of probe 0= 0V $\rightarrow$ 24V; 1= 24V $\rightarrow$ 0V	TRUE	0	0
18118	MM_NUM_GUD_MODULES	Number of data blocks	9	7	7
18120	MM_NUM_GUD_NAMES_NCK	Number of GUD variable names in the control	plus	10	30
18130	MM_NUM_GUD_NAMES_CHAN	Number of GUD variable names per channel	plus	40	130
18150	MM_GUD_VALUES_MEM	Memory space for the values of the GUD variables	plus0	12/16 <sup>1)</sup>	28/32 <sup>1)</sup>
18170	MM_NUM_MAX_FUNC_NAMES	Number of miscellaneous functions (cycles, DRAM)	plus	40	70
18180	MM_NUM_MAX_FUNC_PARAM	Number of miscellaneous functions (cycles, DRAM)	plus	300	600
28020	MM_NUM_LUD_NAMES_TOTAL	Number of LUD variables in total (in all program levels)	plus	200	200
28040	MM_NUM_LUD_VALUES_MEM	Memory for values of the LUD variables	plus	12/25 <sup>1)</sup>	14/27 <sup>1)</sup>
28082	MM_SYSTEM_FRAME_MASK (as from measuring cycle SW 6)	Channel-specific system frames	7FH	21H	21H (Bit0, 5=1)
NC machine	data for measuring in JOG	·			•
11602	ASUP_START_MASK	Ignore stop conditions for ASUB	3	0	1, 3 Bit0=1
11604	ASUP_START_PRIO_LEVEL	Priority for ASUP_START_MASK effective	64H	0	From 1 to 64H
20110	RESET_MODE_MASK	Define control default setting after power-up and RESET	07FFFH	0	min. 4045H (Bit0, 2, 6, 14=1)
20112	START_MODE_MASK	Define control default setting after part	07FFFH	400H	400H (Bit6=0)



Cycle data			
The measu	ring cycle data are stored in bl	ocks GUD5 and GUD6.	
Central val	ues	- <u>-</u>	
Block	Name of identifier	Description	As-delivered value
	_CVAL[]	Number of elements	
GUD6	_CVAL[0]	Number of data fields for tool probe, machine-related	3
GUD6	_CVAL[1]	Number of data fields for workpiece probe	3
GUD6	_CVAL[2]	Number of data fields for gauging block	3
GUD6	_CVAL[3]	Number of data fields for tool probe, tool-related (from	
		measuring cycles SW 6.3)	
	_TP[]	Tool probe (machine-related)	
	Assignment for milling		
GUD6	_TP[k,0]	Trigger point in minus direction X (1st geometry axis)	0
GUD6	_TP[k,1]	Trigger point in plus direction X (1st geometry axis)	0
GUD6	_TP[k,2]	Trigger point in minus direction Y (2nd geometry axis)	0
GUD6	_TP[k,3]	Trigger point in plus direction Y (2nd geometry axis)	0
GUD6	_TP[k,4]	Trigger point in minus direction Z (3rd geometry axis)	0
GUD6	_TP[k,5]	Trigger point in plus direction Z (3rd geometry axis)	0
GUD6	_TP[k,6]	Edge length/disk diameter	0
GUD6	_TP[k,7]	Assigned internally	133
GUD6	_TP[k,8]	Probe type 0: cube	0
		101: disk in XY	
		201: disk in ZX	
		301: disk in YZ	
GUD6	_TP[k,9]	Distance between upper edge of tool probe and lower edge of tool	2
	Assignment for turning		
GUD6	_TP[k,0]	Trigger point in minus direction, abscissa	0
GUD6	_TP[k,1]	Trigger point in plus direction, abscissa	0
GUD6	_TP[k,2]	Trigger point in minus direction, ordinate	0
GUD6	_TP[k,3]	Trigger point in plus direction, ordinate	0
GUD6	_TP[k,4]	Irrelevant	0
	to		
GUD6	_TP[k,9]	Irrelevant	0
	_TPW[ ]	Tool probe (workpiece-related).	
	Assignment for milling		
GUD6	_TPW[k,0]	Trigger point in minus direction X (1st geometry axis)	0
GUD6	_TPW[k,1]	Trigger point in plus direction X (1st geometry axis)	0
GUD6	_TPW[k,2]	Trigger point in minus direction Y (2nd geometry axis)	0
GUD6	_TPW[k,3]	Trigger point in plus direction Y (2nd geometry axis)	0
GUD6	_TPW[k,4]	Trigger point in minus direction <b>Z</b> (3rd geometry axis)	0
GUD6	_TPW[k,5]	Trigger point in plus direction <b>Z</b> (3rd geometry axis)	0
GUD6	_TPW[k,6]	Edge length/disk diameter	0
GUD6	_TPW[k,7]	Assigned internally	133
GUD6	_TPW[k,8]	Probe type 0: cube	0
		101: disk in XY	
		201: disk in ZX	
01.5		301: disk in YZ	
GUD6	_1PW[k,9]	Distance between upper edge of tool probe and lower edge of	2
		1001	





	Assignment for turning		
GUD6	_TPW[k,0]	Trigger point in minus direction, abscissa	0
GUD6	_TPW[k,1]	Trigger point in plus direction, abscissa	0
GUD6	_TPW[k,2]	Trigger point in minus direction, ordinate	0
GUD6	_TPW[k,3]	Trigger point in plus direction, ordinate	0
GUD6	_TPW[k,4]	irrelevant	0
	to		
GUD6	_TPW[k,9]	irrelevant	0
	_WP[]	Workpiece probe	
GUD6	_WP[k,0]	Ball diameter	6
GUD6	_WP[k,1]	Trigger point in minus direction of abscissa	3
GUD6	_WP[k,2]	Trigger point in plus direction of abscissa	-3
GUD6	_WP[k,3]	Trigger point in minus direction of ordinate	3
GUD6	_WP[k,4]	Trigger point in plus direction of ordinate	-3
GUD6	_WP[k,5]	Trigger point in minus direction of applicate	3
GUD6	_WP[k,6]	Trigger point in plus direction of applicate	-3
GUD6	_WP[k,7]	Position deviation abscissa	0
GUD6	_WP[k,8]	Position deviation ordinate	0
GUD6	_WP[k,9]	Calibration status, coded	0
GUD6	_WP[k,10]	Calibration status, coded	0

Block	Name of identifier	Description	As-delivered value
	_KB[]	Gauging block	
GUD6	_KB[k,0]	Groove edge in plus direction, ordinate	0
GUD6	_KB[k,1]	Groove edge in minus direction, ordinate	0
GUD6	_KB[k,2]	Groove base in abscissa	0
GUD6	_KB[k,3]	Groove edge in plus direction, abscissa	0
GUD6	_KB[k,4]	Groove edge in minus direction, abscissa	0
GUD6	_KB[k,5]	Upper edge groove in ordinate	0
GUD6	_KB[k,6]	Groove base in ordinate	0
	_CM[]	Monitoring functions _CBIT[12] = 0	
GUD6	_CM[k,0]	Max. permissible peripheral speed [m/min]/[feet/min]	60
GUD6	_CM[k,1]	Max. permissible speed [rpm]	2000
GUD6	_CM[k,2]	Minimum feedrate for probing [mm/min]	1
GUD6	_CM[k,3]	Required measuring accuracy [mm]	0.005
GUD6	_CM[k,4]	Max. permissible feedrate for probing	20
GUD6	_CM[k,5]	Direction of spindle rotation	4
GUD6	_CM[k,6]	Feed factor 1	10
GUD6	_CM[k,7]	Feed factor 2	0
	_MFS[]	Speed and feedrate _CBIT[12] = 1	
GUD6	_MFS[k,0]	Speed 1st probing	0
GUD6	_MFS[k,1]	Feed 1st probing	0
GUD6	_MFS[k,2]	Speed 2nd probing	0
GUD6	_MFS[k,3]	Feed 2nd probing	0
GUD6	_MFS[k,4]	Speed 3rd probing	0
GUD6	_MFS[k,5]	Feed 3rd probing	0



Central valu	ue for logging		
GUD6	_PROTFORM	Int field for formatting for log	
GUD6	_PROTFORM[0]	Number of line per page	60
GUD6	_PROTFORM[1]	Number of characters per line	80
GUD6	_PROTFORM[2]	First page number	1
GUD6	_PROTFORM[3]	Number of header lines	5
GUD6	_PROTFORM[4]	Number of value lines in the log	1
GUD6	_PROTFORM[5]	Number of characters per column	12
GUD6	_PROTSYM	Separator in the log	
GUD6	_PROTSYM[0]	Separators between the values in the log	"."
GUD6	_PROTSYM[1]	Special characters for identification when tolerance limits are	"; <b>#</b> "
		exceeded	
GUD6	_PMI	Int field for internal flags for logging	
GUD6	_PMI[0]	Current line number	0
GUD6	_PMI[1]	Flag for interim output of log header	0
		1: Log header output	
GUD6	_PMI[2]	Current page number	0
GUD6	_PMI[3]	Number of log files	0
GUD6	_SP_B	Int field for variable column widths	
GUD6	_SP_B[019]	Internal flag	0
GUD6	_DIGIT	Integer number of decimal places	3

Central bits			
Block	Name of	Description	As-delivered
	identifier		value
	_CBIT[]	Central bits	
GUD6	_CBIT[0]	Measurement repetition after violation of dimensional difference and safe area 0: no measurement repeat 1: measurement repeat, up to 4	0
GUD6	_CBIT[1]	Alarm and M0 for measurement repeat with _CBIT[0]=1 0: no alarm, no M0 generated 1: M0 and an alarm are generated before each repeat	0
GUD6	_CBIT[2]	M0 for tolerance alarms "oversize", "undersize", or "permissible dimensional difference exceeded" 0: no generation of M0 for the above alarms 1: generation of M0 for the above alarms	0
GUD6	_CBIT[3]	Central marker for basic dimension system of the control 0: Basic system setting is inches 1: Basic system setting is metric Note. If _CBIT[3] and MD 10240 are not equal, the value in fields TP[], _TPWW[], _KB[], _MV[], _CM, and _MFS are converted.	1
GUD6	_CBIT[4]	currently not assigned	0
GUD6	_CBIT[5]	Tool measurement and calibration in the WCS in CYCLE982 (from measuring cycles SW 5.4) 0: machine-related measurement and calibration 1: workpiece-related measurement and calibration Note: In both cases, the _TP[] field of the probe is used. From measuring cycles SW 6.3, function switchover via _MVAR is available.	0
GUD6	_CBIT[6]	Logging without output of the measuring cycle name and measuring variant (from measuring cycles SW 6.2) 0: Measuring cycle name and measuring variant will be output. 1: These outputs will be suppressed.	0
GUD6	_CBIT[7]	currently not assigned	0
GUD6	_CBIT[8]	Offset of the mono probe setting 0: No compensation 1: offset of spindle by angle _CORA,	0
GUD6	_CBIT[9]	Assigned internally	0
GUD6	_CBIT[10]	currently not assigned	0
GUD6	_CBIT[11]	Selection of log header for logging 0: Standard 1: user-defined	0





GUD6	_CBIT[12]	Feed and speed in CYCLE971 0: calculation by measuring cycle itself 1: set by user in array _MFS[]	0
GUD6	_CBIT[13]	Deletion of the measuring cycle data fields in the GUD6 0: No deletion 1: delete _TP[], _TPW[], _WP[], _KB[], _EV[], _MV[], _CBIT[13]	0
GUD6	_CBIT[14]	Length reference of the workpiece probe in milling measuring cycles (from measuring cycles SW 4.5) 0: length relative to probe ball center 1: length relative to end	0
GUD6	_CBIT[15]	<ul> <li>Transfer of workpiece probe data into the tool offset in CYCLE976</li> <li>(from measuring cycles SW 4.5)</li> <li>0: no transfer</li> <li>1: result of probe ball calculation on <b>calibration</b> will be entered in the geometry memory of the workpiece probe (radius)</li> </ul>	0

Central str	ings		
Block	Name of identifier	Description	As-delivered value
	_SI	Central strings	
GUD6	_SI[0]	currently not assigned	0
GUD6	_SI[1]	Software version	6
Central str	ings for logging		
	_PROTNAME (32 chars)		
GUD6	_PROTNAME [0]	Name of the main program to log from (for log header)	
GUD6	_PROTNAME [1]	Name of the log file to be created	
GUD6	_HEADLINE (80 chars)	Strings for log header (80 chars)	
GUD6	_HEADLINE[09]	The user can enter customized texts in these strings; they are included in the log	
GUD6	_PROTVAL (80 chars)	Strings for log content	
GUD6	_PROTVAL[0]	Content of the header line (line 9)	
GUD6	_PROTVAL[0]	Content of the header line (line 10)	
GUD6	_PROTVAL[25]	Specification of the values to be logged in successive lines	
GUD6	_TXT[100]	String field for formatted strings (12 characters)	

Block	Name of identifier	Description	As-delivered value		
	_EVMVNUM	Number of empirical values and mean values			
GUD6	_EVMVNUM[0]	Number of empirical values	20		
GUD6	_EVMVNUM[1]	Number of mean values	20		
	_SPEED	Traversing velocities for intermediate positioning			
GUD6	_SPEED[0]	Max. rapid traverse in % (only active with collision monitoring switched off, max 100 %)	100		
GUD6	_SPEED[1]	Positioning velocity in the plane with collision monitoring active	1000		
GUD6	_SPEED[2]	Positioning velocity applicate	1000		
GUD6	_SPEED[3]	Fast measuring feed	900		
	_EV	Empirical values			
GUD5	_EV[x]	Empirical value	0		
	_MV	Mean values			
GUD5	_MV[x]	Mean value	0		



Block	Name of identifier	Description	As-delivered value
	_JM_I	I	
GUD6	_ <b>JM_I</b> [0]	Setting for the workpiece probe number	0
		U: set by JMI[1]	
		1: set by tool parameters (ShopMill)	
GUD6	_ <b>JM_I</b> [1]	Probe number for workpiece measurement (_PRNUM) only if JMI[0]=0	1
GUD6	_JM_I [2]	Probe number for tool measurement (_PRNUM)	1
GUD6	_JM_I [3]	Working plane 17: G17 18: G18 19: G19 Every other value for working plane is defined in the machine	17
	164 1 [4]	Odia.	0
GOD6	_JMI_I [4]	0: G500 1: G54	U
		4: G57	
		5: G505	
		100: active ZO is defined in machine data	
GUD6	_JM_I [5] (as from meas. cycle SW 6.3)	Offset 0: standard offset (measuring only, basic reference, settable ZO) 1: extended offset (additional global and channel-specific basic frame)	0
GUD6	_JM_I [6]	Internal copy of _JM_I[4]	0
	(as from meas. cycle SW 6.3)		
GUD6	_JM_I [7] to _JM_I [9]	Reserved	-
	(as from meas. cycle SW 6.3)		





Channel-sp	ecific values (for measuring in	JOG, from measuring cycles SW6.3, GUD7_MC)	
Block	Name of identifier	Description	As-delivered value
GUD7	E MESS IS METRIC	All dimensioned data are metric	1
	E_MESS_IS_METRIC_SPEZ _VAR=1		
GUD7	E_MESS_MS_IN	Measurement input 1 for workpiece measurement	0
GUD7	E_MESS_MT_IN	Measurement input 2 for tool measurement	1
GUD7	E_MESS_D	Internal data item	5
GUD7	E_MESS_D_M	Measuring path for manual measuring [mm] (in front of and behind meas. point)	50
GUD7	E_MESS_D_L	Measuring path for length measurement [mm] (in front of and behind the measuring point) for tool measurement	2
GUD7	E_MESS_D_R	Measuring path for radius measurement [mm] (in front of and behind the measuring point) for tool measurement	1
GUD7	E_MESS_FM	Measuring feed [mm/rev]	300
GUD7	E_MESS_F	Plane feedrate for collision monitoring [mm/min]	2000
GUD7	E_MESS_FZ	Infeed feedrate for collision monitoring [mm/min]	2000
GUD7	E_MESS_CAL_D (from measuring cycles SW 6.3)	Diameter, calibration ring	0
GUD7	E_MESS_CAL_L (from measur. cycles SW 6.3)	Calibration dimension in the feed axis (referred to WCS)	0
GUD7	E_MESS_MAX_V	Max. peripheral speed for measuring with rotating spindle [m/min]	100
GUD7	E_MESS_MAX_S	Max. spindle speed for measuring with rotating spindle [rpm]	1000
GUD7	E_MESS_MAX_F	Max. feedrate for measuring with rotating spindle [mm/min]	20
GUD7	E_MESS_MIN_F	Min. feed for measuring with rotating spindle for the 1st probing [mm/min]	1
GUD7	E_MESS_MIN_F_FAK1	On tool measurement with rotating spindle, traversal with 10 times measuring feed is performed in the 1st probing (limitation by E_MESS_MAX_F) [mm/min]	10
GUD7	E_MESS_MIN_F_FAK2	For tool measurement with rotating spindle, 2nd probing is performed with measuring feed. There is no 3rd probing [mm/min]	0
GUD7	E_MESS_MIN_D	Measuring accuracy for measuring with rotating spindle [mm/min]	0.01
GUD7	E_MESS_MT_TYP[3]	Type of tool probe	0
GUD7	E_MESS_MT_AX[3]	Permissible axis directions for tool probe	133
GUD7	E_MESS_MT_DL[3] 1)	Diameter of tool probe for length measurement	0
GUD7	E_MESS_MT_DR[3] 1)	Diameter of tool probe for radius measurement	0
GUD7	E_MESS_MT_DZ[3]	Infeed for measurement tool probe diameter	2
GUD7	E_MESS_MT_DIR[3]	Approach direction in the plane tool probe	-1
GUD7	E_MESS_MT_D	Calibrate measurement path for tool probe and tool measurement with motionless spindle (before and after expected switching position)	10
GUD7	E_MESS_MT_FM	Calibrate measuring feed for tool probe and tool measurement with motionless spindle	100
GUD7	E_MESS_MT_CF	No tool probe make (manufacturer) specified	0
GUD7	E_MESS_MT_COMP	No offset of the measurement result on tool measurement with rotating spindle	0
GUD7	E_MESS[3]	Internal data item	
GUD7	E_MEAS	Internal data item	
GUD7	E_MESS_RETT	Internal data item	
GUD7	E_MESS_SETT[10]	Field for settings	
GUD7	E_MESS_AM	Internal data item	

Α



1) During installation value input is mandatory here!



Channel-oriented bits

Block	Name of	Description	As-delivered
DIOCK	identifier	Description	value
	CHBIT	Channel hits	Value
GUD6		Measurement input for workpiece measurement	0
0000		0: measuring input 1 1: measurement input 2	Ū
GUD6	_CHBIT[1]	Measurement input for tool measurement 0: measuring input 1 1: measurement input 2	1
GUD6	_CHBIT[2]	Collision monitoring for intermediate positioning 0: OFF 1: ON	1
GUD6	_CHBIT[3]	Tool offset mode for tool measurement 0: first-time measurement (determining geometry) 1: remeasuring (determining wear)	0
GUD6	_CHBIT[4]	Mean value for workpiece measurement with automatic tool correction (_EVNUM>0) 0: no mean value derivation over several parts 1: with mean value formation and calculation	0
GUD6	_CHBIT[5]	Inclusion of empirical value (_EVNUM>0) 0: subtraction of actual value 1: addition to actual value	0
GUD6	_CHBIT[6]	Tool offset mode for workpiece measurement with automatic tool offset 0: Offset in wear 1: offset in geometry, delete wear From meas. cycles SW 6.3: For additive and setup offset and _CHBIT[8]=0: 0: offset in additive offset 1: offset in set-up offset_delete additive offset	0
GUD6	_CHBIT[7]	Measured value offset in CYCLE994 0: use of trigger values of the probe _WP[k,1] 1: use of the active ball diameter of the probe _WP[k 0]	0
GUD6	_CHBIT[8]	From meas. cycles SW 6.3: Offset mode for workpiece measurement with automatic tool offset 0: additive, setup offset according to _CHBIT[6]	0
GUD6	_CHBIT[9]	currently not assigned	0
GUD6	_CHBIT[10]	Measuring result display 0: OFF 1: ON	0
GUD6	_CHBIT[11]	Acknowledgment measurement result screen with NC start 0: OFF (If _CHBIT[18]=0, the display is automatically deselected at end of cycle.) 1: ON (M0 is generated in the cycle.)	0
GUD6	_CHBIT[12]	currently not assigned	0
GUD6	_CHBIT[13]	Coupling spindle position with coordinate rotation in active plane for workpiece measurement with multi probe 0: OFF 1: ON	0
GUD6	_CHBIT[14]	Adapt spindle positioning, if _CHBIT[13]=1 0: according to default 1: adapted angle values	0
GUD6	_CHBIT[15]	Number of measurements on failure to switch 0: up to 5 measurements 1: only 1 measurement	0
GUD6	_CHBIT[16]	Retraction velocity from the measuring point 0: velocity as for intermediate positioning 1: with percentage of rapid traverse velocity (_SPEED[0])(only active with collision monitoring ON) _CHBIT[2]=1)	0
GUD6	_CHBIT[17]	Feed during measurement 0: with feed in _VMS 1: on 1st measurement feed in _SPEED[3] on 2nd measurement with feed in _VMS	0



GUD6	_CHBIT[18]	Static measurement result display	0
		0: effect as set in _CHBIT[11].	
		1: only active if _CHBIT[11]=0:	
		Measuring result display remains until next measuring cycle is called	
GUD6	_CHBIT[19]	(CYCLE974 and CYCLE994 only):	0
		Special treatment of Y axis with G18	
		0: no special treatment	
		1: setpoint setting and parameterization (_SETVAL, _TUL, _TLL, SZO) for the Y	
		The tool offset is applied in the length that is active in the ordinate (X axis)	
		(usually 1) as long as not other length has been set in KNIIM	
		The 70 correction is applied in the specified 70 memory in the ordinate	
		component (X axis)	
GUD6	_CHBIT[20]	(CYCLE982 only):	0
		Suppression of the starting angle positioning _STA1	
		0: suppression OFF	
		1: suppression ON	
GUD6	_CHBIT[21]	(CYCLE974, CYCLE977, CYCLE978, CYCLE979, CYCLE9997 only)	0
		Mode of ZO correction	
		0: offset additive in FINE	
		1: offset in COARSE, delete FINE	
GUD6	_CHBIT[22]	(CYCLE971 only):	0
		with rotating spindle and multiple measurement with rotating spindle and multiple	
		0: last measurement with reduced speed if CBIT[12]=0	
		1: no speed reduction	
GUD6	CHBIT[23]	From measuring cycles SW 6.3 (CYCLE982 only)	0
		Recoding of tool point direction during tool measurement	
		0° no recoding	
		1: internal recording (tool point direction mirroring about X)	
GUD6	CHBIT[24]	Channel-oriented marker for basic dimension system	1
		(On deviation from MD 10240, cycle parameters are converted internally and this	
		bit is reset.)	
		0: The basic system is inch	
		1: The basic system is metric	

Channel-specific bits (for measurement in JOG)				
Block	Name of identifier	Description	Value for meas. cycles	
	_JM_B			
GUD6	_ <b>JM_B</b> [0]	Tool offset mode for tool measurement	0	
		0: offset in geometry, wear is deleted		
GUD6	_ <b>JM_B</b> [1]	Number of measurement attempts	0	
GUD6	_JM_B[2]	Retraction from measuring point in rapid traverse	0	
GUD6	_JM_B[3]	Fast measuring feed	0	
GUD6	_ <b>JM_B</b> [4]	Selection offset frame	0	
GUD6	_JM_B[5]	Currently not assigned	0	
GUD6	_ <b>JM_B</b> [6]	Internal data item	0	



В



ASUB	Asynchronous subroutine
CNC	Computerized Numerical Control
CPU	Central Processing Unit
DIN	Deutsche Industrie-Norm (German Industry Standard)
DOS	Disk Operating System
DRF	Differential Resolver Function (handwheel)
FM-NC	Function Module Numerical Control
GUD	Global User Data
I/O	Input/Output
IBN	Startup
JOG	Jogging: Setup mode
LUD	Local User Data
MCS	Machine coordinate system
MD	Machine data
ММС	Man-Machine Communication: User interface on numerical control systems for operator control, programming and simulation
MS	MicroSoft (software manufacturer)
NC	Numerical Control
NCK	Numerical Control Kernel: NC kernel with block preparation, traversing range, etc.
NCU	Numerical Control Unit: NCK hardware unit
OI	User interface



PCIN	Name of SW for data exchange with the control
PG	Programming device
PLC	Programmable Logic Control: Programmable logic control
RS-232-C	Serial interface (definition of the exchange lines between DTE and DCE)
SR	Subprogram
SW	Software
то	Tool Offset
ΤΟΑ	Tool Offset Active identifier (file type) for tool offsets
WCS	Workpiece coordinate system
ZO	Zero offset



С

Terms

	Important terms are listed in alphabetical order. The symbol "->" precedes terms that are explained in a separate entry in this list.
Α	
Actual/set difference	Difference between measured and expected value.
Asynchronous subroutine	A parts program, which can be started asynchronously to (independently of) the current program status by an interrupt signal (e.g., "rapid NC input" signal).
В	
Blank measurement	The blank measurement ascertains the position, deviation, and zero offset of the workpiece in the result of a -> workpiece measurement.
С	
Calibrating tool	Is a special tool (usually a cylindrical stylus), whose dimensions are known and that is used for precisely determining the distances between the machine zero and the probe trigger point (of the workpiece probe).
Calibration	During calibration, the trigger points of the probe are ascertained and stored in the measuring cycle data in block GUD6.
Collision monitoring	In the context of measuring cycles, this is a function that monitors all intermediate positions generated within the measuring cycle for the switching signal of the probe. When the probe switches, motion is stopped immediately and an alarm message is output.
D	
Data blocks for measuring cycles	Data blocks GUD5.DEF, GUD6.DEF, GUD7DEF and GUD7.MC.DEF contain data required for configuration and execution of the measuring cycles. These blocks must be loaded into the control during start-up. They must then be adapted according to the characteristics of the relevant machine by the machine manufacturer. They are stored in the nonvolatile storage area of the control such that their setting values remain stored even when the control is switched off and on.



Delete distance-to-go	If a measuring point is to be approached, a traverse command is transmitted to the position control loop and the probe is moved towards the measuring point. A point behind the expected measuring point is defined as setpoint position. As soon as the probe makes contact, the actual axis value at the time the switching position is reached is measured and the drive is stopped. The remaining "distance-to-go" is deleted.
Differential measurement	Differential measurement means that the measuring point is measured twice, the first time at the probe position reached and the second time with a spindle reversal of 180° (rotation of probe through 180°).
Dimension difference check	Is a tolerance window. On reaching a limit (_TDIF) the tool will probably be worn and have to be replaced. The dimension difference check has no effect on generation of the compensation value.
E	
Empirical value	The empirical values are used to suppress constant dimensional deviations that are not subject to a trend.
F-K	
L	
Logging measurement results	Measurement results can be optionally be logged in a file located in the part program memory. The log can be output from the control either via RS-232-C or on a diskette.
М	
Mean value	The mean value calculation takes account of the trend of the dimensional deviations of a machining series. The → weighting factor k from which the mean value is derived is selectable. Mean value calculation alone is not enough to ensure constant machining quality. The measured dimensional deviation can be corrected for constant deviations without a trend by an -> empirical value.
Measurement at an angle	A measurement variant used to measure a drill-hole, shaft, groove, or web at random angles. The measurement path is traveled at a certain set angle defined in the WCS.





Measurement path	Measurement path _FA defines the distance between the starting position and the expected switching position (setpoint) of the probe. Always specify _FA in <b>mm</b> .
Measurement result	Measurement result displays can be shown automatically during
display	measuring cycle runtime. Activation of this function depends on the settings in the measuring cycle data.
Measuring accuracy	<ul> <li>The measurement accuracy that can be obtained is dependent on the following factors:</li> <li>Repeat accuracy of the machine</li> <li>Repeatability of the probe</li> <li>Resolution of the measuring system</li> <li>The repeat accuracy of the controls for "on-the-fly measurement" is ±1 μm.</li> </ul>
Measuring in JOG	<ul> <li>It contains the following functions:</li> <li>Semi-automatic calculation of tool lengths and storage in tool offset memory</li> <li>Semi-automatic calculation and setting of reference points and storage in zero offset memory</li> <li>The function is operated with softkeys and input displays.</li> </ul>
Measuring the tool	To perform tool measurement, the changed tool is moved up to the probe which is either permanently fixed or swiveled into the working range. The automatically derived tool geometry is entered in the relevant tool offset data record.
Measuring variant	The measuring variant of each measuring cycle is defined in parameter _MVAR. The parameter can have certain integer values for each measuring cycle, which are checked for validity within the cycle.
Measuring velocity	The measuring speed can be freely selected by means of parameter _VMS. The maximum measuring speed must be selected such that safe deceleration within the probe deflecting path is ensured.
Mono probe	A mono(directional) probe is a probe that can only deflect in one direction. It can only be used for workpiece measurement on milling machines and machining centers with slight limitations.
Multi probe	A multi(directional) probe is one that can deflect in three dimensions.



Multiple measurement at the same location	Parameter _NMSP can be used to determine the number of measurements at the same location. The actual/set difference is determined arithmetically.
N	
0	
Offset angle position	If a -> mono probe is used, the position of the probe can also be corrected for machine-specific reasons using the parameter _CORA.
Offset axis	With some measuring variants, for example, in CYCLE998, positioning in another axis that must be defined, also called offset axis can be performed between measurements in the measuring axis. This is must be defined in parameter _MA with offset axis/measuring axis.
On-the-fly measurement	This method processes the probe signal directly in the NC.
Р	
Paraxial measurement	A measuring variant used for paraxial measurement of a workpieces, such as a drill-hole, shaft, rectangle, etc. The measuring path is traveled paraxially.
Positional deviation	The positional deviation (skew) describes the difference between the spindle center and the probe tip center ascertained by calibration. It is compensated for by the measuring cycles.
Probe ball diameter	The active diameter of the probe ball. It is ascertained during calibration and stored in the measuring cycle data.
Probe type	<ul> <li>In order to measure tool and workpiece dimensions, a touch-trigger probe is required that supplies a constant signal (rather than a pulse) when deflected.</li> <li>Probes are therefore classified in three groups according to the number of directions in which they can be deflected.</li> <li>Multidirectional</li> <li>Bidirectional</li> <li>Monodirectional</li> </ul>
Q	
R	
Reference groove	A groove located in the working area (permanent feature of the machine) whose precise position is known and that can be used to calibrate workpiece probes.



S	
Safe area	The safe area _TSA does not affect the offset value; it is used for diagnosis. If this limit is reached, there is a defect in the probe or the set position is incorrect.
Setpoint	In the measuring procedure "inprocess measurement", a position is specified as the -> setpoint value for the cycle at which the signal of the touch-trigger probe is expected.
т	
Tolerance bottom limit	When measuring a dimensional deviation as the lower tolerance limit (_TLL) ranging between "2/3 tolerance of workpiece" and "Dimensional difference control", this is regarded 100% as tool compensation. The previous average value is erased.
Tolerance top limit	When measuring a dimensional deviation as the upper tolerance limit (_TUL) ranging between "2/3 tolerance of workpiece" and "Dimensional difference control", this is regarded 100% as tool compensation. The previous average value is erased.
Tool environment	As from NCK SW 6.3, you can now save the operating environment of a particular tool you are using. This is to allow you to correct the tool used to measure a workpiece taking into account the operating conditions (environment: G commands, setting data,). You then no longer have to specify the T, D, DL number in the offset explicitly. These are included in the stored tool environment. The name of a tool environment can have up to 32 characters.
Tool name	If tool management is active, the name of the tool can be entered in parameter _TNAME as an alternative to the -> tool number. The tool number is derived from it within the cycle and entered in _TNUM.
Tool number	The parameter _TNUM contains the tool number of the tool to be automatically offset after workpiece measurement.
Trigger point	The trigger points of the probe are ascertained during calibration and stored in block GUD6 for the axis direction.

U



V	
Variable measuring speed	The measuring speed can be freely selected by means of _VMS. The maximum measuring velocity must be selected to ensure safe deceleration within the probe deflecting path> Measuring velocity
w	
Weighting factor for mean value calculation	The weighting factor k can be applied to allow different weighting to be given to an individual measurement. A new measurement result thus has only a limited effect on the new tool offset as a function of k.
Workpiece measurement	For workpiece measurement, a measuring probe is moved up to the clamped workpiece in the same way as a tool. The flexibility of measuring cycles makes it possible to perform nearly all measurements required on a milling or turning machine.
x	
Y	
Z	
Zero offset area	This tolerance range (lower limit _TZL) corresponds to the amount of maximum accidental dimensional deviations. If the absolute value of the actual/set difference is less than the zero offset range, the offset is not applied.
ZO determination	In the result of a measurement, the actual-setpoint value difference is stored in the data set of any settable zero offset.



## D References

	General Documentation	
/BU/	SINUMERIK & SIMODRIVE, Automation Systems for Machine Tools Catalog NC 60 Order No.: E86060-K4460-A101-A9-7600	
/IKPI/	Industrial Communications and Field Devices Catalog IK PI Order No.: E86060-K6710-A101-B2-7600	
/ST7/	SIMATIC Products for Totally Integrated Automation and Micro Automation Catalog ST 70 Order No.: E86060-K4670-A111-A8-7600	
ΙΖΙ	OTION-CONNECT onnections & System Components for SIMATIC, SINUMERIK, ASTERDRIVES, and SIMOTION atalog NC Z rder No.: E86060-K4490-A001-B1-7600	
	Electronic Documentation	
/CD1/	The SINUMERIK System <b>DOC ON CD</b> (includes all SINUMERIK 840D/840Di/810D/802- and SIMODRIVE publications) Order No.: 6FC5298-7CA00-0BG0	(10.04 Edition)



#### **User Documentation**

/AUK/	SINUMERIK 840D/810D Short Guide <b>AutoTurn</b> Order No.: 6FC5298-4AA30-0BP2	(09.99 Edition)
/AUP/	SINUMERIK 840D/810D Operator's Guide <b>AutoTurn Graphic Programming Sy</b> Programming/Setup Order No.: 6FC5298-4AA40-0BP3	(02.02 Edition) /stem
/BA/	SINUMERIK 840D/810D Operator's Guide <b>MMC</b> Order No.: 6FC5298-6AA00-0BP0	(10.00 Edition)
/BAD/	SINUMERIK 840D/840Di/810D Operator's Guide <b>HMI Advanced</b> Order No.: 6FC5298-6AF00-0BP3	(03.04 Edition)
/BAH/	SINUMERIK 840D/840Di/810D Operator's Guide <b>HT 6</b> Order No.: 6FC5298-0AD60-0BP3	(03.04 Edition)
/BAK/	SINUMERIK 840D/840Di/810D Short Guide Operation Order No.: 6FC5298-6AA10-0BP0	(02.01 Edition)
/BAM/	SINUMERIK 810D/840D Operation/Programming <b>ManualTurn</b> Order No.: 6FC5298-6AD00-0BP0	(08.02 Edition)
/BAS/	SINUMERIK 840D/840Di/810D Operation/Programming <b>ShopMill</b> Order No.: 6FC5298-6AD10-0BP3	(10.04 Edition)
/BAT/	SINUMERIK 840D/810D Operation/Programming <b>ShopTurn</b> Order No.: 6FC5298-6AD50-0BP2	(06.03 Edition)
/BEM/	SINUMERIK 840D/810D Operator's Guide <b>HMI Embedded</b> Order No.: 6FC5298-6AC00-0BP3	(03.04 Edition)
/BNM/	SINUMERIK 840D/840Di/810D User's Guide <b>Measuring Cycles</b> Order No.: 6FC5298-7AA70-0BP1	(10.04 Edition)
/BTDI/	SINUMERIK 840D/840Di/810D Motion Control Information System (MCIS) User's Guide <b>Tool Data Information</b> Order No.: 6FC5297-6AE01-0BP1	(06.04 Edition)
/CAD/	SINUMERIK 840D/840Di/810D Operator's Guide <b>CAD Reader</b> Order No.: (included in online help)	(03.02 Edition)
/DA/	SINUMERIK 840D/840Di/810D Diagnostics Guide Order No.: 6FC5298-6AA20-0BP4	(03.04 Edition)

Λ\



/KAM/	SINUMERIK 840D/810D Short Guide <b>ManualTurn</b> Order No.: 6FC5298-5AD40-0BP0	(04.01 Edition)
/KAS/	SINUMERIK 840D/810D Short Guide <b>ShopMill</b> Order No.: 6FC5298-5AD30-0BP0	(04.01 Edition)
/KAT/	SINUMERIK 840D/810D Short Guide <b>ShopTurn</b> Order No.: 6FC5298-6AF20-0BP0	(07.01 Edition)
/PG/	SINUMERIK 840D/840Di/810D Programming Guide <b>Fundamentals</b> Order No.: 6FC5298-7AB00-0BP0	(03.04 Edition)
/PGA/	SINUMERIK 840D/840Di/810D Programming Guide <b>Advanced</b> Order No.: 6FC5298-7AB10-0BP0	(03.04 Edition)
/PGA1/	SINUMERIK 840D/840Di/810D Lists Manual <b>System Variables</b> Order No.: 6FC5298-7AE10-0BP0	(03.04 Edition)
/PGK/	SINUMERIK 840D/840Di/810D Short Guide <b>Programming</b> Order No.: 6FC5298-7AB30-0BP0	(03.04 Edition)
/PGM/	SINUMERIK 840D/840Di/810D Programming Guide ISO Milling Order No.: 6FC5298-6AC20-0BP2	(11.02 Edition)
/PGT/	SINUMERIK 840D/840Di/810D Programming Guide <b>ISO Turning</b> Order No.: 6FC5298-6AC10-0BP2	(11.02 Edition)
/PGZ/	SINUMERIK 840D/840Di/810D Programming Guide <b>Cycles</b> Order No.: 6FC5 298-7AB40-0BP1	(10.04 Edition)
/PI/	PCIN 4.4 Software for Data Transfer to/from <b>MMC Modules</b> Order number: 6FX2060-4AA00-4XB0 (English, German, French) Order from: WK Fürth	
/SYI/	SINUMERIK 840Di <b>System Overview</b> Order No.: 6FC5298-6AE40-0BP0	(02.01 Edition)



a) Lists	Manufacturer/Service Documentation	
/LIS/	SINUMERIK 840D/840Di/810D SIMODRIVE 611D Lists	(03.04 Edition)
	Order No.: 6FC5297-7AB70-0BP0	
b) Hardware /ASAL/	SIMODRIVE Planning Guide General Information for <b>Asynchronous</b> Order No.: 6SN1197-0AC62-0BP0	(10.03 Edition) <b>5 Motors</b>
/APH2/	SIMODRIVE Planning Guide <b>Asynchronous Motors 1PH2</b> Order No.: 6SN1197-0AC63-0BP0	(10.03 Edition)
/APH4/	SIMODRIVE Planning Guide <b>Asynchronous Motors 1PH4</b> Order No.: 6SN1197-0AC64-0BP0	(10.03 Edition)
/APH7S/	SIMODRIVE Planning Guide <b>Asynchronous Motors 1PH7</b> Order No.: 6SN1197-0AC65-0BP0	(04.04 Edition)
/APL6/	MASTERDRIVES VC/MC Planning Guide <b>Asynchronous Motors 1PL6</b> Order No.: 6SN1197-0AC67-0BP0	(03.04 Edition)
/BH/	SINUMERIK 840D/840Di//810D Operator Components Manual (HW) Order No.: 6FC5297-6AA50-0BP0	(04.04 Edition)
/BHA/	SIMODRIVE Sensor User Guide (HW) <b>Absolute Position Sensor with Pro</b> Order No.: 6SN1197-0AB10-0YP3	(03.04 Edition) fibus DP
/EMV/	SINUMERIK, SIROTEC, SIMODRIVE Planning Guide EMC Installation Guide Order No.: 6FC5297-0AD30-0BP2	(03.04 Edition)
	The current Declaration of Conformity is available unde Internet address: <u>http://www4.ad.siemens.de</u>	r the following
	Please enter the ID no. 15257461 in the "Search" field ( on "go".	(top right) and click
/GHA/	SINUMERIK/SIMOTION <b>ADI4 - Analog Drive Interface for 4 Axes</b> Manual Order No.: 6FC5297-0BA01-0BP1	(02.03 Edition)
/PFK6/	SIMODRIVE 611, SIMOVERT MASTERDRIVES Planning Guide <b>1FK6 Three-Phase AC Servomotors</b> Order No.: 6SN1197-0AD05-0BP0	(05.03 Edition)
/PFK7/	SIMODRIVE 611, SIMOVERT MASTERDRIVES Planning Guide <b>1FK7 Three-Phase AC Servomotors</b> Order No.: 6SN1197-0AD06-0BP0	(01.03 Edition)

/Δ`

/PFS6/	SIMOVERT MASTERDRIVES Planning Guide <b>1FS6 Three-Phase AC Servomotors</b> Order No.: 6SN1197-0AD08-0BP0	(07.03 Edition)
/PFT5/	SIMODRIVE Planning Guide <b>1FT5 Three-Phase AC Servomotors</b> Order No.: 6SN1197-0AD01-0BP0	(02.04 Edition)
/PFT6/	SIMODRIVE 611, SIMOVERT MASTERDRIVES Planning Guide <b>1FT6 Three-Phase AC Servomotors</b> Order No.: 6SN1197-0AD02-0BP0	(01.04 Edition)
/PFU/	SINAMICS, SIMOVERT MASTERDRIVES, MICROMASTER SIEMOSYN Motors 1FU8 Order No.: 6SN1197-0AC80-0BP0	(09.03 Edition)
/PHC/	SINUMERIK 810D Configuring Manual CCU (HW) Order No.: 6FC5297-6AD10-0BP1	(11.02 Edition)
/PHD/	SINUMERIK 840D Configuring Manual NCU (HW) Order No.: 6FC5297-6AC10-0BP3	(11.04 Edition)
/PJAL/	SIMODRIVE 611, SIMOVERT MASTERDRIVES Planning Guide <b>Three-Phase Servomotors</b> <b>General Part for 1FT / 1FK Motors</b> Order No.: 6SN1197-0AD07-0BP1	(01.03 Edition)
/PJFE/	SIMODRIVE Planning Guide <b>1FE1 Built-In Synchronous Motors</b> Three-Phase AC Motors for Main Spindle Drives Order No.: 6SN1197-0AC00-0BP4	(02.03 Edition)
/PJF1/	SIMODRIVE Installation Guide <b>1FE1 0511FE1 147</b> . <b>Built-In Synch</b> AC Motors for Main Spindle Drives Order No.: 610.43000.02	(12.02 Edition) ronous Motors
/PJLM/	SIMODRIVE Planning Guide <b>1FN1, 1FN3 Linear Motors</b> ALL General Information about Linear Motor 1FN1 1FN1 Three-Phase AC Linear Motor 1FN3 1FN3 Three-Phase AC Linear Motor CON Connections Order No.: 6SN1197-0AB70-0BP3	(06.02 Edition)
/PJM/	SIMODRIVE 611, SIMOVERT MASTERDRIVES Planning Guide <b>Motors</b> Three-Phase Motors for Feed and Main Spindle Drives Order No.: 6SN1197-0AA20-0BP4	(11.00 Edition)
/PJM2/	SIMODRIVE Planning Guide <b>Servomotors</b> Three-Phase Servo Motors for Feed and Main Spindle I Order No.: 6SN1197-0AC20-0BP0	(07.03 Edition) Drives



/PJTM/	SIMODRIVE Planning Guide <b>1FW6 Built-In Torque Motors 1FW6</b> Order No.: 6SN1197-0AD00-0BP2	(11.03 Edition)
/PJU/	SIMODRIVE 611 Planning Guide <b>Converters</b> Order No.: 6SN1197-0AA00-0BP7	(10.04 Edition)
/PMH/	SIMODRIVE Sensor Configuring/Installation Guide <b>Hollow-Shaft Measuring System SIMAG H</b> Order No.: 6SN1197-0AB30-0BP1	(07.02 Edition)
/PMHS/	SIMODRIVE Installation Guide <b>Measuring System for Main Spind</b> SIZAG2 Toothed-Wheel Encoder Order No.: 6SN1197-0AB00-0YP3	(12.00 Edition) I <b>le Drives</b>
/PMS/	SIMODRIVE Planning Guide <b>ECO Motor Spindle for Main Spindle Drives</b> Order No.: 6SN1197-0AD04-0BP1	(03.04 Edition)
/PPH/	SIMODRIVE Planning Guide <b>1PH2 / 1PH4 / 1PH7 Motors</b> AC Induction Motors for Main Spindle Drives Order No.: 6SN1197-0AC60-0BP0	(12.01 Edition)
/PPM/	SIMODRIVE Planning Guide Hollow-Shaft Motors for <b>1PM4 and 1PM6</b> Main Spindl Order No.: 6SN1197-0AD03-0BP0	(11.01 Edition) e Drives
c) Software /FB1/	<ul> <li>SINUMERIK 840D/840Di/810D/FM-NC</li> <li>Description of Function Basic Machine (Part 1) (the various sections are listed below)</li> <li>Order No.: 6FC5297-7AC20-0BP0</li> <li>A2 Various Interface Signals</li> <li>A3 Axis Monitoring, Protection Zones</li> <li>B1 Continous-Path Mode, Exact Stop and Look A</li> <li>B2 Acceleration</li> <li>D1 Diagnostic Tools</li> <li>D2 Interactive Programming</li> <li>F1 Traverse to Fixed Stop</li> <li>G2 Velocities, Setpoint/Actual Value Systems, Cloc</li> <li>H2 Output of Auxiliary Functions to PLC</li> <li>K1 Mode Group, Channel, Program Operation Mode</li> <li>K2 Axes, Coordinate Systems, Frames, Actual-Value System for Workpiece, External 1</li> <li>K4 Communication</li> <li>N2 EMERGENCY STOP</li> <li>P1 Traverse Axes</li> <li>P3 Basic PLC Program</li> <li>R1 Reference Point Approach</li> <li>Spindles</li> <li>V1 Feeds</li> </ul>	(03.04 Edition) head osed-Loop Control ode Zero Offset



/FB2/

SINUMERIK 840D/840Di/810D	(03.04 Edition)
---------------------------	-----------------

- Description of Functions Extended Functions (Part 2) including FM-NC: Turning, Stepper Motor (the various sections are listed below)
- Order No.: 6FC5297-7AC30-0BP0
- A4 Digital and Analog NCK I/Os
- B3 Several Operator Panels and NCUs
- B4 Operation via PG/PC
- F3 **Remote Diagnostics**
- JOG with/without Handwheel H1
- K3 Compensations
- K5 Mode Groups, Channels, Axis Replacement
- L1 FM NC Local Bus
- M1 **Kinematic Transformation**
- M5 Measurement
- Software Cams, Position Switching Signals N3
- N4 Punching and Nibbling
- P2 **Positioning Axes**
- P5 Oscillation
- R2 Rotary Axes
- S3 Synchronous Spindle
- S5 Synchronized Actions (SW 3 and lower, higher /FBSY/)
- S6 Stepper Motor Control
- S7 Memory Configuration

SINUMERIK 840D/840Di/810D

- T1 Indexing Axes
- W3 **Tool Change**
- W4 Grinding

/FB3/

Description of Functions Special Functions (Part 3)

(08.04 Edition)

(the various sections are listed below) Order No.: 6FC5297-7AC80-0BP1

- F2 3-Axis to 5-Axis Transformation
- Gantry Axes G1
- G3 **Clock Times**
- K6 Contour Tunnel Monitoring
- Coupled Motion and Leading Value Coupling М3
- Constant Workpiece Speed for Centerless Grinding S8
- Т3 Tangential Control
- TE0 Installation and Activation of Compile Cycles
- TE1 **Clearance Control**
- TE2 Analog Axes
- TE3 Master-Slave for Drives
- TE4 Transformation Package Handling
- TE5 Setpoint Exchange
- MCS Coupling TE6
- **Retrace Support** TE7
- TE8 Pulse-Independent Path-Synchronized Switching Signal Output
- V2 Preprocessing
- W5 **3D Tool Radius Compensation**





/FBA/	SIMODRIVE 611D/SINUMERIK 840D/810D Description of Functions <b>Drive Functions</b> (the various sections are listed below) Order No.: 6SN1197-0AA80-1BP1 DB1 Operating Messages/Alarm Reactions DD1 Diagnostic Functions DD2 Speed Control Loop DE1 Extended Drive Functions DF1 Enable Commands DG1 Encoder Parameterization DL1 Linear Motor MD DM1 Calculating Motor/Power Section Parameters an DS1 Current Control Loop DÜ1 Monitors/Limitations	(03.04 Edition) nd Controller Data
/FBAN/	SINUMERIK 840D/SIMODRIVE 611 DIGITAL Description of Functions <b>ANA MODULE</b> Order No.: 6SN1197-0AB80-0BP0	(02.00 Edition)
/FBD/	SINUMERIK 840DDescription of Functions <b>Digitizing</b> Order No.: 6FC5297-4AC50-0BP0DI1Start-UpDI2Scan with Tactile Sensor (scancad scan)DI3Scan with Laser (scancad laser)DI4Milling Program Generation (scancad mill)	(07.99 Edition)
/FBDM/	SINUMERIK 840D/840Di/810D Description of Functions DNC NC Program Managemer Order No.: 6FC5297-1AE81-0BP0	(09.03 Edition) nt
/FBDN/	SINUMERIK 840D/840Di/810D Motion Control Information System (MCIS) Description of Functions <b>DNC NC Program Manageme</b> Order No.: 6FC5297-1AE80-0BP0 DN1 DNC Plant / DNC Cell DN2 DNC IFC SINUMERIK, NC Data Transfer via Ne	(03.03 Edition) ent etwork
/FBFA/	SINUMERIK 840D/840Di/810D Description of Functions <b>ISO Dialects for SINUMERIK</b> Order No.: 6FC5297-6AE10-0BP3	(11.02 Edition)
/FBFE/	SINUMERIK 840D/810D Description of Functions <b>Remote Diagnosis</b> Order No.: 6FC5297-0AF00-0BP2 FE1 Remote Diagnosis ReachOut FE3 Remote Diagnosis pcAnywhere	(04.03 Edition)
/FBH/	SINUMERIK 840D/840Di/810DHMI Configuring PackageOrder No.: (supplied with the software)Part 1User's GuidePart 2Description of Functions	(02.04 Edition)
/FBH1/	SINUMERIK 840D/840Di/810D HMI Configuring Package ProTool/Pro Option SINUMERIK Order No.: (supplied with the software)	(03.03 Edition)

/FBHL/	SINUMERIK 840D/SIMODRIVE 611 digital Description of Functions <b>HLA Module</b> Order No.: 6SN1197-0AB60-0BP3	(10.03 Edition)
/FBIC/	SINUMERIK 840D/840Di/810D Motion Control Information System (MCIS) Description of Functions <b>TDI Ident Connection</b> Order No.: 6FC5297-1AE60-0BP0	(06.03 Edition)
/FBMA/	SINUMERIK 840D/810D Description of Functions <b>ManualTurn</b> Order No.: 6FC5297-6AD50-0BP0	(08.02 Edition)
/FBO/	SINUMERIK 840D/810DDescription of Functions Configuring OP 030 Operator(the various sections are listed below)Order No.: 6FC5297-6AC40-0BP0BAOperator's GuideEUDevelopment Environment (Configuring PackagePSEIntroduction to Configuring of Operator InterfaceIKScreen Kit: Software Update and Configuration	(09.01 Edition) Interface ge)
/FBP/	SINUMERIK 840D Description of Functions <b>C-PLC Programming</b> Order No.: 6FC5297-3AB60-0BP0	(03.96 Edition)
/FBR/	SINUMERIK 840D/810D IT Solutions Description of Functions <b>RPC SINUMERIK Computer I</b> Order No.: 6FC5297-6AD61-0BP0 NFL Host Computer Interface NPL PLC/NCK Interface	(01.04 Edition) L <b>ink</b>
/FBSI/	SINUMERIK 840D/SIMODRIVE Description of Functions SINUMERIK <b>Safety Integrate</b> Order No.: 6FC5297-6AB80-0BP2	(11.03 Edition) d
/FBSP	SINUMERIK 840D/840Di/810D Description of Functions <b>ShopMill</b> Order No.: 6FC5297-6AD80-0BP2	(11.03 Edition)
/FBST/	SIMATIC Description of Functions <b>FM STEPDRIVE/SIMOSTEP</b> Order No.: 6SN1197-0AA70-0YP4	(01.01 Edition)
/FBSY/	SINUMERIK 840D/810D Description of Functions <b>Synchronized Actions</b> Order No.: 6FC5297-7AD40-0BP2	(03.04 Edition)
/FBT/	SINUMERIK 840D/810D Description of Functions <b>ShopTurn</b> Order No.: 6FC5297-6AD70-0BP2	(03.04 Edition)
/FBTC/	SINUMERIK 840D/810D IT Solutions Description of Functions <b>Tool Data Communication Si</b> Order No.: 6FC5297-5AF30-0BP0	(01.02 Edition) inTDC



/FBTD/	SINUMERIK 840D/810D IT Solutions Description of Functions <b>Tool Information System (Si</b> with Online Help Order No.: 6FC5297-6AE00-0BP0	(02.01 Edition) nTDI)
/FBTP/	SINUMERIK 840D/840Di/810D Motion Control Information System (MCIS) Description of Functions <b>TPM Total Productive Mainte</b> Order No.: Document is supplied with the software	(01.03 Edition)
/FBU/	SIMODRIVE 611 universal/universal E Description of Functions Closed-Loop Control Compo for Speed Control and Positioning Order No.: 6SN1197-0AB20-1BP1	(10.04 Edition) onent
/FBU2/	SIMODRIVE 611 <b>universal</b> Installation Guide (enclosed with SIMODRIVE 611 universal)	(04.02 Edition)
/FBW/	SINUMERIK 840D/810D Description of Functions <b>Tool Management</b> Order No.: 6FC5297-6AC60-0BP1	(03.04 Edition)
/HBA/	SINUMERIK 840D/840Di/810D Manual <b>@Event</b> Order No.: 6AU1900-0CL20-0BA0	(03.02 Edition)
/HBI/	SINUMERIK 840Di Manual <b>SINUMERIK 840Di</b> Order No.: 6FC5297-7AE60-0BP0	(03.04 Edition)
/INC/	SINUMERIK 840D/840Di/810D System Description <b>Commissioning Tool SINUMERIK</b> Order No.: (an integral part of the online help for the sta	(06.03 Edition) <b>SinuCOM NC</b> art-up tool)
/PJE/	SINUMERIK 840D/810D Description of Functions <b>Configuring Package HMI Er</b> Software Update, Configuration, Installation Order No.: 6FC5297-6EA10-0BP0	(08.01 Edition) nbedded
	(the document PS Configuring Syntax is supplied with t and available as a pdf file)	he software
/POS1/	SIMODRIVE <b>POSMO A</b> User Guide's <b>Distributed Positioning Motor on PROF</b> Order No.: 6SN2197-0AA00-0BP7	(08.04 Edition) FIBUS DP
/POS2/	SIMODRIVE <b>POSMO A</b> Installation Guide (enclosed with POSMO A)	(05.03 Edition)
/POS3/	SIMODRIVE POSMO SI/CD/CA User Guide's <b>Distributed Servo Drive Systems</b> Order No.: 6SN2197-0AA20-0BP7	(10.04 Edition)
/POS4/	SIMODRIVE <b>POSMO SI</b> Installation Guide (enclosed with POSMO SI)	(08.03 Edition)

∕▲`

10.04	
	-

/POS5/	SIMODRIVE <b>POSMO CD/CA</b> Installation Guide (enclosed with POSMO CD/CA)	(08.03 Edition)
/S7H/	SIMATIC S7-300 Installation Manual <b>Technological Functions</b> Order No.: 6ES7398-8AA03-8BA0 - Reference Manual: CPU Data (HW Description) - Reference Manual: Module Data	(2002 Edition)
/S7HT/	SIMATIC S7-300 Manual <b>STEP 7, Fundamentals, V. 3.1</b> Order No.: 6ES7810-4CA02-8BA0	(03.97 Edition)
/S7HR/	SIMATIC S7-300 Manual <b>STEP 7, Reference Manuals, V. 3.1</b> Order No.: 6ES7810-4CA02-8BR0	(03.97 Edition)
/S7S/	SIMATIC S7-300 FM 353 Positioning Module for Stepper Drive Order together with configuring package	(04.02 Edition)
/S7L/	SIMATIC S7-300 FM 354 Positioning Module for Servo Drive Order together with configuring package	(04.02 Edition)
/S7M/	SIMATIC S7-300 FM 357.2 Multimodule for Servo and Stepper Drives Order together with configuring package	(01.03 Edition)
/SP/	SIMODRIVE 611-A/611-D <b>SimoPro 3.1</b> Program for Configuring Machine-Tool Drives Order number: 6SC6111-6PC00-0BA Order from: WK Fürth	



d) Installation and Start-up /BS/	SIMODRIVE 611 analog Description <b>Start-Up Software for</b> <b>Main Spindle and Asynchronous Motor Modules Ver</b> Order No.: 6SN1197-0AA30-0BP1	(10.00 Edition) rsion 3.20
/IAA/	SIMODRIVE 611A Installation and Start-Up Guide Order No.: 6SN1197-0AA60-0BP6	(10.00 Edition)
/IAC/	SINUMERIK 810D Installation and Start-Up Guide (including description of SIMODRIVE 611D start-up software) Order No.: 6FC5297-6AD20-0BP1	(11.02 Edition)
/IAD/	SINUMERIK 840D/SIMODRIVE 611D Installation and Start-Up Guide (including description of SIMODRIVE 611D start-up software) Order No.: 6FC5297-7AB10-0BP0	(03.04 Edition)
/IAM/	SINUMERIK 840D/840Di/810DInstallation and Start-Up Guide HMI/MMCOrder No.: 6FC5297-6AE20-0BP3AE1Updates/SupplementsBE1Expanding the Operator InterfaceHE1Online HelpIM2Start-Up HMI EmbeddedIM4Start-Up HMI AdvancedTX1Creating Foreign Language Texts	(03.04 Edition)



## E Commands, identifiers

List of input/output variables of the measuring cycles

Name	Stands for	Explanation
_CALNUM	Calibration groove number	
_CBIT[16]	Central Bits	Field for NCK-global bits
_CHBIT[16]	Channel Bits	Field for channel-specific bits
_CM[8]		Field: Monitoring functions for tool
		measurement with rotating spindle with
		8 elements each
_CORA	Correction angle position	Offset angle position
_CPA	Center point abscissa	
_CPO	Center point ordinate	
_CVAL[4]		Field: Number of elements with e
		elements each
_DIGIT		Number of decimal places
_DLNUM		DL number for set-up and total offsets
_EV[20]		20 empirical value memories
_EVMVNUM[2]		Number of empirical values and mean
		values
_EVNUM		Number of empirical value memory
_FA	Factor for multipl. of measurem. path	Measuring path
_HEADLINE[10]		10 strings for protocol headers
_ID	Infeed in applicate	Incremental infeed depth/offset
_INCA	Indexing angle	
_K	Weighting factor for averaging	
_KB[3,7]		Field: Gauging block data with 7
		elements each
_KNUM		Offset number
_MA	Number of <b>m</b> easuring <b>a</b> xis	
_MD	Measuring direction	
_MFS[]		Field: Feedrates and spindle speeds for
		tool measurement with rotating spindle
		with 6 elements each
_MV[20]		20 mean value memory
_MVAR	Measuring variant	
_NMSP	Number of measurements at same spo	ot
_OVI[20]		Field: Output values INT
_OVR[32]		Field: Output values REAL
_PRNUM	Probe type and probe number	
_PROTFORM[6]		Log formatting
_PROTNAME[2]		Name of log file
_PROTSYM[2]		Separator in the log





_PROTVAL[13]		Log header line
_RA	Number of <b>r</b> otary <b>a</b> xis	
_RF	Feedrate for circular interpolation	Feedrate in circular-path programming
_SETVAL	Setpoint value	
_SETV[9]		Setpoint values for measuring rectangle
_SI[3]	System information	System information
_SPEED[4]		Field: Feedrate values
_STA1	Starting angle	
_SZA	Safety zone on workpiece abscissa	Protection zone in abscissa
_SZO	Safety zone on workpiece ordinate	Protection zone in ordinate
_TDIF	Tolerance dimensional difference chec	:k
_TENV		Name of tool environment
_TLL	Tolerance lower limit	
_TMV		Mean value generation with
		compensation
_TNAME	Tool name	Tool name for use in tool management
_TNUM	T number for automatic tool offset	
_TNVL		Limit value for distortion of triangle
_TP[3,10]		Field: Tool probe data with 6 elements
		each
_TPW[3,10]		3 data fields for tool probe, workpiece-
		related
_TSA	Tolerance safe area	
_TUL	Tolerance upper limit	
_TZL	Tolerance zero offset range	Zero offset
_VMS	Variable measuring velocity	
_WP[3,11]		Field: Workpiece probe data with 9
		elements each



Notes



## F Index

#### 1

1-point measurement 1-49, 1-59, 5-217, 6-337, 6-340, 6-343 1-point measurement inside 1-59 1-point measurement outside 1-59 1-point measurement with reversal 1-59, 6-348 2 2-point measurement 1-60, 6-351 2-point measurement at an angle 1-52 2-point measurement on inside diameter 1-60 2-point measurement on outside diameter 1-60 3 3-point measurement at an angle 1-52 Α Alarms 11-461 Angle measurement 7-383 Automatic tool measurement 6-311 Auxiliary parameters 2-64 R Bidirectional probe 1-27 Blank measurement 1-48, 1-49 С Calculating the repeat accuracy 10-459 Calculation of center point and radius of a circle 3-85 Calibrating a tool probe 5-156, 6-298 Calibrating a tool probe 7-378 Calibrating a workpiece probe in the applicate Determinining probe length 5-188 Calibrating a workpiece probe, hole of known center point 5-177 Calibrating a workpiece probe, hole of unknown hole center 5-180 Calibrating groove pair 9-424 Calibrating in the reference nut 6-331 Calibrating the tool probe 6-280 Calibrating the tool probe automatically (from measuring cycle version SW 6.3) 5-162 Calibrating the workpiece probe at an angle 5-175 Calibrating tool 1-33 Calibrating tool probes 1-47 Calibrating tool probes (machine-related) 6-282 Calibrating workpiece probes 1-48, 5-174, 6-329 Calibration 1-56

Calibration at a surface 6-334 Calibration in groove 7-379 Calibration in hole 7-379 Calibration of a workpiece probe on a surface 5-185 Calibration on surface 7-378 Calibration tool 1-29 Central bits 9-426 Central strings 9-429 Central values 9-420 Channel-oriented bits 9-433 Channel-oriented values 9-430 Commissioning measuring cycles for the first time 10-450 Compensation strategy 5-152, 5-153 Compensation value calculation 1-38 Corner measurement Specifying angles 7-384 Corner measurement, specifying points 7-385 Creating a measurement result log, example 7-371 Cycle data 9-413 CYCLE 961 P 7-385 CYCLE 961 W 7-384 CYCLE\_971 7-380 CYCLE\_972 7-380 CYCLE\_974 7-385 CYCLE 976 7-379 CYCLE 977 979A 7-381 CYCLE 977 979B 7-381 CYCLE\_977\_979C 7-382 CYCLE\_978 7-383 CYCLE\_994 7-386 CYCLE\_998 7-383 CYCLE CAL PROBE 7-378 CYCLE CAL TOOLSETTER 7-378 CYCLE PARA 7-377 CYCLE113 7-364 CYCLE116 3-85 CYCLE119 5-274 CYCLE198 3-87 CYCLE199 3-87 CYCLE961 5-252 CYCLE971 5-150, 5-152, 5-156, 5-166 CYCLE973 6-329 CYCLE974 6-337 CYCLE976 5-174

CYCLE977 5-191 CYCLE978 5-211 CYCLE979 5-221 CYCLE982 6-292 CYCLE994 6-351 CYCLE997 5-264 CYCLE998 5-236 D Data block for the measuring cycles 9-413 Deceleration path calculation 1-35 Defining parameters 2-66 Determining dimensions of calibration 6-285 Determining ZO on a surface 5-214 **Dimensional deviations** 1-38 Dimensional difference control 1-42 F Effect of empirical value, mean value and tolerance parameters 1-46 Empirical value 2-80 Example of functional check 8-406 Example of tool measurement 6-359 F Function test 8-405 G Gauging block 1-29 н Handling the log cycles 7-363 Hardware prerequisites 8-400 I I/O interface 8-402 Incremental calibration 6-318 Incremental measurement 6-321 Input parameters 2-63 L Log content of measurement results 7-367 Log format of measurement results 7-369 Log header of measurement results 7-370 Logging the measurement results 7-362 Μ Machine data 9-408 Machine data for adapting the probe 9-412 Mandatory parameters Measuring cycles 2-63 Mean value 1-39, 2-80 Mean value calculation 1-39 Measure groove, web 7-381 Measure hole, shaft 7-381

Measure surface 5-211 Measurement path FA 2-78 Measuring a groove 5-195, 5-203, 5-223 Measuring a hole 1-50, 5-195, 5-203, 5-223 Measuring a rectangle 5-195, 5-203 Measuring a roove 1-50 Measuring a shaft 1-50, 5-195, 5-203 Measuring a shaft 5-223 Measuring a surface 1-53 Measuring a turning tool (machine-related) 6-286 Measuring a web 1-50, 1-51, 5-195, 5-203, 5-223 Measuring a workpiece at an angle 1-52 Measuring accuracy 1-37 Measuring an angle 1-53, 1-54 Measuring angles 1-53 Measuring axis, number 2-66 Measuring cycle subroutines 3-84 Measuring cycle sub-routines 12-472 Measuring cycle support in the program editor (from SW 6.2) 7-387 Measuring cycle support in the program editor (up to SW 5.4) 7-374 Measuring cycle support, files 7-375 Measuring cycle support, loading 7-375 Measuring cycle user programs 3-87 Measuring cycles, call 7-376 Measuring in JOG Function 4-92 General prerequisites 4-92 Measuring tools 4-140 Measuring workpieces (up to SW 6.2) 4-95 Overview of procedure 4-92 Workpiece measurement Calibrating probe (up to SW 6.2) 4-97 End of measurement (from SW 6.3) 4-134 Measuring a corner (from SW 6.3) 4-119 Measuring a corner (up to SW 6.2) 4-101 Measuring a hole (from SW 6.3) 4-122 Measuring a hole (up to SW 6.2) 4-103 Measuring a spigot (from SW 6.3) 4-122 Measuring a spigot (up to SW 6.2) 4-104 Measuring an edge (up to SW 6.2) 4-100 Measuring the edge (from SW 6.3) 4-113 Orienting a plane (from SW 6.3) 4-132 Rejecting the measurement (from SW 6.3) 4-134

Measure rectangle 7-382

Repeating the measurement (from SW 6.3) 4-134 Workpiece measurement Measuring a pocket (from SW 6.3) 4-122 Workpiece measurement (from SW 6.3) 4-106 Measuring milling tools 7-380 Measuring principle 1-34 Measuring probe connection 8-400 Measuring spheres (from SW 6.3) 1-54 Measuring strategy 1-38, 5-152 Measuring the surface 5-213 Measuring the tool 1-56, 5-166, 6-301 Measuring tools 1-47 Measuring tools at lathes 1-56 Measuring turning tools 6-280, 7-380 Measuring variant 2-66 Measuring variants 1-47, 1-56 Measuring velocity 1-35, 2-76 Mono probe 1-27 Monodirectional probe 1-27 Multi probe 1-27 Multiple measurement at the same location 2-81 Ν Number of the measuring axis 2-66 0 Offset angle position 2-76 Offset axis 2-66 Offset for mono probe setting 2-68 Offset number with flat D number structure 2-71 Ρ Package structure of the measuring cycles 3-88 Parameter overview 2-63 Parameters for checking the measurement result and offset 1-41 Planes definition 1-25 Probe number 2-79 Probe type 2-79 Probes that can be used 1-27 R Reference points at machine and workpiece 1-23 Result parameters 2-65 S Safe area 1-42 Selecting log content 7-367 Sequence for probe installation 10-457 Setting additional parameters 7-377

Setting up a corner with definition of distances and angles 5-254 Setting-up a corner, specifying distances and angles 5-252 Single-point measurement 1-49 Single-point measurement 7-383, 7-385 Software prerequisites 8-403 Software versions 8-403 Starting position/Setpoint position 1-34 Start-up of the measuring cycle user interface for MMC 102 (up to SW 4.4) 12-474 Switching edge of probe 9-433 Switching edge probe 9-433 Т Tolerance bottom limit 1-42 Tolerance parameters 2-77 Tolerance top limit 1-42 Tool Measure turning and milling tools 6-292 Tool measurement for drills 5-150 Tool measurement for milling tools 5-150 Tool name 2-67 Tool number 2-67 Tool probe on milling machine 9-421 Two-point measurement 7-386 U Upgrading measuring cycles 10-456 User program After measurement 3-87 Before measurement 3-87 V Variable measuring velocity 2-76 Variables when logging 7-366 W Weighting factor for mean value calculation 2-81 Workpiece 2-angle measurement 5-246 Measure angle, ZO determination 5-236 Measure groove parallel to axes 5-191 Measure hole parallel to axes 5-191 Measure shaft parallel to axes 5-191 Measure web parallel to axes 5-191 Measuring a surface parallel to the axis 5-211 Measuring spheres and ZO determination (from measuring cycles SW 6.3) 5-264 Paraxial measurement of a rectangle 5-191 Setting up a corner with 4 points 5-259



Setting up an internal and external corner 5-252 Workpiece measurement 1-18, 1-49, 6-337 Groove 5-221 Hole 5-221 Shaft 5-221 Web 5-221 Workpiece probe 1-29

# Ζ

Workpiece

Zero compensation area 1-43

1-angle measurement 5-239

ZO determination in a groove 5-207, 5-232

ZO determination in a hole 5-207, 5-232

ZO determination in a rectangle 5-207

ZO determination on a shaft 5-207, 5-232

ZO determination on a web 5-207, 5-232
To SIEMENS AG	Suggestions
A&D MC BMS Postfach 3180	Corrections For Publication/Manual: SINUMERIK 840D/840Di/810D Measuring Cycles
91050 ERLANGEN, GERMANY Phone +49 (0) 180 / 5050-222 [Hotline] Fax +49 (0) 9131 / 98-2176 [Documentation] Email: motioncontrol.docu@erlf.siemens.de	User Documentation
From	User's Guide
Name	Order No.: 6FC5298-7AA70-0BP1 10.04 Edition
Address:	Should you come across any printing errors when reading this publication, please notify us on this sheet.
Zip code: City:	Suggestions for improvement are also welcome.
Phone: /	
Fax: /	

Suggestions and/or corrections



\*) These documents are a minimum requirement

## **Siemens AG**

Automation and Drives Motion Control Systems Postfach 3180 91050 ERLANGEN GERMANY

© Siemens AG, 2004 Subject to change without prior notice Order No.: 6FC5298-7AA70-0BP1

www.siemens.com/motioncontrol